SECURITIES AND EXCHANGE COMMISSION

FORM S-1/A

General form of registration statement for all companies including face-amount certificate companies [amend]

Filing Date: **1997-12-18 SEC Accession No.** 0000899243-97-002385

(HTML Version on secdatabase.com)

FILER

US LEGAL SUPPORT INC

CIK:1046600| IRS No.: 760523238 | State of Incorp.:TX | Fiscal Year End: 1231

Type: S-1/A | Act: 33 | File No.: 333-36575 | Film No.: 97740157

SIC: 8111 Legal services

Mailing Address 1001 FANNIN STREET SUITE 650 HOUSTON TX 77002 Business Address 1001 FANNIN STREET, SUITE 650 HOUSTON TX 77002 7136537100 _____

SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

AMENDMENT NO. 3 TO FORM S-1 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

U.S. LEGAL SUPPORT, INC. (EXACT NAME OF REGISTRANT AS SPECIFIED IN ITS CHARTER)

TEXAS (STATE OR OTHER (PRIMARY STANDARD JURISDICTION OF INCORPORATION OR CLASSIFICATION CODE ORGANIZATION) ORGANIZATION)

7338 INDUSTRIAL NUMBER)

76-0523238 (I.R.S. EMPLOYER IDENTIFICATION NO.)

RICHARD O. LOONEY PRESIDENT AND CHIEF EXECUTIVE OFFICER

1001 FANNIN ST., SUITE 650

(ADDRESS , INCLUDING ZIP CODE, AND (ADDRESS , INCLUDING ZIP CODE, AND
TELEPHONE NUMBER, INCLUDING AREA CODE,
OF REGISTRANT'S PRINCIPAL EXECUTIVE OF AGENT FOR SERVICE) OFFICES)

(NAME, ADDRESS, INCLDING ZIP CODE, AND

Copies to:

W. CLELAND DADE DAN BUSBEE BRACEWELL & PATTERSON, L.L.P. LOCKE PURNELL RAIN HARRELL (A PROFESSIONAL 711 LOUISIANA STREET, SUITE 2900 CORPORATION)
HOUSTON, TEXAS 77002-2781 2200 ROSS AVENUE, SUITE 2200 DALLAS, TEXAS 75201 (713) 221-1314 (214) 740-8000

Approximate date of commencement of proposed sale to the public: As soon as practicable after this Registration Statement becomes effective.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, check the following box. [_]

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. [_]

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. [_]

If this Form is a post-effective amendment filed pursuant to Rule $462\,(\mathrm{d})$ under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. [_]

If delivery of the prospectus is expected to be made pursuant to Rule 434, please check the following box. [_]

THE REGISTRANT HEREBY AMENDS THIS REGISTRATION STATEMENT ON SUCH DATE OR DATES AS MAY BE NECESSARY TO DELAY ITS EFFECTIVE DATE UNTIL THE REGISTRANT SHALL FILE A FURTHER AMENDMENT WHICH SPECIFICALLY STATES THAT THIS REGISTRATION STATEMENT SHALL THEREAFTER BECOME EFFECTIVE IN ACCORDANCE WITH SECTION 8 (A) OF THE SECURITIES ACT OF 1933 OR UNTIL THIS REGISTRATION STATEMENT SHALL BECOME EFFECTIVE ON SUCH DATE AS THE COMMISSION, ACTING PURSUANT TO SAID SECTION 8 (A), MAY DETERMINE.

SUBJECT TO COMPLETION, DATED , 1998 3,500,000 SHARES

LOGO

[LOGO OF U.S. LEGAL SUPPORT, INC. APPEARS HERE]

COMMON STOCK

All of the 3,500,000 shares of Common Stock offered hereby are being offered by U.S. Legal Support, Inc. (the "Company"). The Company and a shareholder have granted to the Underwriters a 30-day option to purchase up to 525,000 additional shares of Common Stock solely to cover over-allotments, if any. If the over-allotment option is exercised, up to 100,000 shares of Common Stock will be offered by a shareholder.

Prior to this offering (the "Offering"), there has been no public market for the Common Stock of the Company. It is currently estimated that the initial public offering price of the Common Stock will be between \$11.00 and \$13.00 per share. See "Underwriting" for a discussion of the factors considered in determining the initial public offering price. The Company has applied for quotation of the Common Stock on the Nasdaq National Market under the symbol "LEGL," subject to official notice of issuance.

SEE "RISK FACTORS" COMMENCING ON PAGE 7 FOR A DISCUSSION OF CERTAIN FACTORS THAT SHOULD BE CONSIDERED BY PROSPECTIVE PURCHASERS OF THE COMMON STOCK OFFERED HEREBY

THESE SECURITIES HAVE NOT BEEN APPROVED OR DISAPPROVED BY THE SECURITIES AND EXCHANGE COMMISSION OR ANY STATE SECURITIES COMMISSION NOR HAS THE SECURITIES AND EXCHANGE COMMISSION OR ANY STATE SECURITIES COMMISSION PASSED UPON THE ACCURACY OR ADEQUACY OF THIS PROSPECTUS. ANY REPRESENTATION TO THE CONTRARY IS A CRIMINAL OFFENSE.

<TABLE> <CAPTION>

		Underwriting Discount (1)	
<\$>	<c></c>	<c></c>	<c></c>
Per Share	\$	\$	\$
Total (3)	\$	\$	\$

 | | |(1) See "Underwriting" for information concerning indemnification of the Underwriters and other matters.

- (2) Before deducting expenses payable by the Company, estimated at \$2,350,000.
- (3) The Company and a shareholder, Mr. Richard O. Looney, President and Chief Executive Officer of the Company, have granted to the Underwriters a 30-day option to purchase up to 525,000 additional shares of Common Stock solely

to cover over-allotments, if any. If the Underwriters exercise this option in full, the Price to Public will total \$, the Underwriting Discount will total \$, the Proceeds to Company will total \$ and the Proceeds to Shareholder will total \$. See "Underwriting."

The shares of Common Stock are offered by the several Underwriters named herein, subject to receipt and acceptance by them and subject to their right to reject any order in whole or in part. It is expected that delivery of the certificates representing such shares will be made against payment therefor at the office of NationsBanc Montgomery Securities, Inc. on or about , 1998.

NATIONSBANC MONTGOMERY SECURITIES, INC.

HAMBRECHT & QUIST

J.C. BRADFORD & CO.

, 1998

LOGO OF U.S. LEGAL SUPPORT, INC. APPEARS HERE

[COLOR MAP OF UNITED STATES WITH COMPANY OFFICE AND NETWORK AFFILIATE LOCATIONS IDENTIFIED.]

A LEADING PROVIDER OF LEGAL SUPPORT AND STAFFING SERVICES

The Company intends to furnish its shareholders with annual reports containing financial statements audited by independent certified public accountants and with quarterly reports containing unaudited summary financial information for each of the first three quarters of each fiscal year.

CERTAIN PERSONS PARTICIPATING IN THIS OFFERING MAY ENGAGE IN TRANSACTIONS THAT STABILIZE, MAINTAIN, OR OTHERWISE AFFECT THE PRICE OF THE COMMON STOCK, INCLUDING STABILIZING BIDS, SYNDICATE COVERING TRANSACTIONS AND THE IMPOSITION OF PENALTY BIDS. FOR A DESCRIPTION OF THESE ACTIVITIES, SEE "UNDERWRITING."

2

PROSPECTUS SUMMARY

The following summary is qualified in its entirety by reference to, and should be read in conjunction with, the more detailed information and the financial statements, including the related notes thereto, appearing elsewhere in this Prospectus. Concurrently with the closing of the Offering made by this Prospectus, the Company will acquire three legal support businesses and one staffing business (collectively, the "Pending Acquisitions") in separate transactions in exchange for shares of Common Stock and cash. Unless the context otherwise requires, the "Company" refers to U.S. Legal Support, Inc., its subsidiaries and the Pending Acquisitions. See "The Company--Pending Acquisitions." Disclosures herein relating to the number of shares of Common Stock to be outstanding after the Offering are estimated, based upon an assumed initial public offering price of \$12.00 per share (the mid-point of the estimated initial public offering price range) and give effect to the conversion of all outstanding securities that are convertible into Common Stock as described herein. See "--The Offering." Unless otherwise indicated, the information in this Prospectus: (i) gives effect to the Pending Acquisitions; (ii) assumes no exercise of the Underwriters' over-allotment option; and (iii) gives effect to the 100-for-one stock split effected on December 16, 1996.

THE COMPANY

The Company is one of the largest providers of legal support and staffing services in the United States, providing court reporting, certified record retrieval, legal placement and staffing and related services to law firms and corporations, including insurance companies, through 40 offices in seven states and the District of Columbia. The Company seeks to become the leading national, full-service provider of legal support and staffing services through a combination of selective acquisitions and internal growth. After succeeding to the operations of Looney in January 1997, the Company has acquired 13 businesses and will acquire four additional businesses concurrently with the Offering. In 1997, the Company has provided court reporting and certified record retrieval services to The Boeing Company, Ford Motor Company, ITT Hartford, CNA and Kemper Insurance, among others. For the year ended December 31, 1996, the Company had pro forma revenues of \$43.4 million and pro forma operating income of \$5.2 million. For the nine months ended September 30, 1997, the Company had pro forma revenues of \$37.3 million and pro forma operating income of \$5.3 million compared with pro forma revenues of \$32.0 million and pro forma operating income of \$3.7 million for the nine months ended September

Based on available industry data, the Company estimates that the market for legal support and staffing services in the United States exceeds \$5.0 billion annually. The industry is highly fragmented, with more than 1,000 court reporting and record retrieval firms and over 400 legal placement and staffing firms. The Company believes that the legal support and staffing services market is growing due to several trends, including an increase in the: (i) outsourcing of legal support services by law firms, corporations and insurance providers, to companies that specialize in providing such services at a lower cost; (ii) use of attorneys on a temporary basis by law firms and corporations; (iii) volume and complexity of litigation; and (iv) national scope of litigation, particularly in class action and product liability lawsuits.

Legal support and staffing services traditionally have been marketed to law firms. Increasingly, corporations, especially insurance providers, who ultimately pay the costs of legal support and staffing services, are seeking to control and reduce the costs associated with lawsuits, centralize their purchasing decisions and ensure consistent service quality. As a result, the Company believes that these companies are more frequently selecting the providers of legal support services, rather than delegating that selection to the law firms engaged to represent them. Currently the industry is highly fragmented and consists primarily of local and regional firms that typically provide a single or limited number of legal support and staffing services. The Company believes that many legal

3

support businesses lack: (i) a full range of legal support services; (ii) regional or national coverage; and (iii) access to capital and effective marketing programs. As a result, the Company believes that many legal support and staffing companies are unable to effectively service large, geographically dispersed clients.

The Company is implementing a focused business strategy that includes establishing full service operations in multiple metropolitan areas; adopting best practices, policies and procedures; achieving operating efficiencies; and managing its business on a decentralized basis, with local management retaining primary responsibility for day-to-day operations and local marketing. The Company believes that allowing local management to retain appropriate autonomy will preserve existing client relationships, provide opportunities for internal growth and enhance the Company's competitiveness in attracting acquisition candidates.

The Company has implemented a strategy designed to continue its growth in existing and new markets based on the following key elements: (i) actively pursue strategic acquisitions; (ii) establish an effective national marketing program; (iii) capitalize on cross-selling opportunities; and (iv) develop and expand new and existing client relationships. The Company's acquisition strategy is to identify, acquire and integrate independent companies with strong management, profitable operating results and recognized local or regional market presence. The Company typically pursues acquisitions that will allow it to accomplish one or more of the following: (i) expand the geographic markets served by the Company; (ii) increase the Company's penetration of existing markets; (iii) establish or enhance customer relationships; and (iv) offer services complementary to those offered by the Company. The Company believes there are numerous attractive acquisition candidates due to the large size and fragmentation of the legal support and staffing services industry, including participants in the Company's referral network of over 130 court reporting affiliates, through which the Company supplies court reporting services to its clients in locations not served directly by the Company.

The Company is a Texas corporation. Its principal executive offices are located at 1001 Fannin Street, Suite 650, Houston, Texas 77002, and its telephone number at that location is (713) 653-7100.

THE OFFERING

Common Stock offered by the	3,500,000 shares				
Common Stock to be outstanding after the Offering	7,813,115 shares (1) (2)				
Use of proceeds	To repay indebtedness, to pay a portion of the purchase price associated with the Pending Acquisitions and to redeem shares of the Company's Series C Preferred Stock. See "Use of Proceeds."				
Proposed Nasdaq National Market symbol	LEGL				

Acquisitions and gives effect to the conversion of: (i) 1,000,000 shares of Series A Convertible Preferred Stock into 1,560,000 shares of Common Stock; (ii) 2,046,667 shares of Series B Convertible Preferred Stock into 183,393 shares of Common Stock; and (iii) \$1.8 million principal amount of Convertible Subordinated Promissory Notes into 225,890 shares of Common Stock, in each case to be effected concurrently with the Offering. See "Capitalization" and Notes 7 and 8 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc.

(2) Excludes (i) 900,000 shares of Common Stock reserved for issuance under the Company's 1997 Stock Incentive Plan and the Company's Stock Option Plan for Non-Employee Directors and (ii) 131,856 shares of Common Stock issuable upon exercise of options granted in connection with completed acquisitions. See "Management," "Capitalization" and "Principal Shareholders."

4

SUMMARY FINANCIAL DATA (1) (IN THOUSANDS, EXCEPT PER SHARE DATA)

<TABLE> <CAPTION>

CAFIION							PRO FORMA		
		AR ENDE EMBER 3		E	MONTHS NDED MBER 30,	YEAR ENDED DECEMBER 31,	EN	MONTHS NDED MBER 30,	
	1994	1995	1996	1996	1997 (2)	1996 (3)(4)	1996 (3)(4)		
<s> STATEMENT OF INCOME DATA:</s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	
Revenues Cost of services	\$8,363 5,589	5,763	4,839	3,726	\$14,549 9,287	\$43,405 25,474	\$32,024 18,575	\$37,277 21,261	
Gross profit Selling, general and administrative			2,828			17,931	13,449	16,016	
expenses (5) Depreciation and	3,043	1,970	2,352	1,310	3 , 855	10,968	8,420	9,365	
amortization (4)	224	231	212	159	332	1,754 	1,332	1,337 	
Operating income (loss) Interest expense	(493) 185	1,140 230		691 178	1,075 1,147	5,209 880	3,697 668	5,314 707	
<pre>Income (loss) before taxes Provisions (benefit) for</pre>	(678)	910	26	513	(72)	4,329	3,029	4,607	
income taxes	(183)	327		174	11	1,847	1,298	1,928 	
Net income (loss) Accretion of preferred	(495)	583		339	(83)	2,482	1,731	2,679	
stock Net income (loss)					(479)				
attributable to common shareholders					\$ (562) =====	\$ 2,482 =====	\$ 1,731 ======	\$ 2,679 =====	
Net income (loss) per common share (6)					\$ (.13) ======	\$.31 =====	\$.22 =====	\$.33 =====	
Weighted average shares outstanding (7)(8)					4,226 ======	8,002 =====	8,002 =====	8,002 =====	

					SEPTEME	BER 30, 1997		
						PRO FORMA (3)		
<pre><s> BALANCE SHEET DATA: Cash</s></pre>	rrent ma	turitie	s)		\$ 559 38,720 3,936 26,339 3,757	\$ 1,238 62,652 75 11,083 \$43,940		

⁽¹⁾ Prior to January 17, 1997, the Company had no business operations. Therefore, the business of Looney & Company, for financial statement

- (2) The Company's Completed Acquisitions (as hereafter defined) have been, and the Pending Acquisitions will be, accounted for as purchases, and therefore, the operations of the acquired businesses are included in the statement of income data from the respective dates of acquisition. See the Consolidated Financial Statements of U.S. Legal Support, Inc. included herein
- (3) Pro forma information gives effect to: (i) the completed acquisitions and completion of the Pending Acquisitions; (ii) an adjustment to compensation expense for the difference between actual compensation paid to certain officers of businesses acquired or to be acquired and employment contract compensation negotiated in connection with the completed acquisitions and the Pending Acquisitions; (iii) amortization expense relating to intangible assets recorded in conjunction with the completed acquisitions and to be recorded in the Pending Acquisitions; and (iv) the sale of the shares offered hereby and the application of the net proceeds thereof, as if such events had occurred on January 1, 1996 (for statement of income data) and as of September 30, 1997 (for balance sheet data). The Pending Acquisitions and the conversion of certain outstanding preferred stock and Convertible Subordinated Promissory Notes described in Note 6 below are contingent upon, and will close concurrently with, completion of the Offering. The pro forma results of operations are not necessarily indicative of the results that would have occurred had these transactions been completed as of such date or the results that may be attained in the future.
- (4) Pro forma depreciation and amortization amounts consist primarily of amortization of goodwill totaling \$1,313,000 and \$985,000 for the periods ended December 31, 1996 and September 30, 1997, respectively, recorded or to be recorded as a result of the completed acquisitions and the Pending Acquisitions. Goodwill is amortized over periods ranging from ten to 40 years and computed on the basis described in Note 2 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc.
- (5) Includes a non-recurring charge of \$360,000 in the fourth quarter of 1996 representing the estimated fair value of ownership interests granted to certain employees by Looney's shareholder.
- (6) Supplementary historical earnings per common share for the nine months ended September 30, 1997, would be \$.09 per share, giving effect to: (i) the 3,893,000 weighted average shares outstanding for the period; (ii) the issuance of 2,908,000 of shares in the Offering, the proceeds from which would be necessary to (a) repay \$16,000,000 of bank indebtedness under the Company's credit facility, (b) redeem \$9,000,000 of Senior Subordinated Notes, (c) repay \$4,979,000 of Subordinated Promissory Notes, (d) redeem 231,250 shares of Series C Preferred Stock, and (e) pay the applicable Offering expenses; and (iii) the reduction of interest expense and dividends.
- (7) Gives effect to: (i) 1,734,564 shares outstanding prior to the Offering; (ii) 3,500,000 shares issued in the Offering; (iii) 609,268 shares to be issued in the Pending Acquisitions; (iv) 1,969,283 shares issuable upon conversion of preferred stock and Convertible Subordinated Promissory Notes; and (v) 188,187 shares issuable upon exercise of outstanding stock options in accordance with SEC Staff Accounting Bulletin Topic 4D. See "Capitalization."
- (8) Shares used in calculating net loss per share for the nine months ended September 30, 1997, include the number of shares, the proceeds from the sale of which would be necessary to repay the portion of the Company's debt that funded the distribution to Richard O. Looney in connection with the reorganization of the Company and Looney & Company.

5

SUMMARY OF INDIVIDUAL COMPANY REVENUES (1)

(IN THOUSANDS)

After succeeding to the operations of Looney in January 1997, the Company has acquired 13 businesses and will acquire four others in the Pending Acquisitions. The following table sets forth a summary of the revenues attributable to Looney, the principal businesses acquired and the businesses to be acquired in the Pending Acquisitions for the fiscal years ended December 31, 1995 and 1996 and for the nine-month periods ended September 30, 1996 and 1997.

<TABLE>

Looney (2).....

<S>

YEAR I	ENDED	NINE MON'	THS ENDED
DECEMBI	ER 31,	SEPTEM	BER 30,
1995	1996	1996	1997
<c></c>	<c></c>	<c></c>	<c></c>
 \$ 9,104	\$ 7,667	\$ 5,886	\$ 5,641

Klein Bury (3). Elaine Dine (4). Legal Enterprise. Reporting Service (5) Jilio (5). Johnson Group (6). Ziskind Greene. Amicus One. Kirby Kennedy (5).	7,302 3,503 2,756 1,906 3,366 1,714 1,442 1,642 1,629	8,526 4,658 3,707 3,012 4,022 2,155 1,841 1,883 1,866	6,524 2,808 2,711 2,171 2,772 1,616 1,593 1,394 1,455	6,460 5,821 3,464 3,067 3,085 1,702 1,653 1,544 1,427
G&G San Francisco Reporting	1,544 1,105	1,517 1,140	1 , 152	1,371 946
Block	1,025	1,317	947	760
Commander Wilson (5)	578	94	84	336
Total			\$32,024	\$37,277

</TABLE>

- (1) See "The Company" for the full names and additional information concerning the completed acquisitions and the Pending Acquisitions.
- (2) The revenues of Looney include from their respective dates of acquisition, the revenues of: (i) Cindi Rogers & Associates (acquired April 3, 1997); (ii) Preferred Records, Inc. (acquired July 31, 1997); (iii) Rocca Reporting Service (acquired August 15, 1997); and (iv) Encore Reporting (acquired August 28, 1997). The table does not include revenues for these entities prior to their acquisition because these amounts are immaterial. See "The Company--Completed Acquisitions."
- (3) Revenues for 1995 are for the year ended September 30, 1995.
- (4) The revenues for the nine months ended September 30, 1997 include the revenues of Elaine Dine Temporary Attorneys, L.L.C. Revenues for 1995 and 1996 represent results for the years ended March 31, 1996 and 1997, respectively.
- (5) To be acquired in a Pending Acquisition concurrently with the completion of the Offering.
- (6) The revenues of Johnson Group include the combined revenues of Goren of Newport, Inc., Medtext, Inc. and Rapidtext, Inc.

6

RISK FACTORS

The factors set forth below should be considered carefully in evaluating an investment in the shares of Common Stock offered by this Prospectus. Further, this Prospectus contains certain forward-looking statements. These forward-looking statements are subject to certain assumptions, risks and uncertainties which may cause actual results to be materially different from those expressed in or implied by such statements.

ABSENCE OF COMBINED OPERATING HISTORY; RISKS OF INTEGRATING ACQUIRED COMPANIES

Since the combination of Looney and the Company in January 1997, the Company has acquired 13 businesses. The Company also has entered into agreements to acquire four additional businesses in the Pending Acquisitions. While the Company intends to continue to emphasize decentralized management of operations and marketing in the acquired businesses, its success will depend, to a large extent, upon its ability to integrate effectively the operations of the acquired businesses. There can be no assurance that the recently assembled management group will be able to implement successfully the Company's business and growth strategies or manage successfully the combined operations of the Company and the businesses acquired. Most of the businesses acquired or to be acquired in the Pending Acquisitions historically have operated with limited financial and other reporting systems. Although the Company believes that its existing financial reporting and accounting control systems are satisfactory for the operation of the combined businesses for the near term, these systems may not be capable of efficiently addressing the rapidly changing needs of the Company as it implements its acquisition strategy. The Company is currently evaluating alternatives designed to address these needs, including the possibility of enhancing existing systems or acquiring new systems. Any changes in the systems currently used by the Company could require the Company to provide additional training to existing personnel or hire additional personnel. If the Company cannot upgrade or replace these systems in a timely manner, or with respect to acquired businesses, if it must rely on the financial reporting and accounting control systems of the businesses acquired, the Company's ability to integrate successfully and manage effectively the combined enterprise could be adversely affected. The pro forma combined historical financial results included herein cover periods during which the businesses acquired were not under common control and may not be indicative of the Company's future financial or operating results. See "Business--Acquisition and Integration Strategy" and "Management."

RISKS ASSOCIATED WITH ACQUISITIONS

The Company's growth strategy is dependent upon a program of continuing acquisitions. However, there can be no assurance the Company will be able to identify attractive acquisition candidates or to negotiate acquisition terms acceptable to the Company, and the failure to do so would have a material adverse effect on the Company's results of operations and its ability to sustain growth. The Company's acquisition strategy involves a number of risks, each of which could affect adversely the Company's reported operating results, including the diversion of management attention from operation of the business, loss of key personnel from acquired businesses and the failure of an acquired business to achieve targeted financial results. In addition, the Company could encounter unanticipated business risks or unanticipated liabilities with respect to the acquired businesses, and significant amortization of acquired intangible assets is likely to be required in most acquisitions. Further, there can be no assurance that the Company's strategy to become a national, full service provider of legal support services will be successful, or that the Company's clients will accept the Company as a provider of such services. The legal support and staffing industry is undergoing consolidation, and the resulting increase in the competition for acquisition candidates could limit the Company's acquisition opportunities or increase the cost of acquisitions. See "Business--Growth Strategy."

The Company will require additional financing for future acquisitions, which may not be available on terms favorable to the Company, if available at all. The Company currently intends to finance future acquisitions using shares of its Common Stock for a significant portion of the purchase price. In the event the Company's Common Stock does not maintain sufficient value or potential acquisition candidates are unwilling to accept Common Stock as consideration for the sale of their businesses, the Company may be required to utilize more of its cash

7

resources, if available, in order to continue its acquisition program. The net proceeds of the Offering will be used primarily to repay existing indebtedness and to partially fund the cash portion of the purchase price of the Pending Acquisitions, and none of such proceeds will be available for future acquisitions. If the Company does not have sufficient cash resources, is unable to borrow the funds required to make acquisitions or is not able to use its Common Stock as consideration for acquisitions, its growth through acquisitions would be limited. See "Management's Discussion and Analysis of Financial Condition and Results of Operations--Liquidity and Capital Resources."

RISKS ASSOCIATED WITH RAPID GROWTH

The Company has experienced rapid growth through acquisitions since it commenced operations in January 1997, which has placed demands on its management, operational capacity and financial resources. The Company's growth strategy provides for a continuing acquisition program, which will place further demands on its management, operational capacity and financial resources and systems. The increased management requirements will necessitate the recruitment and retention of additional qualified management personnel and the purchase and implementation of new management information systems. There can be no assurance that the Company will be able to recruit and retain qualified personnel or expand and manage its operations effectively and profitably. The failure to manage growth effectively could have a material adverse effect on the Company's business, results of operations and financial condition. See "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Business."

REGULATORY INITIATIVES

The Company derives most of its revenues from operations in California, Florida, New York, Pennsylvania and Texas. Legislation or regulations enacted in any of these states, states into which the Company expands or at the federal level relating to lawsuits or other dispute resolution proceedings, or to the provision of court reporting or other legal support and staffing services provided by the Company, could have a material adverse effect on the Company's business and results of operations. A key component of the Company's business strategy is the pursuit of arrangements with insurance providers and major corporations under which the Company is designated as the exclusive or preferred provider of court reporting and certified record retrieval services. Legislation recently was proposed in the State of Texas that could have prohibited such arrangements by making illegal the provision of services by a court reporter under any agreement other than on a case-by-case basis. The proposed legislation also would have prohibited a court reporter from being employed by, or serving as an independent contractor for, a court reporting firm unless a majority of the firm is owned by certified shorthand reporters. While the foregoing provisions were not included in the Texas legislation as enacted, the Company expects that there will continue to be efforts to sponsor the adoption of similar prohibitions in legislative or regulatory action or through the ethics codes governing the conduct of court reporters or attorneys. If enacted, these prohibitions would represent a significant

impediment to the implementation of the Company's current business strategy and could have a material adverse effect on the Company's business and results of operations. Other states have enacted or considered such legislation and may do so in the future. West Virginia recently enacted legislation that prohibits a court reporter from entering into a contract for the provision of court reporting services directly with a party to a lawsuit. In addition, recent federal and certain state legislative proposals have included limitations on the number and length of depositions or proposed the substitution of videotaped reporting for stenographic transcription of certain legal proceedings.

State and national bar associations and committees on legal ethics and professional responsibility have from time to time issued opinions regarding the ethical implications of arrangements involving temporary attorneys. These opinions have suggested that the payment of fees to agencies that place temporary attorneys may constitute, in certain circumstances, the improper splitting of legal fees with a non-lawyer. The applicability of these opinions to the Company's business is uncertain, and there can be no assurance that a state will not determine that the business as conducted by the Company violates ethical or professional responsibility regulations for attorneys. In addition, the practice of placing temporary lawyers with a number of firms may raise conflict of interest issues under applicable ethics codes, particularly when attorneys from a placement firm are placed with opposing parties, or law firms representing such parties, in a lawsuit or business transaction.

8

The Company cannot determine whether legislative or regulatory proposals affecting the Company's operations will be initiated, reproposed or enacted; however, if adopted, certain of such proposals could require the Company to alter the manner in which it conducts its business, and could materially and adversely affect its business and results of operations. See "Business--Regulation."

ALTERNATIVE DISPUTE RESOLUTION

The high cost of litigation in the United States has resulted in the more frequent use of alternative dispute resolution practices, such as arbitration and mediation. The increasing use of such alternative dispute resolution practices could affect the demand for the Company's legal support and staffing services; however, the Company is unable to assess the ultimate effect, if any, that such practices may have on its services.

COMPETITION

The legal support and staffing services industry is highly competitive and fragmented, and limited barriers to entry exist with respect to each component of the Company's business. Although the industry is undergoing consolidation, the principal competition for the Company's court reporting and certified record retrieval businesses currently consists of numerous local and regional firms. In its legal staffing business, the Company competes with national, regional and local firms, some of which have substantially greater resources, offer more diversified services and operate in broader geographic areas than the Company. As the Company seeks to expand into new geographic markets, its growth in those markets will depend upon its ability to gain market share from competitors, and there can be no assurance that additional market share will be obtained. Prices for legal support services generally have remained stable or have declined in many markets over the past several years as law firms, insurance providers and corporations have increased their efforts to reduce expenses by centralizing their purchasing decisions and negotiating lower rates with vendors. As this trend and the related consolidation among legal support service companies continue, the Company anticipates that it will encounter more intense price-based competition which could adversely affect the Company's profitability. See "Business--Competition."

STATUS OF INDEPENDENT CONTRACTORS

The Company typically provides court reporting services through independent contractors and, therefore, does not pay federal or state employment taxes or withhold income taxes for such persons. Further, the Company does not include these independent contractors in its employee benefit plans. From time to time persons engaged as independent contractors are determined to be employees as a result of challenges by the federal Internal Revenue Service ("IRS") and state authorities asserted in administrative proceedings or court action. In certain instances, persons engaged to be independent contractors also could initiate court proceedings asserting that they are employees under state law and should be included in employee benefit plans. Such determinations of employee status under these challenges are made on a case-by-case basis and are based upon the particular facts of each case. In the event persons engaged by the Company as independent contractors are determined to be employees by the IRS or any state taxation department, the Company would be required to pay applicable federal and state employment taxes and withhold income taxes with respect to such persons and could become liable for amounts required to be paid or withheld in prior periods. In addition, the Company could be required to include such persons in its employee benefit plans on a retroactive as well as a current basis. Approximately 500 court reporters are engaged by the Company as independent contractors, and any challenge by the IRS, state authorities or private litigants resulting in a determination that such persons are employees would have a material adverse effect on the Company's business, results of operations and financial condition. In October 1997, a bill was introduced in the U.S. House of Representatives that would amend the Internal Revenue Code to establish more stringent requirements for the engagement of independent contractors. The Company is unable to assess the likelihood that this bill or similar legislation will be enacted. See "Business--Legal Support and Staffing Services" and "--Independent Contractors."

DEPENDENCE UPON KEY PERSONNEL

The Company is dependent on the continuing services of Richard O. Looney, other executive officers and the senior management of the acquired businesses, and the Company likely will depend on the senior management of any significant business it acquires in the future. The business or prospects of the Company could

9

be affected adversely if any of these persons do not continue in their management roles and the Company is unable to attract and retain qualified replacements. The Company does not currently maintain key-man life insurance on any executive officer. See "Management."

ABILITY TO ATTRACT AND RETAIN QUALIFIED PERSONNEL

The Company is dependent on the availability of a sufficient number of licensed court reporters. From time to time, there are shortages of qualified court reporters, and there can be no assurance that the Company will be able to maintain an adequate force of licensed court reporters or that the Company's expenses will not increase as a result of such a shortage. The Company competes with other legal staffing companies, as well as its clients, for qualified attorneys. The Company expects that a substantial number of the Company's temporary legal staffing personnel will terminate their temporary assignments to accept full-time employment with Company clients. The Company also may encounter difficulty in recruiting a sufficient number of qualified attorneys. In addition, during periods of high demand for the Company's services, the Company may incur increased expense associated with recruiting qualified temporary attorneys. The Company's inability to attract and retain a sufficient number of qualified temporary attorneys, or the loss of a significant number of qualified temporary attorneys to clients, could adversely affect the Company's operating results and business strategy. See "Business--Business Strategy."

POTENTIAL LIABILITY TO CLIENTS

The provision of personnel in the legal support and staffing business entails an inherent risk of professional malpractice and similar claims, and the Company could be named as a defendant in litigation in connection with services rendered. Although the Company maintains errors and omissions insurance for court reporting, record retrieval and temporary and permanent legal staffing businesses, there can be no assurance that, if named, the Company would be able to defend such claims within policy limits. The Company's business also involves the handling of clients' documents containing confidential and other sensitive information. There can be no assurance that unauthorized disclosures will not occur, or that clients' documents will not be mishandled or lost, which could have a material adverse effect on the Company's business reputation or results of operations. The Company also may be subject to discrimination and harassment claims for the acts of legal staffing personnel who are placed with the Company's clients. See "Business—Regulation."

TECHNOLOGICAL ADVANCES

The Company's business is subject to changes in technology, which may result in the introduction of new products or services that are competitive with, superior to, or render obsolete the services provided by the Company. For example, voice recognition technology is designed to eliminate the need for manual transcription of oral testimony and has been under development for several years. There can be no assurance that substantial advances will not be made in the area of voice recognition technology or that other technology will not be developed that renders the Company's services obsolete or impractical. These changes in technology could also include conducting certified document retrieval electronically, with deposition notices and subpoenas being transmitted electronically to document custodians and witnesses who similarly would return responses electronically. The Company's ability to compete effectively will depend upon its ability to adapt to such changes and to develop services to satisfy evolving client requirements. There can be no assurance that current technologies, or technologies developed in the future,

will not compete with or replace services provided by the Company, or that any technological advances made by the Company will be responsive to client requirements or represent the best available technology to meet client needs. See "Business--Legal Support and Staffing Services."

CONTROL BY OFFICERS, DIRECTORS AND SHAREHOLDERS

Upon completion of the Offering, the Company's officers, directors and current principal shareholders will beneficially own approximately 42.0% of the outstanding Common Stock. These persons, if acting together, will have substantial control over matters requiring approval of the shareholders of the Company, including the election of directors. See "--Anti-Takeover Effect of Certain Charter Provisions," "Management" and "Principal Shareholders."

10

DILUTION

Purchasers of Common Stock in the Offering will experience immediate dilution in net tangible book value per share of Common Stock of \$12.93 from the assumed initial public offering price per share of \$12.00 and may experience further dilution in that value from issuances of Common Stock in connection with future acquisitions. See "Dilution."

ABSENCE OF PUBLIC MARKET; POSSIBLE VOLATILITY OF STOCK PRICE

Prior to the Offering, there has been no public market for the Common Stock. There can be no assurance that an active trading market for the Common Stock will develop or be sustained after the Offering or that purchasers of the Common Stock will be able to resell their shares at prices equal to or greater than the initial public offering price. The initial public offering price will be determined through negotiations between the Company and the Underwriters and may not be indicative of the prices that will prevail in the public market after the Offering. Numerous factors, including fluctuations in the operating results of the Company or its competitors, fluctuations in the demand for the Company's services or business services in general, and the timing and announcement of acquisitions by the Company or its competitors, could have a significant impact on the price of the Common Stock. In addition, the stock markets have experienced significant price and volume fluctuations in recent years that often have been unrelated or disproportionate to the operating performance of companies. These fluctuations may adversely affect the market price of the Common Stock. See "Underwriting."

SHARES ELIGIBLE FOR FUTURE SALE

Upon completion of the Offering, the Company will have 7,813,115 shares of Common Stock outstanding. Of these shares, all of the shares sold in the Offering will be freely transferable without restriction or limitation under the Securities Act of 1933, as amended (the "Securities Act"), except for any shares purchased by "affiliates" of the Company, as such term is defined in Rule 144 under the Securities Act. The remaining 4,313,115 shares constitute "restricted securities" within the meaning of Rule 144 and the resale of such shares is restricted for one year from the date they were acquired. The holders of such shares have certain rights to have shares registered in the future under the Securities Act pursuant to the terms of agreements between such holders and the Company. The Company, its executive officers, directors and principal shareholders have agreed not to offer or sell any shares of Common Stock for a period of 180 days following the date of this Prospectus without the prior written consent of Montgomery Securities, except that the Company may issue shares of Common Stock in connection with acquisitions and pursuant to the exercise of stock options described in this Prospectus. On the date of this Prospectus, the Company had outstanding options to purchase 609,024 shares of Common Stock granted pursuant to the 1997 Stock Incentive Plan. In addition, the Company intends to grant options to purchase 100,000 shares of Common Stock pursuant to the Company's Stock Option Plan for Non-Employee Directors upon completion of the Offering. The Company intends to register all of the 900,000 shares of Common Stock reserved for issuance pursuant to the Company's 1997 Stock Incentive Plan and pursuant to the Company's Stock Option Plan for Non-Employee Directors under the Securities Act for public resale. On the date of this Prospectus, the Company had outstanding options to purchase 131,856 shares of Common Stock granted in connection with the Completed Acquisitions. Sales of substantial amounts of shares of Common Stock in the public market after this Offering or the perception that such sales could occur may adversely affect the market price of the Common Stock. See "Shares Eligible for Future Sale."

ANTI-TAKEOVER EFFECT OF CERTAIN CHARTER PROVISIONS

The Board of Directors of the Company is empowered to issue preferred stock in one or more series without shareholder action, which could render more difficult or discourage an attempt to obtain control of the Company by means of a tender offer, business combination, proxy contest or otherwise. In addition, certain provisions of the Texas Business Corporation Act also may

1 1

THE COMPANY

The Company was founded in October 1996 as the successor to Looney & Company ("Looney") and to create a leading provider of legal support and staffing services to law firms, insurance providers and major corporations. Since combining with Looney in January 1997, the Company has acquired 13 businesses (together with Looney, the "Completed Acquisitions") and will acquire four additional businesses concurrently with the Offering in the Pending Acquisitions. Prior to the combination with Looney in January 1997, the Company did not conduct any operations. Looney now conducts its operations as a subsidiary of the Company and provides court reporting and certified record retrieval services through operations in Houston, Dallas, San Antonio, Austin, Corpus Christi and Harlingen, Texas. In July 1997, Looney acquired Preferred Records, Inc., a Dallas court reporting and certified record retrieval company and in August 1997, acquired Encore Reporting, a Harlingen-based court reporting and record retrieval company. Looney had 1996 revenues of \$7.7 million. Mr. Richard O. Looney, the founder of Looney, serves as Chairman, President and Chief Executive Officer of the Company.

The Company also provides court reporting services to customers in locations not directly served by the Company through U.S. Court Reporters, Inc., a referral network with over 130 affiliated court reporting firms located throughout the United States. In addition, in April 1997, the Company acquired Cindi Rogers & Associates, a Houston-based court reporting company. In August 1997, the Company acquired Rocca Reporting Service ("Rocca"), a Chicago Illinois court reporting company.

The following is a brief description of each of the Completed Acquisitions and the Pending Acquisitions.

COMPLETED ACQUISITIONS

Klein, Bury & Associates. Klein, Bury & Associates ("Klein Bury") provides court reporting services through operations in Miami, Fort Lauderdale, West Palm Beach, Orlando, Jacksonville and Tampa, Florida and was previously a U.S. Court Reporters network affiliate. Michael Klein, President of Klein Bury, serves on the Board of Directors of the Company. Headquartered in Miami, Florida, Klein Bury had 1996 revenues of \$8.5 million.

G&G Court Reporters. G&G Court Reporters ("G&G") provides court reporting services in southern California. Headquartered in Los Angeles, G&G had 1996 revenues of \$1.5\$ million.

San Francisco Reporting Service. San Francisco Reporting Service ("San Francisco Reporting") provides court reporting services in the San Francisco and northern California markets and was previously a U.S. Court Reporters network affiliate. Headquartered in San Francisco, California, San Francisco Reporting had 1996 revenues of \$1.1 million.

Johnson Court Reporting Group. The Johnson Court Reporting Group ("Johnson Group"), through Johnson Court Reporting, Rapidtext, Inc. and Medtext, Inc., provides court reporting services, closed-captioning services for the hearing impaired, remote classroom captioning services, medical transcription services, and scanning and imaging services primarily in southern California. Headquartered in Los Angeles, California, Johnson Group had 1996 revenues of \$2.2 million.

Legal Enterprise, Inc. Legal Enterprise, Inc. ("Legal Enterprise") provides certified record retrieval services as well as digital scanning, reproduction services and automated subpoena preparation services through seven offices in California. Tony Maddocks, President of Legal Enterprise, serves as Vice President of Sales and Marketing for the Company. Headquartered in Los Angeles, California, Legal Enterprise had 1996 revenues of \$3.7 million.

Amicus One Legal Support Services, Inc. Amicus One Legal Support Services, Inc. ("Amicus One") provides court reporting, computer and still board animation exhibit preparation and image scanning services with operations in Manhattan, White Plains and Brooklyn, New York. Headquartered in New York, New York, Amicus One had 1996 revenues of \$1.9 million.

12

Block Court Reporting, Inc. Block Court Reporting, Inc. and its subsidiary ("Block") provide court reporting services to the Washington, D.C. and northern Virginia markets. Block was previously a U.S. Court Reporters network affiliate. Headquartered in Washington, D.C., Block had 1996 revenues of \$1.3 million.

Ziskind Greene Watanabe & Nason. Ziskind Greene Watanabe & Nason ("Ziskind Greene") provides permanent legal search services to national and California law firms and legal departments of major corporations through offices in Los Angeles, Orange County, San Diego and San Francisco, California. Headquartered in Los Angeles, California, Ziskind Greene had 1996 revenues of \$1.8 million.

Elaine P. Dine, Inc. Elaine P. Dine, Inc. ("Elaine Dine") provides permanent legal search services to national and New York law firms and legal departments of major corporations nationwide. In addition, Elaine Dine, through Elaine Dine Temporary Attorneys, L.L.C., provides temporary lawyer services to its clients. Headquartered in New York, New York, Elaine Dine had revenues of \$4.7 million for the twelve months ended March 31, 1997.

The consideration paid by the Company in the Completed Acquisitions after giving effect to post-closing adjustments to date consisted of: (i) \$21.4 million in cash; (ii) 2,046,667 shares of Series B Preferred Stock; (iii) 231,250 shares of Series C Preferred Stock; (iv) \$5.1 million aggregate principal amount of Subordinated Promissory Notes; (v) \$1.8 million aggregate principal amount of Convertible Subordinated Promissory Notes; and (vi) 703,244 shares of Common Stock. In addition, with respect to certain of the businesses acquired, the Company may be obligated to pay contingent consideration based on improvements in the financial performance of the businesses during specified periods after their acquisition. The Company also granted options to purchase a total of 144,336 shares of Common Stock to the owners or employees of the businesses acquired. These options expire five years after their respective dates of grant and may be exercised for nominal consideration. See "Use of Proceeds" and Notes 7 and 8 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc.

PENDING ACQUISITIONS

Jilio & Associates. Jilio & Associates ("Jilio") provides court reporting services in the Southern California market. Headquartered in Los Angeles, California, Jilio had 1996 revenues of \$4.0\$ million.

Kirby A. Kennedy & Associates. Kirby A. Kennedy & Associates ("Kirby Kennedy") provides court reporting services in the Minneapolis and St. Paul, Minnesota markets and is a U.S. Court Reporters network affiliate. Headquartered in Minneapolis, Minnesota, Kirby Kennedy had 1996 revenues of \$1.9 million.

Reporting Service Associates, Inc. Reporting Service Associates, Inc. ("Reporting Service") provides court reporting services in the mid-Atlantic markets. Headquartered in Philadelphia, Pennsylvania, Reporting Service had 1996 revenues of \$3.0 million.

Commander Wilson, Inc. Commander Wilson, Inc. ("Commander Wilson") provides permanent legal search services to national and Texas law firms and legal departments of major corporations. James Wilson, the owner of Commander Wilson, became Vice President, Legal Staffing of the Company on September 25, 1997. Headquartered in Houston, Texas, Commander Wilson had 1996 revenues of \$94,000. See Note 6 of Notes to Financial Statements of Commander Wilson for a discussion of bankruptcy proceedings involving Commander Wilson. See "Management" and "Certain Transactions."

The aggregate purchase price to be paid by the Company in connection with the Pending Acquisitions, subject to post-closing adjustments, if any, is approximately \$16.9 million in cash and 609,268 shares of Common Stock.

13

USE OF PROCEEDS

The net proceeds to the Company from the sale of shares of Common Stock offered hereby are estimated to be approximately \$36.6 million (\$41.3 million if the Underwriters' over-allotment option is exercised in full), assuming an initial offering price of \$12.00 per share and after deducting estimated underwriting discounts and estimated expenses payable by the Company.

The net proceeds of the Offering will be utilized by the Company as follows: (i) approximately \$17.1 million will be used to repay the outstanding principal and interest under the Company's existing credit agreement with a commercial bank, which was used to finance the Completed Acquisitions and for working capital purposes (the "Bank Credit Agreement"); (ii) approximately \$9.5 million will be used to repay the outstanding principal of, and accrued interest on, the Company's 12% Senior Subordinated Notes due 2005 (the "Senior Subordinated Notes"), which were issued in January 1997 to obtain funds to finance the Completed Acquisitions; (iii) \$231,250 will be used to redeem 231,250 shares of Series C Preferred Stock issued in connection with the acquisition of San Francisco Reporting; (iv) approximately \$5.0 million will be used to repay the outstanding principal balance of, and interest on, the Company's Subordinated Promissory Notes issued in connection with certain of

the Completed Acquisitions; and (v) the remaining net proceeds of approximately \$4.8 million will be applied to the \$16.9 million required to pay the cash portion of the purchase price in the Pending Acquisitions. The additional \$12.1 million required for the cash portion of the Pending Acquisitions is expected to be financed through borrowings under a new credit facility. See "Management's Discussion and Analysis of Financial Condition and Results of Operations--Liquidity and Capital Resources."

Indebtedness under the Bank Credit Agreement currently bears interest at the rate of 8.50% per annum and is required to be accelerated upon completion of the Offering. The Subordinated Promissory Notes become due and payable upon completion of the Offering and bear interest at rates ranging from 6.0% to 10.0% per annum. Approximately \$1.6 million of the principal amount of the Subordinated Promissory Notes is held by two executive officers of the Company. See "Capitalization," "Management's Discussion and Analysis of Financial Condition and Results of Operations--Liquidity and Capital Resources," "Certain Transactions" and Notes 3 and 7 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc.

DIVIDEND POLICY

The Company has not paid a cash dividend on the Common Stock since its incorporation and does not anticipate paying any dividends on Common Stock in the foreseeable future. The Company intends to retain future earnings, if any, to finance its operations and to fund the growth of its business, to repay indebtedness and for general corporate purposes. Any payment of dividends will be at the discretion of the Board of Directors and will depend upon, among other things, the Company's earnings, financial condition, capital requirements, level of indebtedness, contractual restrictions with respect to the payment of dividends and other relevant factors. The Bank Credit Agreement prohibits the payment of dividends. The Company has entered into a new credit facility with a commercial bank, which will be effective concurrently with completion of the Offering, and such agreement contains provisions restricting the payment of dividends. See "Description of Capital Stock."

14

CAPITALIZATION

The following table sets forth, as of September 30, 1997: (i) the historical consolidated capitalization of the Company and (ii) the historical consolidated capitalization of the Company, as adjusted to reflect the sale of the Common Stock in the Offering, the application of the estimated net proceeds therefrom and the Pending Acquisitions. See "Use of Proceeds." This presentation should be read in conjunction with the historical and pro forma consolidated financial statements and related notes thereto included elsewhere herein.

<TABLE>

	AS OF SEPTEMBER 30, 1997		
	ACTUAL	PRO FORMA	
<\$>	(IN THO	,	
Short-term debt (including current portion of long-term debt)	,	\$75 =====	
Long-term debt, excluding current portion: Bank indebtedness (1)	\$12,424 9,000 4,694 1,845 (1,707) 83 26,339	\$11,000 83 	
Redeemable Preferred Stock: Series A Convertible Preferred Stock, 1,000,000 shares outstanding (4)	2,047		
outstanding (5)	231	 78	
· · ·			

Additional paid-in capital		
Total shareholders' equity (deficit)	(2,321)	43,940
Total capitalization	\$27,775	\$55,023
	======	======

</TABLE>

- (1) Actual long-term bank indebtedness does not include borrowings of approximately \$1.1 million incurred after September 30, 1997. Therefore, pro forma long-term bank indebtedness, which represents borrowings to be made under the Company's new credit facility to fund a part of the cash portion of the purchase price of the Pending Acquisitions, are estimated to be approximately \$12.1 million.
- (2) The Convertible Subordinated Promissory Notes were issued in connection with certain of the Completed Acquisitions, and will be converted into an aggregate of 225,890 shares of Common Stock.
- (3) Includes \$1,142,000 attributable to the Subordinated Promissory Notes and \$565,000 attributable to the Subordinated Convertible Promissory Notes. See Note 7 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc.
- (4) The Series A Convertible Preferred Stock will be converted into 1,560,000 shares of Common Stock and the Series B Convertible Preferred Stock will be converted into an aggregate of 183,393 shares of Common Stock, in each case, upon completion of the Offering.
- (5) The Series C Convertible Preferred Stock will be redeemed for \$231,250 out of the net proceeds of the Offering.
- (6) Excludes: (i) 900,000 shares of Common Stock reserved for issuance under the Company's 1997 Stock Incentive Plan and the Company's Stock Option Plan for Non-Employee Directors and (ii) 131,856 shares of Common Stock issuable upon exercise of options granted in connection with Completed Acquisitions. See "Management," "Certain Transactions" and "Principal Shareholders."

15

DILUTION

The net tangible book deficit of the Company at September 30, 1997 was approximately \$30.3 million, or \$(17.47) per share of Common Stock. Net tangible book deficit per share represents the Company's total tangible assets less its total liabilities, divided by 1,734,564 shares of Common Stock outstanding as of September 30, 1997. After giving effect to the sale of the 3,500,000 shares of Common Stock offered hereby (after deducting the underwriting discount and estimated Offering expenses) and consummation of the Pending Acquisitions, the pro forma net tangible book deficit of the Company at September 30, 1997 would have been approximately \$7.3 million, or \$(0.93) per share. This represents an immediate increase in such net tangible book value of \$16.54 per share to existing shareholders at September 30, 1997, consisting of \$10.54 per share attributable to the conversion of convertible subordinated debt and preferred stock, \$8.11 per share attributable to the Offering, offset by a decrease of \$3.22 per share attributable to the Pending Acquisitions. This results in immediate dilution of \$12.93 per share to investors purchasing shares in the Offering. The following table illustrates pro forma dilution to new investors:

<TABLE>

<\$>	<c></c>	<c></c>
Assumed initial public offering price per share Net tangible book deficit as of September 30, 1997		\$12.00
[\$30.3 million]	\$(17.47)	
able to the conversion of convertible subordinated debt and preferred stock [\$4.8 million]	10.59	
Increase in pro forma net tangible book value attributable to the Offering, net of the extraordinary loss on	10.33	
debt [\$35.1 million]	9.26	
Decrease in pro forma net tangible book value per share attributable to the Pending Acquisitions [\$16.7 mil-		
lion]	(3.31)	
Pro forma net tangible book value per share after the Of-		
fering [\$7.3 million]		(0.93)
Dilution per share to new investors		\$(12.93) ======

</TABLE>

The following table presents, after giving effect to the Offering and the Pending Acquisitions, the difference among existing shareholders, sellers

receiving shares in the Pending Acquisitions and new investors with respect to the number of shares purchased from the Company and the total consideration and average price per share paid to the Company, before deducting the underwriting discounts and estimated Offering expenses.

<TABLE> <CAPTION>

	SHARES PU	RCHASED	TOTAL CONSI	PURCHASE PRICE PER	
	NUMBER	PERCENT	AMOUNT	PERCENT	SHARE
<\$>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Existing common shareholders	1,734,564	22%	\$4,246,000	7%	\$2.45
Common shareholders upon conversion of:					
Series A Preferred Stock	1,560,000	20	1,000,000	2	0.64
Series B Preferred Stock Convertible Subordinated	183,393	2	2,046,667	4	11.16
Promissory Notes Sellers receiving shares in	225,890	3	1,844,955	3	8.17
the Pending Acquisitions	609,268	8	6,580,000	11	10.80
New investors	3,500,000	45	42,000,000	73	12.00
Total	7,813,115	100.0%	\$57,717,622	100.0%	
	=======	=====			

 | | | | |The foregoing tables assume no exercise of outstanding options. As of the date of this Prospectus, there are 795,880 shares of Common Stock issuable upon the exercise of stock options at a weighted average exercise price of \$9.16 per share. See "Management."

16

SELECTED FINANCIAL DATA

The financial data set forth below as of and for each of the years in the three year period ended December 31, 1996 were derived from audited financial statements of Looney. The financial data for the years ended December 31, 1992 and 1993 and as of and for the nine months ended September 30, 1996 and 1997 were derived from the unaudited financial statements of Looney and the Company, respectively, which include all adjustments (consisting only of normal recurring adjustments) that, in the opinion of management, are necessary to present fairly the information set forth therein.

The pro forma financial data as of and for the nine months ended September 30, 1996 and 1997 and the year ended December 31, 1996, were derived from the pro forma combined financial statements of the Company appearing elsewhere in this Prospectus. Such pro forma combined financial statements give effect to the completed acquisitions and the Pending Acquisitions, the Offering and the application of the proceeds therefrom, and the related conversion of outstanding securities as if each of these events had occurred on January 1, 1996. The pro forma financial information of the Company does not purport to represent what the Company's results of operations or financial position actually would have been had the completed acquisitions and the Pending Acquisitions occurred on the dates specified, nor is it a projection of the Company's results of operations or financial position for any future period or date. The data presented below should be read in conjunction with the Management's Discussion and Analysis of Financial Condition and Results of Operations of the Company, the financial statements and pro forma combined financial statements and the notes thereto included elsewhere herein.

17

SELECTED FINANCIAL DATA (1)

(IN THOUSANDS, EXCEPT PER SHARE DATA)

<TABLE> <CAPTION>

	YI	EAR ENDE	D DECEMB	ER 31,		NINE MONTHS ENDED YEAR ENDED SEPTEMBER 30, DECEMBER 31,			EN	MONTHS DED BER 30,
	1992	1993	1994	1995	1996	1996	1997 (2)	1996 (3)(4)	1996 (3)(4)	1997 (3)(4)
<pre><s> STATEMENT OF INCOME DATA:</s></pre>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$6,706	\$7,722	\$8,363	\$9,104	\$7,667	\$5,886	\$14,549	\$43,405	\$32,024	\$37 , 277

Cost of services	3,802	4,595	5,589	5,763	4,839	3,726	9,287	25,474	18 , 575	21,261
Gross profit Selling, general and administrative	2,904	3,127	2,774	3,341	2,828	2,160	5 , 262	17,931		16,016
expenses (5) Depreciation and	2,612	2,824	3,043	1,970	2,352	1,310	3,855	10,968	8,420	9,365
amortization (4)	95	110	224	231	212	159	332	1,754	1,332	1,337
Operating income (loss)	197 101	193 181	(493) 185	1,140 230	264 238	691 178	•	5 , 209 880	3,697 668	5,314 707
Income (loss) before taxes	96	12	(678)	910	26	513	(72)	4,329	3,029	4,607
Provisions (benefit) for income taxes		35	(183)	327	10	174	11	1,847	1,298	1,928
Net income (loss)	\$96	\$(23)		\$583	\$16	\$339	(83)	\$2,482	\$1,731	\$2,679 ======
Accretion of preferred stock							(479)			
Net income (loss) attributable to common shareholders							\$(562) ======			
Net income (loss) per common share (6)							\$(.13)	\$0.31 =====	\$0.22 ======	\$0.33 =====
Weighted average shares outstanding (7) (8)							4,226	8,002 =====	8,002 =====	8,002 =====

</TABLE>

<TABLE>

SEPTEMBER 30, 1997

BALANCE SHEET DATA: Cash			
BALANCE SHEET DATA: Cash		ACTUAL	PRO FORMA (3) (7)
Cash. \$ 559 \$ 1,238 Total assets. 38,720 62,652 Short-term debt. 3,936 75 Long-term debt (net of current portion) 26,339 11,083 Redeemable preferred stock. 3,757 Total shareholders' equity (deficit) \$(2,321) \$43,940	<s> BALANCE SHEET DATA.</s>	<c></c>	<c></c>
Long-term debt (net of current portion) 26,339 11,083 Redeemable preferred stock 3,757 Total shareholders' equity (deficit) \$(2,321) \$43,940	Cash		, ,
Redeemable preferred stock		.,	
	Redeemable preferred stock	3,757	,
	Total shareholders' equity (deficit)		

 \$(2,321) | \$43,940 |

- (1) Prior to January 17, 1997, the Company had no business operations. Therefore, the business of Looney & Company, for financial statement purposes, represents the predecessor business.
- (2) The Completed Acquisitions have been, and the Pending Acquisitions will be, accounted for as purchases and therefore, the operations of the acquired businesses are included in the statement of income data from the respective dates of acquisition. See the Consolidated Financial Statements of U.S. Legal Support, Inc. included herein.
- (3) Pro forma information gives effect to: (i) the Completed Acquisitions and completion of Pending Acquisitions; (ii) an adjustment to compensation expense for the difference between actual compensation paid to certain officers of businesses acquired or to be acquired and employment contract compensation negotiated in connection with the completed acquisitions and the Pending Acquisitions; (iii) amortization expense relating to intangible assets recorded in conjunction with the Completed Acquisitions and to be recorded in the Pending Acquisitions; and (iv) the sale of the shares offered hereby and the application of the net proceeds thereof, as if such events had occurred on January 1, 1996 (for statement of income data) and as of September 30, 1997 (for balance sheet data). The Pending Acquisitions and the conversion of certain outstanding preferred stock and Convertible Subordinated Promissory Notes described in Note 6 below, are contingent upon and will close concurrently with completion of the Offering. The pro forma results of operations are not necessarily indicative of the results that would have occurred had these transactions been completed as of such date or the results that may be attained in the future.
- (4) Pro forma depreciation and amortization amounts consist primarily of amortization of goodwill totaling \$1,313,000 and \$985,000 for the periods ended December 31, 1996 and September 30, 1997, respectively, recorded or to be recorded as a result of the Completed Acquisitions and the Pending Acquisitions. Goodwill is amortized over periods ranging from ten to 40

years and computed on the basis described in Note 2 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc.

- (5) Includes a non-recurring charge of \$360,000 in the fourth quarter of 1996 representing the estimated fair value of ownership interests granted to certain employees by Looney's shareholder.
- (6) Supplementary historical earnings per common share for the nine months ended September 30, 1997, would be \$.09 per share, giving effect to: (i) the 3,893,000 weighted average shares outstanding for the period; (ii) the issuance of 2,908,000 of shares in the Offering, the proceeds from which would be necessary to (a) repay \$16,000,000 of bank indebtedness under the Company's credit facility, (b) redeem \$9,000,000 of Senior Subordinated Notes, (c) repay \$4,979,000 of Subordinated Promissory Notes, (d) redeem 231,250 shares of Series C Preferred Stock, and (e) pay the applicable Offering expenses; and (iii) the reduction of interest expense and dividends.
- (7) Gives effect to: (i) 1,734,564 shares outstanding prior to the Offering; (ii) 3,500,000 shares to be issued in the Offering; (iii) 609,268 shares to be issued in the Pending Acquisitions; (iv) 1,969,283 shares issued upon conversion of preferred stock and Convertible Subordinated Promissory Notes; and (v) 188,187 shares issuable upon exercise of outstanding stock options in accordance with SEC Staff Accounting Bulletin Topic 4D. See "Capitalization."
- (8) Shares used in calculating net loss per share for the nine months ended September 30, 1997, include the number of shares, the proceeds from the sale of which would be necessary to repay the portion of the Company's debt that funded the distribution to Richard O. Looney in connection with the reorganization of the Company and Looney & Company.

18

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion should be read in conjunction with the historical financial statements and related notes and the pro forma financial statements and related notes included elsewhere in this Prospectus.

OVERVIEW

The Company is one of the largest providers of legal support and staffing services in the United States providing court reporting, certified record retrieval, legal placement and staffing, and other related services to law firms and corporations, including insurance companies, through 40 offices in seven states and the District of Columbia. The Company seeks to become the leading national, full-service provider of legal support and staffing services in the United States through a combination of selective acquisitions and internal growth. After succeeding to the operations of Looney in January 1997, the Company has acquired 13 businesses and will acquire four additional businesses concurrently with the Offering. For the year ended December 31, 1996, the Company had pro forma revenues of \$43.4 million and pro forma operating income of \$5.2 million. For the nine months ended September 30, 1997, the Company had pro forma revenues of \$37.3 million and pro forma operating income of \$5.3 million, compared with pro forma revenues of \$32.0 million and pro forma operating income of \$3.7 million for the nine months ended September 30, 1996.

In 1996, the Company derived approximately 64.0% of its pro forma revenues from court reporting services, 18.0% from certified record retrieval services, 15.0% from permanent placement and temporary legal staffing and 3.0% from other services. For the nine months ended September 30, 1997, the Company derived 60.0% of its pro forma revenues from court reporting, 17.0% from certified record retrieval, 21.0% from permanent placement and temporary legal staffing and 2.0% from other services. The Company believes that over the past several years prices for court reporting and certified record retrieval services have remained stable or have declined in many markets, while rates for permanent placement and temporary legal staffing have remained relatively constant. The Company also believes that law firms and corporations have increased their efforts to reduce legal expenses by centralizing their purchasing decisions and negotiating lower rates with vendors. As these trends and the consolidation among legal support companies continue, the Company believes that the industry will experience more price-based competition. The Company's strategy, however, is to capitalize on the centralization of purchasing decisions by increasing its share of the market represented by larger accounts and expanding the geographic markets in which it operates.

Charges for court reporting services typically are based upon the number of pages transcribed, with a significant portion of revenues being derived from the production of additional copies. Substantially all of the Company's court reporting services are performed by independent contractors. Under its

standard arrangements with independent court reporters, the Company retains a portion, typically averaging 40.0%, of the total court reporting fee and the independent court reporter receives the balance. The Company also derives court reporting revenues from its network of over 130 affiliated court reporting firms in locations not directly served by the Company. Under these arrangements, the Company refers court reporting assignments to network participants and, upon completion of the assignment, bills the client directly and retains a referral fee ranging from 5.0% to 10.0% of the total court reporting fees. The Company anticipates that its reliance on the network of court reporting affiliates will diminish as the Company acquires additional court reporting businesses, including certain of the network affiliates located in market areas not currently served by the Company.

Charges for certified record retrieval services are based upon the number of subpoenas or other notices initiated and the volume of documents generated in response to these subpoenas and notices. Certified record retrieval services generally are used in personal injury and medical malpractice litigation. Because the number of transcript copies requested by the parties and the volume of certified records retrieved generally increase as the number of parties in a lawsuit increases, revenues derived from court reporting and certified record retrieval services are significantly higher in litigation involving more than two parties.

19

The Company's legal staffing recruiters are compensated on a commission basis. Fees for successful placement of an attorney typically are based upon a percentage, approximately 25.0% to 30.0% of the attorney's compensation earned during the year following the placement, of which 40.0% to 50.0% is customarily paid to the individual recruiter. This fee is subject to a partial refund if the new employment arrangement is terminated prior to the expiration of a negotiated period, usually three to six months. The Company charges its clients an hourly fee for the number of hours worked by attorneys placed with clients on a temporary basis.

Cost of services consists primarily of amounts due to the Company's independent contractors, payroll for field agents, commissions for recruiters, production equipment rental and costs associated with delivery of documents. Selling, general and administrative expenses include payroll for management and administrative employees, office occupancy costs, sales and marketing expenses and other general and administrative costs. Depreciation and amortization relates primarily to depreciation of office furniture and equipment and amortization of intangible assets. The Company pays fees to its independent contractors based on a percentage of the fees billed to clients. Similarly, the Company compensates its temporary staffing attorneys only for the hours actually worked. Consequently, the compensation for such personnel is a variable cost that fluctuates in proportion to revenues.

The Completed Acquisitions have been, and the Pending Acquisitions will be, accounted for under the purchase method of accounting, with the results of operations of businesses acquired being included in the Company's results of operations beginning on the date of acquisition. Upon completion of the Offering, the Company will have recorded approximately \$51.0 million as goodwill, representing the excess of the fair value of the consideration paid over the fair value of the assets to be acquired. Approximately 77.0% of the goodwill is deductible for federal income tax purposes over 15 years. For financial reporting purposes, goodwill is amortized over periods ranging from ten to 40 years. The pro forma effect of this amortization expense is expected to be approximately \$1.3 million annually. Certain owners and employees of the businesses acquired and to be acquired in the Pending Acquisitions have agreed to reductions totaling \$1.3 million from the levels of their 1996 compensation. These reductions have been reflected in the pro forma financial statements included herein.

COMBINED AND PRO FORMA OPERATING DATA

The combined operating data of the Company for the periods presented do not represent combined results of operations prepared in accordance with generally accepted accounting principles, but are only a summation of the revenues, cost of services and gross profit of the individual businesses on a historical basis. The combined results of operations assume that each of the businesses was combined from January 1, 1995.

The pro forma operating data for the year ended December 31, 1996 and the nine months ended September 30, 1997 assumes that the Completed Acquisitions as well as the Pending Acquisitions, were consummated as of January 1, 1996. Pro forma adjustments include a reduction in compensation expense to reflect amounts to be paid under the terms of employment agreements entered into in connection with certain of the acquisitions. Pro forma adjustments also have been made to reflect the amortization of the goodwill recorded in connection with each acquisition.

The following table sets forth certain combined and pro forma operating data for the indicated periods:

<TABLE>

	YEAR ENDED DECEMBER 31,					30		
	1995				1996			
					PRO FORMA			
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Revenue	\$38,616	100.0%	\$43,405	100.0%	\$32,024	100.0%	\$37,277	100.0%
Cost of services					18,575			
Gross profit	\$15,416 ======		\$17,931	41.3	\$13,449	42.0	\$16,016	43.0
Selling, general and administrative								
expenses Depreciation and			10,968	25.3	8,420	26.3	9,365	25.1
amortization			1,754		1,332		•	3.6
Operating income			\$5,209		\$3,697			14.3%

NITHE MONEY CENTER CENTER

</TABLE>

RESULTS OF OPERATIONS--PRO FORMA 1996 AND COMBINED 1995

Nine Months Ended September 30, 1997 Compared to Nine Months Ended September 30, 1996

Revenue. Revenue increased \$5.3 million, or 16.4%, to \$37.3 million for the nine months ended September 30, 1997 from \$32.0 million for the nine months ended September 30, 1996. The increase in revenue was a result of the following: (i) revenue from court reporting services increased by approximately \$1.6 million, primarily due to higher demand in Pennsylvania, California and New York; (ii) the Company's permanent legal search services revenue increased by approximately \$2.3 million as the result of several significant retained searches; and (iii) revenue from certified record retrieval services in California increased by \$754,000 as the result of several new insurance clients. In addition the start-up of temporary attorney staffing services contributed approximately \$1.0 million to the revenue increase. These increases were partially offset by lower revenue from certified record retrieval services in Texas.

Gross Profit. Gross profit increased \$2.6 million, or 19.1%, to \$16.0 million for the nine months ended September 30, 1997 from \$13.4 million for the nine months ended September 30, 1996. The gross profit on court reporting services increased by approximately \$800,000, while the gross profit on permanent legal placement and temporary staffing services increased by approximately \$1.7 million. The gross margin percentage improved to 43.0% during the nine months ended September 30, 1997 from 42.0% during the nine months ended September 30, 1996. The improvement in the gross margin percentage was primarily the result of an increase in the gross margin percentage on court reporting services resulting from a reduction in the percentage of court reporting fees paid to court reporters at one of the Company's locations, offset by a slight decrease in the gross margin percentage on permanent legal search and temporary staffing services as the result of the start-up of temporary attorney staffing services.

Year Ended December 31, 1996 Compared to Year Ended December 31, 1995

Revenue. Revenue increased \$4.8 million, or 12.4%, to \$43.4 million in 1996 from a combined \$38.6 million in 1995. The increase in revenue was a result of the following: (i) revenue from court reporting services increased by approximately \$3.2 million primarily as the result of higher demand in Florida, Pennsylvania and California; (ii) the Company's permanent legal search services revenue increased by approximately \$1.1 million as the result of several significant retained searches; and (iii) revenue from certified record retrieval services in California increased by \$950,000 as the result of several new insurance clients. The increase in revenue was partially offset by lower court reporting and certified record retrieval revenue in Texas resulting from the settlement of a number of large lawsuits in which the Company had been providing services.

Gross Profit. Gross profit increased \$2.5 million, or 16.3%, to \$17.9 million in 1996 from \$15.4 million in 1995. The gross profit on court reporting services increased by approximately \$2.0 million while gross profit on permanent legal placement services increased by approximately \$400,000 due to the higher level of revenue and improved gross margin percentages. The gross margin percentage improved to 41.3% in 1996 from 39.9%

in 1995. The improvement in the gross margin percentage was primarily the result of an increase in the gross margin percentage from court reporting services resulting from a reduction in the percentage of court reporting fees paid to court reporters at one of the Company's locations. Also, the gross margin percentage on certified record retrieval services in California improved as the result of increased volume from new insurance clients.

RESULTS OF OPERATIONS -- THE COMPANY AND LOONEY

The Company provides court reporting, certified record retrieval, legal placement and staffing and related services and derives its revenues from fees associated with these services. Looney provides court reporting and certified records retrieval services. The Company acquired Looney in January 1997.

Nine Months Ended September 30, 1997 Compared to Nine Months Ended September 30, 1996

The Company acquired Klein Bury and eight other businesses during the nine months ended September 30, 1997. Consequently, certain aspects of the Company's results of operations in that period are not comparable to the results of Looney for the nine months ended September 30, 1996.

Revenue. Revenue increased \$8.7 million, or 147.2%, to \$14.5 million for the nine months ended September 30, 1997 from \$5.9 million for the nine months ended September 30, 1996. The increase in revenue was primarily the result of the acquisitions made by the Company during the nine months ended September 30, 1997 which contributed \$8.9 million of additional revenue.

Gross Profit. Gross profit increased \$3.1 million, or 143.6%, to \$5.3 million for the nine months ended September 30, 1997 from \$2.2 million for the nine months ended September 30, 1996. The gross margin percentage decreased by 0.5% to 36.2% for the nine months ended September 30, 1996 are 1997 from 36.7% for the nine months ended September 30, 1996. The slight decrease in the gross margin percentage was primarily the result of higher fees paid to court reporters at the Company's Florida subsidiary as compared to fees paid by Looney.

Selling, General and Administrative Expenses. Selling, general and administrative expenses increased \$2.5 million, or 194.3%, to \$3.9 million for the nine months ended September 30, 1997 from \$1.3 million for the nine months ended September 30, 1996. Approximately \$1.8 million was attributable to acquisitions and \$746,000 was the result of higher expenses related to the addition of corporate personnel to facilitate anticipated future growth. Selling, general and administrative expenses as a percentage of revenue was 26.5% for the nine months ended September 30, 1997 as compared to 22.3% for the nine months ended September 30, 1996 as the result of the additional corporate office personnel.

Depreciation and Amortization. Depreciation and amortization increased by \$173,000, or 108.8%, to \$332,000 for the nine months ended September 30, 1997 from \$159,000 for the nine months ended September 30, 1996. The increase was due to the amortization of goodwill incurred in connection with the 14 acquisitions made during the nine months ended September 30, 1997.

Operating Income. As the result of the foregoing, operating income increased by \$384,000, or 55.5%, to \$1.1 million for the nine months ended September 30, 1997 from \$691,000 for the comparable period in 1996. The operating margin percentage decreased 4.3% to 7.4% for the nine months ended September 30, 1997 from 11.7% for the nine months ended September 30, 1996.

Year Ended December 31, 1996 Compared To Year Ended December 31, 1995

Revenue. Revenue decreased \$1.4 million, or 15.8%, to \$7.7 million in 1996 from \$9.1 million in 1995. The decrease in revenue was primarily a result of the bankruptcy of a major client and the settlement of a number of significant lawsuits in late 1995 and early 1996.

Gross Profit. Gross profit decreased \$512,000, or 15.3%, to \$2.8 million in 1996 from \$3.3 million in 1995. The gross margin percentage remained relatively constant at 36.9% in 1996 compared with 36.7% in 1995.

22

Selling, General and Administrative Expenses. Selling, general and administrative expenses increased \$381,000, or 19.4%, to \$2.4 million in 1996 from \$2.0 million in 1995. The increase was due to a non-recurring charge of \$360,000 related to equity interests in the Company that was granted to certain employees. Excluding this non-recurring charge, selling, general and administrative costs remained relatively constant from 1995 to 1996. Selling, general and administrative expenses as a percentage of revenue increased 9.1%

to 30.7% at December 31, 1996 from 21.6% for December 31, 1995.

Depreciation and Amortization. Depreciation and amortization decreased \$18,000, or 7.8%, to \$212,000 in 1996 from \$231,000 in 1995 due to certain assets becoming fully depreciated.

Operating Income. As a result of the foregoing, operating income decreased by \$875,000, or 76.8%, to \$264,000 in 1996 from \$1.1 million in 1995. The operating margin percentage decreased 9.0% to 3.5% in 1996 as compared to 12.5% in 1995.

Year Ended December 31, 1995 Compared To Year Ended December 31, 1994

Revenue. Revenue increased by \$741,000, or 8.9%, to \$9.1 million in 1995 from \$8.4 million in 1994 as the result of higher business activity attributable to two major lawsuits for which the Company was providing both court reporting and certified record retrieval services.

Gross Profit. Gross profit increased \$567,000, or 20.5%, to \$3.3 million in 1995 from \$2.8 million in 1994. The gross margin percentage increased 3.5% to 36.7% in 1995 from 33.2% in 1994 as a result of improved margins related to major multi-party lawsuits.

Selling, General and Administrative Expenses. Selling, general and administrative expenses decreased by \$1.1 million, or 35.3%, to \$2.0 million in 1995 from \$3.0 million in 1994. The decrease was the result of reduced expenses resulting from cost containment actions in 1995. In addition, in 1994 the Company recorded non-recurring charges totalling \$611,000 related to the settlement of certain employee-related litigation and elimination of certain document copying services. Selling, general and administrative expenses as a percentage of revenue decreased 14.8% to 21.6% in 1995 from 36.4% in 1994 due to the factors discussed above.

Depreciation and Amortization. Depreciation and amortization increased \$7,000, or 3.0%, to \$231,000 in 1995 from \$224,000 in 1994.

Operating Income. As a result of the foregoing, operating income increased \$1.6 million to \$1.1 million in 1995 from an operating loss of \$493,000 in 1994

RESULTS OF OPERATIONS -- KLEIN BURY

Klein Bury provides court reporting services and derives its revenue from court reporting fees. The Company acquired Klein Bury in January 1997.

Twelve Month Period ended December 31, 1996 Compared To the Twelve Month Period Ended September 30, 1995

Revenue. Revenue increased \$1.2 million, or 16.7%, to \$8.5 million in 1996 from \$7.3 million, for the fiscal year ended September 30, 1995. The increase in revenue was a result of higher demand for Klein Bury's court reporting services attributable to Klein Bury obtaining additional clients in the insurance and health care industry, the expansion of services statewide for certain existing clients and the addition of court reporting services for municipal courts in Dade County, Florida which were previously performed by municipal employees. In addition, Klein Bury derived additional revenue from a new office which opened in 1995.

23

Gross Profit. Gross profit increased \$475,000, or 19.3%, to \$2.9 million in 1996 from \$2.5 million in 1995. The gross margin percentage increased slightly to 34.5% in 1996 from 33.7% at September 30, 1995. The improvement in gross margin percentage was the result of slightly lower independent contractor costs in 1996.

Operating Expenses. Operating expenses increased \$515,000, or 22.6%, to \$2.8 million in 1996 from \$2.3 million for the fiscal year ended September 30, 1995. The increase in operating expenses was the result of increased revenue and increased compensation to executive officers. Operating expenses as a percentage of revenue increased to 32.7% in 1996 from 31.2% in fiscal year ended September 30, 1995.

Operating Income. As a result of the foregoing, operating income decreased \$41,000, or 22.1%, to \$144,000 in 1996 from \$185,000 for the fiscal year ended September 30, 1995. Operating margin percentage decreased to 1.7% in 1996 from 2.5% in 1995.

RESULTS OF OPERATIONS--REPORTING SERVICE

Reporting Service provides court reporting services and derives its revenue from court reporting fees. Reporting Service will be acquired concurrently with the Offering.

Nine Months Ended September 30, 1997 Compared to Nine Months Ended September 30, 1996

Revenue. Revenue for the nine months ended September 30 1997 increased \$896,000, or 41.3\$, to \$3.1 million from \$2.2 million for the nine months ended September 30, 1996. The increase in revenue was the result of: (i) increased demand from existing clients; (ii) more multi-party lawsuits which result in higher demand for services as well as an increased number of court reporting transcripts; and (iii) improved pricing of Reporting Service's court reporting services.

Gross Profit. Gross profit for the nine months ended September 30, 1997 increased \$718,000, or 72.4%, to \$1.7 million from \$992,000 for the nine months ended September 30, 1996. The gross margin percentage improved by 10.1% to 55.8% for the nine months ended September 30, 1997 compared with 45.7% at September 30, 1996. The improvement in gross margin percentage primarily resulted from (i) increased volume of copies required by parties in multiparty lawsuits and (ii) cost containment actions taken by Reporting Service and improved pricing of its services.

Operating Expenses. Operating expenses for the nine months ended September 30, 1997 increased \$114,000, or 29.1%, to \$505,000 from \$391,000 for the nine months ended September 30, 1996. The increase in operating expenses resulted from costs attributable to additional administrative personnel and other expenses attributable to the increased demand for Reporting Service's court reporting services. Operating expenses as a percentage of sales decreased 1.5% to 16.5% at September 30, 1997 from 18.0% at September 30, 1996, due to the impact of higher volume without a corresponding increase in fixed overhead costs.

Operating Income. As the result of the foregoing, operating income for the nine months ended September 30, 1997 increased \$603,000 or 101.4%, to \$1.2 million from \$595,000 for the nine months ended September 30, 1996. The operating margin percentage improved 11.7% to 39.1% for the nine months ended September 30, 1997 from 27.4% for the nine months ended September 30, 1996.

Year Ended December 31, 1996 Compared to Year Ended December 31, 1995

Revenue. Revenue increased \$1.1 million, or 58.0%, to \$3.0 million in 1996 from \$1.9 million in 1995. The increase was due to increased demand from existing clients, more multi-party lawsuits and improved pricing.

Gross Profit. Gross profit increased \$581,000, or 77.5%, to \$1.3 million in 1996 from \$749,000 million in 1995. The gross margin percentage increased 4.9% to 44.2% in 1996 from 39.3% in 1995. The improvement in the gross margin percentage was due to cost containment actions and improved pricing.

24

Operating Expenses. Operating expenses increased \$162,000, or 38.2%, to \$586,000 in 1996 from \$424,000 in 1995 as a result of additional administrative personnel and increased demand for Reporting Service's court reporting services. Operating expenses as a percentage of revenue decreased 2.7% to 19.5% at December 31, 1996 from 22.2% at December 31, 1995.

Operating Income. As a result of the foregoing, operating income increased \$419,000, or 128.8%, in 1996, to \$744,000 from \$325,000 in 1995. Operating margin percentage improved 7.6% to 24.7% in 1996 from 17.1% in 1995.

RESULTS OF OPERATIONS--JILIO

Jilio provides court reporting services and derives its revenue from court reporting fees. Jilio will be acquired concurrently with the Offering.

Nine Months Ended September 30, 1997 Compared to Nine Months Ended September 30, 1996

Revenue. Revenue for the nine months ended September 30 1997 increased \$313,000, or 11.3\$, to \$3.1 million from \$2.8 million for the nine months ended September 30, 1996. The increase in revenue was due to an expansion of Jilio's client base as a result of marketing efforts, the impact of large multi-party litigation and the offering of new services, including document depository services, to its clients.

Gross Profit. Gross profit for the nine months ended September 30, 1997 increased \$108,000, or 7.7%, to \$1.5 million from \$1.4 million for the nine months ended September 30, 1996. The gross margin percentage declined 1.6% to 49.1% for the nine months ended September 30, 1997 from 50.7% for the nine months ended September 30, 1996. The decrease in the gross margin percentage was primarily the result of volume pricing discounts related to multi-party lawsuits and start-up costs related to new document depository services.

Operating Expenses. Operating expenses increased \$101,000, or 13.4%, to \$857,000 for the nine months ended September 30, 1997 from \$756,000 for the nine months ended September 30, 1996. Operating expenses as a percentage of revenue increased 0.5% to 27.8% at September 30, 1997 from 27.3% for the nine months ended September 30, 1996. The increase in operating expenses and operating expenses as a percentage of revenue were the result of additional personnel and increased costs related to the initiation of new document depository services and complimentary services related to large multi-party lawsuits.

Operating Income. As the result of the foregoing, operating income for the nine months ended September 30, 1997 increased \$7,000, or 1.0\$, to \$656,000 from \$649,000 for the nine months ended September 30, 1996. Operating margin percentage declined 2.1\$ to 21.3\$ for the nine months ended September 30, 1997 from 23.4\$ at September 30, 1996.

Year Ended December 31, 1996 Compared to Year Ended December 31, 1995

Revenue. Revenue increased \$657,000, or 19.5%, to \$4.0 million in 1996 from \$3.4 million in 1995. The increase in revenue was the result of an increase in the number of clients, additional services from existing clients and the number of multi-party lawsuits.

Gross Profit. Gross profit increased \$384,000, or 23.0%, to \$2.1 million in 1996 from \$1.7 million in 1996. Gross margin percentage increased slightly to 51.0% in 1996 from 49.6% in 1995. The improvement in the gross margin percentage was the result of cost containment actions.

Operating Expenses. Operating expenses increased \$188,000, or 21.1%, to \$1.1 million in 1996 from \$891,000 in 1995 as the result of the higher level of business activity. Operating expenses as a percentage of revenue increased slightly from 26.8% in 1996 from 26.5% in 1995.

25

Operating Income. As a result of the foregoing, operating income increased \$196,000, or 25.2%, to \$973,000 in 1996 from \$777,000 in 1995. Operating margin percentage increased to 24.2% in 1996 from 23.1% in 1995.

RESULTS OF OPERATIONS--ELAINE DINE

Elaine Dine provides permanent placement and temporary legal staffing and derives its revenue from legal placement and staffing fees which are based upon a percentage of the attorney's compensation earned during the year following the placement.

Historically, the operations of Elaine Dine have been focused on the permanent placement of attorneys. In April 1996, Elaine Dine created Elaine Dine Temporaries LLC which provides temporary legal staffing services. Revenue from temporary staffing services are derived from hourly fees charged to clients for the number of hours worked by temporary staffing attorneys.

The Company acquired Elaine Dine in September 1997.

Three Months Ended June 30, 1997 Compared to Three Months Ended June 30, 1996

Revenue. Revenue for the three months ended June 30, 1997 increased \$488,000, or 70.2%, to \$1.2 million from \$695,000 for the three months ended June 30, 1996. The increase was the result of the addition of several new corporate clients and a higher number of searches for corporate in-house attorneys.

Gross Profit. Gross profit for the three months ended June 30, 1997 increased \$210,000, or 62.3%, to \$546,000 from \$337,000 for the three months ended June 30, 1996. During the three months ended June 30, 1997, the gross margin percentage declined 2.3% to 46.2% at June 30, 1997 from 48.5% at June 30, 1996. The decrease in gross margin percentage was the result of an increase in the commission percentages paid to staff recruiters.

Operating Expenses. Operating expenses increased \$58,000, or 35.2%, to \$222,000 for the three months ended June 30, 1997 from \$164,000 for the three months ended June 30, 1996. Operating expenses as a percentage of revenue decreased 4.8% to 18.8% for the three months ended June 30, 1997 from 23.6% for the three months ended June 30, 1996.

Operating Income. As a result of the foregoing, operating income increased \$152,000, or 88.0%, to \$325,000 for the three months ended June 30, 1997 from \$173,000 for the three months ended June 30, 1996. During the three months ended June 30, 1997, the operating margin percentage improved 2.6% to 27.5% for the three months ended June 30, 1997 from 24.9% for the three months ended June 30, 1996.

Year Ended March 31, 1997 Compared to Year Ended March 31, 1996

Revenue. Revenue increased \$1.2 million, or 32.9%, to \$4.7 million for the year ended March 31, 1997 from \$3.5 million for the year ended March 31, 1996. The increase in revenue was due to favorable economic conditions in the financial services industry which created a need for additional attorneys for both law firms and large corporations. In 1996, Elaine Dine added new clients and was awarded several large retained searches, including searches for inhouse attorneys.

Gross Profit. Gross profit increased \$544,000, or 36.9%, to \$2.0 million in 1997 from \$1.5 million in 1996. Gross margin percentage increased slightly to 43.3% in 1997 from 42.1% in 1996.

Operating Expenses. Operating expenses decreased \$116,000, or 9.2%, to \$1.1 million in 1997 from \$1.3 million in 1996. The reduction in operating expenses in 1997 was primarily due to lower health insurance costs and outside professional fees as compared with 1996. Operating expenses as a percentage of revenue decreased 11.4% to 24.6% in 1997 from 36.0% in 1996.

26

Operating Income. As a result of the foregoing, operating income increased \$660,000, or 313.1%, to \$870,000 in 1997 from \$211,000 in 1996. Operating margin percentage improved 12.7% to 18.7% in 1997 from 6.0% in 1996.

RESULTS OF OPERATIONS -- LEGAL ENTERPRISE, INC.

Legal Enterprise provides certified record retrieval services and derives its revenue from fees generated from such services, which typically are based upon the number of subpoenas or other notices initiated and the volume of documents generated in response to these subpoenas. The Company acquired Legal Enterprise in August 1997.

Six Months Ended June 30, 1997 Compared to Six Months Ended June 30, 1996

Revenue. Revenue increased \$512,000, or 29.8%, to \$2.2 million for the six months ended June 30, 1997 from \$1.7 million for the six months ended June 30, 1996. The increase in revenue was attributable to several new clients during 1996 and early 1997, including several large insurance companies. In addition, Legal Enterprise opened a new office in northern California.

Gross Profit. Gross profit increased \$80,000, or 12.5%, to \$722,000 for the six months ended June 30, 1997 from \$642,000 for the six months ended June 30, 1996. Gross margin percentage declined 5.0% to 32.4% from 37.4% for the six months ended June 30, 1996. The decline in gross margin percentage was the result of increased costs attributable to hiring additional personnel and training costs incurred in connection with the initiation of record retrieval services to be provided to new clients. In addition, Legal Enterprise was negatively impacted by servicing clients outside its core geographic area in the six months ended June 30, 1997.

Operating Expenses. Operating expenses increased \$13,000, or 2.4%, to \$584,000 for the six months ended June 30, 1997 from \$571,000 for the six months ended June 30, 1996. Operating expenses as a percentage of revenue decreased 7.0% to 26.2% for the six months ended June 30, 1997 from 33.2% for the six months ended June 30, 1996 as a result of achieving operating leverage.

Operating Income. As a result of the foregoing, operating income increased \$67,000, or 93.5%, to \$138,000 for the six months ended June 30, 1997 from \$71,000 for the six months ended June 30, 1996. Operating margin percentage improved 2.1% to 6.2% for the six months ended June 30, 1997 from 4.1% for the six months ended June 30, 1996.

Year Ended December 31, 1996 Compared to Year Ended December 31, 1995

Revenue. Revenue increased \$950,000, or 34.5%, to \$3.7 million in 1996 from \$2.8 million in 1995. During 1996, Legal Enterprise obtained several new insurance company clients and expanded its business into northern California.

Gross Profit. Gross profit increased \$444,000, or 45.8%, to \$1.4 million in 1996 from \$971,000 in 1995. Gross margin percentage increased 3.0% to 38.2% in 1996 from 35.2% in 1995. The increase in the gross margin percentage was the result of the impact of greater revenue on the fixed component of cost of services.

Operating Expenses. Operating expenses increased \$377,000, or 43.3%, to \$1.2 million in 1996 from \$870,000 in 1995. Operating expenses as a percentage of revenue increased 2.0% to 33.6% in 1996 from 31.6% in 1995, primarily as the result of costs associated with relocating the headquarters and the opening of additional locations in 1996.

Operating Income. As a result of the foregoing, operating income increased

27

FLUCTUATIONS IN QUARTERLY RESULTS OF OPERATIONS

Revenues from the Company's services historically have shown no significant seasonal variations. Revenues can vary from period to period due to the effects of the timing of acquisitions and the timing and magnitude of costs related to such acquisitions, as well as the timing of the commencement, settlement or completion of major lawsuits. The increase in corporate overhead associated with the Completed Acquisitions and the Pending Acquisitions will be reflected in the Company's results of operations in the periods immediately following the Offering, while the realization of benefits, if any, from such acquisitions may not be reflected until future periods.

LIQUIDITY AND CAPITAL RESOURCES

The Company's cash requirements have been primarily to fund the cash portion of the purchase price of the acquisitions. The Company has obtained these funds principally through borrowings under the Bank Credit Agreement, the proceeds from the placement of \$10.0 million of Senior Subordinated Notes and Preferred Stock, and cash provided by operations. For the nine months ended September 30, 1997, cash provided by operating activities was \$615,000. The aggregate cash consideration paid to sellers for acquisitions completed through September 30, 1997 was approximately \$21.4 million, prior to giving effect to any associated working capital adjustments. The cash portion of the purchase price for the Pending Acquisitions will be approximately \$16.9 million, which will be provided from approximately \$4.8 million of the net proceeds of the Offering and borrowings of approximately \$12.1 million under the new credit facility described below.

The Company's existing Bank Credit Agreement provides for a \$4.3 million revolving line of credit of which approximately \$3.2 million has been utilized and a \$14.0 million acquisition line of credit which has been fully utilized. Borrowings under the Bank Credit Agreement bear interest at a base rate plus 0.75%, or LIBOR plus 2.5%, at the Company's option. As of December 15, 1997, the average annual interest rate on borrowings under the credit facilities was 8.50% and such borrowings are secured by substantially all of the assets of the Company, including the capital stock of subsidiaries. The Bank Credit Agreement contains various covenants that, among other matters, restrict or limit the Company's ability to pay dividends, incur indebtedness, make capital expenditures and repurchase capital stock. The Company must also maintain minimum fixed charge coverage and other ratios. All borrowings under the Bank Credit Agreement are required to be repaid concurrently with the completion of the Offering. See "Use of Proceeds."

The Company anticipates that it will require significant amounts of cash to support its acquisition strategy after the Offering. The Company expects to fulfill these requirements for cash primarily through bank borrowings, cash from operations, and from the sale of debt or equity securities of the Company. The Company has entered into a credit agreement (the "New Credit Agreement") with NationsBank of Texas, N.A. ("NationsBank") to provide a new \$40.0 million revolving credit facility upon completion of the Offering. The amount the Company may borrow under the revolving credit facility will be subject to limitations based upon the EBITDA of the Company for the preceding twelve months, adjusted to give effect to all acquisitions completed during such period or pending as of the date of borrowing. The Company expects that, immediately upon completion of the Offering, approximately \$12.1 million will be borrowed and approximately \$10.0 million will be available for borrowing under the New Credit Agreement. From time to time, additional borrowing capacity will be available to the Company under the New Credit Agreement provided that the Company achieves certain levels of financial performance through operations or through acquisitions. Loans under the New Credit Agreement will bear interest at rates based, at the Company's option, on either LIBOR or NationsBank's base rate plus, in each case, an applicable margin. The applicable margin will be contingent upon the ratio of the Company's consolidated funded debt to its consolidated EBITDA (the "Debt Coverage Ratio") and will vary from 1.75% to 2.25% in the case of LIBOR loans and 0% to .50% in the case of base rate loans. In addition, the Company will be required to pay to the lenders a quarterly fee with respect to the unused portion of the credit facility. The unused fee varies depending on the Debt Coverage Ratio and ranges from .30% to .50%. The New Credit Agreement will also permit \$2.5 million of the amount available for borrowings to be used for the issuance of letters of credit. A per annum fee based on the applicable margin for LIBOR loans is payable each quarter with respect to the face amount of letters of credit outstanding during the applicable quarter. See Note 7 of Notes to Financial Statements of U.S. Legal Support, Inc.

Borrowings under the New Credit Agreement will be secured by substantially all of the assets of the Company and its subsidiaries and by a pledge of the stock of the subsidiaries. In addition, subsidiaries of the Company will be required to guarantee the Company's obligations under the New Credit Agreement. The New Credit Agreement also will require the Company and its subsidiaries to comply with various affirmative and negative covenants related to, among other matters: (i) the maintenance of certain financial ratios; (ii) limitations on the incurrence of indebtedness; (iii) restrictions on liens, guarantees, dividends and stock redemptions; and (iv) limitations on mergers and sales of assets. The new revolving credit facility will terminate, and all borrowings will be required to be repaid on December 31, 2000. The Company anticipates that bank borrowings and cash from operations will be sufficient to meet the Company's capital requirements through the end of 1998.

The Company's business has not typically required substantial capital expenditures, and during the nine months ended September 30, 1997, capital expenditures were approximately \$111,000. However, the Company anticipates that capital expenditures will increase to approximately \$150,000 for the remainder of 1997 and will be approximately \$750,000 in 1998. These amounts include approximately \$500,000 expected to be incurred to acquire and implement a centralized management information system designed to standardize financial reporting and accounting controls.

RECENTLY ISSUED ACCOUNTING PRONOUNCEMENTS

In February 1997, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards No. 128, Earnings Per Share ("SFAS 128"). SFAS 128 changes the computation of earnings per share and requires dual presentation of basic and diluted earnings per share. SFAS 128 is effective for financial statements issued for periods ending after December 15, 1997, including interim periods. The Company will adopt SFAS 128 in the fourth quarter of 1997 as required.

In June 1997, the FASB issued Statement of Financial Accounting Standards No. 130, "Reporting Comprehensive Income" ("SFAS 130") and Statement of Financial Accounting Standards No. 131, "Disclosure about Segments of an Enterprise and Related Information" ("SFAS 131").

SFAS 130 establishes standards for reporting and display of comprehensive income and its components (revenues, expenses, gains and losses) in a full set of general-purpose financial statements. It requires: (i) classification of the components other comprehensive income by their nature in a financial statement and (ii) the display of the accumulated balance of the components other comprehensive income separate from retained earnings and additional paid-in-capital in the equity section of a statement of financial position. SFAS 130 is effective for years beginning after December 15, 1997 and is not expected to have a material impact on financial position or results of operations.

SFAS 131 establishes standards for reporting information about operating segments in annual financial statements and requires selected information about operating segments in interim financial reports issued to shareholders. It also establishes standards for related disclosures about products and services, geographic areas and major customers. The Company has not determined the impact of SFAS 131 on its financial reporting practices.

29

BUSINESS

GENERAL

The Company is one of the largest providers of legal support and staffing services in the United States, providing court reporting, certified record retrieval, legal placement and staffing, and other related services to law firms and corporations, including insurance companies, through 40 offices in seven states and the District of Columbia. The Company seeks to become the leading national, full-service provider of legal support and staffing services through a combination of selective acquisitions and internal growth. Since commencing operations in January 1997, the Company has acquired 14 businesses, and will acquire four additional businesses concurrently with the Offering. In 1997, the Company has provided court reporting and certified record retrieval services to clients such as The Boeing Company, Ford Motor Company, ITT Hartford, CNA and Kemper Insurance Company. For the year ended December 31, 1996, the Company had pro forma revenues of \$43.4 million and pro forma operating income of \$5.2 million. For the nine months ended September 30, 1997, the Company had pro forma revenues of \$37.3 million and pro forma operating income of \$5.3 million, compared with pro forma revenues of \$32.0 million and pro forma operating income of \$3.7 million for the nine months ended September 30, 1996.

Legal support and staffing services are frequently used by the participants

in civil litigation, particularly in cases involving personal injury, products liability, workers' compensation and property casualty claims. While most civil proceedings are resolved by settlement or are otherwise terminated prior to trial, most legal support services are required in the pre-trial period. The following presents the general progression of a civil lawsuit or other proceeding and the opportunities for the Company to provide services to the parties:

T.OCC

[Flowchart of basic steps in the progression of a lawsuit or other legal proceeding identifying Company's services at each step as follows:] [Claimant Retains Counsel--Legal Staffing and Placement]--[File Lawsuit]--[Defendant Retains Counsel--Legal Staffing and Placement]--[Gather & Prepare Documents--Certified Record Retrieval]--[Examine Witnesses by Deposition--Court Reporting]--[Trial or Other Proceeding Occurs].

Legal Placement and Staffing. During the course of litigation, particularly in the pre-trial phase of complex, multi-party cases, law firms and corporate legal departments may hire additional attorneys on a permanent or temporary basis to assist in the preparation of the case for trial. In addition, law firms and corporations increasingly are utilizing temporary lawyers in other areas of practice to more effectively manage fluctuating work loads.

Certified Record Retrieval. After the initiation of a lawsuit or other legal proceeding, the parties typically seek medical, employment, financial or other records to assist in the evaluation, furtherance or defense of the claims asserted. These records must be obtained through the issuance of subpoenas or other notices directed to hospitals, physicians, employers, banks or others in accordance with applicable federal and state rules of procedure to ensure the admissibility of the records in the lawsuit. The Company prepares and delivers subpoenas and other document request notices, monitors the recipient's response and coordinates the retrieval and dissemination of documents for litigation.

Court Reporting. Prior to commencement of a trial or other proceeding, attorneys for the parties to the dispute and their insurance providers often seek sworn oral testimony of witnesses. Oral testimony is generally obtained through a deposition transcribed by a court reporter. The persons deposed may include the parties to

30

the proceeding as well as expert witnesses and others, and the depositions may include testimony concerning the content of documents previously obtained through certified record retrieval. Licensed court reporters typically transcribe the testimony given in deposition and written copies are made available to the parties. Testimony given during a deposition often leads to further depositions and to additional certified document retrieval assignments.

INDUSTRY OVERVIEW

Based on available industry data, the Company estimates that the market for legal support and staffing services in the United States exceeds \$5.0 billion annually. The industry is highly fragmented, with more than 1,000 court reporting and record retrieval firms and over 400 legal placement and staffing firms. The Company believes that the legal support and staffing services market is growing due to several trends, including an increase in the: (i) efforts by law firms and corporations, especially insurance providers, to reduce legal expenses; (ii) outsourcing of legal support services to companies that specialize in providing such services at a lower cost; (iii) use of lawyers on a temporary basis by law firms and corporations; (iv) volume and complexity of litigation; and (v) national scope of litigation, particularly in class action and product liability lawsuits.

Legal support services, such as court reporting and certified record retrieval, traditionally have been marketed to law firms. Increasingly, insurance providers and major corporations, who ultimately pay the costs of legal support and staffing services, are seeking to: (i) control and reduce the costs associated with lawsuits; (ii) centralize their purchasing decisions; and (iii) ensure consistent service quality. As a result, these companies are more frequently selecting the providers of legal support services themselves, rather than delegating that decision to the law firms engaged to represent them.

Currently the legal support and staffing services industry is highly fragmented and consists primarily of local and regional firms that typically provide a single or limited number of legal support and staffing services. The Company believes that many legal support and staffing businesses lack: (i) a full range of legal support services; (ii) regional or national coverage; (iii) access to capital; or (iv) effective marketing programs, and are therefore unable to meet the needs of large, geographically dispersed clients. In addition, there are limited opportunities for owners of local legal support and staffing firms to obtain liquidity or to sell their businesses. The

Company consequently believes that many owners of such businesses will be receptive to consolidation.

BUSINESS STRATEGY

The Company seeks to become the leading national, full-service provider of legal support and staffing services in the United States. To achieve this goal, the Company is implementing a focused business strategy that includes the following key elements:

Establish Full Service Operations in Multiple Metropolitan Areas. The Company seeks to capitalize on the trend toward vendor consolidation in the legal support and staffing services industry. The Company believes that national and regional accounts are increasingly contracting with vendors such as the Company that are capable of delivering a full range of high-quality legal support and staffing services on a broad geographic basis. The Company offers an array of complementary services that includes court reporting, certified record retrieval, legal placement and staffing, and related services through 40 offices located in or near major metropolitan markets in the United States. The Company expects to continue to increase the variety and geographic scope of its services in targeted metropolitan areas throughout the United States through selected acquisitions and expansion of its existing businesses.

Adopt Best Practices, Policies and Procedures. The Company evaluates its operating policies and procedures in order to identify and implement practices that best serve the objectives of the Company and its clients ("best practices"). The Company intends to integrate these best practices, including marketing, sales, field operations, human resources policies and recruiting and training programs into the operations of acquired businesses. The Company also intends to evaluate the practices of each acquired company and to implement selected practices of the acquired companies on a Company-wide basis.

31

Achieve Operating Efficiencies. The Company seeks to be a low-cost provider of legal support services through operating efficiencies and cost savings, which the Company believes can be achieved by combining a number of general and administrative functions at the corporate level and by reducing or replacing redundant functions and facilities of acquired companies. In addition, the Company believes that it will be able to reduce costs in the purchase of insurance, management information systems, advertising and other corporates.

Maintain Decentralized Management. Many of the businesses the Company will seek to acquire operate on a local or regional basis. To take advantage of existing relationships between the acquired companies and their clients, the Company intends to manage its business on a decentralized basis, with acquired-company management retaining primary responsibility for day-to-day operations and local marketing. The Company believes that allowing local management to retain appropriate autonomy will provide opportunities for internal growth, enhance competitiveness in attracting acquisition candidates and assist in preserving the entrepreneurial spirit of the acquired-company management.

GROWTH STRATEGY

The Company has implemented a strategy designed to continue its growth in existing and new markets based on the following key elements:

Actively Pursue Strategic Acquisitions. The Company intends to pursue an aggressive strategy of acquiring companies in the fragmented legal support and staffing services industry. Through strategic acquisitions, the Company will seek to serve new geographic markets, expand its presence in existing markets and add complementary services. One source of acquisition candidates has been the court reporting firms that have been affiliated with the Company's national court reporting network, through which the Company provides court reporting services to customers in locations not directly served by the Company. Since inception, the Company has acquired four network affiliates.

Establish Effective National Marketing Program. Insurance providers and major corporations, which ultimately pay the cost of legal support and staffing services, are increasingly seeking to centralize their purchasing decisions on a national and regional level and to control or reduce the costs associated with lawsuits. The Company seeks to capitalize on these initiatives by marketing its services directly to these companies and has established a national sales force to pursue these opportunities. The Company seeks to be designated as the exclusive or preferred provider of legal support services on a regional or national basis, which may result in assignments substantially larger than those obtained through local accounts.

Capitalize on Cross-Selling Opportunities. The Company believes that significant cross-selling opportunities exist which could enhance the

Company's revenue growth. Many legal support and staffing companies are local or regional and specialize in one segment of the legal support services market. The Company believes that its acquisition of such companies will enable it to become a full-service provider of legal support services nationwide and to leverage its existing client relationships by selling such clients a full range of legal support and staffing services. For example, as a result of the Company's acquisition of a certified record retrieval business in California, the Company recently obtained the California certified record retrieval business from a national insurance provider, who previously had utilized only the Company's court reporting services. In addition, the Company works closely with its clients to identify cost-savings opportunities and to develop solutions to their legal support needs.

Develop New and Expand Existing Client Relationships. The Company intends to develop new client relationships and expand existing relationships with law firms, insurance providers and major corporations through aggressive sales and marketing of its services. Sales and marketing efforts are conducted on both a local and national level and are designed to focus potential clients on the Company's: (i) ability to provide a broad range of complementary legal support and staffing services; (ii) national geographic coverage through 40 offices and its network of more than 130 affiliated court reporting firms; and (iii) high-quality services and programs.

32

ACOUISITION AND INTEGRATION STRATEGY

The Company's acquisition strategy is to identify, acquire and integrate companies with strong management, profitable operating results and recognized local or regional market presence. The Company typically pursues acquisitions that will allow it to accomplish one or more of the following: (i) expand the geographic markets served by the Company; (ii) increase the Company's penetration of existing markets; (iii) establish or enhance customer relationships; and (iv) offer services complementary to those offered by the Company.

The Company believes that there are numerous attractive acquisition candidates due to the large size and fragmentation of the legal support and staffing services industry. These candidates include but are not limited to participants in the Company's referral network of over 130 affiliated court reporting firms, through whom the Company supplies court reporting services to its clients in locations not served directly by the Company. The Company has acquired four businesses that were previously part of its affiliated network and the Company believes that others in the affiliated network may be likely acquisition candidates. The Company also pursues smaller "tuck-in" businesses that can be easily assimilated into the Company's existing operations. Such tuck-in acquisitions are intended to enable the Company to benefit from the operating leverage of its existing business by adding market share while eliminating or reducing certain general administrative and operating costs. The Company has made three such acquisitions of court reporting businesses in Texas.

The Company's corporate officers are responsible for identifying acquisition prospects, conducting due diligence, negotiating the terms of acquisitions and integrating acquired companies. The Company expects that the management of acquired companies will actively assist the Company in identifying additional acquisition candidates. The Company also utilizes the services of The GulfStar Group, Inc. ("GulfStar"), an investment banking firm and an affiliate of one of the founding shareholders of the Company, in evaluating acquisition candidates and negotiating acquisition terms. The Company's policy is to include Common Stock as part of the consideration for acquisitions to more closely align the interests of the shareholders and managers of acquired companies with those of the Company. See "Certain Transactions."

The Company has relied on the existing accounting, financial and computer systems of certain of the acquired companies for a transition period following completion of their acquisition. Although the Company believes that its existing financial reporting and accounting control systems are satisfactory for the operation of the combined businesses for the near term, the Company is currently evaluating alternatives in order to address the future needs of the combined businesses and to standardize and centralize accounting and financial procedures of acquired companies and establish a uniform system of accounts and standardized budgeting and reporting processes. The marketing, sales, field operations and personnel programs of the acquired companies also will be evaluated and integrated with those of the Company under its best practices program. Further, the Company seeks to identify any practices of an acquired company that could be beneficial if implemented on a Company-wide basis. Management of each of the acquired companies meet regularly to discuss the assimilation of acquired companies, as well as cross-selling and other opportunities to improve operating efficiency and increase revenue and profitability. The Company has entered into employment agreements with certain of the principal executives of each acquired business and intends to continue to seek to enter into such arrangements with key executives of companies

After succeeding to the operations of Looney in January 1997, the Company has acquired 13 businesses and will acquire four additional businesses in the Pending Acquisitions. Certain information relating to the Company's Completed Acquisitions and Pending Acquisitions is summarized in the following table:

<TABLE>

SERVICES	ACQUIRED COMPANY	HEADQUARTERS	
<s></s>	<c></c>	<c></c>	<c></c>
Court Reporting	Amicus One	New York	3
	Block	Washington, D.C.	3
	G&G	Los Angeles	1
	Jilio (1)	Los Angeles	1
	Johnson Group	Los Angeles	1
	Kirby Kennedy (1)	Minneapolis	1
	Klein Bury	Miami	6
	Looney (2)	Houston	8
	Reporting Service (1)	Philadelphia	1
	Rocca	Chicago	1
	San Francisco Reporting	San Francisco	1
			27
Certified Record			
Retrieval Legal Placement and	Legal Enterprise	Los Angeles	7
Staffing	Elaine Dine	New York	1
	Ziskind Greene	Los Angeles	4
	Commander Wilson (1)	Houston	1
			6
		Total:	40
			===

</TABLE>

- (1) Pending Acquisition to be completed concurrently with the Offering.
- (2) Looney also provides certified record retrieval services. Excludes the operations of Cindi Rogers & Associates, Encore Reporting and Preferred Records, Inc., which were acquired in 1997.

LEGAL SUPPORT AND STAFFING SERVICES

The Company provides court reporting, certified record retrieval, legal placement and staffing and other services, as described below.

Court Reporting. Court reporting is the verbatim transcription of sworn oral testimony, generally for use in legal proceedings. While the transcription of legal proceedings held in a courtroom generally is performed by personnel employed by the federal court system or by state agencies, counties or municipalities, court reporting performed outside a courtroom is generally conducted by private court reporters who transcribe depositions and certain other proceedings. Most states require court reporters to be licensed under regulations that require each candidate to attend a certified court reporting school, pass a written examination and demonstrate transcription proficiency using machine shorthand equipment. Court reporters are officers of the court and subject to ethical codes governing their professional conduct.

The Company believes that its court reporters utilize state-of-the-art technology, including computer-aided transcription ("CAT") systems. CAT systems allow the testimony transcribed by a court reporter to be simultaneously recorded on computer disk. In addition to CAT systems, the Company utilizes the following technologies to provide high-quality court reporting services:

. Transcription on a Real-Time Basis. The Company's court reporters provide instant transcripts of testimony on computer monitors, which may be located where the testimony is taking place, as well as at remote locations. This system also allows attorneys to receive a transcript of the testimony on diskette at the conclusion of the proceeding.

34

. Search and Retrieval Programs. Through the use of computers, court reporters can search, store, index and manage transcripts and other documents. In particular, a search through a transcript for a particular

- . Compressed Transcripts. Compressed transcripts contain an index listing all words in the transcript as well as the frequency and location of the words, thereby simplifying the summarizing of transcripts. This technique also reduces transcript bulk by organizing the text in block and columns.
- . Video Services. Video services include the taping of depositions and other testimony on videotape while the court reporter simultaneously transcribes the testimony on a CAT system. Videotaped testimony is sometimes used in legal proceedings when a witness is unable to appear in person at trial or when capturing the demeanor of a witness is important.

The Company believes that voice recognition technologies currently do not represent a practicable alternative to the Company's court reporting services because of the extremely high accuracy required in the transcription of legal proceedings, the difficulties associated with the electronic recognition of multiple voices, variances in dialect and regional accents and the higher cost of the application of voice recognition technology.

Certified Record Retrieval. The parties to legal proceedings frequently use certified record retrieval services. Certified record retrieval services are labor-intensive and involve the preparation, handling, tracking and delivery of large numbers of written documents. A significant portion of the record retrieval business involves medical records acquired on behalf of insurance companies and their counsel. The Company's record retrieval services include the preparation and delivery of written deposition notices and subpoenas, the monitoring of the recipient's response and the coordination of document retrieval and production for litigation or other legal proceedings.

The Company has developed customized computer software to link record retrieval directly to law firms. This software permits instant communication and promotes efficient litigation management. The Company anticipates that certified document retrieval will increasingly be conducted electronically, with deposition notices and subpoenas being transmitted electronically to document custodians and witnesses, and responses similarly returned electronically.

Legal Placement and Staffing. The Company provides clients with qualified permanent and temporary attorneys through a team of 11 recruiters located in six metropolitan markets. The Company believes it is able to attract clients based on the reputation and relationships of its recruiting personnel, its emphasis on client service and its extensive legal staffing databases, which allow it to match qualified personnel with its clients' needs. The Company's databases include more than 50,000 attorneys and include individuals with practices encompassing a broad range of legal specialties. The Company's permanent search activities consist primarily of the recruitment and placement of attorneys on a permanent basis with law firms and corporate legal departments. The Company also has been engaged from time to time on assignments to establish entire departments or offices. The Company is engaged and paid by the hiring firm or corporation on either an exclusive or nonexclusive basis pursuant to arrangements that may include a non-refundable retainer paid to the Company or entitle the Company to a fee only upon the successful placement of a candidate with the client.

Law firms and corporate legal departments are increasingly using temporary legal personnel to enable them to respond more effectively to fluctuations in work load. In response to this trend, the Company recently has begun to expand its services beyond permanent placement services and supplies attorneys to clients on a temporary basis. Temporary attorney assignments may range from one day to more than a year and may involve one attorney or a team composed of numerous lawyers. The Company does not maintain malpractice insurance to cover the performance of services by its temporary attorneys. Instead, the Company requests that clients agree to include temporary attorneys under their policies and to indemnify the Company from losses associated with the provision of legal services by temporary attorneys. In addition, the Company believes that malpractice insurance coverage maintained by its clients typically includes coverage for temporary attorneys who are supervised by employees of clients.

35

Other Ancillary Services. The Company also offers its clients transcription, closed captioning, translation and document management services, either directly or through relationships or alliances with other companies. Document management services include the electronic recording, storage, coding, indexing and automated retrieval of documents, as well as database management services. The Company markets these services to legal support applications.

CLIENT RELATIONSHIPS

The Company's client base is composed primarily of over 9,000 law firms, insurance providers and corporations. The Company has a broad customer base,

and no customer of the Company accounted for more than 5% of the Company's pro forma revenues for 1996 or for the nine months ended September 30, 1997. Clients typically engage the Company on a case-by-case basis, although the Company intends to market its services increasingly to general counsels, regional or national litigation managers, or other corporate officers in an effort to persuade such persons to retain the Company on an exclusive or "preferred provider" basis. The Company also works closely with its clients to identify cost-saving opportunities and to develop solutions to the client's legal support needs.

SALES AND MARKETING

The Company markets its services through management and sales personnel located in 40 offices. Because the Company derives a majority of its revenues from local or regional accounts, the managers of the Company's individual offices are primarily responsible for sales and marketing in their respective markets. These managers seek to identify leads, qualify prospects and close sales. The Company obtains new clients in local markets primarily through direct sales, client referrals and a variety of local media, including direct mail, Yellow Pages and trade publications. In response to the trend among insurance providers and major corporations to centralize their purchasing decisions and to engage legal support and staffing services on a regional or national basis, the Company also has established a national sales force consisting of six salespersons. These national sales personnel focus on national accounts and seek to have the Company designated as the exclusive or preferred provider of legal support and staffing services. Company operating personnel also participate in marketing efforts by providing advice regarding the Company's operational capabilities. Because the Company offers a variety of services and is seeking to establish national market coverage, sales and operating personnel also seek to capitalize on cross-selling opportunities.

The Company is developing a marketing and advertising program to establish a national brand identification, while preserving the value of the established trade names and customer relationships of the acquired companies. The Company's logo and identifying marks will be featured in promotional materials of the Company, in a manner designed not to detract from the local recognition of an acquired business.

COMPETITION

The market for legal support and staffing services is highly competitive. The Company competes with a large number of local and regional court reporting and certified record retrieval companies, as well as with permanent and temporary legal staffing companies, including national temporary staffing firms. A significant source of competition is also the in-house provision of legal support and staffing services by law firms, insurance providers and major corporations. There can be no assurance that these businesses will continue to increase their outsourcing of legal support and staffing services needs or that such businesses will not bring in-house services that they currently outsource. Certain of the Company's competitors have substantially greater resources and operate in broader geographic areas than the Company. Many larger clients retain multiple legal support and placement and staffing service providers, which exposes the Company to continuous competition. The Company competes primarily on the basis of the quality, breadth and timeliness of service, geographic coverage and price.

The Company believes that further consolidation among legal support and staffing services providers will continue during the next few years and that there will be significant competition for attractive acquisition

36

candidates. This competition could lead to higher prices being paid for businesses. The Company believes that it will have certain advantages in completing acquisitions, including: (i) management's personal relationships with existing legal support and staffing companies; (ii) its decentralized, entrepreneurial management strategy; (iii) its greater size and scope of services; and (iv) its visibility and resources as a public company. However, there can be no assurance that the Company's acquisition program will be successful or that the Company will be able to compete effectively for acquisitions.

INDEPENDENT CONTRACTORS

The Company provides court reporting services through the use of independent contractors who are not employees of the Company. The Company does not pay federal or state employment taxes or withhold income taxes with respect to these independent contractors or include such persons in the Company's employee benefit plans. Independent court reporters are responsible for owning and operating their court reporting equipment. The use of independent contractors as court reporters is widespread industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible

for the taxes required to be paid or withheld and could incur additional costs associated with employee benefits and other employee costs on both a current and a retroactive basis.

EMPLOYEES

The Company currently employs approximately 280 persons and believes that its relationships with employees are good.

FACILITIES

The Company maintains 40 leased office locations in seven states and the District of Columbia. The initial terms of most of the Company's leases range from one to five years and do not contain renewal terms. The Company's leases generally specify a fixed annual rent with fixed increases, or increases based on changes in the Consumer Price Index, at various intervals during the lease term. Generally, the leases are net leases which require the Company to pay all or a portion of the cost of insurance, maintenance and utilities. The Company believes that these facilities are adequate to serve its current level of operations. If additional facilities are required, the Company believes that suitable additional or alternative space will be available as needed on commercially reasonable terms. Substantially all of the leasehold interests and personal property of the Company are subject to a lien under its Bank Credit Agreement. See "Certain Transactions."

REGULATION

Court reporters are subject to significant regulation under state licensing programs. The conduct of court reporters is also subject to ethical and other restrictions imposed by state laws and regulations. While these regulations are not directly applicable to the Company, they affect the Company's court reporting business. In addition, attorneys are subject to significant regulation by committees on legal ethics and professional responsibility of the various state and national bar associations, who from time to time, examine and issue opinions regarding attorney services, including the use of temporary attorneys through a placement agency. Changes in these laws and regulations, particularly in California, New York, Florida, Pennsylvania or Texas, the states from which the Company derives most of its revenues, could have a material effect on the Company, its business, operations and profitability.

LEGAL PROCEEDINGS

The Company is involved in legal proceedings from time to time in the ordinary course of its business. Management believes that no pending legal proceeding will have a material adverse effect on the financial condition or results of operations of the Company.

37

MANAGEMENT

EXECUTIVE OFFICERS AND DIRECTORS

The following table sets forth certain information concerning the executive officers and directors of the Company and certain persons who will become directors or executive officers upon completion of the Offering:

<TABLE>

NAME	AGE	POSITION
<\$>		<c></c>
Richard O. Looney	40	Chairman of the Board, President and Chief Executive Officer
David P. Tusa	37	Executive Vice President and Chief Financial Officer
Tony L. Maddocks	40	Vice President, Sales and Marketing
James M. Wilson	48	Vice President, Legal Staffing
Scott R. Creasman	32	Vice President and Corporate Controller
Michael A. Klein	52	Director; PresidentKlein Bury
Fentress Bracewell	76	Director
Robert J. Cresci	53	Director
G. Kent Kahle	46	Director
<pre>Ronald C. Lassiter</pre>	65	Director

Richard O. Looney has served as Chairman of the Board, President and Chief Executive Officer of the Company since January 1997. Mr. Looney founded Looney & Company in 1988 and served as its President until October 1997. Mr. Looney is an active member of the National Court Reporters Association and the Texas Court Reporters Association.

David P. Tusa joined the Company in November 1997 as Executive Vice President and Chief Financial Officer. From August 1994 until August 1997, Mr. Tusa served as Senior Vice President of Finance and Administration of Serv-Tech, Inc., a publicly held specialty provider of technology driven industrial maintenance services. From 1990 until 1994, Mr. Tusa was with CRSS, Inc., a publicly held diversified engineering and construction services firm, most recently as Corporate Controller.

Tony L. Maddocks joined the Company in August 1997 as Vice President, Sales and Marketing. From June 1993 until August 1997, Mr. Maddocks served as President and Chief Executive Officer of Legal Enterprise, a certified record retrieval company. The Company purchased substantially all of the assets of Legal Enterprise in August 1997. From 1981 until June 1993, Mr. Maddocks served as Vice President of Sales and Marketing of Compex, Inc., a document management and record retrieval firm.

James M. Wilson joined the Company in September 1997 as Vice President, Legal Staffing. Mr. Wilson has been directly involved in the attorney search and consulting industry since 1986. In 1996, Mr. Wilson formed Commander Wilson, Inc. Mr. Wilson has served on the Board of Directors of the National Association of Legal Search Consultants, as well as serving as the organization's Ethics Committee Chairman. In May 1996, Mr. Wilson filed a voluntary petition for Chapter XIII bankruptcy relief following the initiation of a lawsuit filed by a former independent contractor engaged by Commander Wilson. Mr. Wilson has submitted a Chapter XIII plan which has not yet been confirmed by the presiding bankruptcy court. Commander Wilson will sell all of its assets to the Company concurrently with the closing of the Offering, and such sale has been approved by the bankruptcy court.

Scott R. Creasman joined the Company in November 1997 as Vice President and Corporate Controller. From October 1996 until August 1997, Mr. Creasman served as Controller of Brink's Home Security, Inc., a national provider of residential security services. From 1990 until 1996, Mr. Creasman was with Texas Eastern Products Pipeline Company, the operator of a publicly held petroleum products pipeline company, most recently as Controller.

Michael A. Klein has been a director of the Company since January 1997. Mr. Klein is the founder and President of Klein Bury, the Company's Florida-based subsidiary, and has been directly involved in the court reporting industry for over twenty-nine years. The Company purchased all of the capital stock of Klein Bury in January 1997.

38

Fentress Bracewell has agreed to serve as a director of the Company effective upon the closing of the Offering. Mr. Bracewell is a founder of and Senior Counsel to the law firm of Bracewell & Patterson, L.L.P. He also serves as the Chairman of the Board of Directors of First Investors Financial Services, Inc., an automobile finance company and is a member of the Board of Trustees of the Institute of International Education. Mr. Bracewell served as the Chairman of the Port of Houston Authority from 1970 to 1985.

Robert J. Cresci has served as a director of the Company since January 1997. Since September 1990, Mr. Cresci has been a Managing Partner of Pecks Management Partners Ltd., an investment firm. Mr. Cresci currently serves on the boards of Bridgeport Machines, Inc., EIS International, Inc., Sepracor, Inc., Garnet Resources Corporation, HarCor Energy, Inc., Meris Laboratories, Inc., Westbrae Natural, Inc. Arcadia Financial, Ltd., Hitox, Inc. Film Roman, Inc., Educational Medical, Inc., Source Media, Inc. and NetPower, Inc.

G. Kent Kahle has served as a director of the Company since its formation. Mr. Kahle has been a Managing Director of GulfStar since 1990. Prior to joining GulfStar he was a Senior Vice President and Director of Rotan Mosle, Inc., a subsidiary of PaineWebber Inc. Mr. Kahle currently serves on the board of Castle Dental Centers, Inc.

Mr. Ronald C. Lassiter has agreed to serve as a director of the Company effective upon the closing of the Offering. Since January 1997, Mr. Lassiter has served as the Chairman and Chief Executive Officer of Daniel Industries, a manufacturer of fluid measurement and flow control products and services for the oil and gas industry. From October 1992 until June 1997, Mr. Lassiter served as Chairman and Chief Executive Officer of Zapata Protein, Inc. Before joining Zapata Protein, Inc., Mr. Lassiter served as Chairman and Chief Executive Officer of Zapata Corporation. Mr. Lassiter also serves on the Board of Directors of Zapata Corporation.

See "Certain Transactions" for a description of the terms of transactions or other relationships between the Company and certain of its officers and directors.

BOARD OF DIRECTORS

Directors serve for one-year terms and are elected annually by the Company's

shareholders. Pursuant to the terms of an agreement entered into in connection with the Company's acquisition of all of the capital stock of Klein Bury, as long as the Company is indebted to Mr. Klein under the \$1.4 million Subordinated Promissory Note issued in connection with such acquisition, Mr. Klein is entitled to serve as a director of the Company. Mr. Cresci serves as a director of the Company pursuant to the terms of the Securities Purchase Agreement between the Company and the purchasers of the Senior Subordinated Notes and the Series A Preferred Stock. The Securities Purchase Agreement will terminate immediately prior to the completion of the Offering.

The Board of Directors has established two standing committees, the Audit Committee and the Compensation Committee. Pursuant to resolutions of the Board, these committees have the following responsibilities and authority.

Audit Committee. The Audit Committee has the responsibility, among other things, to: (i) recommend the selection of the Company's independent public accountants; (ii) review and approve the scope of the independent public accountants' audit activity and extent of non-audit services; (iii) review with management and the Company's independent public accountants the adequacy of the Company's basic accounting system and the effectiveness of the Company's internal audit plan and activities; (iv) review with management and the independent public accountants the Company's financial statements and exercise general oversight of the Company's financial reporting process; and (v) review litigation and other legal matters that may affect the Company's financial condition. The Audit Committee is composed of Messrs. Kahle (Chair) and Cresci.

Compensation Committee. The Compensation Committee has the responsibility, among other things, to: (i) recommend to the Board the salary rates of officers of the Company and its subsidiaries; (ii) examine periodically the compensation structure of the Company; and (iii) supervise the welfare and pension plans and compensation

39

plans of the Company. The Compensation Committee is composed of all of the non-employee directors, currently Messrs. Kahle (Chair) and Cresci.

The Company's Board also may establish other committees.

DIRECTOR COMPENSATION

Prior to the date of this Prospectus, directors of the Company did not receive compensation for their services as directors or for attending board meetings. Upon completion of the Offering, non-employee directors of the Company will receive options to purchase 25,000 shares of Common Stock at the initial public offering price. In addition, each non-employee director will receive an annual fee of \$6,000, a fee of \$1,000 for each meeting of the Board attended and \$500 for each meeting of a Board committee attended. Each director also will be reimbursed for travel expenses incurred for each non-telephonic meeting of the Board or any committee thereof attended.

EXECUTIVE COMPENSATION; EMPLOYMENT AGREEMENTS

Prior to January 1997, the Company did not conduct any operations, other than activities related to the acquisition of Looney and Klein Bury, and did not pay any compensation. The Company anticipates that during 1997 its most highly compensated executive officers (the "Named Executive Officers") and their annualized base salaries will be: Richard O. Looney, \$250,000; David P. Tusa, \$170,000; Tony L. Maddocks, \$162,500; James M. Wilson, \$150,000; Michael A. Klein, \$175,000; and Scott R. Creasman, \$100,000.

The Company and Mr. Looney have entered into an employment agreement that provides for an annual base salary of \$250,000 and a semi-annual cash bonus based on a percentage of the annual pre-tax profits of the Company, subject to an annual maximum of \$100,000. Mr. Looney's employment agreement expires on January 17, 2000. In the event his employment is terminated without cause, Mr. Looney's annual salary and bonus opportunity will continue until the earlier to occur of one year from the date of such termination or January 17, 2000. Mr. Looney has also agreed not to compete with the Company in a court reporting or litigation support service business within 50 miles of an office served by the Company until the later to occur of January 17, 2002, or three years after his employment terminates.

The Company and Mr. Maddocks have entered into an employment agreement which provides for an annual base salary of \$162,500 and an annual cash bonus not exceeding \$60,000, based on annual new national account sales of the Company. Mr. Maddocks' employment agreement expires on August 29, 2000. In the event his employment is terminated without cause, Mr. Maddocks' annual salary and bonus opportunity will continue until the earlier to occur of one year from the date of termination or August 29, 2000 or Mr. Maddocks' death. Mr. Maddocks has agreed not to compete with the Company until the last to occur of August 29, 2002, or three years after his employment under the agreement

Mr. Wilson has entered into an employment agreement with the Company which provides for an annual base salary of \$150,000 and, commencing in 1998, an annual cash bonus equal to 10% of the amount by which the annual adjusted net profit (as defined in the agreement) derived from legal placement and staffing services exceeds the adjusted net profit in the preceding year. In addition, Mr. Wilson is entitled to receive: (i) commissions based upon the revenues generated from national account sales by Mr. Wilson, not to exceed 3% of such sales and (ii) 10% of any finder's fees or commissions payable by the Company to a third party in connection with any acquisition in which Mr. Wilson provides assistance. The agreement expires September 25, 2000. In the event his employment is terminated without cause, Mr. Wilson's annual salary and bonus opportunity will continue until the earlier to occur of one year from the date of termination or the expiration of the term of the agreement. In addition, Mr. Wilson has agreed not to compete with the Company until the later of March 25, 2002, three years after termination of his employment or three years after expiration of the agreement.

Mr. Michael Klein has entered into an employment agreement with the Company which provides for an annual base salary of \$175,000 and entitles Mr. Klein to a percentage of certain revenues derived from specified

40

clients, a percentage of court reporting fees billed by Mr. Klein and a cash bonus based on the increase in the annual net profits of Klein Bury. Under such bonus, Mr. Klein will receive 10% of the amount, if any, by which the annual net profits of Klein Bury exceed the prior year's net profits. In addition, Mr. Klein is entitled to receive (i) 85% of the amount billed on any eight depositions per month transcribed by him individually in his capacity as a court reporter and (ii) a percentage of any non Florida-based revenue primarily attributable to Mr. Klein's promotional efforts. The agreement expires on January 17, 2000. In the event his employment is terminated without cause, Mr. Klein's annual salary, commissions and bonus opportunity will continue until the earlier to occur of the year from the date of termination or January 17, 2000. In addition, Mr. Klein has agreed not to compete with the Company within 50 miles of any office location operated by the Company until the last to occur of January 17, 2002, or three years after his employment under the agreement terminates.

STOCK OPTION PLANS

1997 Stock Incentive Plan

The Company's 1997 Stock Incentive Plan provides for the granting to eligible employees or directors of the Company and its subsidiaries of options to purchase shares of Common Stock, which may be incentive stock options within the meaning of Section 422(b) of the Internal Revenue Code or nonqualified options. The 1997 Stock Incentive Plan is administered by the Compensation Committee of the Board of Directors, which designates option recipients, the number and type of options granted and their terms and conditions, including the exercise price and vesting schedule. A total of 750,000 shares of Common Stock have been reserved for issuance pursuant to options granted under the 1997 Stock Incentive Plan. Options to purchase 158,915 shares of Common Stock had been granted as of September 30, 1997, with exercise prices ranging from \$6.41 to \$10.20, and options to purchase an additional 565,109 shares of Common Stock will be granted upon completion of the Offering under the 1997 Stock Incentive Plan with exercise prices equal to the initial public offering price set forth on the cover of this Prospectus. All options granted under the 1997 Stock Incentive Plan vest 20% annually, are fully vested on the fifth anniversary of the date of grant and expire 10 years after the date of grant.

No options were granted pursuant to the Company's 1997 Stock Incentive Plan in 1996. The Board of Directors and shareholders have approved the 1997 Stock Incentive Plan. The Company has granted to Messrs. Looney, Tusa, Wilson and Creasman options to purchase 200,000, 150,000, 25,000 and 40,000 shares of Common Stock, respectively, at an exercise price per share equal to the initial public offering price set forth on the cover of this Prospectus.

Messrs. Looney, Tusa, Wilson and Creasman hold no other options to purchase Common Stock. The Company has not granted options to any other Named Executive Officer.

Stock Option Plan for Non-Employee Directors

The Company has adopted the U.S. Legal Support, Inc. Stock Option Plan for Non-Employee Directors (the "Directors' Stock Option Plan"). A total of 150,000 shares of Common Stock have been reserved for issuance under the Directors' Stock Option Plan, which provides for the grant of options to purchase 25,000 shares of Common Stock with an exercise price equal to the fair market value on the date of grant, to each incumbent director and to each person who becomes a director concurrently with his or her first election to

the Board. Options granted under the Directors' Stock Option Plan vest 20% annually and are fully vested on the fifth anniversary of the date of grant. Upon completion of the Offering, each non-employee director of the Company will be awarded options to purchase 25,000 shares of Common Stock.

LIMITATIONS ON DIRECTORS' LIABILITIES AND INDEMNIFICATION

Pursuant to the Company's Articles of Incorporation, as amended, and under Texas law, the directors of the Company are not liable to the Company or its shareholders for monetary damages for breach of fiduciary duty, except for liability in connection with a breach of duty of loyalty, for acts or omissions not in good faith which involve intentional misconduct or a knowing violation of law, for unlawful dividend payments or stock repurchases or any transaction in which a director has derived an improper personal benefit. The Company intends to maintain liability insurance for the benefit of its directors and its officers.

41

CERTAIN TRANSACTIONS

In connection with the initial capitalization of the Company on October 2, 1996, the Company issued 843,840 shares of Common Stock to Mr. Looney, the Chairman of the Board, President and Chief Executive Officer of the Company, in exchange for services rendered to the Company by Mr. Looney, which were valued by the Company at \$8,438.40, the aggregate par value of the shares issued. In January 1997, the Company and Looney entered into a business combination in which the Company acquired the capital stock of Looney for consideration consisting of \$4,087,834 in cash and 2,046,667 shares of Series B Preferred Stock.

On January 17, 1997, the Company acquired all of the outstanding capital stock of Klein Bury from Mr. Klein who became a director of the Company upon consummation of the sale. The purchase price consisted of: (i) \$3,850,397 in cash; (ii) 170,600 shares of Common Stock issued to Mr. Klein; (iii) a Subordinated Promissory Note issued by a subsidiary of the Company in the adjusted principal amount of \$1,424,113, which bears interest at the rate of 10% annually and is due on February 1, 2002; and (iv) options to purchase 40,000 shares of Common Stock granted to various employees and independent contractors of Klein Bury. At the option of Mr. Klein, the payment of the principal and all accrued interest on the note may be accelerated at any time after the Offering. The Company expects to repay the principal and accrued interest on the note with a portion of the proceeds of the Offering. In addition, Mr. Klein may receive contingent consideration equal to 50% of the accounts receivable of Klein Bury that were 120 days past due as of January 16, 1997 but which are collected by the Company. Such contingent consideration is estimated by the Company to be \$500,000, the contractual maximum. In addition, since 1993, Klein Bury has leased its offices in Fort Lauderdale, Florida from Mr. Klein, his wife and Richard Bury. The lease expires on March 31, 1998, and provides for annual rental of approximately \$50,400. The Company believes that the terms of the lease are no less favorable to the Company than would be available under a similar lease entered into with an unaffiliated third party.

On January 17, 1997, the Company sold \$9,000,000 principal amount of Senior Subordinated Notes and 1,000,000 shares of Series A Preferred Stock to three investors for an aggregate purchase price of \$10,000,000. The Company used the proceeds from the sale to finance a portion of the purchase price for the acquisition of Looney and Klein Bury. Mr. Robert Cresci, a director of the Company, is a Managing Partner of Pecks Management Partners Ltd., which provides investment advisory services to each of the investors. Pecks received a \$35,000 structuring fee paid by the Company in connection with the transaction. The Senior Subordinated Notes bear interest at an annual interest rate of 12% and will be due and payable on January 27, 2004, subject to mandatory prepayment two days following the completion of the Offering. The Company intends to repay the Senior Subordinated Notes with a portion of the proceeds of the Offering. The Series A Convertible Preferred Stock will be converted into a total of 1,560,000 shares of Common Stock upon completion of the Offering. The Company and the investors also entered into a Registration Rights Agreement pursuant to which the Company has agreed in certain circumstances to register the shares of Common Stock issued on conversion of the Series A Preferred Stock. See Note 8 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc. and "Shares Eligible For Future Sale--Registration Rights."

On August 29, 1997, the Company acquired substantially all of the assets of Legal Enterprise. Mr. Maddocks, who serves as Vice President, Sales and Marketing of the Company, was the President and a 50% shareholder of Legal Enterprise. The purchase price consisted of: (i) \$1,200,000 in cash, \$1,189,169 as adjusted; (ii) a Subordinated Promissory Note in the principal amount of \$319,340, \$316,458, as adjusted; (iii) a Convertible Subordinated Promissory Note in the principal amount of \$821,160, \$813,748, as adjusted; and (iv) options to purchase 7,000 shares of Common Stock to an employee of

Legal Enterprise. The promissory notes bear interest at a rate of 6.375% per annum, are payable on August 31, 2005 and were issued by a subsidiary of the Company. At the option of Legal Enterprise, the payment of the principal and all interest accrued on the Subordinated Promissory Note may be accelerated at any time after the Offering. The Company expects to repay the principal and accrued interest on the Subordinated Promissory Note with a portion of the proceeds of the Offering. Concurrently with the Offering, the Convertible Subordinated Promissory Note will be converted into 95,735 shares of Common Stock at a conversion rate of \$8.50 per share of Common Stock. Any accrued but unpaid interest on such note will be paid to Legal Enterprise, in cash, upon the completion of the Offering. In addition, Legal Enterprise may receive additional consideration (the "Additional Consideration") based on an amount

42

equal to the excess of six times earnings before taxes, depreciation and amortization of the acquired business over the initial purchase price, subject to adjustment on February 28, 2000. Sixty-five percent of the Additional Consideration is payable in cash and thirty-five percent is payable in shares of Common Stock valued at the average trading price per share of Common Stock over the first ten business days after the end of each period in which Additional Consideration is calculated. Interim payments of the Additional Consideration are to be made every six months until the aggregate amount of such payments is equal to \$934,000. Thereafter, no additional payments will be made until April 15, 2000, at which time any Additional Consideration owed is required to be paid to Legal Enterprise.

On September 4, 1997, the Company acquired substantially all of the non-cash assets of Amicus One, a holder of more than 5% of the currently outstanding Common Stock. The purchase price consisted of: (i) \$1.9 million in cash; (ii) 116,471 shares of Common Stock; and (iii) a Subordinated Promissory Note issued by a subsidiary of the Company in the principal amount of \$560,000, which bears interest at the rate of 6% annually, and is due on July 1, 2002. The Company expects to repay the principal and accrued interest on the note with a portion of the proceeds of the Offering. At the option of Amicus One, the payment of the principal and all accrued interest on the note may be accelerated concurrently with the Offering.

On September 17, 1997, the Company acquired all of the outstanding capital stock of Burton House, Inc, doing business as Ziskind Greene, from Gregg Ziskind and Susan Ziskind. The purchase price consisted of \$1.4 million in cash and 158,824 shares of Common Stock. In addition, the Ziskinds may receive additional consideration beginning on January 1, 1998, in the amount of 20% of the amount by which earnings before taxes, depreciation and amortization of Burton House, Inc. exceeds threshhold amounts ranging from \$150,000 for the period from the date of the acquisition through December 31, 1997, to \$750,000 for the year ended December 31, 2000.

The Company has entered into Registration Rights Agreements with Messrs. Looney, Maddocks, Klein, Ziskind and Susan Ziskind and Amicus One, pursuant to which the Company has agreed to include shares of Common Stock held by them in any registration of securities effected by the Company, subject to certain customary provisions restricting the number of shares to be included.

The Company and James M. Wilson entered into an Asset Purchase Agreement dated September 25, 1997, and on that date Mr. Wilson was appointed Vice President, Legal Staffing of the Company. Under the Asset Purchase Agreement, the Company will acquire from Mr. Wilson the assets of Commander Wilson, a sole proprietorship, in exchange for approximately \$1.4 million in cash and 56,250 shares of Common Stock, valued at the initial public offering price. The Company and Mr. Wilson entered into a letter agreement dated May 7, 1997, pursuant to which Mr. Wilson assisted the Company in identifying acquisition candidates in the legal placement and staffing industry. Pursuant to this agreement, the Company paid Mr. Wilson a non-refundable retainer of \$50,000 and reimbursed him for certain expenses. The letter agreement will be terminated upon completion of the Company's acquisition of Commander Wilson. See Note 6 of Notes to Financial Statements of Commander Wilson.

In October 1996, the Company issued 150,000 shares of its Common Stock to GulfStar Investments, Ltd., an affiliated partnership of Gulfstar, in exchange for investment banking services rendered to the Company by GulfStar. GulfStar has provided merger and acquisition advisory services to the Company since its inception. Mr. Kahle, a director of the Company, is a Managing Director of GulfStar. In addition, the Company paid investment banking fees aggregating \$475,000 to GulfStar for services in connection with the placement of the Senior Subordinated Notes and Series A Convertible Preferred Stock, negotiation of the Company's Bank Credit Agreement, and the acquisitions of Looney and Klein Bury in January 1997. Pursuant to the terms of a letter agreement dated April 24, 1997 (the "Letter Agreement") between GulfStar and the Company, GulfStar has agreed to provide negotiation and other financial advisory services to the Company in connection with the Company's evaluation of acquisitions and will be paid advisory fees equal to 1.0% of the total purchase price for each acquisition, as well as reimbursement of out-of-pocket expenses.

The fees payable to GulfStar under the Letter Agreement in connection with the Completed Acquisitions and the Pending Acquisitions will be \$480,000, of which \$33,000 has been paid.

NationsBank of Texas, N.A. has agreed to act as arranger, syndication agent and a lender in connection with the Company's \$40.0 million credit facility.

43

PRINCIPAL SHAREHOLDERS

The following table sets forth, as of November 30, 1997, certain information with respect to the ownership of shares of Common Stock, before and after the Offering and the Pending Acquisitions, by: (i) each person who is or is expected to be the beneficial owner of 5% or more of the outstanding shares of Common Stock upon consummation of the Offering; (ii) each person who is or is expected to become a director upon completion of the Offering; (iii) each Named Executive Officer; (iv) the Selling Shareholder; and (v) all officers and directors of the Company as a group. All persons listed have an address in care of the Company's principal executive offices at 1001 Fannin Street, Suite 650, Houston, Texas 77002, and have sole voting and investment power with respect to shares beneficially owned by them, unless otherwise noted. Information set forth in the table with respect to beneficial ownership of the Company's equity securities has been provided to the Company by such holders.

<TABLE> <CAPTION>

		PERCENTA(
NAME AND ADDRESS OF BENEFICIAL OWNER	SHARES	BEFORE	AFTER
<\$>	<c></c>	<c></c>	<c></c>
Richard O. Looney (1)			13.1%
David P. Tusa (2)		5.6	1.3
Scott R. Creasman (5)			
Michael A. Klein	170,600	9.8	2.2
Fentress Bracewell (6)	 1,560,000 150,000		20.0 1.9
Ronald C. Lassiter (6) Delaware State Employees' Retirement Fund (9) Declaration of Trust for Defined Benefit Plan of	1,045,200		13.4
ICI American Holdings Inc. (10) Declaration of Trust for Defined Benefit Plan of	304,200	14.9	3.9
Zeneca Holdings Inc. (11)	210,600	10.8	2.7
GulfStar Investments, Ltd. (12)	150,000	8.6	1.9
Gregg M. and Susan L. Ziskind (13)	158,824	9.2	2.0
Legal Enterprise, Inc. (14)	95,735	5.2	1.2
Amicus One (15)	116,471	6.7	1.5
(10 persons) (16)	3,009,808	80.8	38.5

- * Beneficially owns less than 1% of the outstanding shares of Common Stock.
- (1) Includes (a) 183,393 shares of Common Stock issuable upon conversion of the Series B Convertible Preferred Stock; (b) 90,000 shares of Common Stock held by trusts for the benefit of Mr. Looney's children; and (c) 10,000 shares subject to a three-year option granted by Mr. Looney to an officer of Gulfstar. Excludes 200,000 shares of Common Stock subject to unvested options granted pursuant to the Company's 1997 Stock Incentive Plan. Mr. Looney has granted the Underwriters a 30-day option to purchase up to 100,000 shares of Common Stock solely to cover any over-allotments. If this option is exercised in full, Mr. Looney would hold 927,233 shares of Common Stock, representing 12.0% of the outstanding Common Stock after the Offering.
- (2) Excludes 150,000 shares of Common Stock subject to unvested options granted pursuant to the Company's 1997 Stock Incentive Plan.
- (3) Includes 95,735 shares of Common Stock issuable upon conversion of a Convertible Subordinated Promissory Note held by Legal Enterprise, a corporation of which Mr. Maddocks is President and a 50% shareholder.

44

(4) Mr. Wilson owns no shares directly. Excludes 56,250 shares of Common Stock issuable as a portion of the purchase price in a Pending Acquisition and 25,000 shares of Common Stock subject to unvested options granted pursuant to the Company 1997 Stock Incentive Plan to be awarded upon completion of the Offering.

- (5) Excludes 40,000 shares of Common Stock subject to unvested options granted pursuant to the Company's 1997 Stock Incentive Plan.
- (6) Excludes 25,000 shares of Common Stock subject to unvested options granted pursuant to the Directors' Stock Option Plan to be awarded upon completion of the Offering.
- (7) Mr. Cresci owns no shares directly. Includes 1,560,000 shares issuable upon conversion of an aggregate of 1,000,000 shares of Series A Preferred Stock, which is held by three pension or defined benefit plans for whom Pecks Management Partners, Ltd. provides investment management services. Mr. Cresci is a Managing Director of Pecks Management Partners, Ltd. and therefore may be deemed to be a beneficial owner of such shares. Mr. Cresci disclaims beneficial ownership of all such shares. Excludes 25,000 shares of Common Stock subject to unvested options granted pursuant to the Directors' Stock Option Plan to be awarded upon completion of the Offering. The shareholders' addresses of record is c/o Pecks Management Partners Ltd., One Rockefeller Plaza, New York, New York 10020.
- (8) Mr. Kahle owns no shares directly. Includes 150,000 shares held by GulfStar Investments, Ltd., an affiliate of GulfStar. Mr. Kahle serves as a Managing Director of GulfStar. Excludes 25,000 shares of Common Stock subject to unvested options to be awarded upon completion of the Offering pursuant to the Company's Directors' Stock Option Plan.
- (9) Represents shares of Common Stock issuable upon conversion of 670,000 shares of Series A Preferred Stock. The shareholder's address of record is c/o Pecks Management Partners Ltd., One Rockefeller Plaza, New York, New York 10020.
- (10) Represents shares of Common Stock issuable upon conversion of 195,000 shares of Series A Preferred Stock. The shareholder's address of record is c/o Pecks Management Partners Ltd., One Rockefeller Plaza, New York, New York 10020.
- (11) Represents shares of Common Stock issuable upon conversion of 135,000 shares of Series A Preferred Stock. The shareholder's address of record is c/o Pecks Management Partners Ltd., One Rockefeller Plaza, New York, New York 10020.
- (12) The shareholder's address of record is 700 Louisiana Street, Suite 3800, Houston, Texas 77002.
- (13) The shareholder's address of record is 2666 Overland Avenue, Suite 600, Los Angeles, California 90064.
- (14) Represents shares of Common Stock issuable upon conversion of a Convertible Subordinated Promissory Note. The shareholder's address of record is 4232-1 Las Virgenes Road, Suite 100, Calabasas, California 91302.
- (15) The shareholder's address of record is 20 Vesey Street, 9th Floor, New York New York 10007.
- (16) Excludes an aggregate of 415,000 shares of Common Stock subject to unvested options granted pursuant to the Company's 1997 Stock Incentive Plan and 100,000 shares subject to unvested options granted pursuant to the Company's Directors' Option Plan.

45

DESCRIPTION OF CAPITAL STOCK

GENERAL

Under the Company's Articles of Incorporation, as amended (the "Articles"), the Company has authority to issue 110,000,000 shares of capital stock, consisting of 10,000,000 shares of Preferred Stock, par value \$1.00 per share (the "Preferred Stock") and 100,000,000 shares of Common Stock, par value \$.01 per share. As of November 15, 1997, the Company had outstanding 1,734,564 shares of Common Stock and 3,277,917 shares of Preferred Stock (1,000,000 shares of Series A Preferred Stock, 2,046,667 shares of Series B Preferred Stock and 231,250 shares of Series C Preferred Stock). All of the outstanding Preferred Stock will be converted into Common Stock upon completion of the Offering or will be redeemed with the net proceeds of the Offering. See Note 8 of Notes to Consolidated Financial Statements of U.S. Legal Support, Inc. for a description of the terms of each of the outstanding series of Preferred Stock. Shares of Preferred Stock that are redeemed will return to authorized shares of Preferred Stock undesignated as to series.

The following summary description of the capital stock of the Company is intended as a summary only and is qualified in its entirety by reference to the Articles, a copy of which has been filed as an exhibit to this Registration Statement.

PREFERRED STOCK

The Articles authorize the issuance of the Preferred Stock, in one or more series having designations, rights and preferences determined from time to time by the Board of Directors. One of the effects of undesignated Preferred Stock may be to enable the Board of Directors, without approval of holders of Common Stock, to issue Preferred Stock with dividends, liquidation, conversion, voting or other rights that could adversely affect the voting power or other rights of the holders of the Common Stock. In the event of

issuance, the Preferred Stock could be used, under certain circumstances, as a method of discouraging, delaying or preventing a change in control of the Company. Although the Company has no present intention to issue any additional shares of its Preferred Stock, there can be no assurance that it will not do so in the future.

COMMON STOCK

Voting Rights. Holders of Common Stock are entitled to one vote for each share on all matters on which shareholders generally are entitled to vote, including elections of directors. The Articles do not provide for cumulative voting for the election of directors; therefore, the holders of a majority of the voting power of the total number of outstanding shares of Common Stock are able to elect the entire Board of Directors of the Company. Holders of Common Stock have no preemptive, subscription, redemption or conversion rights.

Dividends. Subject to the preferential rights of any outstanding Preferred Stock created by the Board of Directors under the Articles, dividends may be paid to holders of Common Stock when, as and if declared by the Board of Directors out of funds legally available for such purpose. Dividends may be paid by the Company out of "surplus" (as defined under Article 1.02 of the Texas Business Corporation Act) or, if there is no surplus, out of net profits for the fiscal year in which the dividends are declared and/or the preceding fiscal year. Further, dividends may be paid out of any net profits for the current and/or prior fiscal year, if any. The declaration and payment of dividends on Common Stock could be restricted by the terms of any Preferred

Liquidation. In the event of the dissolution or winding up of the Company, after payment or provision for payment of debts and other liabilities of the Company and any other series or class of the Company's securities that rank senior to the Common Stock, the holders of Common Stock will be entitled to share ratably in all remaining assets of the Company. All outstanding shares of Common Stock are, and the shares of Common Stock to be sold by the Company in this Offering will be, duly and validly issued, fully paid and nonassessable.

The transfer agent and registrar for the Common Stock is Harris Trust and Savings Bank, Houston, Texas.

46

STATUTORY BUSINESS COMBINATION PROVISION

The Company is subject to Article 13 of the TBCA ("Article 13") which, with certain exceptions, prohibits a Texas corporation from engaging in a "business combination" (as defined in Article 13) with any shareholder who is a beneficial owner of 20% or more of the corporation's voting power for a period of three years after such shareholder's acquisition of a 20% ownership, unless: (i) the board of directors of the corporation approves the transaction or the shareholder's acquisition of shares prior to the acquisition or (ii) two-thirds of the unaffiliated shareholders of the corporation approve the transaction at a shareholders' meeting held no earlier than six months after the shareholder acquires that ownership. Shares that are issuable pursuant to options, conversion or exchange rights or other agreements are not considered outstanding for purposes of Article 13.

47

SHARES ELIGIBLE FOR FUTURE SALE

Upon completion of the Offering, the Company will have outstanding approximately 7,813,115 shares of Common Stock (8,238,115 shares if the Underwriters' over-allotment option is exercised in full). Of these shares, the 3,500,000 shares (4,025,000 shares if the Underwriters' over-allotment option is exercised in full) sold in the Offering will be freely tradable in the public market without restriction or limitation under the Securities Act, except for any shares purchased by an "affiliate" (as defined in the Securities Act) of the Company. The remaining 4,313,115 shares of Common Stock held by existing shareholders of the Company are "restricted securities" within the meaning of Rule 144 promulgated under the Securities Act of 1933, as amended.

The Company's directors, executive officers and certain shareholders, who hold an aggregate of 3,285,080 shares of Common Stock, have entered into lock-up agreements with the Representatives of the Underwriters. These persons have agreed not to offer, sell, contract to sell, grant any option with respect to, pledge, hypothecate or otherwise dispose of, any shares of Common Stock owned by them until the date occurring 180 days after the date of this Prospectus without the prior written consent of the Representatives. All such shares will become available for sale upon expiration of these lock-up agreements, subject

to compliance with Rule 144 promulgated under the Securities Act. In addition, certain shareholders have the right to have the shares of Common Stock owned by them registered by the Company under the Securities Act as described below.

In general, under Rule 144, as currently in effect, a person (or persons whose shares are required to be aggregated) who has beneficially owned, for a least one year, shares of Common Stock that have not been registered under the Securities Act or that were acquired from an "affiliate" of the Company is entitled to sell within any three-month period the number of shares of Common Stock which does not exceed the greater of one percent of the number of the then outstanding shares or the average weekly reported trading volume during the four calendar weeks preceding the sale. Sales under Rule 144 are also subject to certain notice requirements and to the availability of current public information about the Company and must be made in unsolicited brokers' transactions or to a market maker. A person (or persons whose shares are aggregated) who is not an "affiliate" of the Company under the Securities Act during the three months preceding a sale and who has beneficially owned such shares for at least two years is entitled to sell such shares under Rule 144 without regard to the volume and notice provisions of such Rule. Commencing 90 days after the completion of the Offering, approximately 2,920,000 "restricted" shares of Common Stock will be eligible for resale pursuant to Rule 144, subject to the volume, manner of sale and other limitations thereof. The remaining "restricted" shares will become eligible for resale pursuant to Rule 144 from time to time thereafter.

On the date of this Prospectus, the Company had outstanding options to purchase 649,024 shares of Common Stock granted pursuant to the 1997 Stock Incentive Plan. In addition, the Company intends to grant options to purchase 100,000 shares of Common Stock pursuant to the Company's Stock Option Plan for Non-Employee Directors upon completion of the Offering. Options to purchase at least an additional 150,000 shares of Common Stock are available for grant under the 1997 Stock Incentive Plan and the Non Employee Directors Stock Option Plan. The Company expects to file a registration statement on Form S-8 under the Securities Act to register all of the 900,000 shares of Common Stock issuable upon exercise of options granted under the 1997 Stock Incentive Plan and the Non Employee Directors Stock Option Plan. Accordingly, such shares will be freely tradeable by holders who are not affiliates of the Company and, subject to the volume and manner of sale limitations of Rule 144, by holders who are affiliates of the Company. In addition, on the date of this Prospectus, the Company had outstanding options to purchase 131,856 shares of Common Stock granted in connection with the Completed Acquisitions.

Prior to this Offering, there has been no market for the Common Stock. No predictions can be made of the effect, if any, that market sales of shares of Common Stock or the availability of such shares for sale will have on the market price prevailing from time to time. Nevertheless, sales of significant amounts of Common Stock could adversely affect the prevailing market price of Common Stock, as well as impair the ability of the Company to raise capital through the issuance of additional equity securities.

4.8

REGISTRATION RIGHTS

Pursuant to several Registration Rights Agreements (the "Registration Rights Agreements"), the Company has agreed to register under the Securities Act substantially all of the shares of Common Stock outstanding on the date of this Prospectus (approximately 1,734,564 shares) as well as the 183,393 shares of Common Stock issuable upon conversion of the Series B Preferred Stock, and will enter similar agreements with respect to 609,268 shares of Common Stock to be issued in the Pending Acquisitions. Pursuant to the Registration Rights Agreements, shareholders and their permitted transferees have certain "piggyback" registration rights and will be entitled, subject to certain limitations, to include their shares of Common Stock in a registration of shares of Common Stock by the Company under the Securities Act.

In addition, a Registration Rights Agreement with the holders of the Series A Convertible Preferred Stock provides that following the Offering, any one or more of such shareholders shall twice have the right to require the Company to effect registration of all or any part of the shareholders' shares of Common Stock under the Securities Act. In order to demand registration of the Common Stock, the holder or holders of Common Stock requesting such registration must own more than 50% (by number of shares) of the shares subject to the agreement and the aggregate market value to be registered must be at least \$3.0 million. The number of shares included in any registration is subject to customary provisions providing for a reduction in the number of shares to be registered if in the opinion of the managing underwriter such shares would affect the marketing or the selling price of the securities to be sold.

The Registration Rights Agreements require the Company to pay the expenses associated with any registration other than sales discounts, commissions, transfer taxes and amounts to be borne by underwriters or as otherwise required by law and include customary provisions for indemnification against

49

UNDERWRITING

The underwriters named below (the "Underwriters"), have severally agreed, subject to the terms and conditions in the underwriting agreement (the "Underwriting Agreement") by and between the Company and the Underwriters, to purchase from the Company the number of shares of Common Stock indicated below opposite their respective names, at the initial public offering price less the underwriting discount set forth on the cover page of this Prospectus. The Underwriting Agreement provides that the obligations of the Underwriters are subject to certain conditions precedent and that the Underwriters are committed to purchase all of the shares of Common Stock, if they purchase any.

<TABLE>

UNDERWRITERS	OF SHARES
<\$>	<c></c>
NationsBanc Montgomery Securities, Inc	
Total	3,500,000

NUMBER

</TABLE>

The Underwriters have advised the Company that the Underwriters propose initially to offer the shares of Common Stock to the public on the terms set forth on the cover page of this Prospectus. The Underwriters may allow selected dealers a concession of not more that \$ per share; and the Underwriters may allow to selected dealers, and such dealers may reallow, a concession of not more than \$ per share to certain other dealers. After the Offering, the public offering price and other selling terms may be changed by the Underwriters. The Common Stock is offered subject to receipt and acceptance by the Underwriters, and to certain other conditions, including the right to reject orders in whole or in part.

The Company and a shareholder have granted to the Underwriters an option, exercisable for 30 days from the date of this Prospectus, to purchase up to a maximum of 525,000 additional shares of Common Stock to cover over-allotments, if any, at the same price per share as the initial 3,500,000 shares to be purchased by the Underwriters. To the extent that the Underwriters exercise this option, each of the Underwriters will be committed, subject to certain conditions, to purchase such additional shares in approximately the same proportion as set forth in the above table. The Underwriters may purchase such shares only to cover over-allotments made in connection with the Offering.

The Underwriting Agreement provides that the Company will indemnify the Underwriters against certain liabilities, including civil liabilities under the Securities Act of 1933, as amended, or will contribute to payments the Underwriters may be required to make in respect thereof.

Certain shareholders of the Company and the Company's executive officers and directors have agreed that for a period of 180 days after the date of this Prospectus they will not, without the prior written consent of NationsBanc Montgomery Securities, Inc., offer, sell, or otherwise dispose of any shares of Common Stock, options or warrants to acquire shares of Common Stock or securities exchangeable or exercisable for or convertible into shares of Common Stock. The Company has also agreed not to issue, offer, sell, grant options to purchase or otherwise dispose of any of the Company's equity securities for a period of 180 days after the date of this Prospectus without the prior written consent of NationsBanc Montgomery Securities, Inc., except for: (i) the issuance of shares of Common Stock upon conversion of debt or equity securities of the Company or any of its subsidiaries outstanding as of the date of this Prospectus; (ii) the grant, award, issuance or sale of shares of Common Stock or options or other rights to purchase or acquire Common Stock pursuant to the terms of any stock option, stock bonus or other plan or arrangement referred to in this Prospectus; or (iii) the issuance of shares of Common Stock, Preferred Stock or any security convertible into or exchangeable for Common Stock or Preferred Stock in connection with the acquisition of court reporting, certified record retrieval and legal staffing and placement businesses and related assets.

50

Certain persons participating in this Offering may overallot or effect transactions which stabilize, maintain or otherwise affect the market price of

the Common Stock at levels above those which might otherwise prevail in the open market. Such transactions may include stabilizing bids, effecting syndicate covering transactions or imposing penalty bids. A stabilizing bid means the placing of any bid or effecting any purchase for the purpose of pegging, fixing or maintaining the price of the Common Stock. A syndicate covering transaction means the placing of any bid on behalf of the underwriting syndicate or the effecting of any purchase to reduce a short position created in connection with the Offering. A penalty bid means an arrangement that permits the Underwriters to reclaim a selling concession from a syndicate member in connection with the Offering when shares of Common Stock sold by the syndicate member are purchased in syndicate covering transactions. Such transactions may be effected on the Nasdaq National Market, in the overthe-counter market, or otherwise.

The Underwriters have informed the Company that the Underwriters do not expect to make sales of Common Stock offered by this Prospectus to accounts over which they exercise discretionary authority in excess of 5% of the number of shares of Common Stock offered hereby.

Prior to the Offering, there has been no public trading market for the Common Stock. Consequently, the initial public offering price of the Common Stock has been determined by negotiations between the Company and the Underwriters. Among the factors considered in such negotiations were the results of operations of the businesses acquired in recent periods, the prospects for the Company and the industry in which the Company competes, an assessment of the Company's management, its financial condition, the prospects for future earnings of the Company, the present state of the Company's development, the general condition of the economy and the securities markets at the time of the Offering and the market prices of and demand for publicly traded common stock of comparable companies in recent periods.

LEGAL MATTERS

The legality of the Common Stock offered hereby will be passed upon for the Company by Bracewell & Patterson, L.L.P., Houston, Texas, and for the Underwriters by Locke Purnell Rain Harrell (A Professional Corporation), Dallas, Texas.

EXPERTS

The financial statements and schedules of U.S. Legal Support, Inc., Looney, Klein Bury, G&G, San Francisco Reporting, Legal Enterprise, Elaine Dine, Ziskind Greene, Jilio, Reporting Service, Kirby Kennedy, Johnson Group, Amicus One, Block and Commander Wilson included in this Prospectus and elsewhere in the Registration Statement, to the extent and for the periods indicated in their reports, have been audited by Coopers & Lybrand L.L.P., independent accountants, and are included herein in reliance upon the authority of said firm as experts in accounting and auditing. The report on the financial statements of U.S. Legal Support, Inc. includes an explanatory paragraph regarding the Company's ability to continue as a going concern.

51

AVAILABLE INFORMATION

The Company has filed with the Commission a Registration Statement (which term encompasses any and all amendments thereto) under the Securities Act with respect to the Common Stock offered hereby. This Prospectus, which is filed as part of the Registration Statement, does not contain all the information set forth in the Registration Statement and the exhibits and schedules thereto, certain items of which were omitted in accordance with the rules and regulations of the Commission. Statements made in this Prospectus concerning the contents of any contract, agreement or other document referred to are summaries of the terms of such contract, agreement or other document and are not necessarily complete. With respect to each such contract, agreement or other document filed as an exhibit to the Registration Statement, reference is hereby made to the exhibit for a more complete description of the matter involved, and each such statement shall be deemed qualified in its entirety by such reference. For further information with respect to the Company, reference is hereby made to the Registration Statement and such exhibits and schedules filed as a part thereof, which may be inspected, without charge, at the public reference facilities maintained by the Commission at Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549, and at the following regional offices of the Commission: 7 World Trade Center, Suite 1300, New York, New York 10048; and Citicorp Center, 500 West Madison Street, Suite 1400, Chicago, Illinois 60661. Copies of all or any portion of the Registration Statement may be obtained from the Public Reference facilities of the Commission, upon payment of the prescribed fees. The Registration Statement is also available on the Internet at the Commission's World Wide Web site at http://www.sec.gov.

As a result of the Offering, the Company will be subject to the reporting requirements under the Exchange Act and, in accordance therewith, will file reports, proxy statements, information statements and other information with

the Commission. The Company intends to furnish annual reports to its shareholders containing audited financial statements reported on by an independent certified public accounting firm.

52

INDEX TO FINANCIAL STATEMENTS

U.S. LEGAL SUPPORT, INC.

<TABLE> <CAPTION>

	PAGE
400	<c></c>
<s> U.S. LEGAL SUPPORT, INC.</s>	<c></c>
UNAUDITED PRO FORMA COMBINED FINANCIAL STATEMENTS:	
Basis of Presentation	F-5
Unaudited Pro Forma Combined Balance Sheet as of September 30, 1997	F-7
Notes to Unaudited Pro Forma Combined Balance Sheet	F-8
Unaudited Pro Forma Combined Statement of Income for the Nine Months Ended	
September 30, 1997	F-10
Unaudited Pro Forma Combined Statement of Income for the Nine Months Ended	
September 30, 1996	F-10
Unaudited Pro Forma Combined Statement of Income for the Year Ended	
December 31, 1996	F-11
Notes to Unaudited Pro Forma Combined Statement of Income	F-12
UNAUDITED HISTORICAL FINANCIAL STATEMENTS:	r-12
Report of Independent Accountants	F-13 F-14
Consolidated Statement of Income for the Nine Months Ended September 30,	F-14
1997 (Unaudited)	F-15
Consolidated Statement of Stockholders' Equity for the Nine Months Ended	1 10
September 30, 1997 (Unaudited)	F-16
Consolidated Statement of Cash Flows for the Nine Months Ended September	
30, 1997 (Unaudited)	F-17
Notes to Consolidated Financial Statements (Unaudited)	F-18
LOONEY & COMPANY	
Report of Independent Accountants	F-32
Balance Sheet as of December 31, 1995 and 1996	F-33
Statement of Operations for the Years Ended December 31, 1994, 1995 and	
1996 and for the Nine Months Ended September 30, 1996 (Unaudited)	F-34
Statement of Stockholder's Equity for the Years Ended December 31, 1994,	F-35
1995 and 1996	r-33
1996 and for the Nine Months Ended September 30, 1996 (Unaudited)	F-36
Notes to Financial Statements	F-37
KLEIN, BURY & ASSOCIATES	
Report of Independent Accountants	F-42
Balance Sheet as of September 30, 1995 and December 31, 1996	F-43
Statement of Income for the Years Ended September 30, 1995 and December	
31, 1996	F-44
Statement of Stockholders' Equity for the Years Ended September 30, 1995	
and December 31, 1996	F-45
Statement of Cash Flows for the Years Ended September 30, 1995 and	- 1C
December 31, 1996 Notes to Financial Statements	F-46 F-47
G&G COURT REPORTERS	F-4/
Report of Independent Accountants	F-50
Balance Sheet as of December 31, 1995 and 1996	F-51
Statement of Income for the Years Ended December 31, 1995 and 1996 and for	
the Six Months Ended June 30, 1996 and the Period from January 1, 1997	
through Date of Acquisition, May 19, 1997 (Unaudited)	F-52
Statement of Owner's Equity for the Years Ended December 31, 1995 and 1996	
and the Period from January 1, 1997 through Date of Acquisition, May 19,	
1997 (Unaudited)	F-53
Statements of Cash Flows for the Years Ended December 31, 1995 and 1996	
and the Period from January 1, 1997 through Date of Acquisition, May 19,	D F/
1997 (Unaudited)	F-54 F-55
	1 00
· · · · · · · · · · · · · · · · · · ·	

F-1

<TABLE> <CAPTION>

	PAGE
< \$>	<c></c>
SAN FRANCISCO REPORTING SERVICE	
Report of Independent Accountants	F-57

Balance Sheet as of December 31, 1996 and May 14, 1997	F-58
through Date of Acquisition, May 14, 1997	F-59
1997	F-60
Period from January 1, 1997 through Date of Acquisition, May 14, 1997	F-61
Notes to Financial Statements	F-62
Report of Independent Accountants	F-65
Balance Sheet as of December 31, 1996 and June 30, 1997 (Unaudited)	F-66
Statement of Income for the Years Ended December 31, 1995 and 1996 and the	F-00
Six Months Ended June 30, 1996 and 1997 (Unaudited)	F-67
Statement of Stockholders' Equity for the Years Ended December 31, 1995	1 07
and 1996 and the Six Months Ended June 30, 1996 and 1997 (Unaudited)	F-68
Statement of Cash Flows for the Years Ended December 31, 1995 and 1996 and	1 00
the Six Months Ended June 30, 1996 and 1997 (Unaudited)	F-69
Notes to Financial Statements	F-70
ELAINE P. DINE, INC.	
Report of Independent Accountants	F-72
Balance Sheet as of March 31, 1996 and 1997 and June 30, 1997	
(Unaudited)	F-73
Statement of Income for the Years Ended March 31, 1996 and 1997 and for	
the Three Months Ended June 30, 1996 and 1997 (Unaudited)	F-74
Statement of Stockholder's Equity for the Years Ended March 31, 1996 and	
1997 and the Three Months Ended June 30, 1997 (Unaudited)	F-75
Statement of Cash Flows for the Years Ended March 31, 1996 and 1997 and	
the Three Months Ended June 30, 1996 and 1997 (Unaudited)	F-76
Notes to Financial Statements	F-77
BURTON HOUSE, INC.	
D.B.A. ZISKIND, GREENE, WATANABE & NASON	
Report of Independent Accountants	F-79
Balance Sheet as of December 31, 1995 and 1996 and June 30, 1997	
(Unaudited)	F-80
Statement of Operations for the Years Ended December 31, 1995 and 1996 and	
the Six Months Ended June 30, 1996 and 1997 (Unaudited)	F-81
Statement of Stockholder's Deficit for the Years Ended December 31, 1995	E 00
and 1996 and the Six Months Ended June 30, 1997 (Unaudited)	F-82
the Six Months Ended June 30, 1996 and 1997 (Unaudited)	F-83
Notes to Financial Statements	F-83

 r-04 || / INDUE/ | |
| | |
F-2

<TABLE> <CAPTION>

PAGE <S> <C> JILIO & ASSOCIATES Report of Independent Accountants..... F-87 Balance Sheet as of December 31, 1995 and 1996 and September 30, 1997 (Unaudited)..... Statement of Operations for the Years Ended December 31, 1995 and 1996 and the Nine Months Ended September 30, 1996 and 1997 (Unaudited)...... F-89 Statement of Owner's Equity for the Years Ended December 31, 1995 and 1996 and the Nine Months Ended September 30, 1997 (Unaudited)..... F-90 Statement of Cash Flows for the Years Ended December 31, 1995 and 1996 and the Nine Months Ended September 30, 1996 and 1997 (Unaudited)...... F-91 Notes to Financial Statements..... F-92 REPORTING SERVICE ASSOCIATES, INC. Report of Independent Accountants.... F-95 Balance Sheet as of December 31, 1995 and 1996 and September 30, 1997 $\,$ F-96 (Unaudited)..... Statement of Income for the Years Ended December 31, 1995 and 1996 and F-97 the Nine Months Ended September 30, 1996 and 1997 (Unaudited)...... Statement of Stockholder's Equity for the Years Ended December 31, 1995 and 1996 and the Nine Months ended September 30, 1996 and 1997 F-98 (Unaudited)..... Statement of Cash Flows for the Years Ended December 31, 1995 and 1996 and the Nine Months Ended September 30, 1996 and 1997 (Unaudited)...... Notes to Financial Statements..... F-100 KIRBY A. KENNEDY & ASSOCIATES Report of Independent Accountants..... F-102 Balance Sheet as of December 31, 1995 and 1996 and September 30, 1997 $\,$ Statement of Income for the Years Ended December 31, 1995 and 1996 and the Nine Months Ended September 30, 1996 and 1997 (Unaudited)..... F-104 Statement of Partners' Capital for the Years Ended December 31, 1995 and

1996 and the Nine Months Ended September 30, 1997 (Unaudited)	F-105
Statement of Cash Flows for the Years Ended December 31, 1995 and 1996	
and the Nine Months Ended September 30, 1996 and 1997 (Unaudited)	F-106
Notes to Financial Statements	F-107
JOHNSON COURT REPORTING GROUP	
Report of Independent Accountants	F-109
Combined Balance Sheet as of December 31, 1995 and 1996 and June 30, 1997	
(Unaudited)	F-110
Combined Statement of Income for the Years Ended December 31, 1995 and	
1996 and for the Six Months Ended June 30, 1996 and 1997 (Unaudited)	F-111
Combined Statement of Shareholder's Equity for the Years Ended December	
31, 1995 and 1996 and for the Six Months Ended June 30, 1997	
(Unaudited)	F-112
Combined Statement of Cash Flows for the Years Ended December 31, 1995	
and 1996 and for the Six Months Ended June 30, 1996 and 1997	
(Unaudited)	F-113
Notes to Combined Financial Statements	F-114

 |F-3

<TABLE> <CAPTION>

	PAGE
<\$>	<c></c>
AMICUS ONE LEGAL SUPPORT SERVICES, INC.	107
Report of Independent Accountants	F-117
Balance Sheet as of December 31, 1996 and September 5, 1997	F-118
Statement of Income for the Year Ended December 31, 1996 and the Period	
Ended September 5, 1997	F-119
Statement of Stockholders' Equity for the Year Ended December 31, 1996	
and the Period Ended September 5, 1997	F-120
Statement of Cash Flows for the Year Ended December 31, 1996 and the	- 101
Period Ended September 5, 1997 Notes to Financial Statements	
BLOCK COURT REPORTING, INC.	F-122
Report of Independent Accountants	₽-125
Balance Sheet as of December 31, 1995 and 1996 and June 30, 1997	1 120
(Unaudited)	F-126
Statement of Operations for the Years Ended December 31, 1995 and 1996	
and the Six Months Ended June 30, 1997 and 1996 (Unaudited)	F-127
Statement of Stockholder's Equity (Deficit) for the Years Ended December	
31, 1995 and 1996 and the Six Months Ended June 30, 1997 (Unaudited)	F-128
Statement of Cash Flows for the Years Ended December 31, 1995 and 1996	
and the Six Months Ended June 30, 1997 and 1996 (Unaudited)	
Notes to Financial Statements	F-130
COMMANDER WILSON, INC.	E 100
Report of Independent Accountants	F-133
(Unaudited)	F-134
Statement of Income for the Years Ended December 31, 1995 and 1996 and	1 151
the Nine Months Ended September 30, 1996 and 1997 (Unaudited)	F-135
Statement of Owner's Deficit for the Years Ended December 31, 1995 and	
1996 and the Nine Months Ended September 30, 1997 (Unaudited)	F-136
Statement of Cash Flows for the Years Ended December 31, 1995 and 1996	
and the Nine Months Ended September 30, 1996 and 1997 (Unaudited)	F-137
Notes to Financial Statements	F-138

 |F-4

UNAUDITED PRO FORMA COMBINED FINANCIAL STATEMENTS

BASIS OF PRESENTATION

The following Unaudited Pro Forma Combined Balance Sheet as of September 30, 1997 and the Unaudited Pro Forma Combined Statements of Operations for the year ended December 31, 1996 and the nine months ended September 30, 1997 and 1996 have been prepared to reflect adjustments to the historical financial position and results of operations to give effect to the transactions described below. The Unaudited Pro Forma Combined Balance Sheet reflects such transactions as if they had occurred as of September 30, 1997, and the Unaudited Pro Forma Combined Statements of Operations for the year ended December 31, 1996 and the nine months ended September 30, 1997 and 1996 reflect such transactions as if they had occurred as of January 1, 1996.

U.S. Legal Support, Inc., a Texas Corporation ("U.S. Legal") was formed in October 1996 to continue the business of Looney & Company, a Texas Corporation, ("Looney") and to create a leading provider of legal support and

staffing services to law firms, insurance providers and major corporations. In January, U.S. Legal was combined with Looney in a transaction accounted for as a merger of entities under common control in which Looney became a whollyowned subsidiary of U.S. Legal (Looney and U.S. Legal are hereinafter referred to as the "Company"). Simultaneously, the Company entered into a Securities Purchase Agreement (the "Securities Purchase Agreement") with three investors represented by Pecks Management Partners Ltd. pursuant to which the Company issued 1,000,000 shares of Series A Convertible Preferred Stock (the "Series A Preferred Stock") and \$9.0 million principal amount of 12% Senior Subordinated Notes (the "Senior Subordinated Notes"). Concurrently, the Company entered into a Bank Credit Agreement with a commercial bank, which provided for a revolving credit facility of \$4.0 million. The Company amended the Bank Credit Agreement to, among other things, reduce the revolving credit facility to \$2.0 million and to provide for term loans up to \$14.0 million, for acquisitions. See Note 7 to the Company's financial statements. In October 1997, the Company increased the revolving line of credit to \$3.25 million through December 31, 1997.

On the same date as the transactions described above, the Company acquired all of the outstanding capital stock of Klein, Bury & Associates, Inc. ("Klein Bury") headquartered in Miami, Florida with offices in various locations throughout the state of Florida. In May 1997, the Company acquired certain net assets of G & G Court Reporters and San Francisco Reporting Service headquartered in Los Angeles and San Francisco, California, respectively.

In August 1997, the Company acquired all of the outstanding capital stock of Goren of Newport, Inc., Rapidtext, Inc. and Medtext, Inc. (the "Johnson Court Reporting Group") and acquired certain assets and assumed certain liabilities of Legal Enterprise, Inc. Both companies are headquartered in California.

In September 1997, the Company acquired the assets and assumed certain liabilities of Elaine P. Dine, Inc. and acquired all of the outstanding capital stock of Burton House, Inc. d.b.a. Ziskind Greene Watanabe & Nason headquartered in New York, New York and Los Angeles, California, respectively. Additionally, the Company acquired all of the outstanding capital stock of Block Court Reporting, Inc. and acquired certain assets and assumed certain liabilities of Amicus One Legal Support Services, Inc. headquartered in Washington, D.C. and New York, New York respectively.

The Company intends to sell approximately 3,500,000 shares of common stock, par value \$.01 per share (the "Common Stock") (at an assumed initial price of \$12.00 per share) to the public (the "Offering"). The businesses acquired prior to the Offering ("Completed Acquisitions") were acquired using a combination of cash, preferred stock, Common Stock and subordinated promissory notes. Upon completion of the Offering, all of the shares of preferred stock will be converted into shares of Common Stock, other than 231,250 shares of Series C Preferred Stock which will be redeemed for \$231,250 upon completion of the Offering. The aggregate consideration paid by the Company for the Completed Acquisitions consisted of: (i) \$21,400,000 in cash, (ii) 2,046,667 shares of Series B Convertible Preferred Stock (the "Series B Preferred Stock"); (iii) 231,250 shares of Series C Preferred Stock (the "Series C Preferred Stock"); (iv) \$4,979,000 of Subordinated Promissory Notes; and (v) \$1,845,000 Convertible Subordinated Promissory Notes.

F-5

Additionally, the Company has entered into definitive agreements with respect to the acquisition of Jilio & Associates, Kirby A. Kennedy & Associates, Reporting Service Associates, Inc. and Commander Wilson, Inc. (the "Pending Acquisitions," collectively with the Completed Acquisitions, the "Acquisitions"). All of the Pending Acquisitions are expected to close contemporaneously with, and are conditional upon, the closing of this Offering.

The Unaudited Pro Forma Combined Balance Sheet as of September 30, 1997 gives effect to: (i) the sale of 3,500,000 shares of Common Stock in the Offering and the application of the net proceeds therefrom, as described in "Use of Proceeds"; (ii) the conversion of all outstanding shares of Series A Preferred Stock and Series B Preferred Stock into shares of Common Stock; (iii) the conversion of \$1,845,000 of Convertible Subordinated Promissory Notes into shares of Common Stock; (iv) the redemption of 231,250 shares of Series C Preferred Stock for cash; and (v) the Acquisitions, as if each of such transactions had occurred as of September 30, 1997. The Unaudited Pro Forma Combined Statements of Operations for the year ended December 31, 1996, and the nine months ended September 30, 1996 and 1997, give effect to: (i) the sale of 3,500,000 shares of Common Stock in the Offering and the application of net proceeds therefrom as described in "Use of Proceeds"; (ii) the conversion of all outstanding shares of Series A and Series B Preferred Stock into shares of Common Stock; (iii) the conversion of \$1,845,000 principal amount of Convertible Subordinated Promissory Notes into Common Stock; and (iv) the Acquisitions, as if each of such transactions had occurred as of January 1, 1996.

The pro forma combined financial statements have been prepared by the Company based on the historical financial statements of the Company and the businesses acquired or to be acquired, the financial statements of which are included elsewhere in this Prospectus. These pro forma combined financial statements are presented for illustrative purposes only and are not necessarily indicative of the results that would have been obtained if the transactions had occurred on the dates indicated or that may be realized in the future. The Company is not aware of any known contingencies that could result in adjustments that would have a material effect on the unaudited pro forma financial statements. The pro forma information should be read in conjunction with the Consolidated Financial Statements of U.S. Legal Support, Inc. and the Notes thereto and the historical financial statements of the companies acquired or to be acquired and the notes thereto included elsewhere in this Prospectus.

F-6

U.S. LEGAL SUPPORT, INC.

UNAUDITED PRO FORMA COMBINED BALANCE SHEET

SEPTEMBER 30, 1997

(IN THOUSANDS)

<TABLE> <CAPTION>

<caption></caption>	HISTORICAL			HISTORICAL		
ASSETS	COMPANY (A)	, ,	AS ADJUSTED	PENDING ACQUISITIONS (C)	PENDING ACQUISITIONS ADJUSTMENTS (C)(D)	PRO FORMA
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Current assets:	(0)	(0)	107	(0)	(0)	107
Cash	\$559	\$6,600	\$7 , 159	\$103	\$(103) (5,921)	\$1 , 238
Accounts receivable Prepaid expenses and	6,990		6,990	2,119	(1,458)	7,651
other current assets	189		189	21	(3)	207
Deferred income taxes	48		48			48
Total current						
assets Property and equipment,	7,786	6,600	14,386	2,243	(7,485)	9,144
net	1,153		1,153	109		1,262
Intangibles	27 , 980		27 , 980		22 , 973 285	51,238
Other assets	1,801	(250) (543)	1,008			1,008
Total assets		\$5,807 ======	\$44 , 527	\$2,352	\$15,773 ======	\$62,652
<pre><caption> LIABILITIES AND STOCKHOLDERS' EQUITY</caption></pre>						
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Current liabilities: Accounts payable						
Trade	\$3,689	\$	\$3,689	\$395	\$(167)	\$3,917
Affiliates	316		316			316
Accrued liabilities	2,715		2,715	32	285	3,032
Income taxes payable			34			34
Deferred income taxes Current maturities of long-term						
obligations	3,936	(3,861)	75			75
liabilities						
Total current						
liabilities Long-term obligations, net of current	10,690	(3,861)	6,829	427	118	7,374
maturities	26,339	(26, 256)	83		11,000	11,083
Deferred income taxes Commitments and contingencies Redeemable preferred stock	255		255			255

Series A Convertible							
Preferred Stock	1,479	(1,479)					
Series B Convertible							
Preferred Stock	2,047	(2,047)					
Series C Convertible							
Preferred Stock	231	(231)					
Total redeemable							
preferred stock	3 , 757	(3,757)					
Stockholders' equity							
(deficit):							
Common stock	17	55	72	195	(189)	78	
Additional paid-in							
capital	3,840	41,311	45,151		6 , 574	51 , 725	
Retained earnings							
(deficit)	(6,178)	(1,685)	(7 , 863)	1,730	(1,730)	(7 , 863)	
Total stockholders'							
equity (deficit):	(2,321)	39,681	37,360	1,925	4,655	43,940	
Total liabilities and							
stockholders' equity	***	45.005	*** 505	40.050	445 550	460 650	
(deficit)	\$38,720	\$5,807	\$44,527	\$2,352	\$15 , 773	\$62,652	
/ map; E>	======	======	======	=====	======	======	

</TABLE>

F-7

U.S. LEGAL SUPPORT, INC.

NOTES TO UNAUDITED PRO FORMA COMBINED BALANCE SHEET

- (A) Represents the September 30, 1997 historical consolidated balance sheet of the Company.
- (B) Represents the issuance of 3,500,000 shares of Common Stock offered by the Company hereby at an assumed initial public offering price of \$12.00 per share and the use of proceeds therefrom as follows:

<TABLE> <CAPTION>

2 2 7 2 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	
	IN THOUSANDS
<\$>	<c></c>
Gross proceeds of the Offering	\$42,000
Underwriting discounts and commissions	(2,940)
Expenses related to the Offering	(2,500)
Net proceeds	36 , 560
Repayment of long-term debt, including current portion Redemption of 231,250 shares of Series C preferred stock at	(29 , 979)
\$1.00 per share	(231)
Expenses paid at September 30, 1997 related to Offering	250
Net increase in cash and cash equivalents	\$6 , 600
	======

</TABLE>

In addition, upon completion of the Offering, \$1,845,000 of Subordinated Convertible Promissory Notes, the 1,000,000 shares of Series A Preferred Stock and the 2,046,667 shares of the Series B Preferred Stock will be converted into shares of Common Stock. The principal amount of Subordinated Convertible Promissory Notes convert at prices ranging from \$7.56 to \$8.50 per share. The 1,000,000 shares of Series A Preferred Stock convert into 1,560,000 shares of common stock. The Series B Preferred Stock converts at 93% of the assumed initial public offering price and the Company anticipates recording a \$154,000 noncash dividend, that reflects the discount from the assumed initial offering price at which it will convert to Common Stock.

The Offering and the conversion of the Convertible Subordinated Promissory Notes, Series A Preferred Stock and Series B Preferred Stock will affect the pro forma equity, as follows (in thousands):

<TABLE>

		ADDITIONAL	
	COMMON	PAID-IN-	
	STOCK	CAPITAL	TOTAL
<\$>	<c></c>	<c></c>	<c></c>
Offering	\$35	\$36 , 525	\$36,560
Subordinated Convertible Promissory Notes	2	1,278	1,280
Series A Preferred Stock	16	1,463	1,479

A DD TELTONIA T

	===	======	======
	\$55	\$41,311	\$41,366
Series B Preferred Stock	2	2,045	2,047

</TABLE>

The net repayment of long-term obligations is calculated as follows:

<CAPTION>

110117	
	IN THOUSANDS
<\$>	<c></c>
Other assets	\$(543)(a)
Current portion of long-term obligations	3,861
Long-term debt, net of current maturities	26,256
Common Stock	(2)
Additional paid-in capital	(1,278)
Retained earnings	1,685 (b)
Cash paid	\$29,979
	======

</TABLE>

- (a) To write off deferred financing costs related to the retirement of the Senior Subordinated Notes.
- (b) To reflect the reduction in retained earnings for extraordinary losses on the retirement of indebtedness repaid with cash.

F-8

U.S. LEGAL SUPPORT, INC.

NOTES TO UNAUDITED PRO FORMA COMBINED BALANCE SHEET-- (CONTINUED)

(C) Represents the September 30, 1997 historical combined balance sheets of businesses to be acquired in the Pending Acquisitions, and the purchase adjustments thereto.

The estimated fair value of the assets to be acquired in the Pending Acquisitions is summarized below:

<TABLE>

<CAPTION>

	TOTAL	JILIO	KENNEDY	SERVICE	WILSON
		I	N THOUSAN	IDS	
<\$>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Accounts receivables, net Prepaid expenses and other current	\$661	\$ 404	\$ 257	\$	\$
assets	18	18			
Property and equipment Excess of cost over fair value of net assets acquired, including goodwill of \$22,933 and covenants	109	25	42	42	
not to compete of \$40 Accounts payable and accrued	22 , 973	7,648	3,442	9,858	2,025
liabilities	(260)	(95)	(165)		
(//////////////////////////////////////	\$23,501 =====	\$8,000 =====	\$3,576 =====	\$9,900 =====	\$2,025 =====

</TABLE>

The adjustments also reflect the elimination of certain assets and liabilities not acquired or assumed.

The consideration for the Pending Acquisitions will be as follows:

<TABLE> <CAPTION>

	TOTAL	JILIO		SERVICE	WILSON
			IN THOU	SANDS	
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Cash (includes \$5,921 from Offering proceeds and \$11,000 from new					
borrowings)	\$16,921	\$5,600	\$2,503	\$7,400	\$1,418
Issuance of 609,268 shares of					
Common Stock	6,580	2,400	1,073	2,500	607

KIRBY REPORTING COMMANDER

</TABLE>

The value of the Common Stock to be issued is based upon 90 percent of the assumed initial public offering price of \$12.00 per share. The excess of cost over the fair value of the identifiable net assets acquired in the Pending Acquisitions will be amortized over 40 years.

(D) Includes estimated transaction costs of \$285,000.

F-

U.S. LEGAL SUPPORT, INC.

UNAUDITED PRO FORMA COMBINED STATEMENT OF INCOME FOR THE NINE MONTHS ENDED SEPTEMBER 30, 1997

(IN THOUSANDS)

<TABLE> <CAPTION>

<caption></caption>		TORICAL					HI OBODIONI		
		COMPLETED ACQUISITIONS (B)	ADJUSTMENTS	COMBINED	OFFERING ADJUSTMENTS	SUBTOTAL	~	PENDING ACQUISITIONS ADJUSTMENTS	PRO FORMA
<s> Revenues</s>	<c> \$14,549</c>	<c> \$14,801</c>	<c> \$</c>	<c> \$29,350</c>	<c></c>	<c> \$29,350</c>	<c> \$7,927</c>	<c></c>	<c> \$37,277</c>
Cost of services	9,287	8,199		17,486 		17,486 	3,775 		21,261
Gross profit Selling, general and administrative	5,262	6,602		11,864		11,864	4,152		16,016
expenses Depreciation and	3,855	4,769	(760) (D)	7,864		7,864	1,571	(70) (D)	9,365
amortization	332	161	372 (F)	865		865	35	437 (F)	1,337
Operating income Interest expense	1,075 1,147	1,672 47	388 1,298 (G)	3,135 2,492	 (2,459)(H)	3 , 135	2,546 14	(367) 660 (G)	5,314 707
<pre>Income (loss) before income taxes Provision for income</pre>	(72)	1,625	(910)	643	2,459	3,102	2,532	(1,027)	4,607
taxes	11	757	(425)(I)	343	983 (I)	1,326		602 (I)	1,928
Net (loss) income Accretion of	(83)		(485)	300	1,476	1,776	2,532	(1,629)	2 , 679
preferred stock	(479)			(479)	479 (K)				
Net income (loss) attributable to common									
shareholders	\$ (562)	\$ 868 ======	\$(485) =====	\$ (179) ======	\$1,955 =====	\$ 1,776	\$2,532 =====	\$(1,629)	\$ 2,679
Net income (loss) per share	\$ (0.13)								\$.33
Weighted average outstanding shares	4,226 ======								8,002(J)

 | | | | | | | | |U.S. LEGAL SUPPORT, INC.

UNAUDITED PRO FORMA COMBINED STATEMENT OF OPERATIONS FOR THE NINE MONTHS ENDED SEPTEMBER 30, 1996

(IN THOUSANDS)

<TABLE> <CAPTION>

	HI	STORICAL									
			_						HISTORICAL		
		COMPLETED							PENDING	PENDING	
	COMPANY	ACQUISITION	S			OFFE	RING		ACQUISITIONS	ACQUISITIONS	PRO
	(A)	(B)	ADJUST	TMENTS	COMBINED	ADJUS	TMENTS	SUBTOTAL	(C)	ADJUSTMENTS	FORMA
<s></s>	<c></c>	<c></c>	<c></c>		<c></c>	<c></c>		<c></c>	<c></c>	<c></c>	<c></c>
Revenues Cost of services	\$5,886 3,726	\$19,656 11,423	\$		\$25,542 15,149	\$		\$25,542 15,149	\$6,482 3,426	\$ 	\$32,024 18,575

Gross profit Selling, general and administrative	2,160	8,233		10,393		10,393	3,056		13,449
expenses	1,310	6,533	(830) (D) 494 (E)	7,507		7,507	1,348	(435) (D)	8,420
Depreciation and									
amortization	159	145	548 (F)	852		852	43	437 (F)	1,332
Operating income	691	1,555	(212)	2,034		2,034	1,665	(2)	3,697
Interest expense	177	61	2,253 (G)	2,491	(2,483)(H)	. 8	16	644 (G)	668
Income (loss) before									
income taxes Provision (benefit)	514	1,494	(2,465)	(457)	2,483	2,026	1,649	(646)	3,029
for income taxes	175	102	(374)(I)	(97)	993 (I)	896		402 (I)	1,298
Net income (loss)	\$ 339	\$ 1,392	\$(2,091)	\$ (360)	\$ 1,490	\$ 1,130	\$1,649	\$(1,048)	\$ 1,731
	=====	======	======		======	======	======	======	
Net income per share									\$.22 ======

Weighted average shares outstanding...

8,002(J)

</TABLE>

F-10

U.S. LEGAL SUPPORT, INC.

UNAUDITED PRO FORMA COMBINED STATEMENTS OF INCOME

YEAR ENDED DECEMBER 31, 1996 (IN THOUSANDS)

<TABLE> <CAPTION>

		STORICAL							
		COMPLETED ACQUISITIONS (B)	ADJUSTMENTS	COMBINED	OFFERING ADJUSTMENTS	SUBTOTAL		PENDING ACQUISITIONS ADJUSTMENTS	PRO FORMA
<s> Revenues Cost of services</s>		<c> \$26,743 15,819</c>	<c></c>	<c> \$34,410 20,658</c>	<c></c>	<c> \$34,410 20,658</c>	<c> \$8,995 4,816</c>	<c></c>	<c> \$43,405 25,474</c>
Gross profit Selling, general and administrative		10,924		13,752		13,752	4,179		17,931
expenses	2,352	8,846	(2,280)(D) 659 (E)	9,577		9,577	1,971	(580) (D)	10,968
Depreciation and amortization	212	173	730 (F)	1,115		1,115	56	583 (F)	1,754
Operating income Interest expense	264 238	1,905 63	891 3,021 (G)	3,060 3,322	(3,322) (H)	3,060	2,152 20	(3) 860 (G)	5,209 880
<pre>Income (loss) before income taxes Provision (benefit)</pre>	26	1,842	(2,130)	(262)	3 , 322	3,060	2,132	(863)	4,329
for income taxes	10	232	(232)(I)	10	1,329 (I)	1,339		508 (I)	1,847
Net income (loss)		\$ 1,610	\$(1,898)	\$ (272)	\$1 , 993	\$ 1,721	\$2,132	\$(1,371)	\$ 2,482
Net income per share									\$.31
Weighted average outstanding shares									8,002(J)

</TABLE>

F-11

NOTES TO UNAUDITED PRO FORMA COMBINED STATEMENTS OF INCOME

- (A) Represents the historical Consolidated Statement of Operations data of the Company, which includes the operations of the Completed Acquisitions from respective dates of acquisition through September 30, 1997.
- (B) Represents the combined historical statement of operations data for the

- Completed Acquisitions from January 1, 1997 through their respective dates of acquisition.
- (C) Represents the combined historical statements of operations data of the Pending Acquisitions.
- (D) Represents adjustments to reflect the excess of historical compensation paid to owners of businesses acquired in the Acquisitions over compensation that would have been payable under the terms of employment agreements entered into in connection with each acquisition as follows:

<TABLE>

	NINE MONTE		YEAR ENDED DECEMBER 31,
	1997	1996	1996
<5>	<c></c>	IN THOUSANI	DS) <c></c>
Aggregate owners historical compensation		\$ 2,353	\$4,310
Less: Aggregate Salary Per Post-Acquisition Employment Contract	(946)	(1,088)	(1,450)
Adjustment	\$830	\$1,265	 \$2,860
	======		=====

</TABLE>

- (E) Represents corporate office compensation for personnel-related costs that would have been necessary to perform administrative functions in 1996
- (F) Represents an increase in amortization of goodwill associated with the Acquisitions. The excess of cost over the fair value of the net assets acquired will be amortized over a period of 10 to 40 years.

<TABLE> <CAPTION>

		PTEMI	BER	•	YEAR ENDED	
	199			996	DECEMBER 31, 1996	<u>'</u>
.00				THOUS <i>I</i>	•	
<\$>	<c></c>		<c></c>		<c></c>	
Completed Acquisitions	\$	372	\$	548	\$730	
Pending Acquisitions		437		437	583	
	\$	809		\$985	\$1,313	
	=====		===		======	

</TABLE>

- (G) Represents (i) an adjustment to accrue interest expense on debt incurred in connection with the Completed Acquisitions, such expense computed based on fixed or variable interest rates, as appropriate, at the time the Company entered into each agreement; and (ii) an adjustment to reflect the interest expense on debt to be incurred in connection with the Pending Acquisitions based on pro forma debt of \$11.0 million computed based upon the terms of the New Credit Agreement which bears interest at 8.00% (LIBOR plus 200 basis points). The effect on net income of a 1/8% variance of the interest rate would be \$8,250 (annual basis).
- (H) Represents an adjustment to reduce interest expense on debt that will be repaid with proceeds of the Offering.
- (I) Represents adjustments to accrue income taxes on earnings for certain Acquisitions not previously taxed at the corporate level and to reflect the tax effects of adjustments based on estimated combined federal and state statutory tax rates of 40%. The Company's total effective tax rate approximates 41% and 43% for the nine months ended September 30, 1997 and 1996, and 42% for the year ended December 31, 1996, respectively because of non-deductible portion of the goodwill recorded in connection with the Acquisitions.
- (J) Pro forma earnings per share gives effect to: (i) 1,734,564 shares outstanding prior to the offering, (ii) 609,268 shares to be issued in the Pending Acquisitions; (iii) 3,500,000 shares to be issued in the Offering; and (iv) 1,969,283 shares to be issued upon conversion of preferred stock and Subordinated Convertible Promissory Notes. Also, Common Shares, options and warrants issued within one year prior to the initial public offering have been treated as outstanding for all periods presented.

(K) Represents adjustment to eliminate accretion of redeemable preferred stock which will be converted to Common Stock.

F-12

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders

U.S. Legal Support, Inc.:

We have audited the accompanying consolidated balance sheet of U.S. Legal Support, Inc. as of September 30, 1997, and the related consolidated statements of income, stockholder's deficit and cash flows for the nine months ended September 30, 1997. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audit.

We conducted our audit in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the consolidated financial position of U.S. Legal Support, Inc. as of September 30, 1997 and the consolidated results of its operations and its cash flows for the nine months ended September 30, 1997 in conformity with generally accepted accounting principles.

The accompanying consolidated financial statements have been prepared on the going concern basis of accounting. As more fully described in Note 1 to the consolidated financial statements, proceeds from a planned initial public offering are to be used to repay current maturities of debt. The success of the initial public offering cannot presently be determined. As a result, there is substantial doubt about the Company's ability to continue as a going concern. The consolidated financial statements do not include any adjustments that might result from the outcome of this uncertainty.

COOPERS & LYBRAND L.L.P.

Houston, Texas

December 15, 1997

F-13

U.S. LEGAL SUPPORT, INC.

CONSOLIDATED BALANCE SHEET

<TABLE>

ASSETS	1997	UNAUDITED PRO FORMA SEPTEMBER 30, 1997(1)
<\$>	<c></c>	<c></c>
Current assets:		
Cash	\$ 558,574 6,990,522 188,969 47,632	\$ 558,574 6,990,522 188,969 47,632
Total current assets	7,785,697	7,785,697
Property and equipment, net	1,153,015	1,153,015
Intangibles, net	27,980,167	27,980,167
Deferred charges	1,546,526	1,546,526
Other assets	254,806	254,806
Total assets	\$38,720,211	
<caption></caption>		
LIABILITIES AND STOCKHOLDERS' DEFICIT		
<s></s>	<c></c>	<c></c>

Current liabilities:		
Accounts payable	\$ 3,689,227	\$ 3,689,227
Accounts payable, related parties	315,886	315,886
Accrued liabilities	2,714,558	2,714,558
Income taxes payable	33,637	33,637
Current maturities of long-term obligations	3,935,964	3,935,964
Total current liabilities	10,689,272	10,689,272
Long-term obligations, net of current maturities	26,339,283	25,058,991
Deferred income taxes	255,356	255,356
Redeemable preferred stock:		
Series A Convertible Preferred Stock, \$1.00 par value, 2,000,000 shares authorized; 1,000,000 shares issued		
and outstanding	1,479,000	
Series B Convertible Preferred Stock, \$1.00 par value,		
2,500,000 shares authorized; 2,046,667 shares issued		
and outstanding	2,046,667	
Series C Convertible Preferred Stock, \$1.00 par value,	, ,	
231,250 shares authorized, issued and outstanding	231,250	231,250
. ,		
Total redeemable preferred stock	3,756,917	231,250
Commitments and contingencies		
Stockholders' deficit:		
Preferred Stock, \$1.00 par value, 5,268,750		
authorized, no shares issued or outstanding		
Common Stock, \$.01 par value, 100,000,000 shares		
authorized; 1,734,564 shares issued and outstanding		
(3,703,847 pro forma shares issued and outstanding)	17,346	37,039
Additional paid-in capital	3,840,117	8,626,383
Accumulated deficit		(6,178,080)
Total stockholders' deficit	(2,320,617)	2,485,342
Total liabilities and stockholders' deficit	\$38,720,211	
	========	========

</TABLE>

(1) Gives effect to the conversion of (i) 1,000,000 shares of Series A Convertible Preferred Stock into 1,560,000 shares of Common Stock; (ii) 2,046,667 shares of Series B Convertible Preferred Stock into 183,393 shares of Common Stock; and (iii) \$1,844,955 principal amount (1,280,292 net of discount) of Convertible Subordinated Promissory Notes into 225,890 shares of Common Stock.

The accompanying notes are an integral part of the consolidated financial statements.

F-14

U.S. LEGAL SUPPORT, INC.

CONSOLIDATED STATEMENT OF INCOME

<TABLE> <CAPTION>

	NINE MONTHS ENDED SEPTEMBER 30, 1997
<s> Revenues Cost of services</s>	<c> \$14,549,056</c>
Gross profit	5,261,843 3,854,666
Operating income	1,075,281
Loss before income taxes	(72,314)
Net loss	
Net loss attributable to common shareholders	
Net loss per common share	
Weighted average shares outstanding	3,893,000

========

If the shares necessary to fund the cash portion of the distribution to Richard O. Looney were outstanding for the		
entire period, net loss per common share and weighted average		
shares outstanding would have been as follows:		
Pro forma net loss per common share	\$	(.13)
	=====	=====
Weighted average shares outstanding	4,2	26,000
	=====	=====

</TABLE>

The accompanying notes are an integral part of the consolidated financial statements.

F-15

U.S. LEGAL SUPPORT, INC.

STATEMENT OF STOCKHOLDERS' DEFICIT

<TABLE> <CAPTION>

<caption></caption>			ADDITIONAL PAID-IN	ACCUMULATED	TOTAL STOCKHOLDERS'
			CAPITAL		
<s> Balance as of January 1,</s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
1997 Net assets acquired in connection with reorganization of the Company and Looney &	993,840	\$9,938			\$(62,823)
Company Distribution in connection with reorganization of the Company and Looney &				207,763	207,763
Company Issuance of Common				(6,227,902)	(6,227,902)
Stock	740,724	7,408	4,319,117		4,326,525
StockPreferred Stock			(479,000)		(479,000)
Dividend Net loss					(1,825) (83,355)
Balance as of September 30, 1997	1,734,564	\$17,346 =====	\$3,840,117	\$(6,178,080) ======	

 | | | | |The accompanying notes are an integral part of the consolidated financial statements.

F-16

U.S. LEGAL SUPPORT, INC.

CONSOLIDATED STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

	NINE MONTHS ENDED
	SEPTEMBER 30,
	1997
<\$>	<c></c>
· ·	<c></c>
Cash flows from operating activities:	
Net loss	\$(83,355)
Adjustments to reconcile net loss to net cash provided by operating activities:	
Depreciation and amortization	331,896
Amortization of debt issue costs and debt discount	166,600
Provision for doubtful accounts	154,467
Deferred taxes	(213,026)

Changes in operating assets and liabilities:	
Accounts receivable	(1,186,127)
Prepaid expenses and other current assets	(78 , 069)
Accounts payable and accrued liabilities	2,330,567
Income taxes payable	(183,730)
Deferred charges and other assets	(1,236,783)
Net cash provided by operating activities	2,440
Cash flows from investing activities:	
Capital expenditures	(110,827)
Acquisitions, net of cash acquired	(20,307,527)
Net cash used in investing activities	
Cash flows from financing activities:	
Issuance of redeemable preferred stock	1,000,000
Debt issuance costs	(572,139)
Borrowings under senior credit agreement	16,000,000
Issuance of subordinated debt	9,580,260
Principal payments on long-term obligations	(964,578)
Dividends paid	(1,825)
Distribution to the Looney and Company shareholder	(4,087,835)
Net cash provided by financing activities	20,953,883
Increase in cash and cash equivalents	537,969
Cash and cash equivalents at beginning of period	20,605
Cash and cash equivalents at end of period	\$558,574
Supplemental cash flow information:	
Cash paid for interest	\$606,888
Cash paid for income taxes	360,000

 , |The accompanying notes are an integral part of the consolidated financial statements.

F-17

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. THE COMPANY:

U.S. Legal Support, Inc. (the "Company") was founded in 1996 by Richard O. Looney and GulfStar Investments, Ltd. to continue the business of Looney & Company ("Looney") and create a leading provider of legal support services, including court reporting, certified record retrieval, legal placement and temporary staffing, to law firms, insurance providers and major corporations throughout the United States. As described in Note 3 to these financial statements, the merger with Looney was accounted for as a merger of entities under common control. Accordingly, these financial statements should be read in conjunction with the historical financial statements of Looney included elsewhere in this prospectus. The Company operates in one business segment. The Company, including Looney, has completed the acquisition of 13 companies, and has entered into definitive agreements to acquire an additional four businesses. The Company requires additional financing in order to implement its business strategy. The Company intends to offer and sell 3,500,000 shares of common stock, par value \$.01 per share (the "Common Stock") at an estimated offering price between \$11 and \$13 per share. Additionally, 609,268 shares are expected to be issued in connection with Pending Acquisitions described in Note 3 and 1,969,283 shares are expected to be issued upon the conversion of (i) 1,000,000 shares of Series A Convertible Preferred Stock into 1,560,000 shares of Common Stock; (ii) 2,046,667 shares of Series B Convertible Preferred Stock into 183,393 shares of Common Stock; and (iii) \$1,800,000 principal amount of Convertible Subordinated Promissory Notes into 225,890 shares of Common Stock (see unaudited pro forma balance sheet).

The Company's operations to date have been financed through the issuance of redeemable convertible preferred stock, convertible debt, subordinated debt and bank debt. The proceeds of the planned initial public offering would be used to repay debt, redeem preferred stock and pay the cash portion of pending acquisitions (Note 3). Certain convertible debt and convertible preferred stock is expected to be converted into common stock upon completion of the initial public offering. The failure to raise additional equity in the planned initial public offering would require restructuring of its debt and equity securities. Additionally, without additional equity financing, the Company would be unable to consummate the pending acquisitions and implement its business strategy beyond 1997. Unless the Company were able to obtain financing from other sources or negotiate an extension, the bank could

exercise its rights under the loan agreement giving it additional authority over the Company's operations and the ability to call the outstanding balance of the revolving line of credit and term loans (Note 7). Any impediment to the Company's ability to obtain additional capital could have a material adverse effect on the Company's business, financial position, results of operations and cash flows.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Principles of Consolidation

The consolidated financial statements include the accounts of the Company and its subsidiaries. Unless the context requires otherwise, the term "Company" refers to U.S. Legal Support, Inc. and its consolidated subsidiaries. All significant intercompany balances and transactions have been eliminated.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make certain estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash and Cash Equivalents

Cash and cash equivalents include highly liquid debt instruments purchased with original maturities of three months or less. The Company maintains cash deposits in banks. The balance, at times, may exceed federally insured amounts although management believes the risk of loss is minimal.

F-18

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized, while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in operations.

Intangibles

Intangibles consist primarily of goodwill, which is amortized on a straight line basis over the estimated useful life of 10 to 40 years. Accumulated amortization of goodwill was \$150,613 at September 30, 1997. Amounts allocated to covenants not to compete in the amount of \$105,000 are amortized over the lives of the related contract. Accumulated amortization of such covenants was \$7,750 at September 30, 1997. The Company evaluates, for impairment, the carrying value of intangible assets by comparing the carrying value to the anticipated future undiscounted cash flows from the businesses whose acquisition gave rise to the asset. If the intangible asset is impaired, the asset is written down to fair value.

Debt Issue Costs

Debt issue costs relating to long-term debt are included in other assets and are amortized to interest expense over the scheduled maturity of the debt utilizing the interest method.

Income Taxes

The Company utilizes the liability method of accounting for income taxes. Deferred income taxes are recognized for the tax consequences of differences in the tax bases of assets and liabilities and their financial reporting amounts based on enacted tax laws and statutory tax rates applicable to the periods in which the differences are expected to affect taxable income. A valuation allowance is established, when necessary, to reduce deferred income tax assets to the amount expected to be realized.

Revenue Recognition

The Company recognizes revenues from its court reporting and certified record retrieval services upon delivery of prepared transcripts and as

documents or records are delivered to customers. With respect to the Company's legal placement and staffing services, the Company recognizes revenue from its permanent placement services when candidates accept job offers, and with respect to its temporary placement services, as services are provided to the Company's clients. An allowance is provided for anticipated bad debts, based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit, primarily to law firms, insurance companies, and major corporations. The Company maintains allowances for potential credit losses, and such losses have been within management's estimates.

Earnings per Share Data

Earnings per share data is computed using the weighted average number of common and common equivalent shares outstanding during each year presented. Common equivalent shares consist of convertible debt, convertible preferred stock, and stock options and are computed using the treasury stock method. Under guidelines issued by the Securities and Exchange Commission, common shares, options and warrants issued within one year prior to a public offering at prices below the initial offering price are treated as outstanding for all periods presented (using the Treasury stock method) in computing net earnings (loss) per share. Pro forma earnings (loss) per share have been presented for the nine months ended September 30, 1997 to reflect issuance

F-19

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

of the number of shares that would have been necessary to fund the cash portion of the \$6,227,902 distribution to Richard O. Looney in January 1997 (at an assumed public offering price of \$12.00 per share). Fully diluted earnings per share is not presented because such amounts would be the same as amounts computed for primary earnings per share.

Stock-Based Compensation

The Company applies APB Opinion No. 25, "Accounting for Stock Issued to Employees" and related interpretations in accounting for its stock-based compensation plans.

Financial Instruments

For all financial instruments, including cash and cash equivalents, accounts receivable, accounts payable, and long-term debt, the carrying value is considered to approximate fair value.

New Accounting Standards

In February 1997, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards No. 128, "Earnings Per Share" ("SFAS 128"). SFAS 128 changes the computation of earnings per share and requires dual presentation of basic and diluted earnings per share. SFAS 128 is effective for financial statements issued for periods ending after December 15, 1997, including interim periods.

In June 1997, the FASB issued Statement of Financial Accounting Standards No. 130, "Reporting Comprehensive Income" ("SFAS 130") and Statement of Financial Accounting Standards No. 131, "Disclosure about Segments of an Enterprise and Related Information" ("SFAS 131").

SFAS 130 establishes standards for reporting and display of comprehensive income and its components (revenues, expenses, gains and losses) in a full set of general-purpose financial statements. It requires (a) classification of the components of other comprehensive income by their nature in a financial statement and (b) the display of the accumulated balance of the other comprehensive income separate from retained earnings and additional paid-in capital in the equity section of a statement of financial position. SFAS 130 is effective for years beginning after December 15, 1997 and is not expected to have a material impact on financial position or results of operations.

SFAS 131 establishes standards for reporting information about operating segments in annual financial statements and requires selected information about operating segments in interim financial reports issued to shareholders. It also establishes standards for related disclosures about products and services, geographic areas and major customers. The Company has not determined the impact of SFAS 131 on its financial reporting practices.

3. REORGANIZATION AND BUSINESS COMBINATIONS:

In a reorganization in January 1997, the Company received 100% of the outstanding capital stock of Looney & Company ("Looney"), a Texas-based court reporting and certified records retrieval company. The Company paid Richard Looney approximately \$4,088,000 in cash and issued 2,046,667 shares of the Company's Series B Convertible Preferred Stock, \$1.00 par value (the "Series B Preferred Stock"). Looney has been deemed to be the accounting acquiror because the two companies were under common control and prior to their combination, the Company had no substantive operations. Therefore, the net assets of Looney have been recorded at their historical cost basis. The consideration paid for the net assets of Looney was recorded as a capital distribution to Looney's shareholder.

F-20

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS-- (CONTINUED)

In January 1997, the Company acquired all of the outstanding capital stock of Klein, Bury and Associates, Inc. ("Klein Bury"), a Florida based court reporting company. The purchase price consisted of approximately \$3,850,000 in cash; 170,600 shares of the Company's Common Stock; a \$1,424,000 promissory note payable over five years with interest at 10%; options to purchase 40,000 shares of the Company's Common Stock and 50% of amounts collected in respect of accounts receivable over 120 days past due as of the closing date, up to a maximum of \$500,000. The acquisition was accounted for under the purchase method of accounting. The results of operations of Klein Bury are included from the date of the acquisition. The excess of the cost over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$5,373,000 and is being amortized over 40 years.

In May 1997, the Company acquired the assets of San Francisco Reporting Service ("San Francisco Reporting"), a California-based court reporting company. The purchase price consisted of approximately \$545,000 in cash, 30,608 shares of the Company's Common Stock and 231,250 shares of the Company's Series C Convertible Preferred Stock, \$1.00 par value. The acquisition was accounted for under the purchase method of accounting. The results of operations of San Francisco Reporting are included from the date of acquisition. The excess of the cost over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$811,000 and is being amortized over 40 years.

In May 1997, the Company acquired the assets of G & G Court Reporters ("G&G"), a California-based court reporting company. The purchase price consisted of approximately \$1,268,000 in cash and two subordinated promissory notes in the aggregate amount of \$1,038,000, which were subsequently adjusted to \$996,000. The acquisition was accounted for under the purchase method of accounting. The results of operations of G&G are included from the date of acquisition. The excess of the cost over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$1,919,000 and is being amortized over 40 years.

In August 1997, the Company acquired by merger Goren of Newport, Inc., Rapidtext, Inc. and Medtext, Inc. (the "Johnson Group"), three California-based companies involved in court reporting, closed captioning, and medical transcription. The purchase price consisted of approximately \$983,000 in cash, \$246,000 in subordinated notes and 142,476 shares of the Company's Common Stock. The acquisition has been accounted for under the purchase method of accounting. The results of operations of the Johnson Group are included from the date of acquisition. The excess of the cost over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$2,256,000 and is being amortized over 40 years.

In August 1997, the Company acquired the assets of Legal Enterprise, Inc., ("Legal Enterprise") a California-based document retrieval and data management company. The purchase price consisted of approximately \$1,200,000 in cash, \$1,140,500 in subordinated notes and options to purchase 7,000 shares of Company Common Stock. The acquisition has been accounted for under the purchase method of accounting. The results of operations of Legal Enterprise are included from the date of acquisition. The excess of the cost over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$1,460,000 and is being amortized over 40 years.

In September 1997, the Company acquired the assets of Amicus One Legal Support Services, Inc. ("Amicus"), a New York-based court reporting company. The purchase price was approximately \$1,886,000 in cash, \$560,000 in a subordinated note and 116,471 shares of the Company's common stock. The acquisition has been accounted for under the purchase method of accounting. The results of operations of Amicus are included from the date of acquisition. The excess of the cost over the fair value of the assets acquired less

liabilities assumed attributed to goodwill is approximately \$3,033,000 and is being amortized over 40 years.

In September 1997, the Company acquired the stock of Block Court Reporting ("Block"), a court reporting company serving Washington, D.C., Northern Virginia, and Baltimore. The purchase price was approximately

F-21

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

\$600,000 in cash and \$600,000 in subordinated notes. The acquisition has been accounted for under the purchase method of accounting. The results of operations of Block are included from date of acquisition. The excess of the cost over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$1,018,000 and is being amortized over 40 years.

In September 1997, the Company acquired the stock of Burton House, Inc. d.b.a. Ziskind, Greene, Watanabe and Nason ("Ziskind Greene"), a California-based company providing permanent legal search services. The purchase price was approximately \$1,350,000 in cash and 158,824 shares of the Company's Common Stock. The acquisition has been accounted for under the purchase method of accounting. The results of operations of Ziskind Greene have been included from the date of acquisition. The excess of the costs over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$2,750,000 and is being amortized over 40 years.

In September 1997, the Company acquired the assets of Elaine P. Dine, Inc., ("Elaine Dine"), a New York-based company providing permanent legal search and temporary legal staffing services. The purchase price was approximately \$6,000,000 in cash, \$2,000,000 in subordinated notes and 76,471 shares of the Company's Common Stock and 41,176 options to purchase shares of Company Common Stock. The acquisition was accounted for under the purchase method of accounting. The results of operations of Elaine Dine have been included from the date of acquisition. The excess of the cost over the fair value of the assets acquired less liabilities assumed attributed to goodwill is approximately \$8,687,000 and is being amortized over 40 years. In addition, the Company is obligated to pay \$500,000 for an estimated tax liability.

The following unaudited pro forma summary presents consolidated results of operations information as if the aforementioned acquisitions had been completed at the beginning of the period presented. These results do not purport to be indicative of the results of operations of the Company that might have occurred nor are they indicative of future results.

<TABLE>

<\$>	<c></c>
Revenues	\$29,350,000
Net income	\$300,000
Earnings per common share	\$0.07

 |Adjustments made in arriving at the pro forma unaudited results of operations include interest expense on acquisition debt, amortization of goodwill, compensation reductions and related tax adjustments. No effect has been given to synergistic benefits which may be realized from the acquisition or to the use of proceeds from the Company's proposed initial public offering (the "Offering").

In addition, with respect to certain of the businesses acquired, the Company may be obligated to pay cash or common stock in the form of an "earn-out" payment. In most cases, the earn-out is a function of cash and common stock based on increases in the businesses' future operating income in excess of historical levels. With respect to certain of the businesses acquired, the Company is obligated to pay contingent consideration equal to six times the amount by which pre-tax earnings exceed a specified amount. Contingent consideration paid will be recorded as additional goodwill.

Pending Acquisitions (the "Pending Acquisitions")

All of the Pending Acquisitions will be accounted for under the purchase method of accounting. In addition, the cash consideration to be paid for each of the Pending Acquisitions is subject to post-closing adjustments.

Upon completion of the Offering, the Company will acquire the assets of Reporting Service Associates, Inc., a Philadelphia-based court reporting services firm serving the mid-Atlantic markets. The purchase price will be approximately \$7,400,000 in cash and 231,481 shares of the Company's Common Stock.

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS-- (CONTINUED)

Upon completion of the Offering, the Company will acquire the assets of Jilio & Associates, a Southern California-based court reporting services firm. The purchase price will be approximately \$5,600,000 in cash and 222,222 shares of the Company's Common Stock.

Upon completion of the Offering, the Company will acquire the assets of Kirby Kennedy & Associates, a court reporting services firm which serves the Minneapolis and St. Paul, Minnesota markets. The purchase price will be approximately \$2,500,000 in cash and 99,315 shares of the Company's Common Stock.

Upon completion of the Offering, the Company will acquire the assets of Commander Wilson, Inc., a firm that provides permanent legal search services to national and Texas law firms and legal departments of major corporations. The purchase price will be approximately \$1,400,000 in cash and 56,250 shares of the Company's Common Stock.

4. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE>

	USEFUI	L LIVES	SEPTEMBER 30, 1997
<pre><s> Leasehold improvements</s></pre>		5 years 7 years	
Less accumulated depreciation			2,306,624 (1,153,609)
			\$1,153,015 ======

</TABLE>

The Company has entered into various capital leases. The leases were recorded upon their inception using the interest rate implicit in the lease agreements. The capitalized cost of leased office equipment and related accumulated depreciation was approximately \$324,000 and \$235,000, respectively, at September 30, 1997.

5. ACCRUED LIABILITIES:

Accrued liabilities consisted of the following: <TABLE> <CAPTION>

	SEPTEMBER 30, 1997
<\$>	<c></c>
Accrued interest	\$374,107
Deferred purchase price	500,000
Payroll	574,374
Customer overpayments	507,150
Stock options	438,535
Other	320,392
	\$2,714,558

</TABLE>

Customer overpayments arise primarily when customers make duplicate payments or payments in excess of billed amounts. The customers have generally denied the Company's refund attempts, which the management of the Company believes is due to the significant volume and relatively small amount of each individual billing.

F-23

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

6. INCOME TAXES:

The provision for income taxes consisted of the following: <TABLE>

<CAPTION>

	1997
<\$>	<c></c>
Current	\$ 224,067
Deferred	(213,026)
	\$ 11,041
	=======

</TABLE>

A reconciliation of the differences between income taxes computed at the U.S. federal statutory rate of 34% and the Company's reported provision for income taxes is: <TABLE>

<CAPTION>

	SEPTEMBER 30, 1997
<\$>	<c></c>
Income tax provision at statutory rate	\$(24,587)
State tax provision, net of federal income tax benefit	(2,893)
Nondeductible amortization and other expenses	38,521
	\$11,041

</TABLE>

The components of deferred income tax assets and liabilities were as follows: <TABLE>

<CAPTION>

	SEPTEMBER 30, 1997
<\$>	<c></c>
Deferred tax assets:	
Accrued liabilities	. \$19,000
Allowance for doubtful accounts	. 149,920
Deferred tax assets	
Deferred tax liabilities:	
Conversion from cash to accrual basis for tax reporting	
purposes	. (306,867)
Property and equipment	. (18,387)
Intangibles	. (51,390)
Deferred tax liability	
Net deferred income taxes	\$ (207,724)

 |F-24

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

7. LONG-TERM DEBT:

Long-term debt consisted of the following: <TABLE> <CAPTION>

	SEPTEMBER 30, 1997
<s></s>	<c></c>
Bank notes payable: Revolving line of credit Term loans	\$2,000,000 14,000,000
Senior subordinated notes, interest at 12% payable quarterly, principal due in three annual installments beginning January 2002 but must be repaid within two days of an initial public	
offering of common stock or a change of control	9,000,000

Subordinated Promissory Note, interest at 10% discounted to yield an effective interest rate of 15%, quarterly principal payments of \$71,206 through February 1, 2002 but must be repaid within two days of an initial public offering of common stock or a change of control	1,281,701
paid, or at the holders' option, converted into common stock at the issuance price on the closing date of an initial public offering or a change of control	1,696,873
Convertible Subordinated Promissory Notes, interest at 6.375% payable monthly, principal due in 2005, discounted to yield an effective interest rate of 15% convertible at the holders' option into common stock at conversion prices ranging from \$7.56 to \$8.50, automatically converts into common stock on the closing date of an initial public offering or a change of	
control	1,844,955
date of an initial public offering or a change of control Capital lease obligations	2,000,000 158,911
Total debt Less discount on debt(1)	31,982,440 1,707,193
Long term debt, net of discount	30,275,247 3,935,964
	\$26,339,283

</TABLE>

(1) The debt discount is applicable to the various categories of long-term debt as follows:

<TABLE>

<\$>	<(:>
10% Subordinated Promissory Note	. \$	131,855
6% to 7.5% Subordinated Promissory Notes		347,640
6.375% Convertible Subordinated Promissory Notes		564,663
6.375% Subordinated Promissory Notes		663,035
	\$1	,707,193

</TABLE>

Revolving Credit and Term Loan Agreements

The Company's Bank Credit Agreement provided for a revolving line of credit of \$2,000,000, all of which had been fully utilized at September 30, 1997 and matures on July 10, 1998. The Bank Credit Agreement also provides for an acquisition line of credit of \$14,000,000, all of which had been fully utilized at September 30, 1997. The acquisition line of credit borrowings have been made under term loans with maturity dates through May 2000. Borrowings under this agreement bear interest at the base rate plus 0.75%, or LIBOR plus 2.5%, at the Company's option. At September 30, 1997, the average interest rate under the credit facility was 8.2%. Borrowings under the facility are collateralized by substantially all the assets of the Company and the stock of the subsidiaries. The Bank Credit Agreement contains various financial covenants including the maintenance of certain financial ratios, restrictions on capital expenditures, and a prohibition against the payment of dividends.

F-25

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

In October 1997, the Company amended the Bank Credit Agreement to increase the revolving line of credit from \$2,000,000 to \$3,250,000 and further amended the Agreement in December 1997 to increase the revolving line of credit to \$4,250,000. The additional borrowings of \$2,250,000 mature on February 15, 1998. If the Company is unable to pay such amount, the bank has the ability to call the outstanding balance of both the revolving line of credit and the term loans. The revolving credit and term loans must be repaid within three days of the closing of an initial public offering of Common Stock or a change of control. Also, the holders of the Senior Subordinated Notes agreed to defer the receipt of interest payments in the amount of \$540,000 until the earlier of February 15, 1998 or the completion of an initial public offering of the

The Company has entered into an agreement with a commercial bank for a new \$40.0 million revolving credit facility which will be effective upon the closing of the Company's initial public offering (the "New Credit Facility").

The Company expects that, immediately upon completion of the initial public offering, approximately \$12.1 million will have been borrowed and approximately \$10.0 million will be available for additional borrowings under the New Credit Facility. From time to time, additional borrowing capacity will be available to the Company under the New Credit Agreement provided that the Company achieves certain levels of financial performance through operations or through acquisitions. Loans under the New Credit Facility will bear interest at rates based, at the Company's option, on either LIBOR or the base rate plus, in each case, an applicable margin. The applicable margin will be contingent upon the debt coverage ratio as defined in the New Credit Facility and will vary from 1.75% to 2.25% in the case of LIBOR loans and 0% to .50% in the case of base rate loans. In addition, the Company will be required to pay to the lenders a quarterly fee with respect to the unused portion of the New Credit Facility. The New Credit Agreement will also have a \$2.5 million sublimit for the issuance of letters of credit.

Borrowings under the New Credit Facility will be secured by substantially all of the assets of the Company and the stock of the subsidiaries. The New Credit Facility also will require the Company to comply with various covenants related to, among other matters: (i) the maintenance of certain financial ratios, (ii) limitations on the incurrence of indebtedness, (iii) restrictions on liens, guarantees, dividends and stock redemptions, and (iv) limitations on mergers and sales of assets. The New Credit Facility will terminate, and all borrowings will be required to be repaid on December 31, 2000.

Maturities of debt, excluding capital lease obligations of \$158,911, are as follows:

<TABLE>

<\$>	<c></c>
1997 (three months)	\$ 100,373
1998	4,624,743
1999	3,458,040
2000	9,603,850
2001	3,658,032
2002	3,214,197
Thereafter	7,164,294
	\$31,823,529

</TABLE>

8. PREFERRED STOCK:

Series A Preferred Stock

Pursuant to the terms of the Articles of Incorporation, the Board of Directors has created a series of Preferred Stock consisting of 2,000,000 shares of Series A Convertible Preferred Stock (the "Series A Preferred Stock"). As of September 30, 1997, 1,000,000 shares of Series A Preferred Stock were issued and outstanding. The Series A Preferred Stock (a) has a liquidation preference of \$1.00 per share, plus accrued but unpaid

F-26

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

dividends and (b) entitles the holder, concurrently with each dividend paid on the Common Stock, to dividends in the same amount payable on the number of shares of Common Stock then issuable on conversion of the Series A Preferred Stock. Holders of Series A Preferred Stock vote together with the Common Stock on all matters submitted to a vote of the shareholders, and each share of Series A Preferred Stock entitles the holder to one vote for each share of Common Stock issuable on conversion thereof. The Series A Preferred Stock ranks on a parity with the Common Stock with respect to dividends and senior to Common Stock and the Series B Preferred Stock described below as to distributions of assets upon liquidation. Shares of Series A Preferred Stock may be converted into Common Stock, at the option of the holder, at an initial conversion rate of 1.56 shares of Common Stock for each share of Series A Preferred Stock, subject to adjustment. The Company may elect to convert all outstanding Series A Preferred Stock into shares of Common Stock at the foregoing conversion rate, at any time after the closing of an underwritten public offering of equity securities of the Company resulting in gross proceeds of at least \$15,000,000 (a "Qualifying Public Offering"), provided the \$9,000,000 principal amount of 12% Senior Subordinated Notes issued by a

subsidiary of the Company are no longer outstanding. Under the terms of the Securities Purchase Agreement pursuant to which the Series A Preferred Stock was issued, the holders have the right to require the Company to redeem any or all shares of Series A Preferred Stock upon the occurrence of a Change of Control (as defined in the Agreement). The redemption price is payable in cash in an amount equal to the Market Price (as defined) at the time notice of the Change of Control is given, together with a premium in the maximum aggregate amount of \$2,700,000. If at any time after January 17, 2003, there is no Liquid Secondary Market (as defined), the holders of the Series A Preferred Stock shall have the right to require the Company to redeem any or all of the outstanding shares of Series A Preferred Stock in three annual installments, at a redemption price equal to the Fair Market Value (as defined) of the shares of Common Stock into which the Series A Preferred Stock are then convertible, together with interest on the unpaid balance at an annual rate of 12.0%. For the nine months ended September 30, 1997, the Company has accreted \$479,000 toward the redemption value of the Series A Preferred Stock. Subject to the rights of the holders of the Series A Preferred Stock to convert their shares into Common Stock, the Company may redeem the Series A Preferred Stock, in whole but not in part, at any time on or after a Qualifying Public Offering, provided the Senior Subordinated Notes are no longer outstanding. The redemption price for shares redeemed at the option of the Company shall be \$.001 for each share of Common Stock issuable upon conversion of the Series A Preferred Stock so redeemed (subject to adjustments).

Series B Preferred Stock

Pursuant to the terms of the Articles of Incorporation, the Board of Directors has created a series of Preferred Stock consisting of 2,500,000 shares of Series B Convertible Preferred Stock (the "Series B Preferred Stock"). As of September 30, 1997, 2,046,667 shares of Series B Preferred Stock were issued and outstanding. The Series B Preferred Stock (a) has a liquidation preference of \$1.00 per share, (b) ranks junior to the Series A Preferred Stock with respect to distributions of assets on liquidation and (c) is convertible into Common Stock in the event of a Qualifying Public Offering at a conversion price equal to 93% of the price at which shares of Common Stock are offered to the public in the Qualifying Public Offering. Holders of Series B Preferred Stock are not entitled to receive any dividends. Within five business days after the receipt of notice from the Company that the Company has filed a Registration Statement with the Securities and Exchange Commission (the "Commission") related to a Qualifying Public Offering, the holder is required to notify the Company if such holder elects to cause the Company to redeem all or part of the Series B Preferred Stock for a cash redemption price of \$1.00 per share or to convert such shares into Common Stock at the rate specified above. In the event the holder fails to give the Company notice of its election, all conversion and redemption rights shall immediately terminate. Any shares of Series B Preferred Stock not redeemed or converted into Common Stock, must be redeemed by the Company quarterly, during the 20 fiscal quarters following the quarter in which a Qualifying Public Offering occurs. The Series B Preferred Stock does not entitle the holders to vote on any matter submitted to the Company's shareholders, except as required

F-27

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Series C Preferred Stock

The Board of Directors has created a series of Preferred Stock consisting of 231,250 shares of Series C Convertible Preferred Stock (the "Series C Preferred Stock"). As of September 30, 1997, 231,250 shares of Series C Preferred Stock were issued and outstanding. The Series C Preferred Stock has a liquidation preference of \$1.00 per share, plus accrued and unpaid dividends. The Series C Preferred Stock ranks senior to the Series A Preferred Stock and the Common Stock with respect to dividends and distributions of assets upon liquidation, and ranks senior to the Series B Preferred Stock with respect to distributions of assets on liquidation. The Series C Preferred Stock carries an annual dividend equal to \$.06 per share, payable quarterly in respect of any quarter in which aggregate net income before taxes, depreciation, and amortization of the business represented by SFRS exceeds \$160,000. Dividends of \$1,825 were paid on the Series C Preferred Stock in the nine months ended September 30, 1997. The Series C Preferred Stock is convertible into Common Stock upon the occurrence of a Qualifying Public Offering, at a conversion rate equal to the initial public offering price. Within 10 business days after the receipt of notice from the Company that the Company has filed a Registration Statement with the Commission related to a Qualifying Public Offering, the holder is required to notify the Company if such holder elects to cause the Company to redeem all or part of the Series C Preferred Stock for a cash redemption price of \$1.00 per share or to convert such shares into Common Stock at the rate specified above. In the event the

holder fails to give the Company notice of its election, all conversion and redemption rights shall immediately terminate. On or after May 15, 1998, any holder of Series C Preferred Stock may require the Company to redeem all such holders of shares of Series C Preferred Stock quarterly over the 16 fiscal quarters ended after notice of such election is delivered to the Company. The Series C Preferred Stock does not entitle the holders to vote on any matter submitted to the Company's shareholders, except as required by law.

9. COMMITMENTS AND CONTINGENCIES:

Leases

The Company leases various office space under noncancelable operating leases with remaining terms of up to four years. The Company also leases certain office and computer equipment under operating leases. Rental expense associated with these operating leases for the nine months ended September 30, 1997 was approximately \$355,000.

The future minimum rental payments under noncancelable operating leases from 1997 through 2002 are as follows (in thousands):

<TABLE>

<s> 1997 (three months)</s>	723,699 658,839 528,387 231,121
2002	46,832
	\$2,378,624

</TABLE>

Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

F-28

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS-- (CONTINUED)

Legal Proceedings

The Company is involved in certain lawsuits and claims arising in the normal course of business. In the opinion of management, uninsured losses, if any, resulting from the ultimate resolution of these matters will not have a material adverse effect on the Company's financial position, results of operations or cash flows.

10. STOCK OPTION PLANS:

1997 Stock Incentive Plan

The Company's 1997 Stock Incentive Plan provides for the granting to eligible employees or directors of the Company and its subsidiaries of options to purchase shares of Common Stock, which may be incentive stock options within the meaning of Section 422(b) of the Internal Revenue Code or nonqualified options. The 1997 Stock Incentive Plan is administered by the Compensation Committee of the Board of Directors, which designates the employees who will receive options, the type of options, the number of shares of Common Stock subject to these options and their terms and conditions, including their exercise price and vesting schedule. A total of 750,000 shares of Common Stock have been reserved for issuance pursuant to options granted under the 1997 Stock Incentive Plan. As of September 30, 1997 options to purchase a total of 138,915 shares of Common Stock had been granted under the 1997 Stock Incentive Plan with exercise prices ranging from \$6.41 to \$10.20. All options granted under the 1997 Stock Incentive Plan vest 20% annually, are fully vested on the fifth anniversary of the date of grant and expire 10 years after the date of grant. Subsequent to September 30, 1997, the Company granted an additional 470,109 stock options under the 1997 Stock Incentive Plan with an exercise price equal to the initial public offering price.

Stock Option Plan for Non-Employee Directors

The Company also has adopted the U.S. Legal Support, Inc. Stock Option Plan for Non-Employee Directors (the "Directors' Stock Option Plan"). A total of 150,000 shares of Common Stock have been reserved for issuance under the Directors' Stock Option Plan which provides for the grant of options to purchase 25,000 shares of Common Stock, with an exercise price equal to the fair market value on the date of grant, to each incumbent director and to each person who becomes a director concurrently with his or her first election to the Board. Options granted under the Directors' Stock Option Plan vest 20% annually and are fully vested on the fifth anniversary of the date of grant. Upon completion of the Offering, each director of the Company will be awarded options to purchase 25,000 shares of Common Stock with an exercise price equal to the per share price set forth on the cover page of this Prospectus.

The following is a summary of stock option activity for the nine months ended September 30, 1997:

<TABLE>

	# OF SHARES OF	
	UNDERLYING	WEIGHTED AVERAGE
	OPTIONS	EXERCISE PRICES
<\$>	<c></c>	<c></c>
Outstanding at beginning of the year		
Granted	138,915	\$7.94
Exercised		
Forfeited		
Expired		
Outstanding at end of period	138,915	\$7.94
Exercisable at end of period		
Weighted average fair value of options		
granted	\$	2.54

 | |The fair value of each stock option granted is estimated on the date of grant using The Black-Scholes pricing model with the following assumptions for grants during the nine months ended September 30, 1997: expected dividend yield of 0%; risk-free interest rates of 6%; and weighted average expected life of five years. The minimum value method has been applied, which excludes expected volatility.

F-29

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

Options outstanding as of September 30, 1997 are summarized below:

<TABLE> <CAPTION>

OPTIONS OUTSTANDING			OPTIONS EXE	RCISABLE	
RANGE OF EXERCISE PRICES	NUMBER OUTSTANDING	WGTD. AVG. REMAINING LIFE (IN YEARS)	WGTD. AVG. EXERCISE PRICE	NUMBER EXERCISABLE	WGTD. AVG. EXERCISE PRICE
<\$>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
\$6.41 to \$8.50	98,915	9	\$7.02		
\$10.20	40,000	9	\$10.20		
\$6.41 to \$10.20 					

 138,915 | 9 | \$7.94 | | |Had the compensation expense for the 1997 Stock Incentive Plan been recognized in accordance with SFAS 123, the Company's net loss and net loss per common share for the nine months ended September 30, 1997 would have been as follows (in thousands except per share data):

<TABLE>

AS	
REPORTED	PRO FORMA
/C>	<c></c>

Net loss attributable to common shareholders..... \$ (562,355) \$ (586,583)
Net loss per common share.................. \$ (.14) \$ (.15) \$ </TABLE>

The effects of applying SFAS 123 for only 1997 option grants may not be representative of the pro forma impact in future years.

Purchase Price Options

In connection with the acquisitions of Klein Bury, Legal Enterprise and Elaine Dine, the Company granted non-compensatory stock options as partial consideration. The fair value of the options granted was determined at the date of acquisition and accounted for as part of the purchase price. Additionally, options were granted to employees of Looney in connection with the combination of Looney and compensation expense of approximately \$20,000 was recognized.

Other Options outstanding as of September 30, 1997 are summarized below:

<TABLE>

	OPTION	OPTIONS	OPTIONS
GRANTED	PRICE	EXERCISED	OUTSTANDING
<s></s>	<c></c>	<c></c>	<c></c>
104,336	\$.01	12,480	91,856
40,000	\$.10		40,000

 | | |

11. RELATED PARTY TRANSACTIONS:

To finance the distribution to Richard Looney and the acquisition of Klein Bury, the Company issued \$9,000,000 principal amount of Senior Subordinated Notes to three investors managed by Pecks Management Partners Ltd. in January 1997. A director of the Company serves as a Managing Partner of Pecks Management Partners Ltd.

The GulfStar Group, Inc. ("GulfStar") has provided merger and acquisition advisory services to the Company since its inception. A director of the Company is a Managing Director of GulfStar. In October 1996, GulfStar Investments, Ltd., an affiliate of GulfStar was issued 150,000 shares of Common Stock at a price of \$.01 per share, or \$1,500, in exchange for services rendered in connection with the organization of the Company and strategic planning. The estimated fair value of the shares received by GulfStar was approximately \$60,000, or \$.40 per share, based on a January 17, 1997 valuation which considers the combination with Looney. Upon completion of the initial public offering, such amount will be recorded as a simultaneous increase and decrease to paid-in capital. If the offering is not successful, such amount will be charged to expense. GulfStar also received an investment banking fee aggregating \$475,000 in exchange for services in connection with the placement of \$9,000,000 of Senior Subordinated Notes, the placement of the Series A Preferred Stock, the

F-30

U.S. LEGAL SUPPORT, INC.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS--(CONTINUED)

negotiation of the Bank Credit Agreement and the acquisition of Looney and Klein Bury in January 1997. Pursuant to the terms of a letter agreement dated April 24, 1997, between GulfStar and the Company, GulfStar agreed to provide negotiation and other financial advisory services to the Company in connection with the Company's evaluation of acquisitions and will be paid advisory fees equal to 1.0% of the total purchase price of each acquisition as well as reimbursement of out-of-pocket expenses. GulfStar has received a total of \$33,000 under the letter agreement and will be entitled to an additional \$450,000 upon completion of the initial public offering.

12. SUPPLEMENTAL CASH FLOW INFORMATION:

The following represents supplemental noncash investing and financing activities:

<TABLE>

SEPTEMBER 30, 1997

<S>

Common control exchange of shares between Looney and the Company

<C>

Net assets acquired in	
connection with	
reorganization Issuance of Series B	\$207,763
Preferred Stock	2,046,667
Acquisition of Klein Bury	
Net assets acquired, net of cash	510,521
Grant of stock options	12,000
Issuance of a note	1 404 110
payable Issuance of Common	1,424,113
Stock	68,240
Acquisition of San	
Francisco Reporting Net assets acquired, net	
of cash	112,814
Issuance of Preferred Stock	231,250
Issuance of Common	231,230
Stock	12,243
Acquisition of G&G Net assets acquired, net	
of cash	152,012
Issuance of notes	
payableAcquisition of Legal	995,568
Enterprise	
Net assets acquired, net	
of cashGrant of stock options	625,707 59,500
Issuance of notes	39,300
payable	1,140,500
Acquisition of Ziskind Greene	
Net assets acquired, net	
of cash	4,259
Issuance of Common Stock	1,350,000
Acquisition of Block	, ,
Net assets acquired, net	(22, 404)
of cashIssuance of notes	(32,494)
payable	600,000
Acquisition of Elaine Dine	
Net assets acquired, net of cash	286,000
Issuance of Common	
StockGrant of stock options	650,000 350,000
Issuance of notes	330,000
payable	2,000,000
Deferred purchase price	500,000
Acquisition of Johnson	300,000
Group	
Net assets acquired, net of cash	199,868
Issuance of Common	133,000
Stock	1,211,046
Issuance of notes payable	245,760
Acquisition of Amicus One	,
Net assets acquired, net	262.000
of cashIssuance of Common	363,029
Stock	990,000
Issuance of a note	560 000
payable/TABLE>	560,000

F-31

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Looney & Company as of December 31, 1995 and 1996, and the related statements of operations, stockholder's equity and cash flows for each of the three years in the period ended December 31, 1996. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Looney & Company as of December 31, 1995 and 1996, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 1996 in conformity with generally accepted accounting principles.

COOPERS & LYBRAND L.L.P.

DECEMBER 31,

Houston, Texas September 5, 1997

F-32

LOONEY & COMPANY

BALANCE SHEET

<TABLE> <CAPTION>

	2202112.	DI. 01,
100770		1996
ASSETS		
<\$>	<c></c>	<c></c>
Current assets: Cash	\$	\$20,605
Accounts receivable: Trade, net of allowance of \$142,581 and \$223,331,		
respectively	2,095,032	1,307,330
Related parties		319,302
Prepaid expenses and other current assets		45,926
Deferred income taxes		26,742
Total current assets	2,220,612	1,719,905
Property and equipment, net	590,101	426,296
Other assets	51,041	27 , 500
Deferred income taxes		18,500
Markal acceptance		
Total assets		\$2,192,201
LIABILITIES AND STOCKHOLDER'S EQUITY		
Current liabilities:		
Accounts payable	\$452,125	\$183,389
Accrued liabilities	897,265	1,140,163
Income taxes payable	123,303	
Deferred income taxes	15,188	
Current maturities of long-term obligations		529,413
Total current liabilities		
Long-term obligations, net of current maturities		131,473
Deferred income taxes	122,274	
Commitments and contingencies		
Stockholder's equity:		
Common stock, \$1 par value, 100,000 shares authorized, 1,000 shares issued and outstanding	1 000	1,000
Retained earnings		206,763
Recained earnings		206,763
Total stockholder's equity		207,763
Total liabilities and stockholder's equity	\$2,861,754	\$2,192,201

 | ======= || // TUDHE/ | | |
The accompanying notes are an integral part of the financial statements.

F-33

LOONEY & COMPANY

STATEMENT OF OPERATIONS

CAFILON		DED DECEMBE		NINE MONTHS ENDED SEPTEMBER 30,
		1995		
<s> Revenues Cost of services</s>	<c> \$8,362,920 5,589,599</c>			(UNAUDITED) <c> \$5,886,022 3,725,852</c>
Gross profit Selling, general and administrative expenses Depreciation	2,773,321 3,042,999 223,680	1,970,310	2,828,607 2,351,669 212,277	2,160,170
Operating income (loss) Interest expense	(493,358) 184,414	229,969		
Income (loss) before income taxes	(677,772) (182,979)	909,901 327,177		
Net income (loss)	\$ (494,793) ======	\$ 582,724	\$ 15,699	\$ 338,864 ======

The accompanying notes are an integral part of the financial statements.

F-34

LOONEY & COMPANY

STATEMENT OF STOCKHOLDER'S EQUITY

<TABLE> <CAPTION>

CONTITUM		RETAINED EARNINGS	TOTAL STOCKHOLDER'S EQUITY
<s> Balance as of January 1, 1994 Net loss</s>	\$1,000		
Balance as of December 31, 1994 Net income		(391,660) 582,724	(390,660) 582,724
Balance as of December 31, 1995 Net income		191,064 15,699	192,064 15,699
Balance as of December 31, 1996	\$1,000	\$206,763	\$207 , 763

The accompanying notes are an integral part of the financial statements.

F-35

LOONEY & COMPANY

STATEMENT OF CASH FLOWS

	YEAR ENDE	NINE MONTHS ENDED SEPTEMBER 30,		
	1994	1995	1996	1996
<pre><s> Cash flows from operating</s></pre>	<c></c>	<c></c>	<c></c>	(UNAUDITED) <c></c>
activities: Net income (loss) Adjustments to reconcile net	\$(494,793)	\$582,724	\$15,699	\$338,864

income (loss) to net cash				
<pre>provided by operating activities:</pre>				
Depreciation	223,680	230 353	212,277	159,208
Provision for doubtful				
accounts	28,566			
Deferred income taxes		75,272	(182,704)	(19,639)
Loss on disposal of equipment.	47,345			
Changes in operating assets and liabilities:				
Accounts receivable	(161, 513)	(256, 538)	706.952	982,208
Other receivables, related	(101/010)	(200,000)	.00,302	302,200
parties		34,407	(249,771)	(217,235)
Prepaid expenses and other				
current assets	(12,116)	(1,199)	10,123	(96,102)
Accounts payable and accrued				
liabilities		(305 , 706)		
Income taxes payable			(123,303)	(33,967)
Other assets	148,305	3,010		19 , 226
Not sook manided by				
Net cash provided by operating activities	95 315	568 399	569 702	519,462
operating activities	95,315			319,402
Cash flows from investing activities:				
Purchase of property and				
equipment	(182,331)	(7,654)	(51,028)	(41,180)
- 1				
Net cash used in investing				
activities	(182,331)	(7,654)	(51,028)	(41,180)
Cash flows from financing				
activities:	070 517			
Proceeds from borrowings Principal payments on long-term	972 , 517			
obligations	(1 013 769)	(491 404)	(396 093)	(280 881)
Other			(101,976)	
Net cash provided by (used				
in) financing activities				
	87,016			
Increase in cash			20,605	95 , 425
Cash and cash equivalents at				
beginning of period				
Cash and cash equivalents at end				
of period	\$	\$	\$ 20,605	\$ 95,425
Cash paid for interest	\$252,906	\$229,969	\$238,251	
Cash paid for income taxes		120,000	297,590	
Non cash investing and financing				
activities:				
Equipment acquired under capital	100 105	44 501	10 051	
leases	123,126	44,504	13,051	

The accompanying notes are an integral part of the financial statements.

F-36

LOONEY & COMPANY

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Looney & Company (the "Company"), a Texas corporation, was founded in 1988 and operates in six Texas offices, providing litigation support services primarily to insurance companies, law firms and large corporations. The Company's primary business is court reporting, the transcription of spoken legal testimony into the written word, the retrieval of records used in conjunction with the investigation and litigation of legal proceedings, and copying services. Looney is the predecessor to U.S. Legal Support, Inc. ("USLS"). The interim financial statements for the nine months ended September 30, 1996 should be read in conjunction with the interim financial statements of USLS included elsewhere in this prospectus.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the nine months ended September 30, 1996,

reflect all adjustments that are, in the opinion of management, necessary or a fair presentation of the results for the period. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

Cash is maintained in two banks. The balances, at times, may exceed federally insured amounts although management believes that risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statement of operations.

Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to its customers. An allowance is provided for anticipated bad debts based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

Income Taxes

Deferred income taxes are provided for the accumulated temporary differences in the bases of assets and liabilities for financial reporting and income tax purposes using enacted tax rates and laws in effect in the years in which the differences are expected to reverse.

F-37

LOONEY & COMPANY

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

3. RECEIVABLES, RELATED PARTIES:

Receivables, related parties, consisted of the following at December 31:

<TABLE> <CAPTION>

	1995	1996
<\$>	<c></c>	<c></c>
Shareholder	\$69,531	
U.S. Legal Support, Inc		\$258,988
Klein, Bury & Associates		60,314
	\$69,531	\$319,302

</TABLE>

In 1996, the Company paid, on behalf of Klein, Bury & Associates ("Klein Bury") and U.S. Legal Support, Inc. ("USLS"), costs incurred related to the January 1997 acquisitions of the Company and Klein Bury by USLS (see Note 9).

4. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following at December 31:

	USEFU	JL LIVES	1995	1996
<\$>	<c></c>		<c></c>	<c></c>
Furniture, fixtures and				
equipment	5 to	7 years	\$1,304,264	\$1,367,342
Vehicles		5 years	56 , 640	21,782
			1,360,904	1,389,124
Less accumulated				
depreciation			(770,803)	(962,828)
			\$ 590,101	\$ 426,296
			=======	

The Company has entered into various capital leases. The leases were recorded upon their inception using the interest rate implicit in the lease agreements. The capitalized cost of leased office equipment was approximately \$581,000 and \$594,000 at December 31, 1995 and 1996, respectively.

F-38

LOONEY & COMPANY

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

5. LONG-TERM OBLIGATIONS:

Long-term obligations consisted of the following at December 31:

<TABLE> <CAPTION>

CAF I TON	DECEMBE	•
	1995	1996
<\$>	<c></c>	
Note payable to a bank under line of credit, providing borrowings up to \$1,500,000, with interest at the prime rate plus 2.25%, collateralized by substantially all		
assets of the Company not otherwise pledged Obligations under capital leases of certain equipment, due in monthly installments through July 2000, with	\$657 , 422	\$438,853
implicit interest rates ranging from 8.0% to 20% Notes payable to bank, due in monthly installments, including interest at 7% to 9.25%, through January 1997,	319,024	218,533
collateralized by certain equipment	83,089	3,500
Less current maturities	1,059,535	
	\$227 , 378	\$131,473 ======

</TABLE>

At December 31, 1996, future minimum payments under long-term obligations were as follows:

<TABLE> <CAPTION>

YEAR ENDED	NOTE PAYABLE	CAPITAL LEASES
IEAR ENDED	PAIABLE	LEASES
<\$>	<c></c>	<c></c>
1997		\$109,516
1998		83 , 560
1999		59,179
2000		9,093
	442,353	261,348
Less amounts representing interest		42,815
	\$442,353	\$218,533

</TABLE>

6. INCOME TAXES:

The provision for income taxes consisted of the following for the years ended December $31\colon$

<TABLE>

YEAR ENDED DECEMBER 31,

	1994	1995	1996
<\$>	<c></c>	<c></c>	<c></c>
Current	\$(223,242)	\$251,905	\$193,415
Deferred	40,263	75,272	(182,704)
	\$(182,979)	\$327,177	\$ 10,711
			=======
,			

</TABLE>

F-39

LOONEY & COMPANY

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

A reconciliation of the differences between income taxes computed at the U.S. federal statutory rate of 34% and the Company's reported provision (benefit) for income taxes follows:

<TABLE> <CAPTION>

	YEAR ENDED DECEMBER 31,		
	1994	1995	1996
<s></s>	<c></c>	<c></c>	<c></c>
Income tax provision (benefit) at statutory rate	\$(230,442)	\$309,366	\$8,979
benefit	.,	27,297	792
Nondeductible expenses and other	27,130	(9,486)	940
	\$(182,979) ======	\$327 , 177	\$10,711 =====

</TABLE>

The components of deferred income tax assets and liabilities were as follows:

<TABLE>

<CAPTION>

	DECEMBER	- /
	1995	1996
<\$>	<c></c>	<c></c>
Deferred tax assets:		
Accrued liabilities	,	\$84,884 82,632
Deferred tax assets Deferred tax liabilities:	107,086	167,516
Conversion from cash to accrual basis for tax reporting		
purposes	244,548	122,274
Net deferred tax asset (liability)	\$(137,462)	\$45,242
	=======	======

 | |

7. ACCRUED LIABILITIES:

Accrued liabilities consisted of the following:

<TABLE>

<CAPTION>

10.12 2 2 0 1 7	DECEMBER 31,		
		1996	
<\$>	<c></c>		
Customer overpayments	\$602,361	\$ 519,711	
Payroll	154,904	160,452	
Litigation settlements and other	140,000	100,000	
Ownership interests		360,000	
	\$897,265	\$1,140,163	
	======	========	
/ man = = = :			

</TABLE>

DECEMBER 31

The Company recorded a charge of \$360,000 in the fourth quarter of 1996 for the fair value of ownership interests granted to certain employees by the Company's shareholder. The obligation was satisfied in January 1997 in connection with the acquisition of the Company's stock (Note 9).

Customer overpayments arise primarily when customers make duplicate payments or payments in excess of billed amounts. The customers have generally denied the Company's refund attempts, which the management of the Company believes is due to the significant volume and relatively small amount of each individual billing. Legal counsel has advised the Company that any claims by third parties for overpayments are subject to statute-of-limitation laws and related interpretations, which vary by state, and the ultimate resolution of any such third-party claims, if made, is not certain.

In 1997, the Company paid all outstanding amounts owed under litigation settlements.

F-40

LOONEY & COMPANY

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

8. COMMITMENTS AND CONTINGENCIES:

Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

<TABLE>

<\$>	<c></c>
1997	\$274,314
1998	242,035
1999	197,163
2000	143,637
2001	73,110
	\$930,259

</TABLE>

Rent expense recorded in 1995 and 1996 totaled approximately \$295,000 and \$260,000, respectively. Certain rental agreements provide for additional rent based on the lessors' operating expenses.

Legal Proceedings

The Company is involved in certain lawsuits and claims arising in the normal course of business. In the opinion of management, uninsured losses, if any, resulting from the ultimate resolution of these matters will not have a material adverse effect on the Company's financial position, results of operations or cash flows.

9. FORMATION OF NEW COMPANY:

In October 1996, the Company's shareholder, along with an investment firm, formed U.S. Legal Support, Inc. ("USLS") to create a nationwide a leading provider of legal support and staffing services to law firms, insurance providers and major corporations. In January 1997, the Company's stock was exchanged for stock of USLS as part of a common control combination (see Note 1 of the U.S. Legal Support, Inc. financial statements).

F-41

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Klein, Bury & Associates, Inc. as of September 30, 1995 and December 31, 1996, and the related statements of income, stockholders' equity and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Klein, Bury & Associates, Inc. as of September 30, 1995 and December 31, 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

COOPERS & LYBRAND L.L.P.

SEPTEMBER 30. DECEMBER 31.

Houston, Texas August 15, 1997

F-42

KLEIN, BURY & ASSOCIATES, INC.

BALANCE SHEET

<TABLE>

		1995	1996
ASSETS			
<\$>	<c></c>		<c></c>
Current assets:			
Cash	\$		\$315,437
and \$235,823, respectively		,908 , 325	2,120,265
Total current assets	1,	908,325	2,435,702
Property and equipment, net		61,278	30,411
Other assets		4,405	4,405
Total assets	\$1,	974,008	\$2,470,518 ======
<caption></caption>			
LIABILITIES AND STOCKHOLDERS' EQUITY			
<\$>	<c></c>		<c></c>
Current liabilities:	_	042 577	¢1 167 707
Accounts payable	ş	843,577 4,296	\$1,167,787 4,638
Income taxes payable		4,290	140,482
Deferred income taxes		425,407	358,474
Total current liabilities	1,	273,280	1,671,381
Deferred income taxes		25,605	26,507
Stockholders' equity:			
Common stock, \$1 par value, 500 shares authorized,		F00	F00
issued and outstanding		500 30 , 000	500 30 , 000
Retained earnings		644,623	742,130
Recarried carnings			
Total stockholders' equity		675 , 123	772,630
Total liabilities and stockholders' equity	\$1,	974,008	\$2,470,518

 === | ====== | ======= |The accompanying notes are an integral part of the financial statements.

KLEIN, BURY & ASSOCIATES, INC.

STATEMENT OF INCOME

<TABLE> <CAPTION>

	YEAR ENDED SEPTEMBER 30, 1995	1996
<s> Revenues Cost of services</s>	<c> \$7,302,368</c>	<c> \$8,525,386 5,586,241</c>
Gross profit Selling, general and administrative expenses Depreciation	2,263,112	2,939,145 2,779,564 15,367
Income before income taxes	,	144,214 55,065
Net income	\$ 108,158 ======	\$ 89,149 ======

The accompanying notes are an integral part of the financial statements.

F-44

KLEIN, BURY & ASSOCIATES, INC.

STATEMENT OF STOCKHOLDERS' EQUITY

A DDTMTOMAT

попат

<TABLE> <CAPTION>

	COMMON STOCK	ADDITIONAL PAID-IN CAPITAL		TOTAL STOCKHOLDERS' EQUITY
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
October 1, 1994	\$500	\$	\$536,465	\$536 , 965
Capital contributions		30,000		30,000
Net income			108,158	108,158
September 30, 1995 Net income, three months ended	500	30,000	644,623	675,123
December 31, 1995			8,358	8,358
December 31, 1995	500	30,000	652,981	683,481
Net income			89,149	89,149
December 31, 1996	\$500 ====	\$30,000	\$742 , 130	\$772 , 630

 | | | |The accompanying notes are an integral part of the financial statements.

F-45

KLEIN, BURY & ASSOCIATES, INC.

STATEMENT OF CASH FLOWS

<TABLE>

<cap1< th=""><th>PION></th><th></th><th></th></cap1<>	PION>		
		YEAR ENDED SEPTEMBER 30, 1995	•
<s></s>		<c></c>	<c></c>
	flows from operating activities: income	\$108,158	\$89,149

Adjustments to reconcile net income to net cash provided by operating activities:	4.5 0.40	45.065
Depreciation	16,343	
Provision for doubtful accounts	86,801	64,366
Deferred tax expense	76 , 901	92,528
Accounts receivable	(453,406)	(428,690)
Accounts payable	12,451	
Accrued liabilities	141,336	, -
Net cash provided by operating activities	(11,416)	13,112
Cash flows from investing activities:		
Capital expenditures		(2,261)
Net cash used in investing activities		(2,261)
Cash flows from financing activities:		
Capital contribution	30,000	
Cash overdraft	(18,616)	
Net cash provided by financing activities	11,384	
Net cash provided by financing activities		
Increase (decrease) in cash	(32)	10,851
Cash and cash equivalents at beginning of year	32	304,586
Cash and cash equivalents at end of year	\$	\$315,437
	======	=======

The accompanying notes are an integral part of the financial statements.

F-46

KLEIN, BURY & ASSOCIATES, INC.

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Klein, Bury & Associates, Inc., (the "Company"), a Florida corporation, was founded in 1977 as a court reporting business based in Miami, Florida, with four additional offices in Florida. The Company provides general court reporting services, the transcription of spoken legal testimony into the written word, and has particular expertise in handling cases involving medical malpractice.

The Company changed its fiscal year end for reporting purposes from September 30 to December 31. The results of operations for the three months ended December 31, 1996 have been included in the statement of stockholders' equity.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

Cash is maintained at one bank. The balances, at times, may exceed federally insured amounts, although management believes that risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statement of income.

Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to its customers. An allowance is provided for anticipated bad debts, based

primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

Income Taxes

Deferred income taxes are provided for the accumulated temporary differences in the bases of assets and liabilities for financial reporting and income tax purposes using enacted tax rates and laws in effect in the years in which the differences are expected to reverse.

F-47

KLEIN, BURY & ASSOCIATES, INC.

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

	USEFUL LIVES	SEPTEMBER 30, 1995	DECEMBER 31, 1996
<s></s>	<c></c>	<c></c>	<c></c>
Leasehold improvements Furniture, fixtures and	5 years	\$26,085	\$26,085
equipment	5 to 7 years	116,393	118,654
Less accumulated depreciation		142,478 (81,200)	144,739 (114,328)
		\$61,278	\$30,411
		======	=======

</TABLE>

4. INCOME TAXES:

The provision for income taxes consisted of the following:

<TABLE> <CAPTION>

	SEPTEMBER 30, 1995	DECEMBER 31, 1996
<s> Current Deferred</s>	<c> \$ 76,901</c>	<c> \$ (37,463) 92,528</c>
	\$76,901 =====	\$ 55,065 ======

</TABLE>

A reconciliation of the differences between income taxes computed at the U.S. federal statutory rate of 34% and the Company's reported provision for income taxes follows:

<TABLE> <CAPTION>

	SEPTEMBER 30, 1995	DECEMBER 31, 1996
<\$>	<c></c>	<c></c>
Income tax provision at statutory rate State tax provision, net of federal income	\$62,920	\$49,033
tax benefit	6,847	5,336
Nondeductible expenses and other	7,134	696
	\$76,901 =====	\$55,065 ======

</TABLE>

The components of deferred income tax assets and liabilities were as follows:

	SEPTEMBER 30, 1995	DECEMBER 31, 1996
<\$>	<c></c>	<c></c>
Deferred tax liabilities:		
Conversion from accrual to cash basis	\$425 , 407	\$358 , 474
Property and equipment	25 , 605	26 , 507
	\$451,012	\$384,981
		=======

5. RELATED-PARTY TRANSACTIONS:

During the years ended September 30, 1995 and December 31, 1996, the Company incurred rent expense of approximately \$41,000 and \$45,600, respectively, for office space leased from a partnership in which an interest is held by the Company's president and majority shareholder.

F-48

KLEIN, BURY & ASSOCIATES, INC.

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

6. COMMITMENTS AND CONTINGENCIES:

Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

<TABLE>

<\$>	<c></c>
1997	. \$281,398
1998	. 244,222
1999	. 225,855
2000	. 232,702
2001	. 177,381
	\$1,161,558
	========

</TABLE>

Rent expense for the years ended September 30, 1995 and December 31, 1996 totaled approximately \$240,000 and \$276,000, respectively. Certain rental agreements provide for additional rent based on the lessors' operating expenses.

Legal Proceedings

The Company is involved in certain lawsuits and claims arising in the normal course of business. In the opinion of management, uninsured losses, if any, resulting from the ultimate resolution of these matters will not have a material adverse effect on the Company's financial position, results of operations or cash flows.

7. SALE OF THE COMPANY:

In January 1997, the Company's stock was acquired by U.S. Legal Support, Inc.

F-49

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of G & G Court Reporters (a Sole Proprietorship) as of December 31, 1995 and 1996, and the related statements of income, owner's equity and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of G & G Court Reporters as of December 31, 1995 and 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

COOPERS & LYBRAND L.L.P.

Houston, Texas September 4, 1997

F-50

G & G COURT REPORTERS

BALANCE SHEET

<TABLE>

<caption></caption>	DECEMBI	ER 31,
		1996
ASSETS <s></s>	<c></c>	<c></c>
Current assets: Accounts receivable, net of allowance of \$15,500 in 1995 and 1996	\$281,925	\$251,103
Total current assets	1,693	•
Total assets	\$283,618	
<caption></caption>		
LIABILITIES AND OWNER'S EQUITY		
<\$>	<c></c>	<c></c>
Current liabilities:		
Cash overdraft		
Accounts payable		
Accrued liabilities		
Total current liabilities		15 , 626
Owner's equity	251,756	236,880
Total liabilities and owner's equity		\$252,506 ======

 | |The accompanying notes are an integral part of the financial statements.

F-51

G & G COURT REPORTERS

STATEMENT OF INCOME

<TABLE>

	YEAR ENDED DECEMBER 31,		ENDED	JANUARY 1, THROUGH	
	1995	1996	1996	1997	
				JDITED)	
<\$>	<c></c>	<c></c>	<c></c>	<c></c>	
Revenues	\$1,544,379	\$1,517,377	\$745,412	\$543,282	
Cost of services		837,774	•		
Gross profit	730,733	679 , 603	332 , 562	221,566	
expenses	281,725	286,861	138,900	98,112	
Depreciation			•	145	
Net income	\$ 448,690	\$ 392,452	\$193,517	\$123,309	

 ======= | ======= | ====== | ====== |The accompanying notes are an integral part of the financial statements.

F-52

G & G COURT REPORTERS

STATEMENT OF OWNER'S EQUITY

<table> <s> Balance, January 1, 1995. Distributions. Net income</s></table>	(462,087)
Balance, December 31, 1995. Distributions. Net income.	(407,328)
Balance, December 31, 1996. Distributions (unaudited)	(199,594)
Balance, May 19, 1997 (unaudited)	\$160,595 ======

</TABLE>

The accompanying notes are an integral part of the financial statements.

F-53

G & G COURT REPORTERS

STATEMENT OF CASH FLOWS

CCAPITON	DECEMBE	NDED R 31,	ENDED	
	1995	1996	1996	1997
			(UNAUD	ITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Cash flows from operating activities: Net income	\$448,690	\$392,452	\$193,517	\$123,309
activities: Depreciation	318	290	145	145
Accounts receivable	22,479	30,822	73,191	29,636
liabilities	4,203	6,248	4,619	45 , 795

Net cash provided by operating activities	475,690		271,472	•
Cash flows from financing activities: Cash overdraft Distributions	(462,087)	(22,484)	9,670 (281,142)	709
Net cash used in financing activities		(429,812)	. , ,	
Change in cash				
of period				
Cash and cash equivalents at end of period	\$ ======	\$ ======	\$ ======	\$ ======

 | | | |The accompanying notes are an integral part of the financial statements.

F-54

G & G COURT REPORTERS

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

G & G Court Reporters, a Sole Proprietorship (the "Company"), operates in California, providing litigation support services primarily for insurance companies, law firms and large corporations. The Company's primary business is court reporting, the transcription of spoken legal testimony into the written word.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The unaudited financial statements for the periods ended June 30, 1996 and May 19, 1997 reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash is maintained primarily in one bank. The balances, at times, may exceed federally insured amounts, although management believes that risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and depreciated on the straightline basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statements of income.

Income Taxes

The Company is organized as a sole proprietorship. No provision for federal income taxes is provided in these financial statements because the Company's income is included in the owner's separate income tax return.

Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to its customers. An allowance is provided for anticipated bad debts, based primarily on historical experience and current estimates.

The Company grants credit to various companies primarily in the legal and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

F-55

G & G COURT REPORTERS

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

NOAF I TONY			DECEMBER	R 31,
	USEFUL	LIVES	1995	1996
<pre><s> Furniture, fixtures and equipment Less accumulated depreciation</s></pre>		years		<c> \$51,181 (49,778)</c>
			\$ 1,693	\$ 1,403

</TABLE>

4. COMMITMENTS AND CONTINGENCIES:

Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

Operating Leases

Rent expense totalled approximately \$59,000 and \$56,000 for the years ended December 31, 1995 and 1996, respectively. The Company's office lease expired in June 1997. The Company entered a new office lease in June 1997 expiring July 2002 with annual rent of approximately \$44,000.

Legal Proceedings

The Company is involved in certain lawsuits and claims arising in the normal course of business. In the opinion of management, uninsured losses, if any, resulting from the ultimate resolution of these matters will not have a material adverse effect on the Company's financial position, results of operations or cash flows.

5. SALE OF THE BUSINESS:

On May 19, 1997, certain of the Company's net assets were sold to U.S. Legal Support, Inc.

F-56

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of San Francisco Reporting Service (a California Partnership) as of December 31, 1996 and May 14, 1997, and the related statements of income, partners' capital and cash flows for the year ended December 31, 1996 and the period from January 1, 1997 through May 14, 1997. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of San Francisco Reporting Service as of December 31, 1996 and May 14, 1997, and the results of its operations and its cash flows for the year ended December 31, 1996 and the period from January 1, 1997 through May 14, 1997 in conformity with generally accepted accounting principles.

COOPERS & LYBRAND L.L.P.

Houston, Texas September 19, 1997

F-57

SAN FRANCISCO REPORTING SERVICE

BALANCE SHEET

<TABLE> <CAPTION>

CAFILON	DECEMBER 31, 1996	1997
ASSETS		
<\$>	<c></c>	<c></c>
Current assets:		
Cash	\$	\$19,309
and 1997 Prepaid expenses and other current assets	112,772 5,296	197,543 8,762
1		
Total current assets	118,068 42,939	225,614 38,222
Total assets	\$161,007 ======	\$263,836 ======
<caption></caption>		
LIABILITIES AND PARTNERS' CAPITAL		
<\$>	<c></c>	<c></c>
Current liabilities:		
Note payable to bank	\$21,195	\$63,425
Accounts payable	65,703 11,000	144,166
Due to related parties	28,709	
Current maturities of capital lease obligation	2,477	2,477
Total current liabilities	129,084	210,068
Capital lease obligation	3,580	2,586
Commitments and contingencies Partners' capital	•	51,182
Total liabilities and partners' capital	\$161,007	\$263 , 836

 ====== | ====== |The accompanying notes are an integral part of the financial statements.

F-58

SAN FRANCISCO REPORTING SERVICE

STATEMENT OF INCOME

	SIX MONTHS	PERIOI
YEAR ENDED	ENDED	ENDED
DECEMBER 31,	JUNE 30,	MAY 14,
1996	1996	1997

		(UNAUDITED)	
<\$>	<c></c>	<c></c>	<c></c>
Revenues	\$1,139,538	\$586 , 754	\$467,424
Cost of services	745,336	406,321	289,842
Gross profit	394,202	180,433	177,582
expenses	356,284	168,696	141,682
Depreciation	11,800	5,900	4,717
Operating income	26,118 4,993	5,837 3,571	31,183
Net income	\$ 21,125	\$ 2,266	\$ 28,839
	=======	======	======

The accompanying notes are an integral part of the financial statements.

F-59

SAN FRANCISCO REPORTING SERVICE

STATEMENT OF PARTNERS' CAPITAL

<table></table>	
	<c></c>
Balance as of January 1, 1996	
Distributions	
Net income	21,125
Balance as of December 31, 1996	28,343
Distributions	, ,
Net income	28,839
Balance as of May 14, 1997	\$51,182
	======
/ PADIES	

 $</{\tt TABLE}>$

The accompanying notes are an integral part of the financial statements.

F-60

SAN FRANCISCO REPORTING SERVICE

STATEMENT OF CASH FLOWS

10111 1 1 0 117			
			MAY 14, 1997
	(U:	NAUDITED)	
<\$>	<c></c>	<c></c>	<c></c>
Cash flows from operating activities:			
Net income	\$21,125	\$2,266	\$28,839
Depreciation	11,800	5,900	4,717
Provision for doubtful accounts	2,000		
Accounts receivable	63,171	19,543	(84,771)
Prepaid expenses and other current assets	(176)	(133)	(3,466)
Accounts payable	(47,687)	(12,903)	67,463
Net cash provided by operating			
activities	50,233	14,673	12,782
Cash flows from investing activities:			
Capital expenditures	(18,573)	(18,573)	
Net cash used in investing activities	(18,573)		
Cash flows from financing activities:			

Change in note payable to bank	(39,412) (1,660) (19,000) 28,412	15,002 (536) (14,000) 3,434	42,230 (994) (6,000) (28,709)
Net cash provided by (used in) financing			
activities	(31,660)	3,900	6 , 527
Increase in cash			19,309
Cash and cash equivalents at beginning of			
period			
Cash and cash equivalents at end of period	\$	\$	\$19,309
	======	======	======
Cash paid for interest	\$ 4,993	\$3,571	\$ 2,344
	======	======	======

The accompanying notes are an integral part of the financial statements.

F-61

SAN FRANCISCO REPORTING SERVICE

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

San Francisco Reporting Service (the "Company"), a California partnership, operates in California, providing litigation support services primarily for insurance companies, law firms and large corporations. The Company's primary business is court reporting, the transcription of spoken legal testimony into the written word. An additional component of the Company's litigation support services is the retrieval of records used in conjunction with the investigation and litigation of legal proceedings. The Company also provides copying services. On May 14, 1997, certain of the Company's net assets were sold to U.S. Legal Support, Inc.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the six month period ended June 30, 1996, reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

Cash is maintained in one bank. The balances, at times, may exceed federally insured amounts although management believes the risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and depreciated on the straightline basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized, while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statement of operations.

Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to its customers. An allowance is provided for anticipated bad debts, based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

No provision for federal income taxes is provided in these financial statements, because the Company's income or loss is included in the partners' separate income tax returns.

F-62

SAN FRANCISCO REPORTING SERVICE

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

	USEFUL	LIVES	DECEMBER 31, 1996	MAY 14,
<\$>	<c></c>		<c></c>	<c></c>
Furniture, fixtures and equipment	5 to 7	years	\$60,603	\$60,603
Less accumulated depreciation			(17,664)	(22,381)
			\$42,939	\$38,222
				======

</TABLE>

4. NOTE PAYABLE TO BANK:

At December 31, 1996, the note payable to bank represented amounts borrowed under a \$75,000 revolving line of credit agreement with interest at 3.75% above the bank's prime rate (8.25% at December 31, 1996), maturing in December 1997. The note was guaranteed by the partners. The note was repaid in May 1997.

5. CAPITAL LEASE OBLIGATION:

In 1996, the Company acquired office equipment for 7,717 financed by a long-term capital lease. Future minimum lease payments under the capital lease are as follows:

<TABLE>

<\$> May 15, 1997 through December 31, 1997	3,116
Less: Amount representing interest	5,713 (650)
Present value of net minimum capital lease payments Less: Current portion	
Long-term portion	\$2,586 =====

</TABLE>

6 . COMMITMENTS AND CONTINGENCIES:

Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

F-63

SAN FRANCISCO REPORTING SERVICE

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases and negotiated renewal options with lease terms in excess of one year as of December 31, 1996, are as follows: $\begin{tabular}{c} \begin{tabular}{c} \begin{tabular}{$

<\$>	<c></c>
May 15 through December 31, 1997	\$17,043
1998	36,156
1999	37,326
2000	28,602
	\$119,127
	=======

</TABLE>

Rent expense totaled approximately \$30,000 and \$14,000 for the year ended December 31, 1996 and for the period from January 1, 1997 through May 14, 1997, respectively.

7. RELATED PARTY TRANSACTIONS:

For the year ended December 31, 1996, the Company paid the partners approximately \$95,000 for court reporting services. The balance due to the partners for court reporting services at December 31, 1996, of approximately \$11,000 was included in accounts payable. There were no payments due to partners at May 14, 1997.

F-64

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Legal Enterprise, Inc. as of December 31, 1995 and 1996, and the related statements of income, stockholders' equity and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Legal Enterprise, Inc. as of December 31, 1995 and 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P.

Houston, Texas September 5, 1997

F-65

LEGAL ENTERPRISE, INC.

BALANCE SHEET

<TABLE>

		JUNE 30, 1997
<c></c>	<c></c>	(UNAUDITED)
\$	\$27,958	\$47,368
	1995 	

\$9,640, \$20,960 and \$18,086, respectively Prepaid expenses and other current assets	339,414 11,808	513,365 15,536	528,039 13,860
Total current assets Property and equipment, net Other assets	351,222 161,386 13,492	556,859 193,559 14,351	589,267 252,301 14,376
Total assets		\$764,769	\$855,944
LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities:			
Accounts payable	\$48,494 20,959 354,270	43,084	\$50,102 56,606 345,319
Total current liabilities	423 , 723	522 , 592 	452,027 38,615
Common stock, \$1 par value, 75,000 shares authorized, 2,000 shares issued and			
outstanding	48,000	2,000 48,000 192,177	2,000 48,000 315,302
Total stockholders' equity	102,377	242,177	365,302
Total liabilities and stockholders' equity			\$855,944

 ======= | | ====== |The accompanying notes are an integral part of the financial statements.

F-66

LEGAL ENTERPRISE, INC.

STATEMENT OF INCOME

<TABLE> <CAPTION>

CALITON	YEAR ENDED DECEMBER 31,		SIX MONTHS	
	1995	1996 1996 19		1997
			(UNAU	DITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$2,756,466	\$3,706,782	\$1,718,985	\$2,231,106
Cost of services	1,785,906	2,292,138	1,076,707	1,508,737
Gross profit	970,560	1,414,644	642 , 278	722,369
administrative expenses	830,261	1,206,578	554,495	559,412
Depreciation	39 , 826	40,579	16,487	
Operating income		167,487	•	
Interest expense	10,767	27 , 687	14,888	14,832
Net income	\$ 89,706	\$ 139,800	\$ 56,408	\$ 123,125

 | | | |The accompanying notes are an integral part of the financial statements.

F-67

LEGAL ENTERPRISE, INC.

STATEMENT OF STOCKHOLDERS' EQUITY

							PAID-		TOTAL
						COMMON	IN-	RETAINED	STOCKHOLDERS'
						STOCK	CAPITAL	EARNINGS	EQUITY
<s></s>						<c></c>	<c></c>	<c></c>	<c></c>
Balance	as	of	January	1,	1994	\$2,000	\$48,000	\$(37,329)	\$12,671

Net income			,	89,706
Balance as of December 31, 1995	2,000	48,000	52,377	102,377
Net income			139,800	139,800
Balance as of December 31, 1996	2,000	48,000	192,177	242,177
Net income (unaudited)			123,125	123,125
Balance as of June 30, 1997				
(unaudited)	\$2,000	\$48,000	\$315,302	\$365,302
	=====	======	=======	=======
< /map = 0.5				

The accompanying notes are an integral part of the financial statements.

F-68

LEGAL ENTERPRISE, INC.

STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

0.1.225.0	DECEMBE	ARS ENDED R 31,	FOR THE SIX MONTHS ENDED JUNE 30,		
		1996	1996	1997	
<\$>	20 2	<c></c>	(UNAUD	ITED)	
Cash flows from operating activities: Net income		\$139,800			
activities: Depreciation Provision for doubtful accounts Changes in operating assets and liabilities:	39 , 826 	40,579 11,320	16,487 11,320	25,000 (2,874)	
Accounts receivable Prepaid expenses and other	(166,197)	(185,271)	(93,382)	(11,800)	
current assetsAccounts payable and accrued	(915)	(3,728)	(3,750)	1,676	
liabilities Other assets	(13,392)	82,935 (859)		2,667 (25)	
Net cash provided by (used in) operating activities	(55,432)	84,776	52,231		
Cash flows from investing activities: Capital expenditures		(72,752)			
Net cash used in investing activities		(72,752)			
Cash flows from financing activities: Principal payments on notes payableshareholder Proceeds from notes payable		(62,037)	(43,326)	(81,002)	
shareholderOther		95,000 (17,029)			
Net cash provided by (used in) financing activities	112,762	15,934			
Increase (decrease) in cash Cash and cash equivalents at		27,958		19,410	
beginning of period	7,574			27 , 958	
Cash and cash equivalents at end of period		\$ 27 , 958			
Cash paid for interest	\$10,767	\$3,605	\$2,847	\$7 , 596	

 ~~-~~ | ~~-~~ | ~~-~~ | |The accompanying notes are an integral part of the financial statements.

LEGAL ENTERPRISE, INC.

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Legal Enterprise, Inc. (the "Company"), a California corporation, operates in California offices, providing litigation support services primarily for insurance companies, law firms and large corporations. The Company's primary business is the retrieval of records used in conjunction with the investigation and litigation of legal proceedings. The Company also provides copying services.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the six month periods ended June 30, 1996 and 1997 reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cach

Cash is maintained in one bank. The balances, at times, may exceed federally insured amounts although management believes the risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and depreciated on the straightline basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statements of operations.

Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to its customers. An allowance is provided for anticipated bad debts, based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

Income Taxes

The Company is an S corporation under the Internal Revenue Code and thus for federal tax purposes is not considered to be a tax paying entity but instead taxes are paid at the shareholder level. The provision for income taxes consists of California state income taxes.

F-70

LEGAL ENTERPRISE, INC.

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

DECEMBER 31,

USEFUL LIVES 1995 1996

<\$>	<c></c>	<c></c>	<c></c>
Furniture, fixtures and equipment	5 to 7 years	\$218,213	\$283,592
Less accumulated depreciation		(56 , 827)	(90,033)
		\$161,386	\$193,559
		=======	=======

4. NOTES PAYABLE -- SHAREHOLDER:

At December 31, 1996, the note payable--shareholder is due on demand. The interest rate on the loan is prime rate plus 1%.

In May 1997, the Company obtained an additional loan from a shareholder in the amount of \$46,385 related to the purchase of certain equipment which is pledged as collateral for the loan. The loan bears interest at 9.5% and is payable in monthly principal and interest installments to May 2002.

5. COMMITMENTS AND CONTINGENCIES:

Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

<TABLE>

<\$>	<c></c>
1997	\$101,016
1998	101,016
1999	101,016
2000	101,016
2001	53,495
	\$457,559
	=======

</TABLE>

Rent expense totaled approximately \$100,000 for the years ended December 31, 1995 and 1996. Certain rental agreements provide for additional rent based on the lessors' operating expenses.

6. EMPLOYEE BENEFITS:

The Company has adopted a contributory profit sharing plan pursuant to Internal Revenue Code Section 401(k) covering substantially all employees. Each year the Company determines, at its discretion, the amount of matching contributions. Contributions charged to operations was \$7,327 for the year ended December 31, 1996. There were no contributions charged to operations for the year ended December 31, 1995.

7. SALE OF THE BUSINESS:

On August 28, 1997, certain of the Company's net assets were sold to U.S. Legal Support, Inc. Summary financial information pertaining to the Company's operations from January 1, 1997 through the acquisition date is as follows:

<TABLE>

	(UNAUDITED)
<\$>	<c></c>
Revenues	\$3,050,086
Gross profit	985 , 178
Operating income	200,399
Income before income taxes	180,426

 |F-71

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Elaine P. Dine, Inc. as of March 31, 1996 and 1997, and the related statements of income, changes in stockholder's equity and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of

material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Elaine P. Dine, Inc. as of March 31, 1996 and 1997, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P.

Houston, Texas August 29, 1997

F-72

ELAINE P. DINE, INC.

BALANCE SHEET

<TABLE> <CAPTION>

CALITON	MAR	TUNE 20	
	1996	1997	
<s> ASSETS</s>		<c></c>	(UNAUDITED)
Current assets Cash and cash equivalents Accounts receivable Prepaid expenses and other current assets	295,907	1,227,258	1,075,126
Total current assets Property and equipment, net Other assets	125,821	123,044	112,195
Total assets		\$1,761,426	
LIABILITIES AND STOCKHOLDER'S EQUITY Current liabilities Accounts payable	\$165,383	\$529,871	\$550 , 846
Accrued liabilities		158,059	
Total current liabilities Commitments and contingencies Stockholder's equity Capital stock, no par value; 11,612 shares			
authorized issued and outstanding		13,112 1,060,384	
Total stockholder's equity	462,897	1,073,496	928,285
Total liabilities and stockholder's equity			

 ====== | ======= | ======= |The accompanying notes are an integral part of the financial statements.

F-73

ELAINE P. DINE, INC.

STATEMENT OF INCOME

NOAL LLOW		EARS ENDED H 31,	MONTHS I	HE THREE ENDED JUNE 30,
	1996	1997	1996	1997
			(UNA	JDITED)
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$3,503,580	\$4,657,760	\$694,650	\$1,182,334

Cost of services	2,030,058	2,640,374	357,959	635,942
Gross profitSelling, general and administrative	1,473,522	2,017,386	336,691	546,392
expenses Depreciation	1,210,636 52,183	1,120,540 26,461	153,380 10,628	210,923 10,849
Income before income taxes	210,703 22,621	870,385 96,563	172,683 18,539	324,620 33,089
Net income	\$ 188,082 ======	\$ 773,822 =======	\$154,144 ======	\$ 291,531
/ / M 3 D 1 D 3				

The accompanying notes are an integral part of the financial statements.

F-74

ELAINE P. DINE, INC.

STATEMENT OF STOCKHOLDER'S EQUITY

<TABLE> <CAPTION>

	STOCK	RETAINED EARNINGS	TOTAL STOCKHOLDER'S EQUITY
<pre><s> Balance on April 1, 1995 Distributions to stockholder Net income</s></pre>	<c> \$13,112</c>	<c> \$458,534 (196,831)</c>	<c> \$471,646 (196,831)</c>
Balance on March 31, 1996		(163,223)	•
Balance on March 31, 1997			1,073,496 (436,742)
Balance on June 30, 1997 (unaudited)		\$915,173	

() IIIDDD

The accompanying notes are an integral part of the financial statements.

F-75

ELAINE P. DINE, INC.

STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

<caption></caption>	FOR THE YE.	ARS ENDED	FOR THE MONTHS ENI 30,	DED JUNE
	1996	1997	1996	1997
			(UNAUD	ITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Cash flows from operating activities: Net income	\$188,082	\$773,822	\$154,144	\$291,531
Depreciation	52,183	26,461	10,628	10,849
Accounts receivable	189,229	(931,351)	14,105	152,132
Other assets	(5,195)	(73,768)	(73,468)	(188, 280)
Accounts payable	2,864	364,488	(165,383)	20,975
Accrued liabilities	14,018	113,022	(28,180)	(61,138)

Net cash provided by (used

<pre>in) operating activities Cash flows from investing activities:</pre>	441,181	272 , 674	(88,154)	226,069
Capital expenditures	(1,758)	(18,524)	(6,367)	
Net cash used in investing activities	(1,758)	(18,524)	(6,367)	
Cash flows from financing activities:				
Cash overdraft Distributions to stockholder		(163,223)	20,385 (171,258)	(436,742)
Net cash used in financing activities	(196,831)	(163,223)	(150,873)	(436,742)
Increase (decrease) in cash and cash equivalents	242,592	90 , 927	(245,394)	(210,673)
beginning of period	2,802	245,394	245,394	336,321
Cash and cash equivalents at the end of the period		\$336,321 ======		
Cash paid for income taxes				\$ 8,306

 | | | |The accompanying notes are an integral part of the financial statements.

F-76

ELAINE P. DINE, INC.

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Elaine P. Dine, Inc. (the "Company"), a New York corporation, founded in 1983, provides litigation recruitment services primarily to law firms and large corporations.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The unaudited financial statements for the periods ended June 30, 1996 and 1997 reflect all adjustments that are in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash and Cash Equivalents

Cash is maintained in one bank. The balance, at times, may exceed federally insured amounts although management believes the risk of loss is minimal. Cash equivalents are short-term highly liquid investments with maturities of 90 days or less.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Art work which is included in property and equipment is not depreciated. Expenditures for improvements that extend the life of such assets are capitalized, while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in operations.

Revenue Recognition

The Company records revenue when candidates accept a job offer. An allowance is provided for bad debts, based primarily on historical experience and

current estimates.

Concentration of Credit Risk

The Company grants credit to primarily law firms and large corporations which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses and such losses have been within management's expectations.

Income Taxes

The Company is an S corporation under Internal Revenue Code and thus for federal tax purposes is not considered to be a tax paying entity. Income taxes are paid at the shareholder level. Section 7519 of the Internal Revenue Code requires prepayment of taxes if a company elects to pay taxes for a period other than the required taxable calendar year. As the Company has elected not to pay taxes based on a calendar year, it has deposited \$74,201 for related prepayments as of March 31, 1997. No such deposits were made as of March 31, 1996.

Deferred state income taxes are provided for the accumulated temporary differences in the bases of assets and liabilities for financial reporting and income tax purposes using enacted tax rates and laws in effect in the years in which the differences are expected to reverse.

F-77

ELAINE P. DINE, INC.

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

Employee Benefit Plan

The Company has adopted a contributory profit sharing plan pursuant to Internal Revenue Code Section 401(k) covering substantially all employees. Each year the Company determines, at its discretion, the amount of matching contributions. Contributions charged to operations for the years ended March 31, 1996 and 1997 were \$5,253 and \$6,494, respectively.

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE>

(6)12 12 (10)		MARCH	31,
	USEFUL LIVES	1996	1997
<pre> <s> Furniture and fixtures Art Office equipment Leasehold improvements </s></pre>	5 to 7 years 5 to 7 years	46,925 107,549	•
Less accumulated depreciation		,	998,671 (875,627) \$123,044

</TABLE>

4. INCOME TAXES:

The provision for income taxes consisted of New York state income taxes as follows:

<TABLE>

(dil 110ii)	MARCH	31,
	1996	1997
<s> Current Deferred</s>		\$31,058
	\$22,621	\$96,563

</TABLE>

Temporary differences arise primarily from the conversion from accrual to

cash basis of accounting and depreciation.

5. COMMITMENTS AND CONTINGENCIES:

Operating Leases

At March 31, 1997, aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

<TABLE>

<\$> 1998	<c> \$82,223</c>
1999	- ,
	\$219,261

</TABLE>

6. SALE OF THE BUSINESS:

On September 17, 1997, certain of the Company's net assets were sold to U.S. Legal Support, Inc. Summary financial information pertaining to the Company's operations from January 1, 1997 through the acquisition date is as follows:

<TABLE> <CAPTION>

		(UNAUDITED
	<\$>	<c></c>
	Revenues	\$4,553,763
	Gross profit	2,510,436
	Operating income	1,057,269
	<pre>Income before income taxes</pre>	1,064,258

 | |F-78

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Burton House, Inc., d.b.a. Ziskind, Greene, Watanabe, & Nason, as of December 31, 1995 and 1996, and the related statements of operations, stockholder's deficit and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Burton House, Inc., d.b.a. Ziskind, Greene, Watanabe & Nason, as of December 31, 1995 and 1996 and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P.

Houston, Texas September 5, 1997

F-79

BURTON HOUSE, INC., D.B.A. ZISKIND, GREENE, WATANABE, & NASON

BALANCE SHEET

DECEMBE	R 31,	
		JUNE 30,
1995	1996	1997

<s> ASSETS</s>	<c></c>	<c></c>	(UNAUDITED) <c></c>
Current assets:			
Cash	\$1,936	\$22,901	\$116,514
\$27,500 and \$27,500, respectively	92,000	•	240,792
Deferred income taxes Prepaid expenses and other current assets	16 , 170 800	3,941 800	2,730 800
Total current assets	110,906	46,392	360,836
Other assets	9,360	9,735	9,735
Total assets		\$56 , 127	
LIABILITIES AND STOCKHOLDER'S DEFICIT Current liabilities:			
Accounts payable	111,973	89,354	
Income taxes payable		1,712	62 , 548
Total current liabilities Note payable to bank Commitments and contingencies	•	110,764 100,000	311,140 100,000
Stockholder's deficit: Common stock, no par value, 1,000,000 shares authorized, 8,500 shares issued and			
outstanding	18,493	18,493	18,493
Accumulated deficit	,	(173,130)	
Total stockholder's deficit		(154,637)	(40,569)
Total liabilities and stockholder's deficit			

 ====== | ====== | ====== |The accompanying notes are an integral part of the financial statements.

F-80

BURTON HOUSE, INC., D.B.A. ZISKIND, GREENE, WATANABE, & NASON

STATEMENT OF OPERATIONS

<TABLE> <CAPTION>

	FOR THE YEARS ENDED DECEMBER 31,			
	1995	1996	1996	1997
			•	DITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$1,442,257	\$1,841,309	\$1,026,686	\$1,178,643
Cost of services	964,069	1,125,329	654,674	636,005
Gross profit	478,188	715,980	372,012	542,638
administrative expenses	588,039	704 , 785	348,282	357 , 393
Operating income (loss)	(109.851)	11.195	23,730	185,245
Interest expense				
Income (loss) before income				
taxes	(117,495)	936	18,292	177,826
<pre>Income tax expense (benefit)</pre>	(23,644)	14,741	7,317	63,758
Net income (loss)	\$ (93,851)	\$ (13,805)	\$ 10,975	\$ 114,068

 | | | |The accompanying notes are an integral part of the financial statements.

F-81

BURTON HOUSE, INC., D.B.A. ZISKIND, GREENE, WATANABE, & NASON

STATEMENT OF STOCKHOLDER'S DEFICIT

		DEFICIT	TOTAL STOCKHOLDER'S DEFICIT
<pre><s> Balance as of January 1, 1995 Net loss</s></pre>	\$18,493		\$(46,981)
Balance as of December 31, 1995 Net loss			
Balance as of December 31, 1996 Net income (unaudited)		(173,130) 114,068	, - , ,
Balance as of June 30, 1997 (unaudited)	\$18,493 ======	\$(59,062) =====	\$(40,569) ======

The accompanying notes are an integral part of the financial statements.

F-82

BURTON HOUSE, INC., D.B.A. ZISKIND, GREENE, WATANABE, & NASON

STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

<caption></caption>	YEARS ENDED DECEMBER 31,			ONTHS ENDED	
		1996		1997	
			(UNAUD		
<\$>	<c></c>	<c></c>	<c></c>	<c></c>	
Cash flows from operating activities: Net income (loss)	\$(93,851)	\$(13,805)	\$10,975	\$114,068	
Provision for doubtful accounts Changes in operating assets and liabilities:		27 , 500			
Accounts receivable					
Deferred income taxes	(16,170)	12,229			
Income taxes payable	(9,340)	1,712		60,836	
Prepaid expenses and other current					
assets		(375)			
Accounts payable and accrued liabilities	(101,407)	(52,046)			
Net cash provided by (used in) operating activities Cash flows from investing activities Cash flows from financing activities:		20 , 965 	7,005 	93,613	
Proceeds from note payable	100,000				
Increase in cash	1,394 542	20,965	7,005 1,936	93,613 22,901	
Cash at end of period	\$ 1,936	\$ 22,901	\$ 8,941	\$116,514	
Cash paid for interest	\$ 7,644	\$ 10,259	\$ 5,438	\$ 7,419	
Cash paid for taxes		\$ 800	\$ 800	\$ 8,957	

 ======= | ======= | ===== | |The accompanying notes are an integral part of the financial statements.

F-83

BURTON HOUSE, INC., D.B.A. ZISKIND, GREENE, WATANABE, & NASON

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Burton House, Inc., d.b.a. Ziskind, Greene, Watanabe, & Nason (the "Company"), a California corporation, was founded in 1982 and operates in California, providing placement services primarily for law firms and corporations.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The unaudited financial statements for six months ended June 30, 1996 and 1997 reflect all adjustments that, in the opinion of management, are necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

Cash is maintained in one bank. The balance, at times, may exceed federally insured amounts although management believes the risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets which was 5 years. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in operations. All property and equipment is fully depreciated.

Revenue Recognition

The Company recognizes revenue when candidates accept a job offer. An allowance is provided for anticipated bad debts, based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit primarily to law firms and major corporations which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

Federal Income Taxes

Deferred income taxes are provided for the accumulated temporary differences in the bases of assets and liabilities for financial reporting and income tax purposes using enacted tax rates and laws in effect in the years in which the differences are expected to reverse. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be realized.

F-84

BURTON HOUSE, INC., D.B.A. ZISKIND, GREENE, WATANABE, & NASON

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE>

		DECEMBER 31,		
	USEFUL LIVES	1995	1996	
<pre><s> Furniture and fixtures Office equipment Leasehold improvements</s></pre>	5 years	<c> \$47,429 33,734 7,576</c>	<c> \$47,429 34,234 7,576</c>	

	\$		\$	
Less accumulated depreciation	(88)	3,739)	(89	,239)
	88	739	89	,239

4. ACCRUED LIABILITIES:

Accrued liabilities consisted of the following:

<TABLE> <CAPTION>

	DECEMBI	ER 31,
	1995	1996
<s> Commissions Compensationstockholder</s>		\$10,313
	\$111,973	\$89,354

</TABLE>

5. NOTE PAYABLE TO BANK:

The note payable to a bank under a line of credit provides borrowings up to \$100,000 with interest at the prime rate plus 1.50%, collateralized by the assets of the Company. The line of credit has no set maturity date. Based on the terms of the agreement, the principal balance of the line of credit must be fully repaid in twenty-four equal consecutive monthly installments, together with accrued monthly interest and any other charges, beginning approximately 30 days after the bank terminates the Company's right to obtain loans under the existing agreement. Until terminated, the Company is required only to pay interest.

F-85

BURTON HOUSE, INC., D.B.A. ZISKIND, GREENE, WATANABE, & NASON

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

6. INCOME TAXES:

The significant components of the Company's deferred tax assets and liabilities were as follows:

<TABLE>

<CAPTION>

CAPITON	DECEMBI	ER 31,
	1995	1996
<\$>	<c></c>	<c></c>
Deferred tax assets:		
Allowance for bad debts	17,194	6,894
Accided Habilities	51,170	3,010
Total deferred tax assets	48,370	20,129
Deferred tax liabilities:		
Accounts receivable	32,200	•
Total deferred tax liabilities		16,188
Net deferred income taxes		

The provision (benefit) for income taxes consisted of the following:

<TABLE>

Note 11019	DECEMBE	R 31,
	1995	1996
<\$>	<c></c>	<c></c>
Current		

Total.....\$ (23,644) \$14,741

</TABLE>

The reconciliation of the provision (benefit) for income taxes to the income tax expense (benefit) resulting from the application of the federal statutory tax rate to pretax income is as follows:

<TABLE>

	DECEMBE:	R 31,
	1995	1996
<\$>	<c></c>	<c></c>
Tax at statutory rate	\$(41,123)	\$328
Nondeductible travel and entertainment	6,341	13,893
State income taxes	800	520
Nondeductible life insurance premiums	1,325	
Other	9,013	
Total	\$(23,644)	\$14,741

</TABLE>

7. SALE OF THE BUSINESS:

On September 17, 1997, the Company sold certain of its net assets to U.S. Legal Support, Inc. Summary financial information pertaining to the Company's operations from January 1, 1997 through the acquisition date is as follows:

<TABLE> <CAPTION>

	(UNAUDITED)
<\$>	<c></c>
Revenues	 \$1,623,300
Gross profit	 690,005
Operating income	 223,936
Income before income taxes	 214,439

</TABLE>

F-86

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Jilio & Associates (a Sole Proprietorship) as of December 31, 1995 and 1996, and the related statements of operations, owner's equity and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Jilio & Associates as of December 31, 1995 and 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P

Houston, Texas September 5, 1997

F-87

JILIO & ASSOCIATES

BALANCE SHEET

	DECEM	BER 31,	SEPTEMBER 30,	
		1996		
<s> ASSETS</s>	<c></c>		(UNAUDITED)	
Current assets: Cash	\$22 , 982	\$	\$	
respectively Prepaid expenses and other current assets	10,205		17,939	
Total current assets	634,226 61,026	1,104,223	825,717 25,089	
Total assets	\$695,252		\$850,806	
LIABILITIES AND OWNER'S EQUITY Current liabilities: Cash overdraft				
Accounts payable			34,319 31,305	
Total current liabilities Commitments and contingencies	33 , 939	141,898	81,153	
Owner's equity	661,313	995,013	769 , 653	
Total liabilities and owner's equity		\$1,136,911		

 | | |F-88

JILIO & ASSOCIATES

STATEMENT OF OPERATIONS

<TABLE> <CAPTION>

	DECEMBI	ER 31,	SEPTEM	BER 30,
	1995	1996	1996	1997
			(UNAU	DITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$3,365,754	\$4,022,445	\$2,771,986	\$3,084,611
Cost of services	1,697,290	1,970,094	1,366,435	1,571,057
Gross profit	1,668,464	2,052,351	1,405,551	1,513,554
administrative expenses	874,734	1,062,058	742,606	847,846
Depreciation	16 , 706	17,401	13,506	9,642
Net income	\$ 777,024	\$ 972,892	\$ 649,439	\$ 656,066

 | | | |The accompanying notes are an integral part of the financial statements.

F-89

JILIO & ASSOCIATES

STATEMENT OF OWNER'S EQUITY

<table></table>	
<\$>	<c></c>
Balance as of January 1, 1995	\$671,702
Owner's distributions	(787,413)
Net income	777,024
Balance as of December 31, 1995	661,313
Owner's distributions	(639.192)

Net income	972,892
Balance as of December 31, 1996. Owner's distributions (unaudited). Net income (unaudited).	(881,426)
Balance as of September 30, 1997 (unaudited)	\$769,653

 |F-90

JILIO & ASSOCIATES

STATEMENT OF CASH FLOWS

DECEMBER 31, SEPTEMBER 30,

<TABLE> <CAPTION>

	DECELIE		OBI IBIID	
	1995	1996	1996	1997
			(UNAUD	ITED)
<\$>	<c></c>	<c></c>		<c></c>
Cash flows from operating activities:				
Net income		•	•	,
Depreciation	16,706			
Provision for doubtful accounts Loss on disposal of property and	41,256	64,921	44,547	72,543
equipment		13,428	2,375	
Accounts receivable Prepaid expenses and other	(5,291)	(565,297)		
current assets	(5,301)	7,397	2,397	(15, 131)
Accounts payable	(372)	9 , 530	1,784	(4,578)
Accrued liabilities	4,572	30,341		
Net cash provided by operating activities	828 , 594	550,613		936,028
Cash flows from investing activities: Capital expenditures	(20,931)	(2,491)	(2,493)	(2,043)
Net cash used in investing activities	(20,931)			
Cash flows from financing activities: Owner distributions	(787,413)		23,081	(52,559)
Net cash used in financing activities	(787,413)	(571,104)	(484,111)	(933 , 985)
Increase (decrease) in cash	20,250	(22,982)	(22,982)	
of period	2,732	22,982	22,982	
Cash and cash equivalents at end of period	\$ 22,982	\$	\$	\$
	=======			

 | | | |The accompanying notes are an integral part of the financial statements.

F-91

JILIO & ASSOCIATES

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Jilio & Associates (the "Company"), a California sole proprietorship, was

founded in 1984 and operates in Costa Mesa, California, providing litigation support services primarily for insurance companies and law firms. The Company's primary business is court reporting, the transcription of spoken legal testimony into the written word.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the nine month periods ended September 30, 1996 and 1997 reflect all adjustments which are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

Cash is maintained in one bank. The balance, at times, may exceed federally insured amounts although management believes that risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statement of operations.

Revenue Recognition

The Company recognizes revenue after the documents or records have been shipped. An allowance is provided for anticipated bad debts based primarily on historical experience and current estimates.

Income Taxes

The Company is a sole proprietorship and thus is not a tax-paying entity. Income taxes on the Company's earnings are paid by the owner.

Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for credit losses, and such losses have been within management's expectations.

F-92

JILIO & ASSOCIATES

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following: <TABLE> <CAPTION>

		DECEMBE	- '	
	USEFUL LIVES	1995	1996	
<\$>	<c></c>	<c></c>	<c></c>	
Property and equipment: Furniture and fixtures Office equipment and computers	-			
Less accumulated depreciation		144,753 (83,727)	90,567 (57,879)	
Total		\$61,026	\$32,688	

The Company had fully depreciated assets totaling approximately \$31,000 and \$2,000 at December 31, 1995 and 1996, respectively.

4. COMMITMENTS AND CONTINGENCIES:

Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

<TABLE>

<s> 1997</s>	60,546 10,416 10,416
2001	7,812
	\$219,917
	=======

</TABLE>

Rent expense in 1995 and 1996 totaled approximately \$94,000 and \$140,000, respectively. Certain rental agreements provide for additional rent based on the lessors' operating expenses.

Legal Proceedings

The Company is involved in certain lawsuits and claims arising in the normal course of business. In the opinion of management, uninsured losses, if any, resulting from the ultimate resolution of these matters will not have a material adverse effect on the Company's financial position, results of operations or cash flows.

F-93

JILIO & ASSOCIATES

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

5. LINE OF CREDIT:

The Company has a line of credit with a bank for up to \$100,000 at an interest rate of 10%. As of December 31, 1995 and 1996, there were no funds drawn on the line of credit.

6. SALE OF THE BUSINESS:

In August 1997, the Company's owner agreed to sell certain of the Company's net assets to U.S. Legal Support, Inc.

F-94

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Reporting Service Associates, Inc. as of December 31, 1995 and 1996, and the related statements of income, stockholder's equity and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing

standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Reporting Service Associates, Inc. as of December 31, 1995 and 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P.

Houston, Texas August 29, 1997, except as to the information presented in Notes 4 and 6, for which the date is September 12, 1997

F-95

REPORTING SERVICE ASSOCIATES, INC.

BALANCE SHEET

<TABLE> <CAPTION>

CAPTION	DECEMBER 31,		SEPTEMBER 30,	
	1995	1996	1997	
<\$> ASSETS		<c></c>	(UNAUDITED)	
Current assets Cash	\$54,304	\$268,118	\$29,221	
\$6,200, \$16,000 and \$24,000, respectively Prepaid expenses and other current assets		656 , 168 	2,570	
Total current assets	397,354 22,242	924,286	1,086,302 42,309	
Total assets	\$419,596	\$963,068		
LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities: Accounts payable	200,000	200,000		
Total current liabilities		398,075		
Commitments and contingencies Stockholders' equity: Common stock, \$10 par value, 100 shares authorized, issued and outstanding Retained earnings	63,963	563,993	959,882	
Total stockholder's equity	64,963		960,882	
Total liabilities and stockholder's equity	\$419,596	\$963,068	\$1,128,611	

 | _====== | _======= |The accompanying notes are an integral part of the financial statements.

F-96

REPORTING SERVICE ASSOCIATES, INC.

STATEMENT OF INCOME

<TABLE>

YEARS ENDED NINE MONTHS ENDED DECEMBER 31, SEPTEMBER 30,

	1995	1996	1996	1997
			(UNAUI	DITED)
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$1,905,831	\$3,012,153	\$2,171,005	\$3,067,261
Cost of services	1,156,566	1,682,021	1,179,007	1,356,939
Gross profit	749,265	1,330,132	991,998	1,710,322
administrative expenses	420,994	578,957	391,437	505,467
Depreciation	3,006	6,965	-,	6,124
Operating income	325,265 19,902	744,210 18,922	•	1,198,731 14,300
Net income	\$ 305,363	\$ 725,288 =======	\$ 581,093	\$1,184,431

</TABLE>

The accompanying notes are an integral part of the financial statements.

F-97

REPORTING SERVICE ASSOCIATES, INC.

STATEMENT OF STOCKHOLDER'S EQUITY

<TABLE> <CAPTION>

	TOTAL COMMON RETAINED STOCKHOLDER'S STOCK EARNINGS EQUITY
<pre><s> Balance as of January 1, 1995 Net income Distributions</s></pre>	305,363 305,363
Balance as of December 31, 1995 Net income Distributions	725,288 725,288
Balance as of December 31, 1996 Net income (Unaudited) Distributions (Unaudited)	1,184,431 1,184,431
Balance as of September 30, 1997 (Unaudite	d) \$1,000 \$959,882 \$960,882

</TABLE>

The accompanying notes are an integral part of the financial statements.

F-98

REPORTING SERVICE ASSOCIATES, INC.

STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

	DECEMBER 31,		SEPTEMBER 30,	
	1995	1996	1996	1997
			(UNAU	DITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Cash flows from operating activities: Net income	\$305,363	\$725,288	\$581,093	\$1,184,431
Depreciation	3,006	6,965	5,223	6,124
Provision for doubtful accounts Changes in operating assets and liabilities:	6,200	9,800	4,800	8,000
Accounts receivable Prepaid expenses and other	(63,575)	(322,918)	(272,295)	(406,343)
current assets				(2,570)

Accounts payable	10,533		41,619	(30,346)
Net cash provided by operating activities Cash flows from investing activities:	261,527			759 , 296
Capital expenditures	(25,248)	(23,505)	(19,811)	(9,651)
Repayment of note payable				(200,000)
Stockholder distributions	(272,323)	(225, 258)	(162,567)	(788,542)
Net cash used by financing				
activities	(272,323)	(225, 258)	(162,567)	(988,542)
Increase (decrease) in cash Cash and cash equivalents at the	(36,044)	213,814	178,062	(238,897)
beginning of period	90,348	54,304	54,304	268,118
Cash and cash equivalents at the end				
-	\$54,304	\$268,118	\$232,366	\$29 , 221
Cash paid for interest				\$14,300
4 (ma pa pa				

</TABLE>

The accompanying notes are an integral part of the financial statements.

F - 99

REPORTING SERVICE ASSOCIATES, INC.

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Reporting Service Associates, Inc. (the "Company"), a Pennsylvania corporation founded in 1982 and located in Philadelphia, provides litigation support services primarily for insurance companies, law firms and large corporations in the Philadelphia and Southern New Jersey areas. The Company's primary business is court reporting, the transcription of spoken legal testimony into the written word. An additional component of the Company's litigation support services is the retrieval of records used in conjunction with the investigation and litigation of legal proceedings. The Company also provides copying and videotaping services.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the nine month periods ended September 30, 1996 and 1997 reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

Cash is maintained in one bank. The balance, at times, may exceed federally insured amounts although management believes the risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized, while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statement of income.

Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to its customers. An allowance is provided for anticipated bad debts based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit primarily to insurance companies, law firms and major corporations which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

Income Taxes

The Company is an S corporation under the Internal Revenue Code and thus for federal tax purposes is not considered to be a tax paying entity. Income taxes are paid at the stockholder level.

F-100

REPORTING SERVICE ASSOCIATES, INC.

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

		DECEMBER 31,	
	USEFUL LIVES	1995	1996
<pre><s> Furniture, fixtures and equipment Computer hardware and software</s></pre>	-		
Less accumulated depreciation			48,753 (9,971)
		\$22,242 =====	\$38,782 =====

</TABLE>

4. NOTE PAYABLE:

The Company has outstanding a demand promissory note for \$200,000. The note is personally guaranteed by the sole stockholder of the Company, and collateralized by the stockholder's deposits held at the bank. Interest on unpaid borrowings accrues at the bank's prime rate and is payable monthly. In September 1997, the Company repaid the note.

5. COMMITMENTS AND CONTINGENCIES:

Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

<TABLE>

<\$>	<c></c>
1997	\$63,860
1998	73,148
1999	75,724
2000	39,660
2001	25,783
Thereafter	1,712
	\$279,887

</TABLE>

-------- 01

Rent expense totaled approximately \$35,000 in each of the years ended December 31, 1995 and 1996.

6. SALE OF THE BUSINESS:

In September 1997, the Company's stockholder agreed to sell certain of the Company's net assets to U.S. Legal Support, Inc.

F-101

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Kirby A. Kennedy & Associates (a Minnesota Partnership) as of December 31, 1995 and 1996, and the related statements of income, partners' capital and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Kirby A. Kennedy $\ensuremath{\text{\&}}$ Associates as of December 31, 1995 and 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P.

Houston, Texas August 29, 1997

F-102

KIRBY A. KENNEDY & ASSOCIATES

BALANCE SHEET

<TABLE> <CAPTION>

	DECEMBI	,	SEPTEMBER 30,	
	1995	1996	1997	
<s></s>	<c></c>		(UNAUDITED)	
ASSETS Current assets:				
Cash	133,372		257,126	
Total current assets	133,372 73,896	176,959	331,133 41,355	
Total assets	\$207,268		\$372,488	
LIABILITIES AND PARTNERS' CAPITAL Current liabilities:				
Cash overdraft	86,692 294 23,500	127,463 1,138	177,549 1,175 	
Total current liabilities				
Total liabilities and partners' capital	\$207,268	\$254,678 ======	\$372,488	

 | | |F-103

KIRBY A. KENNEDY & ASSOCIATES

STATEMENT OF INCOME

<TABLE> <CAPTION>

	DECEMBER 31,		SEPTEMI	BER 30,
	1995	1996	1996	1997
			J)	JNAUDITED)
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$1,629,448	\$1,866,355	\$1,455,153	\$1,427,674
Cost of services	1,094,237	1,164,405	880,242	846,706
Gross profit	535,211	701,950	574,911	580,968
administrative expenses	191,858	261,025	179,972	181,435
Depreciation	27,563	32,160	24,120	19,270
Operating income	315,790	•	370,819	
Interest expense	1,412	1,527	1,131	356
Net income	\$ 314,378 =======	\$ 407,238	\$ 369,688 ======	\$ 379,907

</TABLE>

The accompanying notes are an integral part of the financial statements.

F-104

KIRBY A. KENNEDY & ASSOCIATES

STATEMENT OF PARTNERS' CAPITAL

<table></table>	
<\$>	<c></c>
Partners' capital, January 1, 1995	\$59 , 859
Distributions	(299,398)
Net income	314,378
Partners' capital, December 31, 1995	74 , 839
Distributions	(363,137)
Net income	407,238
Partners' capital, December 31, 1996	118,940
Distributions (Unaudited)	(305,083)
Net income for the nine months ended	
September 30, 1997 (Unaudited)	379 , 907
Partners' capital, September 30, 1997	
(Unaudited)	\$193,764
	======

</TABLE>

The accompanying notes are an integral part of the financial statements.

F-105

KIRBY A. KENNEDY & ASSOCIATES

STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

DECEMBER 31,		SEPTEM	BER 30,
1995	1996	1996	1997
		(UNAU	DITED)
<c></c>	<c></c>	<c></c>	<c></c>

Cash flows from operating activities: Net income	\$314,378	\$407,238	\$369,688	\$379,907
Depreciation Changes in operating assets and liabilities:	27,563	32,160	24,120	19,270
Accounts receivable	(14,886)	(43,587)	(10,978)	(80,167)
Accounts payable		40,771		
Accrued liabilities	(6,001)	844		37
Net cash provided by operating				
activities	316,487	437,426		
Cash flows from investing activities:				
Capital expenditures	(27,810)	(35,983)	(5,472)	(3,536)
Net cash used in investing				
activities	(27,810)		(5,472)	
Cash flows from financing activities:				
Cash overdraft	(12,779)	(14,806)	(21,943)	(7,137)
Proceeds from note payable				
Principal payments on note payable		(23,500)		
Distributions	(299,398)	(363,137)	(261,789)	(284,453)
Net cash used in financing				
activities	(288,677)	(401,443)	(307,232)	(291,590)
Increase in cash			87,761	74,007
of period				
Cash and cash equivalents at end of	^	•	A 07 761	A 74 007
period		Ş =======		
Cash paid for interest		\$ 1,527		\$ 356
Supplemental schedule of noncash investing and financing activities: Distributions of automobiles to				
Partners at net book value	ş	\$	т	\$ 20,630

 ======= | ======= | ====== | ====== |F-106

KIRBY A. KENNEDY & ASSOCIATES

NOTES TO FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Kirby A. Kennedy & Associates, a Minnesota Partnership (the "Partnership"), was founded in 1979 and provides litigation support services primarily for law firms and insurance companies in the United States. The Partnership's primary business is court reporting, the transcription of the spoken legal testimony into the written word. The Partnership also provides copying services, video services, special exhibit handling and complete case management. Income is allocated 75% to Jeanne Kennedy and 25% to Kirby Kennedy.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the nine month periods ended September 30, 1996 and 1997 reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash in maintained at one bank. The balance, at times, may exceed federally insured amounts, although management believes the risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized, while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statement of income.

Income Taxes

No provision for federal income taxes is provided in these financial statements because the Partnership's income or loss is included in the partners' separate income tax returns.

Revenue Recognition

The Partnership recognizes revenue as the documents or records are delivered to its customers. An allowance is provided for anticipated bad debts based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Partnership grants credit primarily to law firms and insurance companies which may be affected by economic or other external conditions.

F-107

KIRBY A. KENNEDY & ASSOCIATES

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

		DECEMBER 31,		
	USEFUL LIVES	1995	1996	
<pre><s> Furniture, fixtures and equipment Vehicles</s></pre>	-			
Less accumulated depreciation			347,209 (269,490)	
		\$73,896 =====	\$77 , 719	

</TABLE>

In the third quarter of 1997, automobiles having a book value of \$20,630 at December 31, 1996 were distributed to the partners.

4. COMMITMENTS AND CONTINGENCIES:

Independent Contractors

The Partnership, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Partnership. The Partnership does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Partnership to control costs. In the event the Partnership were required to treat these court reporters as its employees, the Partnership could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

Operating Leases

Aggregate minimum rental commitments under a noncancelable operating lease

with a term in excess of one year are as follows:

<TABLE>

<\$>	<c></c>
1997	\$35,127
1998	35,127
1999	35,127
	\$105,381

</TABLE>

Rent expense totaled approximately \$35,000 for the years ended December 31, 1995 and 1996. The rental agreement provides for additional rent based on yearly increases in the lessors' operating expenses.

5. SALE OF THE BUSINESS:

In August 1997, the Partnership agreed to sell certain of its net assets to U.S. Legal Support, Inc.

F-108

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying combined balance sheet of the Johnson Court Reporting Group as of December 31, 1995 and 1996, and the related combined statements of income, stockholders' equity and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the combined financial position of the Johnson Court Reporting Group as of December 31, 1995 and 1996, and the combined results of their operations and their cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P.

Houston, Texas September 19, 1997

F-109

JOHNSON COURT REPORTING GROUP

COMBINED BALANCE SHEET

<TABLE> <CAPTION>

	DECEMBER 31,		TIME 20
		1996	JUNE 30, 1997
<s> ASSETS</s>	<c></c>	<c></c>	(UNAUDITED) <c></c>
Current assets:			
Cash	\$22,253	\$38,905	\$13,084
Accounts receivable	246,636	252,480	383,399
Prepaid expenses and other current assets	17,789	10,100	20,431
Total current assets	286,678	301,485	416,914
Property and equipment, net	185,821	169,140	168,119
Other assets	16,216	42,050	64,115
Total assets	\$488,715	\$512,675	\$649,148
	=======	=======	======

LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities: Accounts payable	\$73 , 121	\$94 , 558	\$169,121
Income taxes payable	9,152	89,027	101,561
Accrued liabilities	88,212	72,289	85,045
Deferred income taxes	63,370	72,077	119,341
Total current liabilities Notes payable, including related parties of	233,855	327,951	475 , 068
\$103,734, \$28,995 and \$15,911, respectively	133,099	68,265	77,113
Deferred income taxes	14,866		13,450
Commitments and contingencies Stockholders' equity:	·	,	·
Common stock	77,000	77,000	77,000
Retained earnings		25,928	6,517
Total stockholders' equity	106,895	102,928	83,517
Total liabilities and stockholders' equity		\$512,675	\$649,148

 | | |F-110

JOHNSON COURT REPORTING GROUP

COMBINED STATEMENT OF INCOME

<TABLE>

CCAPITON	FOR THE YEARS ENDED DECEMBER 31,		FOR THE SIX MONTHS ENDED JUNE 30,	
	1995	1996	1996	1997
			(UNAU)	DITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues Cost of services				
Gross profit	867,899	1,115,057	573 , 902	641,146
tive expenses	642,756 19,670	835,786 38,401	•	
Operating income	205,473 11,438		69,199 8,383	
Income before income taxes Income taxes	194,035 83,498		60,816 24,507	
Net income	\$ 110,537	\$ 132,748	\$ 36,309	\$ 109,567

 | | | |The accompanying notes are an integral part of the financial statements.

F-111

JOHNSON COURT REPORTING GROUP

COMBINED STATEMENT OF STOCKHOLDERS' EQUITY

<TABLE> <CAPTION>

	COMMON STOCK	RETAINED EARNINGS	TOTAL STOCKHOLDERS' EQUITY
<s> Balance as of January 1, 1995 Dividends Net income</s>		<c> \$22,923 (103,565) 110,537</c>	\$99,923 (103,565)
Balance as of December 31, 1995 Dividends Net income		(136,715) 132,748	132,748
Balance as of December 31, 1996	77,000	25 , 928	102,928

Dividends (unaudited)		(128,978) 109,567	(128,978) 109,567
Balance as of June 30, 1997 (unaudited)	\$77,000	\$6,517	\$83,517

 | | |F-112

JOHNSON COURT REPORTING GROUP

COMBINED STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

	FOR THE YEARS ENDED DECEMBER 31,		FOR THE	DED JUNE
		1996	1996	1997
<\$>	<c></c>	<c></c>	(UNAUD	ITED)
Cash flows from operating activities: Net income	\$110,537	\$132,748	\$36,309	\$109,567
Depreciation Deferred income taxes Changes in operating assets and liabilities:		38,401 7,372		26,080 47,183
Accounts receivable Prepaid expenses and other Accounts payable Income taxes payable Accrued liabilities	(20,481) 35,071 16,400	(18,145) 21,437 79,875 (15,923)	(16,062) 16,834 (10,093) 3,406 29,398	(32,396) 50,381 12,534 12,756
Net cash provided by operating activities	147,944	239,921		
Cash flows from investing activities: Capital expenditures	(102,347)	(21,720)	(10,813)	(25,059)
Net cash used in investing activities			(10,813)	
Cash flows from financing activities: Notes payable Dividends Other	62,275	(64,834) (136,715)	(43,444)	8,848 (128,978) 24,182
Net cash used in financing activities	(41,290)			
Increase (decrease) in cash	4,307	16,652	(6,427) 22,253	(25,821)
Cash and cash equivalents at end of period			\$ 15,826 ======	
Cash paid for interest		\$ 19,023	\$ 8,383 ======	\$ 5,896
Cash paid for taxes	\$ 3,764 ======		\$ 1,369	\$ 1,289

 | | | |The accompanying notes are an integral part of the financial statements.

F-113

JOHNSON COURT REPORTING GROUP

NOTES TO COMBINED FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS:

Johnson Court Reporting Group (the "Company"), operating through Goren of Newport, Inc., Rapidtext, Inc. and Medtext, Inc., provides litigation support services, closed-captioning services for the hearing impaired, remote classroom captioning services, medical transcription services, and scanning and imaging services.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Basis of Presentation

The accompanying combined financial statements include the accounts of Goren of Newport, Inc., Rapidtext, Inc. and Medtext, Inc. all of which are operated under common management. All intercompany amounts have been eliminated.

Preparation of Interim Financial Statements

The financial statements for the six month periods ended June 30, 1996 and 1997 reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Management Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

Cash is maintained in two banks. The balances, at times, may exceed federally insured amounts although management believes that risk of loss is minimal.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statement of income.

Income Taxes

The Company provides for deferred income taxes utilizing the liability method whereby deferred income taxes are recognized for the tax consequences in future years of differences in the tax bases of assets and liabilities and their financial reporting amounts based on enacted tax laws and statutory tax rates applicable to the periods in which the differences are expected to affect taxable income. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be realized.

Revenue Recognition

The Company recognizes revenue after the documents or records have been prepared for shipment or the services have been provided. An allowance is provided for anticipated bad debts based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal, medical and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations. During the six-month period ended June 30, 1997, approximately 16% of revenues was from one company.

F-114

JOHNSON COURT REPORTING GROUP

NOTES TO COMBINED FINANCIAL STATEMENTS-- (CONTINUED)

3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

			ECEMBER		
	USEFUL LIVES	1	995	199	6
<s></s>	<c></c>			<c></c>	
Property and equipment: Furniture and fixtures Office equipment and computers		25	8,852	277,	265
Less accumulated depreciation		26	 6,950 1,129)		807
Total			 5,821		

	=====				4. INCOME TAXES:					
The components of deferred income tax assets a following:	and liabilitie	es c	onsiste	d of	the					
					_					
			DECEM	BER 3						
			1995	19						
~~Deferred income tax assets~~										
Accrued liabilities				4 \$28						
Total deferred income tax assets			35**,**28							
Deferred income tax liabilities Accounts receivable Property and equipment			14,86	6 13	, 53					
Total deferred income tax liabilities				0 114	, 52					
Net deferred income tax liability				6 \$85	,60					
			=====	= ===						
The provision for income taxes consisted of the	ne following:									
					_					
			DECEM	BER 3						
			1995	19	96					
```     Current     Deferred ```					,72					
			\$83,49	8 \$89	, 09					
			=====	= ===	===					
The difference between the Company's provision amount that would have been derived by applying to pretax income for the years ended December 31 as follows:	the federal s	stat	utory t	ax ra						
			DECEME	ER 31						
			1995	199	6					
~~Tax at federal statutory rate~~										
</TABLE>

\$83,498 \$89,099

#### JOHNSON COURT REPORTING GROUP

#### NOTES TO COMBINED FINANCIAL STATEMENTS -- (CONTINUED)

#### 5. COMMITMENTS AND CONTINGENCIES:

#### Independent Contractors

The Company provides court reporting, captioning, transcription, and scanning and imaging services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent contractors are responsible for acquiring and operating their equipment. The use of independent contractors is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these independent contractors as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

#### Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

#### <TABLE>

<s></s>	<c></c>
1997	\$5 <b>,</b> 885
1998	6,147
1999	2,832
2000	2,832
2001	2,832
	\$20,528
	======

#### </TABLE>

Rent expense in 1995 and 1996 totaled \$57,030 and \$68,967, respectively. Certain rental agreements provide for additional rent based on the lessors' operating expenses.

#### Legal Proceedings

The Company is involved in certain lawsuits and claims arising in the normal course of business. In the opinion of management, uninsured losses, if any, resulting from the ultimate resolution of these matters will not have a material adverse effect on the Company's financial position, results of operations or cash flows.

## 6. NOTES PAYABLE:

The Company has lines of credit with two financial institutions for up to \$225,000 at an interest rate of 13%, collateralized by the Company's accounts receivable. These lines are drawn on when needed and are normally repaid within the fiscal year. As of December 31, 1995 and 1996, the Company had \$29,365 and \$39,270 borrowed under the lines of credit.

The Company has notes payable of \$103,734 and \$28,995 to shareholders at December 31, 1995 and 1996.

# 7. SALE OF COMMON STOCK:

On August 19, 1997, the Company's common stock was acquired by U.S. Legal Support, Inc. Summary financial information pertaining to the Company's operations from January 1, 1997 through the acquisition date is as follows:

# <TABLE>

CALITON.		
		(UNAUDITED)
	<\$>	<c></c>
	Revenues	\$1,471,072
	Gross profit	766,808
	Operating income	161,028
	<pre>Income before income taxes</pre>	154,482

  |  |F-116

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders

#### U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Amicus One Legal Support Services, Inc. as of December 31, 1996 and September 5, 1997 and the related statements of income, stockholders' equity and cash flows for the year ended December 31, 1996 and the period from January 1, 1997 through September 5, 1997. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Amicus One Legal Support Services, Inc. as of December 31, 1996 and September 5, 1997, and the results of its operations and its cash flows for the year ended December 31, 1996 and the period from January 1, 1997 through September 5, 1997 in conformity with generally accepted accounting principles.

Coopers & Lybrand 1.1.p.

COMBINED

Houston, Texas October 21, 1997

F-117

#### AMICUS ONE LEGAL SUPPORT SERVICES, INC.

#### BALANCE SHEET

<TABLE>

	1996	COMPANY SEPTEMBER 5, 1997
<\$>	<c></c>	
ASSETS		
Current assets:		
Cash	\$1,626	\$249 <b>,</b> 058
and \$100,000, respectively	513,608	321,991
Prepaid expenses and other current assets	11,398	1,467
Total current assets	526,632	572,516
Property and equipment, net	65,334	96,998
Other assets	34,014	15,032
Total assets	\$625 <b>,</b> 980	\$684,546
	======	=======
LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities:		
Current maturities of long-term obligations	\$24,400	\$253,978
Advance from U.S. Legal Support, Inc		277,043
Accounts payable	30,945	53,988
Accrued liabilities	52,467	33,742
Total current liabilities	107,812	618,751
Long-term obligations	10,423	2,774
Commitments and contingencies Stockholders' equity:		
Common stock	11,750	49,499
Retained earnings	495,995	13,522
Total stockholders' equity	507,745	63,021
Total liabilities and stockholders' equity	\$625 <b>,</b> 980	\$684,546

 ====== | ====== |The accompanying notes are an integral part of the financial statements.

## AMICUS ONE LEGAL SUPPORT SERVICES, INC.

## STATEMENT OF INCOME

<TABLE> <CAPTION>

COALITON	YEAR ENDED	COMBINED PREDECESSORS NINE MONTHS ENDED	
	DECEMBER 31, 1996	SEPTEMBER 30,	SEPTEMBER 5, 1997
<\$>	<c></c>	(UNAUDITED) <c></c>	<c></c>
Revenues  Cost of services		695,302	
Gross profit			
expenses Depreciation		562,027 22,323	598,918 23,607
Operating income			38,836 12,689
Income before income taxes			26,147 625
Net income	\$ 139,466 ======	\$ 108,617	\$ 25,522 ======

  |  |  |The accompanying notes are an integral part of the financial statements.

F-119

AMICUS ONE LEGAL SUPPORT SERVICES, INC.

STATEMENT OF STOCKHOLDERS' EQUITY

<TABLE> <CAPTION>

.0 2.20.0	COMMON STOCK	RETAINED EARNINGS	TOTAL STOCKHOLDERS' EQUITY
<\$>	<c></c>	<c></c>	<c></c>
Combined Predecessors:			
Balance as of January 1, 1996		(34,608)	(34,608)
Balance as of December 31, 1996	11,750	495,995	507,745
distribution of certain net assets to owners	37,749	(495,995)	(458,246)
Net income			25,522
Distributions		(12,000)	(12,000)
Balance as of September 5, 1997		\$ 13,522	\$ 63,021 ======

  |  |  |The accompanying notes are an integral part of the financial statements.

F-120

AMICUS ONE LEGAL SUPPORT SERVICES, INC.

STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

COMBINED COMBINED PREDECESSORS

	YEAR ENDED DECEMBER 31, 1996	NINE MONTHS ENDED SEPTEMBER 30, 1996	PERIOD ENDED SEPTEMBER 5, 1997
		(UNAUDITED)	
<\$>	<c></c>	<c></c>	<c></c>
Cash flows from operating activities: Net income	\$139,466	\$108,617	\$25,522
Depreciation  Provision for doubtful accounts  Changes in operating assets and liabilities:	39,580 24,152	22,323 24,152	23,607 90,160
Accounts receivable  Prepaid expenses and other	(81,766)	(30,565)	(325,334)
current assets	1,661	1,562	(69)
Other assets	17,366	(250)	(11,893)
Accounts payable	866	26,036	31,678
Accrued liabilities	17 <b>,</b> 339	14,747	(27,123)
Net cash provided by (used in)			
operating activities	158,664 	166,622 	(193,452)
Cash flows from investing activities: Capital expenditures Investment in note receivable from	(85 <b>,</b> 272)	(65,058)	(55,271)
officer		(4,927)	
Net cash used in investing activities	(85,272)	(69,985)	(55,271)
Cash flows from financing activities:			
DistributionsProceeds from borrowings	(34,608)	(33,928)	(12,000) 820,898
Repayment of borrowings	(38,647)	(23,789)	(311,926)
Net cash (used in) provided by financing activities	(73,255)	(57,717)	496 <b>,</b> 972
Increase in cash	137	38,920	248,249
of period	1,489	1,489	809
Cash and cash equivalents at end of period	\$ 1,626 ======	\$ 40,409 ======	\$249 <b>,</b> 058
Cash paid for interest	\$ 6,613	\$ 5,730 ======	\$ 12,689 ======
Cash paid for income taxes	\$ 9,040 =====	\$ 244	\$ 825 ======
Issuance of common stock for non-cash			
assets			\$157 <b>,</b> 906
4 (maps m)			

</TABLE>

The accompanying notes are an integral part of the financial statements.

F-121

AMICUS ONE LEGAL SUPPORT SERVICES, INC.

#### NOTES TO FINANCIAL STATEMENTS

# 1. ORGANIZATION AND BUSINESS:

Amicus One Legal Support Services, Inc. (the "Company"), a New York Subchapter S Corporation, was formed on January 1, 1997 through the contribution of certain assets at predecessor cost by three shareholders who together owned two court reporting businesses known as Cardinal Reporting Co., Inc. ("Cardinal") and AM Court Reporting, Ltd. ("AM"). The owners of the "combined predecessors" have presented combined financial statements of the combined predecessor for 1996 to reflect the financial position and results of operations for the periods on a comparable basis. The Company operates in New York providing litigation support services primarily for insurance companies and law firms. The Company's primary business is court reporting, the transcription of spoken legal testimony into written word.

The Company has 200 shares of no par value capital stock authorized, issued and outstanding at September 5, 1997. At December 31, 1996, Cardinal and AM were organized as a C Corporation and a S Corporation, respectively. Cardinal

and AM had 2,500 and 200 shares, respectively, of no par value capital stock authorized, issued and outstanding at December 31, 1996.

#### 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the periods ended September 30, 1996 and January 5, 1997 reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the period. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cach

Cash is maintained in several banks. The balance, at times, may exceed federally insured amounts although management believes that risk of loss is minimal

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statements of operations.

Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to customers. An allowance is provided for anticipated bad debts, based primarily on historical experience and current estimates.

Income Taxes

The Company is and AM was an S Corporation under the Internal Revenue Code and thus, for federal tax purposes, are not considered to be tax paying entities. Cardinal provided for deferred income taxes utilizing the liability method whereby deferred income taxes are recognized for the tax consequences in future years of

F-122

AMICUS ONE LEGAL SUPPORT SERVICES, INC.

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

differences in the tax bases of assets and liabilities and their financial reporting amounts based on enacted tax laws and statutory tax rates applicable to the periods in which differences are expected to affect taxable income. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be utilized.

Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal and insurance industries which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses, and such losses have been within management's expectations.

#### 3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

		DECEMBER 31	, SEPTEMBER 5,
	USEFUL LIVE	S 1996	1997
<\$>	<c></c>	<c></c>	<c></c>
Furniture and fixtures	5 to 7 year	s \$112,812	\$142,627
Office equipment and computers	5 years	19,274	44,730

	132,086	187,357
Less accumulated depreciation	(66,752)	(90,359)
Total	\$ 65,334	\$ 96,998
	=======	=======

</TABLE>

#### 4. COMMITMENTS AND CONTINGENCIES:

#### Independent Contractors

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

## Operating Leases

Aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

# <TABLE>

<\$>	<c></c>
September 6, 1997 through December 31, 1997	 \$31,612
1998	 96,221
1999	 89,719
2000	 67,144
2001	 68,838
Thereafter	 59,933
	\$413,467
	======

</TABLE>

Rent expense totaled approximately \$103,000 and \$68,000 for the year ended December 31, 1996 and the period ended September 5, 1997. Certain rental agreements provide for additional rent based on the lessors' operating expenses.

F-123

AMICUS ONE LEGAL SUPPORT SERVICES, INC.

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

# 5. LONG-TERM OBLIGATIONS:

Long-term obligations consisted of the following:

<TABLE> <CAPTION>

	1996	SEPTEMBER 5, 1997
<\$>	<c></c>	<c></c>
Line of credit with a bank for up to \$50,000 with interest at prime plus .25%. The line has no set maturity date	\$14,888	\$ <b>-</b> -
1997, collateralized by accounts receivable. The line has no set maturity date	 19,935	243,677 13,075
Less current portion	34,823 (24,400)	256,752 (253,978)
	\$10,423 =====	\$2,774 ======
/mapre>		

</TABLE>

The \$300,000 line of credit was repaid in September 1997 with funds advanced

by U.S. Legal Support Inc. (See Note 1).

6. SALE OF NET ASSETS:

On September 6, 1997, the Company's net assets were sold to U.S. Legal Support, Inc.

F-124

#### REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Block Court Reporting, Inc. as of December 31, 1995 and 1996, and the related statements of operations, stockholder's equity (deficit) and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Block Court Reporting, Inc. as of December 31, 1995 and 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand L.L.P.

DECEMBER 31,

Houston, Texas September 19, 1997

F-125

BLOCK COURT REPORTING, INC.

BALANCE SHEET

<TABLE> <CAPTION>

	DECEMBER 31,		- TIME 30	
		1996		
<s> ASSETS</s>	<c></c>	<c></c>	(UNAUDITED) <c></c>	
Current assets: Cash	\$994	\$1 <b>,</b> 852	\$4,437	
\$10,000, \$15,000 and \$15,000, respectively Deferred income taxes		207,911	6,074	
Total current assets  Property and equipment, net Other assets	125,271	89,830	84,489	
Total assets		\$306,161 ======		
LIABILITIES AND STOCKHOLDER'S EQUITY (DEFICIT) Current liabilities: Accounts payable	6,332 109,905 45,910	17,129 68,218 14,107	\$149,921 56,228 62,990 	
Total current liabilities		,	,	
authorized, issued, and outstanding Additional paid in capital	10 990		10 990	

Retained earnings (accumulated deficit)	90,996	26,369	(26,357)
Total stockholder's equity (deficit)	91,996	27 <b>,</b> 369	(25,357)
Total liabilities and stockholder's equity (deficit)	\$274,112	\$306,161	\$243,782

 ====== | ====== | ====== |F-126

BLOCK COURT REPORTING, INC.

## STATEMENT OF OPERATIONS

<TABLE> <CAPTION>

1995   1996   1996   1997	CAPTION		CARS ENDED	SIX MONTHS ENDED JUNE 30,		
CS>       CC>       C		1995	1996	1996	1997	
Revenues				(UNAUD	ITED)	
Cost of services	<\$>	<c></c>	<c></c>	<c></c>	<c></c>	
Gross profit	Revenues	. \$1,024,931	\$1,316,782	\$693,458	\$517,838	
Selling, general and administrative expenses.       524,864       740,279       339,110       331,936         Operating income (loss)       3,766       (85,335)       (22,408)       (76,553)         Interest expense       14,197       13,302       8,833       4,306         Loss before income taxes       (10,431)       (98,637)       (31,241)       (80,859)         Income tax benefit       (3,575)       (34,010)       (11,055)       (28,133)         Net loss       \$ (6,856)       \$ (64,627)       \$ (20,186)       \$ (52,726)	Cost of services	. 496,301	661,838	376,756	262,455	
Operating income (loss)	-	. 528,630	654,944	316,702	255,383	
Interest expense	3. 3	. 524,864	740,279	339 <b>,</b> 110	331,936	
Interest expense	Operating income (loss)	. 3,766	(85,335)	(22,408)	(76,553)	
Income tax benefit						
Net loss\$ (6,856) \$ (64,627) \$ (20,186) \$ (52,726)	Loss before income taxes	. (10,431)	(98,637)	(31,241)	(80,859)	
=======================================	Income tax benefit	. (3,575)	(34,010)	(11,055)	(28,133)	

 Net loss |  |  |  | \$ (52,726) ||  |  |  |  |  |  |
The accompanying notes are an integral part of the financial statements.

F-127
BLOCK COURT REPORTING, INC.

# STATEMENT OF STOCKHOLDER'S EQUITY (DEFICIT)

<TABLE>

CCAPITON	COMMON STOCK	ADDITIONAL PAID-IN CAPITAL	(ACCUMULATED	TOTAL STOCKHOLDER'S EQUITY (DEFICIT)
<s> January 1, 1995</s>	10 (100)	\$ 2,790 990		\$100,742
December 31, 1995		990 	90,996 (64,627)	91,996 (64,627)
December 31, 1996 Net loss (Unaudited)		990	26,369 (52,726)	27,369 (52,726)
June 30, 1997 (Unaudited)	\$ 10 =====	\$ 990 =====	\$(26,357) =====	\$(25,357) =====
-,				

The accompanying notes are an integral part of the financial statements.

#### STATEMENT OF CASH FLOWS

<TABLE> <CAPTION>

VO.22 2 2 0 AV	31	ECEMBER	FOR TH MONTHS JUNE	ENDED 30,
	1995	1996	1996	1997
<\$>	<c></c>		(UNAUD	ITED)
Cash flows from operating activities:  Net income				
operating activities:  Depreciation	36,464	35,441	11,271	9,388
Accounts receivable	33,555 (56,414)	156,573	152,600 9,920	65,697 (14,292) 39,099 (20,181)
Due from shareholder  Income taxes payable  Other assets		  	(48,934) (30,157) 37,879	
Net cash (used in) provided by operating activities		39,749		
Cash flows from investing activities: Capital expenditures	(21,483)			(4,047)
Net cash used in investing activities	(21,483)			(4,047)
Cash flows from financing activities: Net payments of long-term obligations	(414)	(38,891)	(26,554)	
Net cash used in financing activities	(414)	(38,891)		
Net (decrease) increase in cash				
of year	64,825	994		•
Cash and cash equivalents at end of year	\$ 994		\$ 69,990	\$ 4,437
Cash paid for interest		\$ 13,302	\$ 8,833	\$ 4,306
Cash paid for income taxes				<b></b>

  |  |  |  |The accompanying notes are an integral part of the financial statements.

F-129

# BLOCK COURT REPORTING, INC.

# NOTES TO FINANCIAL STATEMENTS

## 1. ORGANIZATION AND BUSINESS:

Block Court Reporting, Inc. (the "Company"), a District of Columbia corporation, is a court reporting business based in Washington, D.C. The Company provides general court reporting services, the transcription of spoken legal testimony into the written word as well as video captioning services to the Washington, D.C., Northern Virginia, and Baltimore, Maryland markets.

## 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The financial statements for the six month periods ended June 30, 1996 and 1997 reflect all adjustments that are, in the opinion of management, necessary

for a fair presentation of the results for the period. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

#### Management Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

#### Cash

Cash is maintained primarily in one bank. The balance, at times, may exceed federally insured amounts although management believes that risk of loss is minimal.

# Property and Equipment

Property and equipment is recorded at cost and is depreciated on the straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in the statements of operations.

#### Revenue Recognition

The Company recognizes revenue as the documents or records are delivered to customers. An allowance is provided for anticipated bad debts, based primarily on historical experience and current estimates.

#### Concentration of Credit Risk

The Company grants credit to various companies primarily in the legal industry which may be affected by economic or other external conditions. The Company maintains allowance for potential credit losses, and such losses have been within management's expectations.

F-130

BLOCK COURT REPORTING, INC.

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

#### 3. PROPERTY AND EQUIPMENT:

Property and equipment consisted of the following:

<TABLE> <CAPTION>

		DECEMBE	•
	USEFUL LIVES	1995	
<\$>	<c></c>	<c></c>	<c></c>
Property and equipment:  Vehicles	7 years		•
Less accumulated depreciation		161,735 (36,464)	159,868 (70,038)
Total		\$125,271 ======	\$89,830
/ /man = ma			

</TABLE>

The Company had fully depreciated assets totaling approximately \$19,000\$ at December 31, 1996.

# 4. COMMITMENTS AND CONTINGENCIES:

# ${\tt Independent\ Contractors}$

The Company, like most court reporting firms, provides court reporting services through the use of independent contractors, who are not employees of the Company. The Company does not pay or withhold federal or state employment taxes with respect to these independent contractors. Independent court reporters are responsible for acquiring and operating their court reporting

equipment. The use of independent contractors as court reporters is consistent with industry practice and allows the Company to control costs. In the event the Company were required to treat these court reporters as its employees, the Company could become responsible for the taxes required to be withheld and could incur additional costs associated with employee benefits and other employee costs.

Operating Lease

The Company leases office space under a noncancelable operating lease which expires on July 31, 1997. The remaining rental commitment under this lease at December 31, 1996 was approximately \$50,000. Subsequent to year-end, the Company entered into new noncancelable operating lease for office space commencing on October 1, 1997 and ending on September 30, 2000. The minimum rental commitment under this lease for each of the next four years approximates \$10,000, \$42,000, \$42,000, and \$32,000. Both agreements provide for additional rent based on increases determined from indices specified within the lease agreements. Additionally, the Company is required to pay its pro rata share of increases in real estate taxes.

Rent expense in both 1995 and 1996 totalled approximately \$97,000.

#### 5. INCOME TAXES:

The benefit for income taxes consisted of the following:

<TABLE> <CAPTION>

	FOR THE YEARS ENDED DECEMBER 31,		
	1995	1996	
<s> Current.  Deferred.</s>	\$(49,485)		
Total	\$ (3,575) ======	\$(34,010)	

F-131

BLOCK COURT REPORTING, INC.

NOTES TO FINANCIAL STATEMENTS--(CONTINUED)

The components of deferred income tax liabilities and assets are as follows:

<TABLE> <CAPTION>

	DECEMBER 31,	
	1995	1996
<\$>	<c></c>	<c></c>
Deferred tax assets:  Accounts payable  Deferred tax liabilities:	\$2,600	\$ 55,832
Accounts receivable	48,510	69 <b>,</b> 939
Net deferred income taxes	\$(45,910) ======	\$(14,107) ======

</TABLE>

## 6. LONG-TERM OBLIGATIONS:

The Company has a line of credit with a bank for up to \$25,000 at an interest rate equal to the prime lending rate plus two percentage points. As of December 31, 1996 and June 30, 1997, there was \$18,000 and \$23,000 drawn on this line of credit, respectively. These borrowings have been classified as current.

The Company has a promissory note with a bank at an interest rate equal to the prime lending rate plus one percentage point. At December 31, 1996, the maturities of the promissory note for each of the next two years approximated \$38,000\$ and \$15,000\$.

The Company leases vehicles under long-term capital leases which expire during 1997. At December 31, 1996, future minimum payments under these leases approximated \$13,000.

#### 7. RELATED PARTY TRANSACTIONS:

The Company subleases certain office space to an affiliate for approximately \$2,000 per month.

#### 8. SALE OF THE COMPANY:

On September 5th 1997, the Company's stock was acquired by U.S. Legal Support, Inc. Summary financial information pertaining to the Company's operations from January 1, 1997 through the acquisition date is as follows:

<TABLE>

	(UNUADITED)
<\$>	<c></c>
Revenues	. \$701,401
Gross profit	426,178
Operating loss	. (90,940)
Income before income taxes	. (93,448)

  |F-132

#### REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders U.S. Legal Support, Inc.:

We have audited the accompanying balance sheet of Commander Wilson, Inc. (a sole proprietorship) as of December 31, 1995 and 1996, and the related statements of income, owner's deficit and cash flows for the years then ended. These financial statements are the responsibility of management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Commander Wilson, Inc. (a sole proprietorship) as of December 31, 1995 and 1996, and the results of its operations and its cash flows for the years then ended in conformity with generally accepted accounting principles.

Coopers & Lybrand 1.1.p.

Houston, Texas September 22, 1997

F-133

COMMANDER WILSON, INC.

BALANCE SHEET

<TABLE> <CAPTION>

<caption></caption>	DECEMBE	R 31,	000000000000000000000000000000000000000	
		1996	DELIBRIDER SO,	
<s> ASSETS</s>	<c></c>	<c></c>	(UNAUDITED) <c></c>	
Current assets Prepaid expenses	\$1,969 	\$ 	\$ 	
Total current assets	1,969			
Total assets	\$1,969 =====	\$ ======	\$ ====	
LIABILITIES AND OWNER'S DEFICIT Current liabilities Accounts payable and accrued liabilities	\$36,000	\$37,303	\$	

Total current liabilities	36,000	37,303	
Commitments and contingencies			
Owner's deficit	(34,031)	(37,303)	
Total liabilities and owner's deficit	\$1,969	\$	\$
	======	======	====

  |  |  |F-134

COMMANDER WILSON, INC.

STATEMENT OF INCOME

<TABLE> <CAPTION>

	DECEMBER 31,		MONTHS ENDED	
		1996	1996	1997
				DITED)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$577,733	\$94,291	\$84,291	\$335,540
Cost of services	,			
Gross profit	477,163	94,291	84,291	335,540
expenses		68,528		
Net income	\$192 <b>,</b> 993	\$25,763	\$33,270 =====	\$273 <b>,</b> 777

  |  |  |  |The accompanying notes are an integral part of the financial statements.

F-135

COMMANDER WILSON, INC.

STATEMENT OF OWNER'S DEFICIT

<TABLE> <CAPTION>

CAFILON	OWNER'S EQUITY (DEFICIT)
<s> Balance on January 1, 1995  Net income  Distributions to owner, net</s>	192,993
Balance on December 31, 1995  Net income  Distributions to owner, net	25,763
Balance on December 31, 1996  Net income (unaudited)  Distributions to owner, net (unaudited)	273,777
Balance on September 30, 1997 (unaudited)	\$

The accompanying notes are an integral part of the financial statements.

F-136

COMMANDER WILSON, INC.

STATEMENT OF CASH FLOWS

	FOR THE YEARS ENDED DECEMBER 31,		FOR THE NINE MONTHS ENDED SEPTEMBER 30,	
		1996	1996	
			(UNAUD	
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Cash flows from operating activities: Net income	\$192,993	\$25,763	\$33,270	\$273,777
Change in operating assets and				
liabilities: Accounts payable Prepaid expenses			 1,969	(37,303)
Net cash provided by operating activities Cash flows from investing	232,824	29,035	35,239	236,474
activities				
Cash flows from financing activities: Distributions to owner	(232,824)	(29,035)	(35,239)	(236,474)
Net cash used in financing activities	(232,824)	(29,035)	(35,239)	(236, 474)
Change in cash				
beginning of period				
Cash and cash equivalents at the end of the period	\$	\$	\$	\$

</TABLE>

The accompanying notes are an integral part of the financial statements.

F-137

# COMMANDER WILSON, INC.

# NOTES TO FINANCIAL STATEMENTS

## 1. ORGANIZATION AND BUSINESS:

Commander Wilson, Inc. (the "Company"), a sole proprietorship, provides legal recruitment services to law firms and corporations primarily in Texas.

# 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES:

Preparation of Interim Financial Statements

The unaudited financial statements for the periods ended September 30, 1996 and 1997 reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the periods. Such adjustments are considered to be of a normal recurring nature unless otherwise identified.

Use of Estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Cash

The Company does not maintain a cash account. All cash receipts are deposited into the owner's bank account and recorded as distributions to the owner. All cash disbursements are made from the owner's bank account and recorded as contributions to the Company.

Property and Equipment

Property and equipment is recorded at cost and is depreciated on the

straight-line basis over the estimated useful lives of the assets. Expenditures for improvements that extend the life of such assets are capitalized, while maintenance and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the related accounts and any resulting gain or loss is included in operations. All property and equipment is fully depreciated.

Revenue Recognition

The Company records revenue when candidates accept a job offer. Non-refundable retainers to provide recruitment services on an exclusive basis are earned upon receipt. An allowance is provided for bad debts, based primarily on historical experience and current estimates.

Concentration of Credit Risk

The Company grants credit to primarily law firms and corporations which may be affected by economic or other external conditions. The Company maintains allowances for potential credit losses and such losses have been within management's expectations. During the years ended December 31, 1995 and 1996 approximately 53% and 93% of revenues was earned from two and five companies, respectively. During the six months ended June 30, 1997 approximately 48% of revenues was earned from two companies.

Income Taxes

The Company is organized as a sole proprietorship. No provision for federal income taxes is provided in these financial statements because the Company's income is included in the owner's tax return.

F-138

COMMANDER WILSON, INC.

NOTES TO FINANCIAL STATEMENTS-- (CONTINUED)

3. COMMITMENTS AND CONTINGENCIES:

Operating Leases

At December 31, 1996, aggregate minimum rental commitments under noncancelable operating leases with lease terms in excess of one year are as follows:

<TABLE>

<\$> 1997	
	\$28,147

</TABLE>

# 4. LITIGATION:

In September 1994, an independent contractor filed a lawsuit against the Company claiming breach of contract for approximately \$600,000. In April 1997, the Company settled the lawsuit for approximately \$36,000. Legal expenses of approximately \$57,000 are included in selling, general and administrative expenses for the year ended December 31, 1995.

5. SALE OF THE BUSINESS:

In September 1997, the owner agreed to sell the Company's business to U.S. Legal Support, Inc.

# 6. BANKRUPTCY:

In May 1996, the Company's owner filed a voluntary petition for Chapter XIII bankruptcy relief following the initiation of a lawsuit filed by a former independent contractor engaged by the Company. The Company's owner has submitted a Chapter XIII plan which has not yet been confirmed by the presiding bankruptcy court. The ultimate outcome and the effect, if any, of this matter on the Company cannot be determined.

F-139

[Logo of U.S. Legal Support, Inc. appears here]

A leading provider of court reporting,

# certified record retrieval and legal placement and staffing services

Court Reporting

[Photo of court reporter transcribing]

Certified Record Retrieval

[Photo of certified records retrieval]

Legal Placement and Staffing

[Photo of attorney working]

_____

No dealer, salesman or other person has been authorized to give any information or to make any representations other than those contained in this Prospectus and, if given or made, such information or representations must not be relied upon as having been authorized by the Company or any of the Underwriters. This Prospectus does not constitute an offer to sell or the solicitation of an offer to buy any security other than the Common Stock offered by this Prospectus, nor does it constitute an offer to sell or the solicitation of an offer to buy the shares of Common Stock to anyone in any jurisdiction in which such offer or solicitation is not authorized, or in which the person making such offer or solicitation is not qualified to do so, or to any person to whom it is unlawful to make such other or solicitation. Neither the delivery of this Prospectus nor any sale made hereunder shall, under any circumstances, create any implication that the information contained herein is correct as of any time subsequent to the date hereof.

-----

## TABLE OF CONTENTS

_____

<TABLE> <CAPTION>

	Page
<\$>	<c></c>
Prospectus Summary	3
Risk Factors	7
The Company	12
Use of Proceeds	14
Dividend Policy	14
Capitalization	15
Dilution	16
Selected Financial Data	17
Management's Discussion and Analysis of Financial Condition and Results	
of Operations	20
Business	31
Management	39
Certain Transactions	43
Principal Shareholders	45
Description of Capital Stock	47
Shares Eligible for Future Sale	49
Underwriting	51
Legal Matters	52
Experts	52
Available Information	53
Index to Financial Statements	F-1
\/ INDHH/	

Until , 1998 (25 days after the date of this Prospectus), all dealers effecting transactions in the Common Stock offering hereby, whether or not participating in this distribution, may be required to deliver a Prospectus. This is in addition to the obligation of dealers to deliver a Prospectus when acting as Underwriters and with respect to their unsold allotments or subscriptions.

_____

-----

_____

3,500,000 SHARES

COMMON STOCK

-----

PROSPECTUS

-----

NationsBanc Montgomery Securities, Inc.

Hambrecht & Quist

J.C. Bradford & Co.

, 1997

-----

#### PART TT

#### INFORMATION NOT REQUIRED IN PROSPECTUS

ITEM 13. OTHER EXPENSES OF ISSUANCE AND DISTRIBUTION.

The following table sets forth the costs and expenses, other than underwriting discounts and commissions, payable by the Company in connection with the sale of the securities being registered. All amounts are estimates except for the fees payable to the Commission and the NASD and the listing fee.

<TABLE>

	AMOUNT
<\$>	<c></c>
Securities and Exchange Commission registration fee	\$ 15,950.00
National Association of Securities Dealers, Inc. filing	
fee	5,732.50
NASDAQ listing fee	50,000.00
Printing and engraving expenses	350,000.00
Legal fees and expenses	600,000.00
Accounting fees and expenses	1,250,000.00
Blue sky fees and expenses	1,500.00
Transfer Agent fees	10,000.00
Miscellaneous	16,817.50
TOTAL	\$2,350,000.00

</TABLE>

ITEM 14. INDEMNIFICATION OF OFFICERS AND DIRECTORS.

The Registrant's Articles of Incorporation, as amended (the "Articles of Incorporation") and Bylaws require the Registrant to indemnify officers and directors of the Registrant to the fullest extent permitted by Article 2.02-1 of the Business Corporation Act of the State of Texas (the "TBCA"). The Articles of Incorporation and Bylaws are filed as Exhibits 3.1 and 3.2 to the Registration Statement. Generally, Article 2.02-1 of the TBCA permits a corporation to indemnify a person who was, is, or is threatened to be made a named defendant or respondent in a proceeding because the person was or is a director or officer if it is determined that such person (i) conducted himself in good faith; (ii) reasonably believed (a) in the case of conduct in his official capacity as a director or officer of the corporation, that his conduct was in the corporation's best interests, or (b) in other cases, that his conduct was at least not opposed to the corporation's best interests; and (iii) in the case of any criminal proceedings, had no reasonable cause to believe that his conduct was unlawful. In addition, the TBCA requires a corporation to indemnify a director or officer for any action that such director or officer is wholly successful in defending on the merits.

The Articles of Incorporation provide that a director of the Registrant will not be liable to the corporation for monetary damages for an act or omission in the director's capacity as a director, except to the extent not permitted by law. Texas law does not permit exculpation of liability in the case of (i) a breach of the director's duty of loyalty to the corporation or its shareholders; (ii) an act or omission not in good faith that involves intentional misconduct or a knowing violation of the law; (iii) a transaction

from which a director received an improper benefit, whether or not the benefit resulted from an action taken within the scope of the director's office; (iv) an act or omission for which the liability of the director is expressly provided by statute; or (v) an act related to an unlawful stock repurchase or payment of dividend.

The Form of Underwriting Agreement filed herewith as Exhibit 1.1, under certain circumstances, provides for indemnification by the Underwriters of the directors, officers and controlling persons of the Company.

The Company has purchased liability insurance policies covering the directors and officers of the Company, including, to provide protection where the Company cannot legally indemnify a director or officer

TT-1

and where a claim arises under the Employee Retirement Income Security Act of 1974 against a director or officer based on an alleged breach of fiduciary duty or other wrongful act.

ITEM 15. RECENT SALES OF UNREGISTERED SECURITIES.

All information set forth in this Item 15 gives effect to a 100-for-one stock split with respect to the Common Stock and Preferred Stock of the Company effected on December 16, 1996.

In connection with the initial capitalization of the Company, on October 2, 1996, the Company issued 150,000 shares of Common Stock to GulfStar Investments, Ltd. at a price of \$.01 per share for services rendered valued at \$1,500 and 843,840 shares of Common Stock to Richard O. Looney at a price of \$.01 per share for services rendered valued at \$8,438.40. On January 17, 1997, the Company issued: (i) 135,000 shares of Series A Convertible Preferred Stock to the Trust for Defined Benefit Plans of Zeneca Holdings Inc. at a price of \$1.00 per share for an aggregate sales price of \$135,000; (ii) 670,000 shares of Series A Convertible Preferred Stock to the Delaware State Employees' Retirement Fund at a price of \$1.00 per share for an aggregate sales price of \$670,000; and (iii) 195,000 shares of Series A Convertible Preferred Stock to the Trust for Defined Benefit Plans of ICI American Holdings Inc. at a price of \$1.00 per share for an aggregate sales price of \$195,000. On July 22, 1997, the Company issued 25,000 shares to David W. Pfleghar at a price of \$.01 per share for an aggregate sales price of \$250.00. All of such sales were completed without registration under the Securities Act in reliance upon the exemption provided by Section 4(2) of the Securities Act, no public offering being involved.

On January 17, 1997, the Company acquired all the issued and outstanding capital stock of Looney & Company in exchange for cash and the issuance of 2,000,000 (2,046,667 as a result of a post-closing adjustment on June 23, 1997) shares of Series B Convertible Preferred to Richard O. Looney. On that date, the Company also acquired all the issued and outstanding capital stock of Klein, Bury & Associates in exchange for cash and the issuance of 170,600 shares of Common Stock to Michael Klein. On April 3, 1997, the Company acquired all the assets of Cindi Rogers & Associates, Inc. in exchange for the payment of cash and the issuance of 5,000 shares of Common Stock to Cynthia A. Rogers. In connection with its acquisition of all the assets of San Francisco Reporting Service, the Company paid cash and issued 15,304 shares of Common Stock each to Jay Harbidge and Richard Posner on May 14, 1997, and 115,625 shares of Series C Convertible Preferred Stock each to Jay Harbidge and Richard Posner on June 23, 1997. On May 19, 1997, the Company acquired all the assets of G&G Court Reporters in exchange for cash and the issuance of a \$345,750 Convertible Subordinated Promissory Note and a \$691,750 Convertible Subordinated Promissory Note to the Giammanco Family Trust. On August 19, 1997, the Company acquired all the issued and outstanding capital stock of Goren of Newport, Inc. d/b/a/ Johnson Court Reporting in exchange for cash and the issuance of a \$78,401 Subordinated Promissory Note and 46,118.117 shares of Common Stock to Glory Johnson. On that date the Company also acquired all the issued and outstanding capital stock of RapidText, Inc. in exchange for cash and the issuance of a \$22,738 Subordinated Promissory Note and 13,375.529 shares of Common Stock to Glory Johnson and a \$37,598 Subordinated Promissory Note and 22,116.471 shares of Common Stock to Seaguestor Trust. On that date the Company also acquired all the issued and outstanding capital stock of MedText, Inc. in exchange for cash and the issuance of a \$107,023 Subordinated Promissory Note and 60,865.764 shares of Common Stock to Seaquestor Trust. On August 28, 1997, the Company acquired all the assets of Encore Court Reporting in exchange for cash and the issuance of 2,941 shares of Common Stock to Jan Coldren. On August 29, 1997, the Company acquired all the assets of Legal Enterprise, Inc. in exchange for cash and the issuance of a \$319,340 Subordinated Promissory Note and a \$821,160 Convertible Subordinated Promissory Note to Legal Enterprise, Inc. On September 4, 1997, the Company acquired all the capital stock of Block Court Reporting, Inc. in exchange for cash and the issuance of a \$240,000 Subordinated Promissory Note and a \$360,000 Convertible Subordinated Promissory Note to Martin H. Block. On that date, the Company also acquired all the assets of Amicus One Legal Support

Services, Inc. in exchange for cash and the issuance of a \$560,000 Subordinated Promissory Note and 116,471 shares of Common Stock to Amicus One. On September 17, 1997, the Company acquired all the issued and outstanding capital stock of Burton House, Inc. d.b.a. Ziskind, Greene, Watanabe & Nason in exchange for cash and the issuance of 158,824 shares of Common Stock to Gregg M. Ziskind and Susan L. Ziskind. On that date the Company also acquired all

II-2

the assets of Elaine P. Dine, Inc. and Elaine P. Dine Temporary Attorneys, L.L.C. in exchange for cash and the issuance of a \$2,000,000 Junior Subordinated Promissory Note and 76,471 shares of Common Stock to such companies. All of such sales were completed without registration under the Securities Act in reliance upon the exemption provided by Section 4(2) of the Securities Act, no public offering being involved.

On January 17, 1997, the Company granted options to purchase 96,160 shares of Common Stock to certain employees or other optionees of Looney & Company and Klein Bury at exercise prices ranging from \$.01 to \$.10 in exchange for options previously granted to such employees and optionees by such companies. On September 17, 1997, the Company granted options to purchase 41,176 shares of Common Stock to certain employees of Elaine P. Dine, Inc. at an exercise price of \$.01 per share in exchange for options previously granted to such employees by such company. On September 25, 1997, options to purchase 12,480 shares of Common Stock were exercised at a price of \$.01 per share. The Company received \$124.80 in aggregate consideration upon the exercise of such options. All of such sales were completed without registration under the Securities Act in reliance upon the exemption provided by Section 4(2) of the Securities Act, no public offering being involved. In addition, the Company believes that the exemption provided by Rule 701 promulgated under the Securities Act is applicable to such sales.

On September 8, 1997, the Company sold a number of shares of Common Stock equal to \$1,072,604 divided by 90% of the initial public offering price per share to Kirby A. Kennedy & Associates in connection with a definitive agreement in which the Company will acquire all of the assets of Kirby A. Kennedy & Associates in exchange for cash and the issuance of such shares. On September 15, 1997, the Company sold a number of shares of Common Stock equal to \$2,400,000 divided by 90% of the initial public offering price per share to Colleen Jilio in connection with a definitive agreement in which the Company will acquire all of the assets of Jilio & Associates in exchange for cash and the issuance of such shares. On September 25, 1997, the Company sold a number of shares of Common Stock equal to \$2,500,000 divided by 90% of the initial public offering price per share to Reporting Services Associates, Inc. in connection with a definitive agreement in which the Company will acquire all of the assets of Reporting Service Associates in exchange for cash and the issuance of such shares. On that date the Company also sold a number of shares of Common Stock equal to \$607,500 divided by 90% of the initial public offering price per share to James M. Wilson in connection with a definitive agreement in which the Company will acquire all of the assets of Commander Wilson Incorporated in exchange for cash and the issuance of such shares. All of such sales were completed without registration under the Securities Act in reliance upon the exemption provided by Section 4(2) of the Securities Act, no public offering being involved. The shares of Common Stock referred to in this paragraph will be issued and delivered simultaneously with the consummation of the Offering.

ITEM 16. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES.

Service, dated May 14, 1997.

# (a) Exhibits

<CAPTION>
EXHIBIT

<TABLE>

NUMBER	DESCRIPTION OF EXHIBIT
<c></c>	<\$>
1.1*	Form of Underwriting Agreement between the Company and the Underwriters named therein.
2.1	Stock Purchase Agreement by and between Litigation Resources of America, Inc. and Richard O. Looney, dated as of January 17, 1997.
2.2	Stock Purchase Agreement by and between Litigation Resources of America, Inc. and Michael Klein, dated as of January 17, 1997.
2.3	Securities Purchase Agreement by and among Litigation Resources of America, Inc. (the "Company"), the Subsidiaries of the Company listed on the signature pages thereto and the Investors listed on the signature pages thereto, dated as of January 17, 1997, as amended.
2.4	Agreement of Purchase and Sale of Assets by and between Litigation

Resources of America--California, Inc. and San Francisco Reporting

2.5 Agreement of Purchase and Sale of Assets by and between Litigation Resources of America--California, Inc., Litigation Resources of America, Inc., Peter Giammanco and Leslie Giammanco, individuals d/b/a G&G Court Reporters, and Peter Giammanco and Leslie Giammanco as Trustees of the Giammanco Family Trust, dated as of May 19, 1997.

</TABLE>

II-3

<TABLE>

<caption></caption>
EXHIBIT
NUMBER

# DESCRIPTION OF EXHIBIT

<C> <S>

- Letter Agreement by and between Sandra Rocca and Litigation 2.6 Resources of America--Midwest, Inc., dated August 15, 1997
- Plan and Agreement of Reorganization and Merger by and among 2.7 Litigation Resources of America--California, Inc., Goren of Newport, Inc. d/b/a Johnson Court Reporting, Glory Johnson and Litigation Resources of America, dated as of August 19, 1997.
- 2.8 Plan and Agreement of Reorganization and Merger by and among Litigation Resources of America--California, Inc., RapidText, Inc., Seaquestor Trust, Glory Johnson and Litigation Resources of America, dated as of August 19, 1997.
- 2.9 Plan and Agreement of Reorganization and Merger by and among Litigation Resources of America--California, Inc., MedText, Inc., Seaquestor Trust and Litigation Resources of America, dated as of August 19, 1997.
- 2.10 Resources of America--California, Inc., Litigation Resources of America, Inc., Legal Enterprise, Inc., Tony L. Maddocks and Alan Simon, dated as of August 29, 1997.
- 2.11 Agreement of Purchase and Sale of Assets by and among Litigation Resources of America -- Northeast, Inc., Litigation Resources of America, Inc., Amicus One Legal Support Services, Inc., Richard A. Portas, Joseph N. Spinozzi, Carl Anderson and Howard Breshin, dated as of September 4, 1997.
- 2.12 Stock Purchase Agreement by and between Litigation Resources of America, Inc., Litigation Resources of America--Northeast, Inc., and Martin H. Block, dated as of September 4, 1997.
- 2.13 Agreement of Purchase and Sale of Assets by and among Litigation Resources of America--Midwest, Inc., Litigation Resources of America, Inc., Kirby A. Kennedy & Associates, Kirby A. Kennedy and Jeanne M. Kennedy, dated as of September 8, 1997.
- 2.14 Agreement of Purchase and Sale of Assets by and among Litigation Resources of America--California, Inc., Litigation Resources of America, Inc., and Colleen Jilio, a resident of California d.b.a. Jilio & Associates, dated as of September 15, 1997.
- 2.15 Agreement of Purchase and Sale of Assets by and among Litigation Resources of America--Northeast, Inc., Litigation Resources of America, Inc., Elaine P. Dine, Inc., Elaine P. Dine Temporary Attorneys, L.L.C., Elaine P. Siegel ane Laurie Becker, dated as of September 17, 1997.
- 2.16 Stock Purchase Agreement by and between Litigation Resources of America, Inc., Litigation Resources of America--California, Inc., Gregg M. Ziskind and Susan L. Ziskind, dated as of September 17, 1997.
- 2.17 Agreement of Purchase and Sale of Assets by and among Looney & Company, U.S. Legal Support, Inc. and James M. Wilson, a resident of Houston, Texas d.b.a Commander Wilson, Incorporated, dated as of September 25, 1997.
- 2.18+ Agreement of Purchase and Sale of Assets by and among Litigation Resources of America--Northeast, Inc., Litigation Resources of America, Inc., Reporting Services Associates, Inc. and Lee Goldstein, dated as of September 25, 1997.
- 3.1 Articles of Incorporation, as amended, of the Company.
- 3.2 Bylaws of the Company, as amended.
- Registration Rights Agreement between the Company and Richard O. 4.1 Looney, dated January 17, 1997.

</TABLE>

II-4

<TABLE> <CAPTION> EXHIBIT

NUMBER

DESCRIPTION OF EXHIBIT

<C> <S>

4.2 Registration Rights Agreement between the Company and Michael Klein, dated January 17, 1997. 4.3 Registration Rights Agreement between the Company and certain purchasers, dated January 17, 1997. 4.4 Form of Registration Rights Agreement among the Company and certain holders of the Common Stock. Securityholders Agreement among Litigation Resources of America, 4.5 Inc., the Investors named therein and the Shareholders named therein, dated January 17, 1997, as amended. Registration Rights Agreement between the Company and Gregg M. 4.6 Ziskind and Susan L. Ziskind, dated September 17, 1997. 5.1* Opinion of Bracewell & Patterson, L.L.P. as to the validity of the Common Stock being offered. 10.1 U.S. Legal Support, Inc. 1997 Stock Incentive Plan. 10.2 Form of Option Agreement for 1997 Stock Incentive Plan. U.S. Legal Support, Inc. 1997 Non-Employee Directors Stock Option 10.3 Plan. 10.4 Form of Option Agreement for Non-Employee Directors Stock Option Plan. 10.5 Employment Agreement by and among the Company, Looney & Company and Richard O. Looney, dated January 17, 1997. 10.6 Employment Agreement by and among the Company, Klein, Bury & Associates and Michael Klein, dated January 17, 1997. Employment Agreement by and among the Company, Litigation Resources of America--California, Inc. and Tony L. Maddocks, dated August 29, 1997. 10.8 Employment Agreement dated September 25, 1997 by and among the Company and James M. Wilson dated September 25, 1997. 10.9 Letter Agreement dated May 7, 1997 by and between James M. Wilson d.b.a. Commander Wilson, Inc. and the Company. Termination of Letter Agreement dated May 7, 1997, between James 10.10 M. Wilson d.b.a. Commander Wilson, Inc. and the Company, dated September 25, 1997. 10.11 Letter Agreement dated April 24, 1997 by and between the Company and The GulfStar Group, Inc. 10.12 Fourth Amended and Restated Bank Credit Agreement among the Company, its subsidiaries and Texas Commerce Bank, National Association. 10.13+ First Amendment to Employment Agreement dated October 1, 1997 by and among the Company, Looney & Company and Richard O. Looney. 10.14+ First Amendment to Employment Agreement dated October 1, 1997 by and among Klein, Bury and Associates, Inc., Michael A. Klein and the Company. 10.15+ Letter Agreement dated March 4, 1996 by and between The Gulf Star Group, Inc. and Looney & Company. </TABLE>

II-5

```
<TABLE>
```

<C> <S>

EXHIBIT

NUMBER

### -----

- 11.1 Computation of Historical and SAB No. 55 Earnings Per Share. 11.2 Computation of Pro Forma Earnings Per Share.
- 21.1 Subsidiaries of the Company.
- 23.1* Consent of Bracewell & Patterson, L.L.P. (included in its opinion filed as Exhibit 5 hereto).

DESCRIPTION OF EXHIBIT

- 23.2+ Consent of Coopers & Lybrand L.L.P.
- 23.3 Consent of Fentress Bracewell.
- 23.4 Consent of Ronald C. Lassiter.
- 24.1 Powers of Attorney (See page II-7).
- 27 Financial Data Schedule

</TABLE>

+Filed herewith.

*To be filed by amendment.

(b) Financial Statement Schedules

The following financial statement schedules are included herein.

Schedule II--Valuation and Qualifying Accounts.

All other schedules for which provision is made in the applicable accounting regulations of the Commission are not required under the related instructions, are inapplicable, or the information is included in the consolidated financial statements, and therefore have been omitted.

ITEM 17. UNDERTAKINGS.

(a) The undersigned registrant hereby undertakes to provide to the

Underwriters at the closing specified in the underwriting agreements certificates in such denominations and registered in such names as required by the Underwriters to permit prompt delivery to each purchaser.

- (b) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Company pursuant to the provisions described in Item 14, or otherwise, the Company has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than payment by the Company of expenses incurred or paid by a director, officer or controlling person of the Company in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Company will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.
  - (c) The undersigned Registrant hereby undertakes that:
  - (1) For purposes of determining any liability under the Securities Act of 1933 the information omitted from the form of prospectus filed as part of this Registration Statement in reliance upon Rule 430A and contained in a form of prospectus filed by the registrant pursuant to Rule 424(b)(1) or (4) or 497(h) under the Securities Act shall be deemed to be part of this Registration Statement as of the time it was declared effective.
  - (2) For the purpose of determining any liability under the Securities Act of 1933, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

II-6

#### SIGNATURES

PURSUANT TO THE REQUIREMENTS OF THE SECURITIES ACT OF 1933, U.S. LEGAL SUPPORT, INC. HAS DULY CAUSED THIS REGISTRATION STATEMENT OR AMENDMENT THERETO TO BE SIGNED ON ITS BEHALF BY THE UNDERSIGNED, THEREUNTO DULY AUTHORIZED, IN THE CITY OF HOUSTON, STATE OF TEXAS, ON DECEMBER 17, 1997.

U.S. Legal Support, Inc.

/s/ Richard O. Looney

By:

Richard O. Looney

President and Chief Executive

Officer

## POWER OF ATTORNEY

Each person whose signature appears below hereby revokes the Power of Attorney dated September 25, 1997 and constitutes and appoints Richard O. Looney and David P. Tusa, or either of them, the undersigned's true and lawful attorney-in-fact and agent, with full power of substitution and resubstitution, for the undersigned and in the undersigned's name, place and stead, in any and all capacities (until revoked in writing), to sign this Registration Statement, any Registration Statement filed pursuant to Rule 462(b), and any and all amendments (including post-effective amendments) thereto, to file the same, together with all exhibits thereto and documents in connection therewith, with the Securities and Exchange Commission, to sign any and all applications, registration statements, notices and other documents necessary or advisable to comply with the applicable state securities authorities, granting unto said attorney-in-fact and agent, or their substitute or substitutes, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises in order to effectuate the same as fully to all intents and purposes as the undersigned might or could do if personally present, thereby ratifying and confirming all that said attorneys-in-fact and agents, or their substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

PURSUANT TO THE REQUIREMENTS OF THE SECURITIES ACT OF 1933, THIS REGISTRATION STATEMENT OR AMENDMENT THERETO HAS BEEN SIGNED BELOW BY THE FOLLOWING PERSONS IN THE INDICATED CAPACITIES ON DECEMBER 17, 1997.

<TABLE>

SIGNATURE TITLE

<S>

/s/ Richard O. Looney Director, President and Chief Executive Officer ----- (principal executive officer)

Richard O. Looney

/s/ David P. Tusa Executive Vice President and Chief Financial ----- Officer (principal financial officer)

David P. Tusa

/s/ Scott R. Creasman Vice President and Corporate Controller (principal accounting officer)

Scott R. Creasman

/s/ Michael A. Klein Director

Michael A. Klein

/s/ Robert J. Cresci Director -----

Robert J. Cresci

/s/ G. Kent Kahle Director

_____

G. Kent Kahle

</TABLE>

II-7

#### REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Shareholders of U.S. Legal Support, Inc.:

In connection with our audits of the financial statements of Looney & Company as of December 31, 1995 and 1996, and for each of the three years in the period December 31, 1996, and the financial statements of U.S. Legal Support, Inc. as of September 30, 1997 and for the nine months ended September 30, 1997, which financial statements are included in the Prospectus, we have also audited the financial statement schedule listed on S-2 herein.

In our opinion, the financial statement schedule, when considered in relation to the basic financial statements taken as whole, presents fairly, in all material respects, the information required to be included herein.

COOPERS & LYBRAND L.L.P.

Houston, Texas

December 15, 1997

S-1

#### SCHEDULE II--VALUATION AND QUALIFYING ACCOUNTS (IN THOUSANDS)

<TABLE> <CAPTION>

BALANCE CHARGED DESCRIPTION OF PERIOD EXPENSES DEDUCTIONS OTHER (1) OF PERIOD -----______ ____ <S> <C> <C> <C> <C> <C> LOONEY & COMPANY Year ended December 31, 1994: Allowance for uncollectible \$ --\$ 32 \$ 28 \$ --\$ 60 accounts..... ==== ==== ==== ==== ==== Year ended December 31, 1995: Allowance for uncollectible \$ 60 \$ 83 \$ --\$ -accounts..... \$143 ==== ==== ==== Year ended December 31, 1996: Allowance for uncollectible \$223 \$143 \$ 80 \$ -- \$ -accounts..... ====

U.S. LEGAL SUPPORT, INC.

Nine months ended September

Allowance for uncollectible

accounts......\$223 \$154 \$(36) \$436 \$777

</TABLE>

(1) Acquired allowances for uncollectible accounts in acquisitions.

#### AGREEMENT OF PURCHASE AND SALE OF ASSETS

This Agreement of Purchase and Sale of Assets (this "Agreement") is entered into and effective as of September 25, 1997 by and among LITIGATION RESOURCES OF AMERICA-NORTHEAST, INC., a New York corporation (the "Buyer"), LITIGATION RESOURCES OF AMERICA, INC., a Texas corporation and the parent of Buyer (the "Parent"), REPORTING SERVICES ASSOCIATES, INC., a Pennsylvania corporation (the "Seller"), and Lee Goldstein, a resident of Florida, individually ("Goldstein") (Goldstein referred to sometimes as the "Stockholder"). Buyer, Parent, Seller and the Stockholder are hereinafter sometimes referred to collectively as the "Parties" or singularly as a "Party."

W I T N E S S E T H :

WHEREAS, the Seller is the owner of various assets associated with the Business (as hereinafter defined);

WHEREAS, the Buyer desires to purchase all of the Assets (as hereinafter defined) owned by the Seller and used in the Business (such purchase of the Assets being sometimes herein referred to as the "Acquisition"), and the Seller desires to sell such Assets to the Buyer;

WHEREAS, in connection with the purchase and sale of the Assets, the Parties desire to set forth in this Agreement the terms and conditions with respect to the transfer of such Assets; and

WHEREAS, the parties understand that the Parent or its Affiliates may enter into other agreements similar or dissimilar to this Agreement (the "Other Agreements") for the acquisition by the Parent or such Affiliates of the assets or stock of other entities (collectively, the "Other Acquired Businesses," and each of those entities, individually, an "Other Acquired Business"), which Other Agreements will be among those entities and their equity owners, the Parent and Affiliates of the Parent;

NOW, THEREFORE, for and in consideration of the mutual covenants, agreements, representations and warranties contained in this Agreement, and other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

ARTICLE I

#### CERTAIN DEFINITIONS

As used herein, the following terms shall have the following meanings:

ACCOUNTS RECEIVABLE. The term "Accounts Receivable" shall mean all of the accounts receivable, notes receivable, trade receivables and intercompany receivables relating to the Business and existing as of the Effective Date.

ACQUISITION. The term "Acquisition" shall have the meaning set forth in the preamble hereto.

AFFILIATE. The term "Affiliate" of a person shall mean, with respect to that person, a person who directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, or is acting as agent on behalf of, or as an officer or director of that person. As used in the definition of Affiliate, the term "control" (including the terms "controlling," "controlled by," or "under common control with") means the possession, direct or indirect, of the ability to affect the management and policies of a person whether through the ownership of voting securities, by contract, through the holding of a position as a director or officer of such person, or otherwise. As used in this definition, the term "person" means an individual, a corporation, a limited liability company, a partnership, an association, a joint stock company, a trust, an incorporated organization, or a Governmental Authority.

AGREEMENT. The term "Agreement" shall the meaning set forth in the preamble hereto.

ANCILLARY AGREEMENTS. The term "Ancillary Agreements" shall have the meaning set forth in Section 3.2.

ASSETS. The term "Assets" shall have the meaning set forth in Section 2.1.

ASSUMED LIABILITIES. The term "Assumed Liabilities" shall have the meaning set forth in Section 2.6.

BALANCE SHEET DATE. The term "Balance Sheet Date" shall have the meaning ascribed to it in Section 2.6.

BILL OF SALE. The term "Bill of Sale" shall have the meaning set forth in Section 6.3(xiii).

BOOKS AND RECORDS. The term "Books and Records" shall have the meaning set forth in Section 2.1(c).

BUSINESS. The term "Business" shall mean the court reporting and litigation support business of the Seller as presently conducted.

BUYER. The term "Buyer" shall have the meaning set forth in the preamble

hereto.

BUYER INDEMNIFIED PARTIES. The term "Buyer Indemnified Parties" shall have the meaning set forth in Section 7.1A.

-2-

CAPITAL STOCK. The term "Capital Stock" shall mean, with respect to: (a) any corporation, any share, or any depositary receipt or other certificate representing any share, of an equity ownership interest in that corporation; and (b) any other Entity, any share, membership or other percentage interest, unit of participation or other equivalent (however designated) of an equity interest in that Entity.

CASH PURCHASE PRICE. The term "Cash Purchase Price" shall have the meaning set forth in Section 2.3(a).

CLOSING. The term "Closing" shall have the meaning set forth in Section 6.1.

CLOSING DATE. The term "Closing Date" shall have the meaning set forth in Section 6.1.

CLOSING MEMORANDUM. The term "Closing Memorandum" shall mean the form of closing memorandum to be prepared by the Buyer and the Parent and approved by the Seller (which approval will not be unreasonably withheld) for the Closing under this Agreement in which are included the forms of officers certificates, certificates of Stockholder, opinions of counsel and certain other documents to be delivered at the Closing as provided herein.

CODE. The term "Code" shall mean the Internal Revenue Code of 1986, as amended.

CONTRACTS. The term "Contracts" shall have the meaning as contained in Section 2.1(b).

CUSTOMERS. The term "Customers" shall have the meaning as contained in Section 3.25.

DAMAGES. The term "Damages" shall have the meaning set forth in Section 7.1A.

EFFECTIVE DATE. The term "Effective Date" shall mean 12:01 a.m. on the "Closing Date."

EMPLOYEE. The term "Employee" shall mean any employee of the Seller who, as of the Effective Date, is employed or otherwise performs work or provides services in connection with the operation of the Business, including those, if any, on disability, sick leave, layoff or leave of absence, who, in accordance with the Seller's applicable policies, are eligible to return to active status,

but shall not include any independent contractor providing court reporting services to Seller from time to time.

EMPLOYMENT AGREEMENT. The term "Employment Agreement" shall have the meaning ascribed to it in Section 3.18.

ENTITY. The term "Entity" shall mean any sole proprietorship, corporation, partnership of any kind having a separate legal status, limited liability company, business trust, unincorporated organization or association, mutual company, joint stock company or joint venture.

ENVIRONMENTAL, HEALTH & SAFETY LAWS. The term "Environmental, Health & Safety Laws" shall mean all laws (including rules and regulations, codes, plans, injunctions, judgments, orders, decrees, rulings, and charges thereunder) of federal, state, local, and foreign

-3-

Governmental Authorities concerning pollution or protection of the environment, public health and safety, or employee health and safety.

EQUIPMENT. The term "Equipment" shall have the meaning as contained in Section 2.1(a).

ERISA. The term "ERISA" shall have the meaning as contained in Section 3.17.

EXCHANGE ACT. The term "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

EXCLUDED ASSETS. The term "Excluded Assets" shall have the meaning as contained in Section 2.2.

EXPIRATION DATE. The term "Expiration Date" shall have the meaning set forth in the introductory paragraph to Article III.

FINAL PROSPECTUS. The term "Final Prospectus" shall mean the prospectus included in the Registration Statement at the time it becomes effective, except that if the prospectus first furnished to the Underwriter after the Registration Statement becomes effective for use in connection with the IPO differs from the prospectus included in the Registration Statement at the time it becomes effective (whether or not that prospectus so furnished is required to be filed with the SEC pursuant to Securities Act Rule 424(b)), the prospectus so first furnished is the "Final Prospectus."

FINANCIAL STATEMENTS. The term "Financial Statements" shall mean the balance sheet, income statement and statement of changes of financial position of the Seller.

GAAP. The term "GAAP" shall mean generally accepted accounting principles

of the Accounting Principles Board of the American Institute of Certified Public Accountants and the Financial Accounting Standards Board that are applicable from time to time.

GENERAL INTANGIBLES. The term "General Intangibles" shall have the meaning set forth in Section 2.1(g).

GOLDSTEIN. The term "Goldstein" shall have the meaning ascribed to it in the preamble hereto.

GOLDSTEIN EMPLOYMENT AGREEMENT. The term "Goldstein Employment Agreement" shall have the meaning ascribed to it in Section 6.2(x).

GOVERNMENTAL APPROVAL. The term "Governmental Approval" shall mean at any time any authorization, consent, approval, permit, franchise, certificate, license, implementing order or exemption of, or registration or filing with, any Governmental Authority, including any certification or licensing of a natural person to engage in a profession or trade or a specific regulated activity, at that time.

-4-

GOVERNMENTAL AUTHORITY. The term "Governmental Authority" shall mean (a) any national, state, county, municipal or other government, domestic or foreign, or any agency, board, bureau, commission, court, department or other instrumentality of any such government, or (b) any Person having the authority under any applicable Governmental Requirement to assess and collect Taxes for its own account.

GOVERNMENTAL REQUIREMENT. The term "Governmental Requirement" shall mean at any time (a) any law, statute, code, ordinance, order, rule, regulation, judgment, decree, injunction, writ, edict, award, authorization or other requirement of any Governmental Authority in effect at that time, or (b) any obligation included in any certificate, certification, franchise, permit or license issued by any Governmental Authority or resulting from binding arbitration, including any requirement under common law, at that time.

GUARANTEE OF PERFORMANCE. The term "Guarantee of Performance" shall have the meaning set forth in Section 6.3(xv).

INTELLECTUAL PROPERTY. The term "Intellectual Property" shall have the meaning as contained in Section 2.1(e).

IPO. The term "IPO" shall mean the first time a registration statement filed under the Securities Act with respect to a primary offering by the Parent to the public of Parent Shares is declared effective under the Securities Act and the shares registered by that registration statement are issued and sold by the Parent (otherwise than pursuant to the exercise by the Underwriter of any overallotment option).

IPO CLOSING. The term "IPO Closing" shall mean the delivery to the Underwriters of Parent Shares against the receipt of payment therefor pursuant to the IPO.

IPO CLOSING DATE. The term "IPO Closing Date" shall mean the date on which the Parent first receives payment for the Parent Shares it sells to the Underwriter in the IPO.

IPO PRICE. The term "IPO Price" shall mean the price per share of Parent Shares which is set forth as the "price to public" on the cover page of the Final Prospectus.

LEASED ASSETS. The term "Leased Assets" shall have the meaning ascribed thereto in Section 3.6.

LITIGATION. The term "Litigation" shall mean any action, case, proceeding, claim, grievance, suit or investigation or other proceeding conducted by or pending before any Governmental Authority or any arbitration proceeding.

MATERIAL. The term "Material" shall mean, as applied to any Entity or the Business, material to the business, operations, property or assets, liabilities, financial condition or results of operations of the Business or that Entity and its Subsidiaries considered as a whole, as the case may be.

-5-

MATERIAL ADVERSE EFFECT. The term "Material Adverse Effect" shall mean, with respect to the consequences of any fact or circumstance (including the occurrence or non-occurrence of any event) to the Business, that such fact or circumstance has caused, is causing or will cause, directly, indirectly or consequentially, singly or in the aggregate with other facts and circumstances, any material damages.

MINIMUM CASH AMOUNT. The term "Minimum Cash Amount" shall have the meaning set forth in Section 6.2(iv).

NOTICE OF ACTION. The term "Notice of Action" shall have the meaning set forth in Section 7.1C.

NOTICE OF ELECTION. The term "Notice of Election" shall have the meaning set forth in Section 7.1C.

OFFSET. The term "Offset" shall have the meaning set forth in Section 8.2.

ORDINARY COURSE OF BUSINESS. The term "Ordinary Course of Business" shall mean the ordinary course of Seller's Business consistent with past custom and practice (including with respect to quantity and frequency).

OTHER ACQUIRED BUSINESSES. The term "Other Acquired Businesses" shall have the meaning set forth in the preamble hereto.

OTHER AGREEMENTS. The term "Other Agreements" shall have the meaning set forth in the preamble hereto.

OTHER FINANCING SOURCES. The term "Other Financing Sources" shall have the meaning set forth in Section 6.2(iv).

PARENT. The term "Parent" shall have the meaning set forth in the preamble hereto.

PARENT SHARES. The term "Parent Shares" shall mean any of the shares of common stock of the Parent.

PARTY. The terms "Party" and "Parties" shall have the meanings set forth in the preamble hereto.

PBGC. The term "PBGC" shall mean the Pension Benefits Guaranty Corporation.

PERSON. The term "Person" shall mean any natural person, Entity, estate, trust, union or employee organization or Governmental Authority.

PRICING. The term "Pricing" shall have the meaning set forth in Section 6.1.

-6-

PURCHASE PRICE. The term "Purchase Price" shall mean the consideration payable to the Seller for the Assets as set forth or contemplated in Section 2.3.

REGISTRATION RIGHTS AGREEMENT. The term "Registration Rights Agreement" shall have the meaning set forth in Section 6.2(xi).

REGISTRATION STATEMENT. The term "Registration Statement" shall mean the registration statement, including (a) each preliminary prospectus included therein prior to the date on which that registration statement is declared effective under the Securities Act (including any prospectus filed with the SEC pursuant to Securities Act Rule 424(b)), (b) the Final Prospectus and (c) any amendments and all supplements and exhibits thereto, filed by the Parent with the SEC to register the Parent Shares under the Securities Act for public offering and sale in the IPO.

RETAINED LIABILITIES. The term "Retained Liabilities" shall mean all liabilities of the Seller other than the Assumed Liabilities.

SEC. The term "SEC" means the Securities and Exchange Commission or any successor Governmental Authority.

SECURITIES ACT. The term "Securities Act" shall mean the Securities Act of 1933, as amended.

SELLER. The term "Seller" shall have the meaning set forth in the preamble hereto.

SELLER INDEMNIFIED PARTIES. The term "Seller Indemnified Parties" shall have the meaning set forth in Section 7.1B.

SELLER INDEMNITORS. The term "Seller Indemnitors" shall have the meaning set forth in Section 7.1A.

SELLER'S NAME. The term "Seller's Name" shall have the meaning set forth in Section 2.1(i).

SIDE LETTER. The term "Side Letter" shall have the meaning set forth in Section 9.5.

STOCK PLEDGE AGREEMENT. The term "Stock Pledge Agreement" shall have the meaning set forth in Section 6.3(xiv).

STOCKHOLDER. The term "Stockholder" shall have the meaning set forth in the preamble hereto.

SUBSIDIARY. The term "Subsidiary," of any specified Person at any time, shall mean any Entity a majority of the Capital Stock of which is at that time owned or controlled, directly or indirectly, by the specified Person.

SUPPLEMENTAL INFORMATION. The term "Supplemental Information" shall have the meaning set forth in Section 5.11.

-7-

TAXES. The term "Taxes" shall mean any and all local, state or federal taxes imposed upon the Business, the Seller or the Stockholder, including, without limitation, tax obligations, tax claims, tax charges, tax fines or any related tax liabilities, regardless of the source, cause or origin of such tax liabilities, including taxes imposed as a result of the consummation of the Acquisition.

UNDERWRITER. The term "Underwriter" shall mean collectively (a) the investment banking firms that prospectively may enter into the Underwriting Agreement, and (b) from and after the IPO Pricing Date, the investment banking firms party to the Underwriting Agreement.

UNDERWRITING AGREEMENT. The term "Underwriting Agreement" shall mean the underwriting agreement to be entered into between the Parent and the Underwriter with respect to the IPO.

ARTICLE II
PURCHASE OF ASSETS AND PURCHASE PRICE

- 2.1 SALE OF ASSETS. Subject to the terms and conditions set forth in this Agreement, the Seller agrees to sell, convey, transfer, assign and deliver to the Buyer, and the Buyer agrees to purchase from the Seller on the Closing Date, all assets owned by Seller and used in or derived from the Business (other than those specifically excluded under Section 2.2 below) including the following (such assets to be referred to herein as the "Assets"):
  - (a) All office equipment, furniture, artwork, service equipment, supplies, computer hardware, data processing equipment, tools and supplies (the "Equipment"), including the Equipment described on SCHEDULE 2.1(A);
  - (b) All contracts, documents, franchises, instruments, and other written or oral agreements relating to the Business of Seller to which Seller is a party or by which Seller or any of the Assets may be bound as well as all rights, privileges, claims and options relating to the foregoing (the "Contracts"), including the Contracts described on SCHEDULE 2.1(B);
  - (c) All customer and supplier files and databases, customer and supplier lists, accounting and financial records, invoices, and other books and records relating principally to the Business (the "Books and Records"), including the Books and Records described on SCHEDULE 2.1(C);
  - (d) Seller's Employee files for those Employees actually hired by Buyer;
  - (e) All right, title and interest of Seller, in, to and under all service marks, trademarks, trade and assumed names, principally related to the Business together with the right to recover for infringement thereon, if any (the "Intellectual Property"), and other marks and/or names described on SCHEDULE 2.1(E);
  - (f) All advertising materials and all other printed or written materials related to the conduct of the Business;

-8-

- (g) All of the Seller's general intangibles, claims, rights of set off, rights of recoupment and other proprietary intangibles, licenses and sublicenses granted and obtained with respect thereto, and rights thereunder, which are used in the Business, and remedies against infringements thereof, and rights to protection of interests therein under the laws of all jurisdictions (the "General Intangibles"), including the General Intangibles described on SCHEDULE 2.1(G);
- (h) All goodwill, going concern value and other intangible properties related to the Business;
- (i) The exclusive right to use the name "Reporting Services Associates, Inc.", any similar name or derivative thereof, and any past or present assumed names in connection with the Business or Seller's use of

- 2.2 EXCLUDED ASSETS. Seller is not selling and Buyer is not purchasing any of the following excluded assets related to the Business ("Excluded Assets"): (i) cash, cash deposits, rights to receive cash refunds, and other cash equivalents, (ii) cash investments, (iii) Accounts Receivable, notes receivable and trade receivables accrued on or before the IPO Closing Date, all as more specifically described on SCHEDULE 2.2.
- 2.3 PURCHASE PRICE. Upon the terms and subject to the conditions contained herein and as consideration for the sale of the Assets and the performance by the Seller of various other matters as provided herein, the Buyer shall pay or deliver to the Seller, on the IPO Closing Date and as soon as practicable after the IPO Closing, the aggregate amount \$9,500,000.00, payable by delivery of the following consideration (collectively the "Purchase Price"):
  - (a) Subject to the provisions of Section 2.5, a cash sum in the amount of Seven Million and No/100 Dollars (\$7,000,000) (the "Cash Purchase Price"), paid by the wire transfer of immediately available funds; and
  - (b) Such number of whole Parent Shares on the IPO Closing Date as, when multiplied by 90% of the IPO Price, will most nearly approximate, but not exceed, \$2,500,000.00.
- 2.4 ASSUMPTION OF LIABILITIES; LEASES. Upon the terms and subject to the conditions contained herein, the Buyer agrees that on the Closing Date, it will not assume any liabilities of Seller (except for those liabilities related to Seller's premise and equipment lease payments as described on Schedule 3.6,) ("Assumed Liabilities").
- 2.5 ALLOCATION OF PURCHASE PRICE. For all federal, state and local tax purposes, the Purchase Price shall be allocated among the various Assets in the manner indicated in SCHEDULE 2.7 hereto. None of the Parties shall file any tax return or report or take any position with any Governmental Authority which is inconsistent with the foregoing allocation, except to the extent

-9-

mandated by a Governmental Authority in a determination binding upon one Party provided that such Party has given written notice and reasonable opportunity to the other Party, at its expense, to contest and appeal such determination on behalf of both Parties and such determination has nevertheless become final. Within ninety (90) days after the Closing Date, the Parties shall prepare for filing with the Internal Revenue Service a Form 8594 in accordance with the foregoing allocation.

2.6 TAXES. Seller shall be liable for the payment of all sales and use taxes arising out of the sale and transfer or removal of the Assets, if any, and the assumption of the Assumed Liabilities. Within one year from the Closing Date, the Seller agrees to furnish to the Buyer certificates from the state

taxing authorities, and any related certificates that the Buyer may reasonably request, as evidence that all sales and use tax liabilities of the Seller accruing before the Effective Date have been fully provided for or satisfied. The Buyer shall not be responsible for any business, occupation, withholding or similar tax, or any taxes of any kind of the Seller, related to any period before the Effective Date and, in the event that any taxes are found to be due or owing by Seller within one year of the Closing Date, Seller shall pay all such amounts and Buyer shall have the right of offset against the Parent Shares pledged under the Stock Purchase Agreement for such amounts, if any, as provided in Section 8.2 hereof.

2.7 TITLE TO ASSETS AND RISK OF LOSS. At the Effective Time, title to the Assets and risk of loss or damage to the Assets by casualty (whether or not covered by insurance) will pass to the Buyer.

# ARTICLE III REPRESENTATIONS AND WARRANTIES OF SELLER

Seller and the Stockholder jointly and severally represent and warrant that all of the following representations and warranties in this Article III are true at the date of this Agreement and shall be true at the time of Closing and the IPO Closing Date, and that such representations and warranties shall survive the IPO Closing Date for a period of twelve months (the last day of such period being the "Expiration Date"), except that (i) the warranties and representations set forth in Section 3.29 hereof shall survive until such time as the limitations period has run for all tax periods ended on or prior to the IPO Closing Date, which shall be deemed to be the Expiration Date for Section 3.29, and (ii) solely for purposes of determining whether a claim for indemnification under Section 7.1 hereof has been made on a timely basis, and solely to the extent that in connection with the IPO, the Seller or the Stockholder actually incur liability under the Securities Act, the Exchange Act, or any other federal or state securities laws, the representations and warranties set forth herein shall survive until the expiration of any applicable limitations period, which shall be deemed to be the Expiration Date for such purposes.

3.1 TITLE TO ASSETS. The Seller has good, marketable and indefeasible title to the Assets, free and clear of restrictions or conditions to transfer or assignment, mortgages, liens, pledges, charges, encumbrances, equities, claims, easements, rights-of-way, covenants, conditions or restrictions, except with respect to the Leased Assets as otherwise disclosed on SCHEDULE 2.1(A). The Seller is in possession of all of the Leased Assets leased to it from others. Except for the

-10-

Excluded Assets, the Assets constitute all of the material property, whether real, personal, mixed, tangible or intangible, that are used in the Business by the Seller.

- CONTRACTS AND AGREEMENTS. SCHEDULE 2.1(B) lists all of the material contracts, agreements, and other written or oral arrangements relating to the Business to which the Seller is a party, or by which the Seller or the Assets are bound. As of the Closing Date, each of the Contracts is valid and in full force and effect, and there has not been any default by the Seller or, to the best of Seller's Knowledge, by any other party to any of the Contracts, or any event that with notice or lapse of time or both, would constitute a default by the Seller or, to the best of Seller's Knowledge, any other party to any of the Except as shall be disclosed in SCHEDULE 2.1(B), each Contract is assignable to the Buyer without the consent of any other party. obtain and deliver at Closing all of the requisite consents relating to the items set forth on SCHEDULE 2.1(B). Seller has not received notice that any party to any of the Contracts intends to cancel or terminate any of the Contracts or exercise or not exercise any options that they might have under any of the Contracts. In the event any of the Contracts are, or are later determined to be, non-assignable, and the other party to any such Contracts refuses to consent to the assignment of same, then the Seller shall subcontract to the Buyer or its designee, if the Buyer so desires, the remaining work on such Contracts, and the Seller shall forward to the Buyer or its designee all proceeds of such Contracts received by the Seller provided however, that Seller shall be reimbursed for any reasonable out-of-pocket expenses incurred by it.
- 3.3 EQUIPMENT. All of the Equipment owned or leased by the Seller is described on SCHEDULE 2.1(A) attached hereto. Except as disclosed on SCHEDULE 2.1(A), none of the Equipment will be, at the Effective Time, held under any security agreement, conditional sales contract, or other title retention or security arrangement or is located other than in the possession of the Seller except for Equipment that is out of Seller's possession at certain job sites and/or with certain of Seller's agents. To the best of Seller's Knowledge, the Equipment is in good operating condition and repair, ordinary wear and tear excepted.
- 3.4 LICENSES. Seller does not own any licenses or have any rights to use any licenses in connection with the Business. Seller has not infringed, and is not now infringing, on any license belonging to any other person, firm, or corporation.
- 3.5 EMPLOYMENT CONTRACTS. Except as set forth in SCHEDULE 3.5, Seller does not have any employment contracts and collective bargaining agreements to which the Seller is a party or by which the Seller is bound relating to any Employee. No Employees are represented by any labor organization, and, as of the date hereof, no labor organization or group of Employees has made a written demand to the Seller for recognition, or filed a petition with the National Labor Relations Board, or announced its intention to hold an election of a collective bargaining representative. There is no pending, or to the best Knowledge and reasonable belief of Seller and Stockholder, threatened, labor dispute, strike or work stoppage affecting or potentially affecting the Business.
- 3.6 COMPLIANCE WITH LAWS. The Seller has complied with, and is not in violation of, applicable federal, state or local statutes, laws, and regulations

or indirectly, any of the Assets or the Business. The Seller has all permits, licenses, and authorizations necessary to the conduct of the Business in the manner and in the locations in which the Business is presently conducted, and all such permits, licenses, or other authorizations are valid and in full force and effect. To the best of Seller's Knowledge, there are not any uncured violations of federal, state or municipal laws, ordinances, orders, regulations or requirements affecting any portion of the Assets or the Business.

- 3.7 LITIGATION. Except as disclosed in SCHEDULE 3.7, there is no suit, action, arbitration or legal, administrative or other proceeding or governmental investigation pending or, to the best of Seller's Knowledge, threatened against or affecting the Seller, its Affiliates, the Stockholder, the Assets, or the Business that could result in a material adverse effect on the Business.
- 3.8 CONSENTS AND APPROVALS; NO BREACH OR DEFAULT. Except as set forth on SCHEDULE 8(A), no consent, approval or authorization of, or filing or registration with, any Person or Entity, is required to be made or obtained by Seller in connection with the execution, delivery or performance of this Agreement, or the consummation by Seller of the transactions contemplated hereby. Except as set forth on SCHEDULE 3.8(B), neither the execution and delivery of this Agreement or the Ancillary Agreements by Seller, nor the consummation of the transactions contemplated herein by Seller, will (a) violate any constitution, statute, regulation, rule, injunction, judgment, order, decree, ruling, charge, or other restriction of any Governmental Authority to which Seller is, or the Assets are, subject, or (b) conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice under, any agreement, contract, lease, license, instrument, promissory note, conditional sales contract, partnership agreement or other arrangement to which Seller or any of Seller's Affiliates is a party, or by which Seller is bound, or to which the Assets are subject, or (c) conflict with or violate the articles of incorporation, bylaws or other charter documents of Seller.
- 3.9 AUTHORITY. The Seller has the full right, power, legal capacity and authority to execute, deliver and perform its obligations under this Agreement, and all agreements ancillary to this Agreement which are part of the underlying transaction made the basis of this Agreement and executed in connection herewith including but not limited to the Exhibits hereto ("Ancillary Agreements"). The sale of the Assets and the execution, delivery and performance of this Agreement and the Ancillary Agreements by Seller have been duly authorized by the Stockholder.
- 3.10 PERSONNEL. SCHEDULE 3.10(A) sets forth a complete and accurate list of the names, addresses, hire dates, dates of birth and job descriptions of all Employees employed by Seller in connection with the Business, current rates of

compensation including, if determined, bonuses payable to each following the Closing.__SCHEDULE 3.10(B) is a complete and accurate list of the names, addresses, hire dates, dates of birth and services provided by all independent contractors used by Seller during the preceding one (1) year, stating their rates and method of compensation.

3.11 VALID AND BINDING OBLIGATIONS. Upon execution and delivery, this Agreement, the Ancillary Agreements and each document, instrument and agreement to be executed by the Seller, or the Stockholder in connection herewith, will constitute the legal, valid, and binding obligation of the Seller, or Stockholder, enforceable against the Seller and/or Stockholder in accordance with its

-12-

terms, except as limited by bankruptcy laws, insolvency laws, and other similar laws affecting the rights of creditors generally.

- 3.12 FINANCIAL STATEMENTS. The Financial Statements (a) are true, complete, and correct in all material respects, (b) fairly and accurately present the financial position of Seller as of the periods described therein, and the results of the operations of Seller for the periods indicated, and (c) have been prepared consistently and in accordance with the Seller's historical customs and practices.
- 3.13 EMPLOYEE BENEFITS. SCHEDULE 3.13 is a true, correct and complete list of each "employee benefit plan,' within the meaning of Section 3(3) of Employee Retirement Income Security Act of 1974, as amended ("ERISA") which has ever been maintained or sponsored by Seller or any of its Affiliates. Each such employee benefit plan (and each related trust, insurance contract, or fund) is in full force and effect, and complies in form and in operation in all respects with the applicable requirements of ERISA, the Code, and other applicable laws. Neither Seller, Stockholder nor any Affiliate is in default under any of the plans, there have been no claims of default, and there are no facts, conditions or circumstances which if continued, or on notice, will result in a default, under any plan. None of the plans will, by its terms or under applicable law, become binding upon or become an obligation of the Buyer. No assets of any plan are being transferred to Buyer or to any plan of Buyer. Seller does not contribute to, and has never contributed to, and has never been required to contribute to, any multiemployer plan, and Seller does not have, and has never had, any liability (including withdrawal liability) under any multiemployer plan.
- 3.14 ABSENCE OF CERTAIN CHANGES OR EVENTS. Except as disclosed in SCHEDULE  $3.14\,(A)$  with regard to the Business and the Assets, since the Balance Sheet Date, there has been no:
  - (i) material adverse change in the condition, financial or otherwise, of the Seller, the Assets or the Business;
    - (ii) waiver of any right of or claim held by the Seller;

- (iii) material loss, destruction or damage to any property of the Seller, whether or not insured;
- (iv) material change in the personnel of the Seller or the terms or conditions of their compensation or employment;
- (v) acquisition or disposition of any assets (or any contract or arrangement therefor), nor any other transaction by the Seller otherwise than for value and in the ordinary course of business;
- (vi) transaction or disbursement of funds or assets by the Seller except in the ordinary course of business;
- (vii) capital expenditure by the Seller exceeding \$10,000 except in the ordinary course of business;

-13-

- (viii) change in accounting methods or practices (including, without limitation, any change in depreciation or amortization policies or rates) by the Seller;
  - (ix) re-valuation by the Seller of any of its Assets;
- (x) amendment, modification or termination of any Contract to which the Seller is a party, except in the ordinary course of business;
  - (xi) mortgage, pledge or other encumbrance of any of the Assets;
- (xii) litigation or facts or circumstances that could result in litigation that, if adversely determined, might reasonably be expected to have a material adverse effect on Seller, Seller's financial condition, Seller's prospects, the Business or the Assets;
- (xiii) increase in salary or other compensation payable or to become payable by the Seller to any of its officers, directors or employees, or the declaration, payment or commitment or obligation of any kind for the payment, by the Seller, of a bonus or other additional salary or compensation to any such person;
  - (xiv) loan by the Seller to any person or entity, or guaranty by the Seller of any loan;
  - (xv) other event or condition of any character that has or might reasonably be expected to have a material adverse effect on the Business, the Assets or the financial condition of the Seller; or
  - (xvi) agreement by the Seller to do any of the things described in the preceding clauses (i) through (xv).

Except as disclosed in SCHEDULE 3.14(B), there have been no contractual commitments by Seller to spend more than \$25,000 per contractual commitment over a continuous 12-month period.

- 3.15 BROKERS. Neither Seller, the Stockholder, nor any of the Seller's or the Stockholder's Affiliates has employed any broker, agent, or finder, or incurred any liability for any brokerage fees, agent's fees, commission or finder's fees in connection with the transactions contemplated herein.
- 3.16 SALE OF ASSETS. For purposes of determining whether a sales and use tax charge is applicable, the sale of the Assets constitutes: (i) the sale of all of the operating assets of a business or of a separate division, branch, or identifiable segment of a business, and (ii) a sale outside the ordinary course of Seller's Business, and represents an isolated or occasional sale by a seller who does not regularly engage in such business. The income and expenses of the Business can be separately established from the Books and Records in the same manner as previously provided to Buyer.
- 3.17 ABSENCE OF CERTAIN BUSINESS PRACTICES. Neither the Seller, the Stockholder, nor any agent of the Seller, or the Stockholder, nor to Seller's or the Stockholder's best Knowledge any

-14-

other person acting on Seller's or the Stockholder's behalf, has, directly or indirectly, within the past five years, given or agreed to give any gift or similar benefit to any customer, supplier, government employee of the United States or any state or foreign government, or other person who is or may be in a position to help or hinder the Business which (1) would subject the Seller to any damage or penalty in any civil, criminal or governmental litigation or proceeding, (2) if not given in the past, would have an adverse effect on the Business, or (3) if not continued in the future, would have a material adverse effect on the Business or the Assets, or which would subject the Seller to suit or penalty in a private or governmental litigation or proceeding.

- 3.18 LIENS ON ASSETS. Except as set forth on SCHEDULE 3.18, all liens or security interests of any third party as to any of the Assets have been removed on or before the Closing Date, and the Seller has furnished evidence thereof to Buyer.
- 3.19 CUSTOMERS. To the best of Seller's Knowledge, SCHEDULE 3.19 contains a true and correct list of all customers of the Business within the last year (the "Customers"). The Seller has no information, nor is the Seller aware of any facts, indicating that any of the material Customers intend to cease doing business with the Seller.
- 3.20 INSURANCE POLICIES. SCHEDULE 3.20 to this Agreement is a description of all insurance policies held by the Seller concerning the Business and Assets. The Seller maintains insurance protection on all its Assets and the Business of

such types and in such amounts as are customarily insured by similar companies in the same industry, covering property damage and loss by fire, theft, or other casualty.

- 3.21 INTEREST IN CUSTOMERS, SUPPLIERS AND COMPETITORS. Except as set forth in SCHEDULE 3.21, neither the Seller, the Stockholder, nor any Affiliate of the Seller or Stockholder, has any direct or indirect interest in any competitor, supplier or customer of the Seller, or in any person from whom or to whom the Seller leases any property, or in any other person with whom the Seller is doing business.
- 3.22 ADVERSE INFORMATION. Neither Seller nor Stockholder has any actual Knowledge of any change contemplated in any applicable laws, ordinances or restrictions, or any judicial or administrative action, or any event, fact or circumstance which will, or could be reasonably expected to, have a material adverse effect on the Seller or its condition, financial or otherwise, the Assets, or the condition, value or operation thereof.
- 3.23 ORGANIZATION. The Seller is a Pennsylvania corporation duly organized, validly existing and in good standing under the laws of the State of Pennsylvania, has all necessary powers to own the Assets and properties and to operate the Business as now owned and operated by it, and is qualified to do business in the State of Pennsylvania.
- 3.24 CONDITION. All of the Assets are in good operating condition and repair, ordinary wear and tear excepted, and, as applicable, good working order. To the best of the Seller's and the Stockholder's Knowledge, the buildings, fixtures, and improvements leased by the Seller, including but not limited to the parking areas, roofs, foundations, plumbing, electrical, air conditioning, heating

-15-

and ventilating systems, doors, building exteriors, landscaping, and interior areas are in good condition, ordinary wear and tear excepted, and, as applicable, good working order.

3.25 INTELLECTUAL PROPERTY. All of the Intellectual Property owned by the Seller is described on SCHEDULE 2.1(E) attached hereto. The Seller is the sole owner of all of the Intellectual Property, free and clear of any liens, encumbrances, restrictions, or legal or equitable claims of others. The Seller has registered all trade names, trademarks, and service marks in all jurisdictions necessary to evidence and protect its ownership thereof, and to permit the Seller to conduct its business in the manner in which it is currently conducted, or otherwise has all rights or licenses necessary to use the same. The Seller has all patents or patent applications and copyrights registered in all jurisdictions necessary to evidence and protect the ownership thereof and to permit the Seller to conduct its business in the manner in which it is currently conducted, or otherwise has all rights or licenses necessary to use same. Except as disclosed in this Agreement, all of the patents of the Seller are

valid and in full force and effect and are not subject to any taxes, maintenance fees, or actions which have not been currently paid. None of the Intellectual Property infringes upon any patents, trade or assumed names, trademarks, service marks, or copyrights belonging to any other person or other entity. The Seller is not a party to any license, agreement, or arrangement, whether as licensor, licensee, or otherwise, with respect to any of the Intellectual Property. The Seller does not have a license or a right to use any other patents, service marks, trademarks, trade and assumed names, trade secrets and royalty rights and other proprietary intangibles in connection with the Business, other than the Intellectual Property.

- 3.26 POWERS OF ATTORNEY. No person or other entity holds a general or special power of attorney from the Seller.
- 3.27 NO SEVERANCE PAYMENTS. Except as set forth in SCHEDULE 3.27, the Seller will not owe a severance payment or similar obligation to any of its Employees, officers, or directors, as a result of the transactions contemplated by this Agreement, nor will any of such persons be entitled to an increase in severance payments or other benefits as a result of the transactions contemplated hereby, nor in the event of the subsequent termination of their employment.
- 3.28 EMPLOYEES. Except as has occurred in the ordinary course of business, the Seller has not, nor has it agreed to do in any unusual or extraordinary amount or manner, any of the following acts with respect to its Employees in the Business: (i) grant any increase in salaries payable or to become payable by it, (ii) increase benefits, (iii) modify any collective bargaining agreement to which it is a party or by which it may be bound, or (iv) declared any bonuses for any of its Employees.
- 3.29 TAXES. Seller has filed all tax returns that Seller was required to file, and all such tax returns were correct and complete in all respects. All Taxes owed by Seller (whether or not shown on any tax return) have been paid. Seller is not the beneficiary of any extension of time within which to file any tax return, and Seller has not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency. Seller has withheld and paid all Taxes required to have been withheld or paid in connection with amounts paid or owing to the Stockholder and any Employee, independent contractor, creditor, or other third party. Neither Seller, an Employee responsible for Tax matters, nor the Stockholder of Seller has reason to believe

-16-

that any authority might assess any additional Taxes for any period for which tax returns have been filed. There is no dispute or claim concerning any Tax liability of Seller.

3.30 HIRING OF EMPLOYEES. As of the Closing Date, Seller shall permit Buyer to offer employment to all of the Employees. At or prior to Closing,

Seller shall have paid such Employees all compensation and benefits to which they are entitled by reason of their previous employment with the Seller on such date, and Buyer shall have no liability with respect thereto. Seller shall use Seller's best efforts to assist Buyer in any reasonable manner in the hiring by Buyer of the Employees that Buyer desires to hire. Buyer shall have the right, but not the obligation, to offer employment to such Employees that it desires to hire in its sole discretion. Seller shall be solely responsible and liable for all severance pay, if any, to the extent that any of the Employees are not offered employment with Buyer or do not accept an offer of employment. Under no circumstances shall the Seller or any of Seller's Affiliates be permitted to employ or offer employment to any of the Employees after the Closing Date, without the prior written consent of Buyer.

- 3.31 OPERATIONS OF THE SELLER. Except as disclosed in SCHEDULE 3.31, since the Balance Sheet Date:
- (i) Seller has used its best efforts to preserve the business organization of the Seller intact, to keep available to the Business the Employees, and to preserve its present relationships with suppliers, Customers and others having business relationships with it;
- (ii) Seller has maintained its existing insurance as to the Business and the Assets, and otherwise maintained and operated the Business in a good and businesslike manner in accordance with good and prudent business practices;
- (iii) Seller has not entered into any agreement or instrument which would constitute an encumbrance of the Assets, which would bind Buyer, the Seller or the Assets after Closing, other than in the ordinary course of business, or which would be outside the normal scope of maintaining and operating the Business and the Assets in the ordinary course of business;
- (iv) Seller has performed all of the Seller's material obligations under all Contracts and commitments applicable to the Seller, the Business, and the Assets, and has maintained the Seller's Books and Records in the usual, regular and customary manner;
- (v) to the best of Seller's Knowledge, the Seller has complied with all statutes, laws, ordinances and regulations applicable to the Seller, the Assets, and the conduct of the Business;
- (vi) Seller has not removed or disposed of, nor permitted the removal or disposal of, any Assets unless such Assets were replaced with an item of at least equal value that is properly suited for its intended purpose;
- (vii) Seller has paid all bills and other payments due with respect to the ownership, use, insurance, operation and maintenance of the Business and the Assets in the usual, regular and customary manner consistent with its prior practices, and has taken all action necessary or prudent

to prevent liens or other claims for the same from being filed or asserted against any part of the Assets; and

- (viii) all revenues received by the Seller relating to the Business have been deposited in the Seller's account relating to the Business.
- 3.32 FULL DISCLOSURE. This Agreement, the Ancillary Agreements and the Schedules hereto, and all other documents and written information furnished by the Seller to the Buyer pursuant hereto or in connection herewith, are true, complete and correct, and do not include any untrue statement of a material fact or omit to state any material fact necessary to make the statements made herein and therein not misleading. To the best of Seller's Knowledge, there are no facts or circumstances relating to the Assets or the Business which adversely affect or might reasonably be expected to adversely affect the Assets, the Business (including the prospects or operations thereof), or the ability of the Seller to perform this Agreement or any of Seller's obligations hereunder.

# ARTICLE IV REPRESENTATIONS AND WARRANTIES OF BUYER

Parent and Buyer jointly and severally represent and warrant that all of the following representations and warranties in this Article IV are true at the date of this Agreement and shall be true at the time of Closing and the IPO Closing Date, and that such representations and warranties shall survive the IPO Closing Date until the Expiration Date, except that solely for purposes of determining whether a claim for indemnification under Section 7.1 hereof has been made on a timely basis, and solely to the extent that in connection with the IPO, the Seller actually incurs liability under the Securities Act, the Exchange Act, or any other federal or state securities laws, the representations and warranties set forth herein shall survive until the expiration of any applicable limitations period, which shall be deemed to be the Expiration Date for such purposes.

- 4.1 ORGANIZATION. The Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of New York and has all the necessary corporate powers to own its properties and to carry on its business as now owned and operated by it. The Parent is a corporation duly organized, validly existing and in good standing under the laws of the State of Texas and has all the necessary corporate powers to own its properties and to carry on its business as now owned and operated by it.
- 4.2 AUTHORITY. Each of Buyer and Parent, as applicable, has the right, power, legal capacity, and authority to execute, deliver and perform this Agreement and the Ancillary Agreements to which it is a party. The execution, delivery and performance of this Agreement and any Ancillary Agreements by Buyer and Parent, as applicable, have been duly authorized by all necessary corporate action.
  - 4.3 CAPITAL STOCK OF PARENT AND BUYER. The respective designations and

issued and outstanding shares will be set forth in the Registration Statement, and 10,000,000 shares of preferred stock, \$.01 par value, of which no shares will be issued and outstanding, and a number of shares of restricted voting common stock, \$.01 par value, to be determined by Parent in good faith, all of which will be issued and outstanding except as otherwise set forth in the Registration Statement, and (ii) as of the date of this Agreement, the authorized capital stock of Buyer consists of 10,000 shares of common stock, all of which shares are issued and outstanding.

- 4.4 TRANSACTIONS IN CAPITAL STOCK; ORGANIZATION ACCOUNTING. Except for the Other Agreements and except as set forth on in the Registration Statement, (i) no option, warrant, call, conversion right or commitment of any kind exists which obligates the Parent or the Buyer to issue any of their respective authorized but unissued capital stock, and (ii) neither the Parent nor the Buyer has any obligation (contingent or otherwise) to purchase, redeem or otherwise acquire any of its equity securities or any interests therein or to pay any dividend or make any distribution in respect thereof. SCHEDULE 4.4 includes complete and accurate copies of all stock option or stock purchase plans, including a list of all outstanding options, warrants or other rights (excluding the Other Agreements) to acquire Parent Shares.
- 4.5 COMMON STOCK. At the time of issuance thereof and delivery to the Seller, the Parent Shares to be delivered to the Seller pursuant to this Agreement will constitute valid and legally issued Parent Shares, fully paid and nonassessable, and with the exception of restrictions upon resale (a) set forth in Section 9.2 hereof and (b) contained in the Stock Pledge Agreement, will be identical in all substantive respects (which do not include the form of certificate upon which it is printed or the presence or absence of a CUSIP number on any such certificate) to the Parent Shares issued and outstanding as of the date hereof by reason of the provisions of the Texas Business Corporation Except as provided in the previous sentence, the Parent Shares issued and delivered to the Seller shall at the time of such issuance and delivery be free and clear of any liens, claims or encumbrances of any kind or character. Parent Shares to be issued to the Seller pursuant to this Agreement will not be registered under the 1933 Act, except as provided in the Registration Rights Agreement.
- 4.6 ASSUMED LIABILITIES. Buyer shall assume the Assumed Liabilities as defined in Section 2.6 herein.
- 4.7 ABSENT CERTAIN CHANGES. Since the Balance Sheet Date, there have been no (a) material adverse changes in the business, financial condition, assets, operations or prospects of Parent, (b) amendments, modifications or terminations

of any material contract applicable to Parent, (c) any changes in the accounting methods or practices of Parent or (d) any litigation or facts or circumstances that could result in litigation that, if adversely determined, might reasonably be expected to have a material adverse effect on Parent or on its business, financial condition or prospects.

4.8 LITIGATION. Except as disclosed in Schedule 4.8, there is no suit, action, arbitration or legal, administrative or other proceeding or governmental investigation pending or, to the best of the Parent's Knowledge, threatened against or affecting the Parent or the Buyer that could result in a material adverse effect on the Business.

-19-

- 4.9 NO BREACH OR VIOLATION. As of the Effective Time and except as set forth on Schedule 4.9, the consummation of the transactions contemplated by this Agreement will not result in or constitute any of the following: (i) a default or an event that, with notice or lapse of time or both, would be a default, breach or violation, or give rise to a right of modification, termination, cancellation or acceleration of any obligation or to a loss of a benefit under, except for third party consents described in this Agreement or any schedule prepared and delivered in connection herewith, of any lease, license, promissory note, conditional sales contract, commitment, indenture, mortgage, deed of trust, security agreement, concession, franchise, permit or other agreement, instrument or arrangement by which the Parent or the Buyer may be affected, or to which the Parent or the Buyer may be bound, (ii) the creation or imposition of any lien, charge, or encumbrance on any of the assets of the Parent or the Buyer, or (iii) a breach of any term or provision of this Agreement.
- 4.10 VALID AND BINDING OBLIGATIONS. Upon execution and delivery, each of this Agreement and the Ancillary Agreements will constitute the legal, valid, and binding obligation of Buyer or Parent, as applicable, enforceable in accordance with its terms, except as limited by bankruptcy laws, insolvency laws, and other similar laws affecting the rights of creditors generally.
- 4.11 BROKERS. Except for The GulfStar Group, Inc. neither Buyer nor any of its respective Affiliates, officers, directors, or employees, has employed any broker, agent, or finder, or incurred any liability for any brokerage fees, agent's fees, commissions or finder's fees in connection with the transactions contemplated herein.
- 4.12 CONSENTS AND APPROVALS. No consent, approval or authorization of, or filing or registration with, any Person or Entity, is required to be made or obtained by Buyer in connection with the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby.

ARTICLE V
COVENANTS OF THE PARTIES

Buyer and Seller covenant and agree as follows:

- 5.1 CONDUCT OF THE BUSINESS. Except as otherwise permitted by this Agreement or consented to by Buyer in writing, Seller shall through the IPO Closing Date conduct the Business in the ordinary course in substantially the same manner as heretofore, using its best efforts to preserve intact its present business organization, to keep available the services of its Employees, and to preserve its relationships with Customers, suppliers and others having business dealings with it.
- 5.2 CERTAIN CHANGES. Except as otherwise permitted by this Agreement or consented to by Buyer in writing, Seller shall not: (a) subject any of the Assets to any lien or encumbrance; (b) dispose of any of the Assets; or (c) grant any increase in compensation or benefits to any Employee; (d) materially modify any of the liabilities, or (e) with respect to the Business, perform any act outside the Ordinary Course of Business except as otherwise contemplated by this Agreement.

-20-

- 5.3 NOTICE. Seller will notify Buyer immediately in writing if (i) any of Seller's representations or warranties set forth in this Agreement are or become untrue prior to the IPO Closing Date, (ii) Seller fails to fully perform all of the covenants of Seller set forth in this Agreement, or (iii) there occurs any Material adverse development in the Business or Seller's market position, sales, profit trends, labor regulations, litigation or insurance claims or otherwise.
- 5.4 RECORDS. From the date of execution of this Agreement until the Closing, Seller shall permit Buyer the right, during normal business hours, to inspect any documents, Books and Records or other information pertaining to the Assets.
- 5.5 U.C.C. SEARCHES. Within five (5) days from the date of execution hereof, Buyer, at Buyer's sole cost and expense, shall deliver to Seller current searches of all Uniform Commercial Code financing statements filed with the Office of the Secretary of State of Pennsylvania and in any other state, or county in which the Seller has an office, against Seller. Any and all liens, pledges, mortgages, security interests and encumbrances affecting the Assets, regardless of whether same are disclosed in such lien searches, shall be released and discharged by Seller at or prior to Closing.
- 5.6 BULK SALES. It may not be practicable to comply or attempt to comply with the procedures of the "Bulk Sales Act" or similar law in any or all of the states in which the Assets are situated or of any other state which may be asserted to be applicable to the transactions contemplated hereby. Accordingly, to induce Buyer to waive any requirements for compliance with any or all of such laws, Seller hereby agrees that except for the Assumed Liabilities, the indemnity provisions of Article VII hereof shall apply to any Damages of Buyer arising out of or resulting from the failure of Buyer or Seller to comply with any such laws or any similar law which may be asserted to be applicable.

- 5.7 NON-COMPETITION AGREEMENT. The Seller agrees that, for the period beginning on the Closing Date and continuing for five (5) years following the Closing Date, neither Seller, Lee Goldstein, individually, nor any Affiliates, shall, either directly or indirectly, individually or separately, for themselves or as an owner, stockholder, joint venturer, promoter, consultant, manager, independent contractor, agent, or in some similar capacity for any reason whatsoever:
  - A. Enter into, engage in, or be connected with any business or business operation or activity which consists in whole or in part of the Business within the following Counties:

(list counties in Pennsylvania);

B. Call upon any customer whose account is or was serviced in whole or in part by the Seller in relation to the Business or the Buyer with the intent of selling or attempting to sell to any such customer any services similar to the services provided by the Buyer; and

-21-

C. Intentionally divert, solicit or take away any customer, supplier or employee of the Buyer, or the patronage of any customer or supplier of the Buyer, or otherwise interfere with or disturb the relationship existing between the Buyer and any of its customers, suppliers, or employees, directly or indirectly.

In the event the Buyer ceases operation of the Business other than in a merger, consolidation, sale of assets or similar transaction, or upon the filing of a bankruptcy or receivership proceeding against the Buyer, or upon the appointment of a liquidator for the Buyer, the provisions of this Section 5.7 will not be applicable to the conduct of Seller subsequent thereto.

It is mutually understood and agreed that if any of the provisions relating to the scope, time or territory in this Section 5.7 are more extensive than is enforceable under applicable law or are broader than necessary to protect the goodwill and legitimate business interests of the Buyer, then the Parties agree that they will reduce the degree and extent of such provisions by whatever minimal amount is necessary to bring such provisions within the ambit of enforceability under applicable law.

The Parties acknowledge that the remedies at law for breach of Seller's covenants contained in this Section 5.7 are inadequate, and they agree that the Buyer shall be entitled, at its election, to injunctive relief (without the necessity of posting bond against such breach or attempted breach), and to specific performance of said covenants in addition to any other remedies at law or equity that may be available to the Buyer.

5.8 TERMINATION OF EMPLOYMENT OF SELLER'S EMPLOYEES. Buyer anticipates

extending an offer of employment to substantially all of the Employees of the Seller on substantially the same terms and conditions as the Employees currently are employed by Seller. Notwithstanding the foregoing, nothing herein shall imply or guarantee employment of any Employee of Seller by Buyer. If Seller's Employees desire employment with Buyer, they will be interviewed in conjunction with the applicants from other sources and given strong consideration for available positions with Buyer, at the wages, hours, and conditions of employment established by Buyer prior to hiring any Employees. Seller agrees to use its best efforts to make available the Employees to the Buyer that Buyer desires to hire for the purpose of operating the Business. Notwithstanding anything to the contrary contained herein, Buyer agrees that it will offer to employ substantially all of Seller's Employees. Nothing shall prohibit Buyer from terminating any of Seller's Employees subsequent to their employment by Buyer.

5.9 COOPERATION IN CONNECTION WITH THE IPO. The Seller and the Stockholder will (a) provide the Parent and the Underwriter with all the Information concerning Seller and the Stockholder which is reasonably requested by the Parent and the Underwriter from time to time in connection with effecting the IPO and (b) cooperate with the Parent and the Underwriter and their respective representatives in the preparation and amendment of the Registration Statement (including the Financial Statements) and in responding to the comments of the SEC staff, if any, with respect thereto, to the extent that any of the foregoing concern or reasonably relate to the Seller or the Stockholder. The Seller and the Stockholder agree promptly to (a) advise the Parent if, at any time during the period in which a prospectus relating to the IPO is required to be delivered under the Securities Act, any information contained in the then current Registration Statement prospectus

-22-

concerning the Seller or the Stockholder becomes incorrect or incomplete in any material respect and (b) provide the Parent with the information needed to correct or complete that information.

- 5.10 ADDITIONAL FINANCIAL STATEMENTS. Seller shall promptly furnish to Buyer a copy of all Financial Statements for each additional month-end period beyond the Balance Sheet Date as soon as same is regularly prepared by Seller in the Ordinary Course of Business. All such additional Financial Statements shall be subject to the same representations and warranties as contained in Section 3.23 of this Agreement. The Parties acknowledge and agree that Buyer shall be permitted to conduct at its sole cost and expense an audit of Seller for the fiscal years ended 1994, 1995 and 1996 and 1997 (or any portion thereof) in connection with the IPO. Buyer shall pay Seller's reasonable costs incurred in preparing such monthly financial statements.
- 5.11 SUPPLEMENTAL INFORMATION. The Seller and the Stockholder agree that, with respect to the representations and warranties of that party contained in this Agreement, that party will have the continuing obligation through the IPO Closing to provide the Parent promptly with such additional supplemental

Information (collectively, the "Supplemental Information"), in the form of (a) amendments to then existing Schedules or (b) additional Schedules, as would be necessary, in the light of the circumstances, conditions, events and states of facts then known to the Seller or such Stockholder, to make each of those representations and warranties true and correct as of the Closing and on the IPO Closing Date. For purposes only of determining whether the conditions to the obligations of the Parent and Buyer which are specified in Section 6.3 have been satisfied, the Schedules as of the Closing and on the IPO Closing Date shall be deemed to be the Schedules and the Investor Representation Letter as of the date hereof as amended or supplemented by the Supplemental Information provided to the Parent prior to the Effective Date pursuant to this Section 5.11; provided, however, that (a) if the Supplemental Information so provided discloses the existence of circumstances, conditions, events or states of facts which, in any combination thereof, have had a Material Adverse Effect or, (b) based upon the advice of the Underwriter the Parent has determined that subsequent events that were revealed through RSA's submission of Supplemental Information pursuant to this Section 5.11 (which shall be conclusive for purposes of this Section 5.11 and 8.3(a)(iv), but not for any purpose of Article VII), are having or will have a Material Adverse Effect, the Parent will be entitled to terminate this Agreement pursuant to Section 8.3(a)(iv); and provided, further, that if the Parent is entitled to terminate this Agreement pursuant to Section 8.3(a)(iv), but elects not to do so, it will be entitled to treat as Buyer Indemnified Losses (which treatment will not prejudice the right of the Seller or the Stockholder to contest Damage claims made by the Parent in respect of those Buyer Indemnified Losses) all Damages to the Business which are attributable to the circumstances, conditions, events and state of facts first disclosed herein after the date hereof in the Supplemental Information. The Parent will provide the Seller and the Stockholder with copies of the Registration Statement, including all pre-effective amendments thereto, promptly after the filing thereof with the SEC under the Securities Act.

5.12 INSURANCE. Seller shall assist, and shall cause its Affiliates to assist, Buyer in transferring to Buyer any insurance applicable to the Assets or the Leased Assets which Buyer elects to maintain in effect.

-23-

5.13 CONFIDENTIALITY. Seller will not, and will not permit any of its Affiliates to, disclose any information of a confidential or proprietary nature concerning the Assets or the Business to any third parties, and in no event shall Seller use, or allow any of its Affiliates to use, such confidential or proprietary information for its or his own benefit or to the detriment of Buyer or the Business. No public or private announcement shall be made of the transactions contemplated herein, nor the terms hereof, by Seller or any of its Affiliates, without the prior written approval of Buyer as to timing, form and content.

ARTICLE VI
THE CLOSING

- PRICING. At or prior to Pricing, the Parties shall take all actions 6.1 necessary to (a) complete the Acquisition (including the execution and delivery of this Agreement and the Ancillary Agreements which shall be placed in escrow under the control of the Parent for release to the Parties on the IPO Date), and (b) effect the delivery of the Parent Shares referred to in Section 2.3 hereof, provided however, that such actions shall not include the actual completion of the Acquisition or the delivery of the Common Stock and funds referred to in Section 2.3 hereof, each of which actions shall only be taken upon the IPO Closing Date as herein provided. For purposes of this Article VI, the term "Pricing" shall mean the date of determination by Parent and the Underwriter of the public offering price of the Parent Shares in the IPO; the Parties contemplate that the Pricing shall take place on the Closing Date. The escrow agreement relating to this Agreement and the Ancillary Agreements shall provide that in the event that there is no IPO Closing Date, and this Agreement terminates as provided in Section 8.3(b)(ii), the Agreement and the Ancillary Agreements shall not be delivered to the Parties. The taking of the actions described in clauses (a) and (b) above (the "Closing") shall take place on the closing date (the "Closing Date") at the offices of Boyer, Ewing & Harris Incorporated, 9 Greenway Plaza, Suite 3100, Houston, Texas 77046. Closing Date, all transactions contemplated by this Agreement, including the delivery of the Parent Shares, the wire transfer of the cash portion of the Purchase Price which Seller is entitled to receive pursuant to Section 2.3 hereof, and the closing of the IPO shall occur and be completed. Except as otherwise provided in Section 8.3 hereof, during the period from the Closing Date to the IPO Closing Date, this Agreement may only be terminated by the Parties if the Underwriting Agreement is terminated pursuant to the terms thereof.
- 6.2 CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER. The obligations of the Seller with respect to actions to be taken on the Closing Date are subject to the satisfaction or waiver on or prior to the Closing Date of all of the following conditions other than the conditions set forth in this Section 6.2(i) and (viii) that cannot be satisfied prior to the IPO Closing Date. The obligations of the Seller with respect to actions to be taken on the IPO Closing Date are subject to the satisfaction or waiver on or prior to the IPO Closing Date of the conditions set forth in this Section 6.2(i) and (viii). As of (a) the Closing Date if any such conditions have not been satisfied other than the conditions set forth in this Section 6.2(i) and (viii) that cannot be satisfied prior to the IPO Closing Date or (b) the IPO Closing Date, if any such conditions have not been satisfied, the Seller shall have the right to terminate this Agreement, or in the alternative, waive any condition not so satisfied. Any act or

-24-

action of the Seller in consummating the Closing on the Closing Date to the extent set forth in the first sentence of Section 6.1 shall constitute a waiver of any conditions not so satisfied other than the conditions set forth in this Section 6.2(i) and (viii) that cannot be satisfied prior to the IPO Closing

Date. However, no such waiver shall be deemed to affect the survival of the representations and warranties of Parent and Buyer contained in Article IV hereof.

- (i) REPRESENTATIONS AND WARRANTIES; PERFORMANCE OF OBLIGATIONS. All representations and warranties of the Parent and the Buyer contained in Article IV shall be true and correct in all Material respects as of the Closing Date and the IPO Closing Date as though such representations and warranties had been made as of that time; all of the terms, covenants and conditions of this Agreement to be complied with and performed by the Parent and the Buyer on or before the Closing Date and the IPO Closing Date shall have been duly complied with and performed in all Material respects; and certificates to the foregoing effect dated the Closing Date and the IPO Closing Date, respectively, and signed by each of the Parent and the Buyer shall have been delivered to the Seller.
- (ii) NO LITIGATION. No action or proceeding before a Governmental Authority shall have been instituted or threatened to restrain or prohibit the Acquisition or the IPO and no Governmental Authority shall have taken any other action or made any request of the Parent or the Buyer as a result of which the management of the Seller deems it inadvisable to proceed with the transactions hereunder.
- (iii) OPINION OF COUNSEL. The Seller shall have received an opinion from counsel for the Parent dated the Closing Date, in the form attached as Exhibit G-1 hereto.
- REGISTRATION STATEMENT. The Registration Statement, as amended to cover the offering, issuance and sale by Parent of such number of Parent Shares at the IPO Price (which need not be set forth in the Registration Statement when it becomes effective under the Securities Act) as shall yield aggregate cash proceeds to the Parent from that sale (net of Underwriter's discount or commissions) in at least the amount (the "Minimum Cash Amount") that is sufficient, when added to the funds, if any, available from other sources (if any, and as set forth in the Registration Statement when it becomes effective under the Securities Act) (the "Other Financing Sources"), to enable the Parent to pay or otherwise deliver on the IPO Closing Date (i) the total cash portion of the Purchase Price then to be delivered pursuant to Article II; (ii) the total cash portion of the acquisition consideration then to be delivered pursuant to the Other Agreements as a result of the consummation of the acquisition transactions contemplated thereby, and (iii) the total amount of indebtedness of the Seller, each Other Acquired Business and the Parent which the Registration Statement discloses at the time it becomes effective under the Securities Act will be repaid with proceeds received by the Parent from the IPO and Other Financing Sources shall have been declared effective by the SEC.
- (v) CONSENTS AND APPROVALS. All necessary consents of and filings with any Governmental Authority relating to the consummation of the transactions contemplated herein

shall have been obtained and made and no action or proceeding shall have been instituted or threatened to restrain or prohibit the Acquisition.

- (vi) GOOD STANDING CERTIFICATES. Parent and the Buyer each shall have delivered to the Seller certificates, dated as of a date no later than ten days prior to the Closing Date, duly issued by the Texas and New York Secretaries of State, respectively, and in each state in which the Parent and Buyer is authorized to do business, showing that each of the Parent and the Buyer is in good standing and authorized to do business and that all state franchise and/or income tax returns and taxes for the Parent and the Buyer, respectively, for all periods prior to the Closing have been filed and paid.
- (vii) NO MATERIAL ADVERSE CHANGE. No event or circumstance shall have occurred with respect to the Parent or the Buyer which would constitute a Material Adverse Effect.
- (viii) CLOSING OF IPO. The closing of the sale of the Parent Shares to the Underwriters in the IPO shall have occurred simultaneously with the IPO Closing Date hereunder.
- (ix) SECRETARY'S CERTIFICATE. The Seller shall have received a certificate or certificates, dated the Closing Date and signed by the secretary of each of the Parent and the Buyer, certifying the truth and correctness of attached copies of the Parent's and the Buyer's respective resolutions of their boards of directors and, if required, the stockholders of the Parent and the Buyer approving the Parent's and the Buyer's entering into this Agreement and the consummation of the transactions contemplated hereby.
- (x) GOLDSTEIN EMPLOYMENT AGREEMENT. The Buyer shall have entered into an employment agreement with Goldstein in the form attached as Exhibit A ("Goldstein Employment Agreement").
- (xi) REGISTRATION RIGHTS AGREEMENT. Parent shall have entered into the Registration Rights Agreement with the Seller in the form attached as Exhibit B ("Registration Rights Agreement").
- (xii) AUDITED FINANCIALS. Parent shall have delivered to Seller a certified copy of the audited reports, including a signed and certified opinion letter, prepared by Coopers & Lybrand, L.L.P. for the calendar years 1995, 1996 and for the twelve months ended June 30, 1997 of the operations of Seller's Business.
- 6.3 CONDITIONS PRECEDENT TO OBLIGATIONS OF THE PARENT AND THE BUYER. The obligations of the Parent and the Buyer with respect to actions to be taken on the Closing Date are subject to the satisfaction or waiver on or prior to the

conditions set forth in this Section 6.3(i) and (viii) that cannot be satisfied prior to the IPO Closing Date have not been satisfied or (b) as of the IPO Closing Date, if any such conditions have not been satisfied, Parent and Buyer shall have the right to terminate this Agreement, or waive any such condition, but no such waiver shall be deemed to affect the survival of the representations and warranties contained in Article III hereof.

- (i) REPRESENTATIONS AND WARRANTIES; PERFORMANCE OF OBLIGATION. All the representations and warranties of the Seller and the Stockholder contained in this Agreement shall be true and correct in all Material respects as of the Closing Date and the IPO Closing Date with the same effect as though such representations and warranties had been made on and as of such date; all of the terms, covenants and conditions of this Agreement to be complied with or performed by the Seller or the Stockholder on or before the Closing Date or the IPO Closing Date, as the case may be, shall have been duly performed or complied with in all Material respects; and the Seller and the Stockholder shall have delivered to the Buyer certificates dated the Closing Date and the IPO Closing Date, respectively, and signed by them to such effect.
- (ii) NO LITIGATION. No action or proceeding before a Governmental Authority shall have been instituted or threatened to restrain or prohibit the Acquisition or the IPO and no Governmental Authority shall have taken any other action or made any request of Parent as a result of which the management of Parent deems it inadvisable to proceed with the transactions hereunder.
- (iii) OPINION OF COUNSEL. Parent shall have received an opinion from Counsel to the Seller, dated the Closing Date, substantially in the form attached as Exhibit G-2 hereto.
- (iv) REGISTRATION STATEMENT. The Registration Statement, as amended to cover the offering, issuance and sale by Parent of such number of Parent Shares at the IPO Price as shall yield aggregate cash proceeds to the Parent from that sale (net of Underwriter's discount or commissions) in at least the Minimum Cash Amount shall have been declared effective by the SEC.
- (v) CONSENTS AND APPROVALS. All necessary consents of and filings with any Governmental Authority relating to the consummation of the transactions contemplated herein shall have been obtained and made; all consents and approvals of third parties shall have been obtained; and no

action or proceeding shall have been instituted or threatened to restrain or prohibit the Acquisition.

(vi) GOOD STANDING CERTIFICATES. Subject to Section 2.6, Seller shall have delivered to the Buyer certificates, dated as of a date no later than ten days prior to the Closing Date, duly issued by the Pennsylvania Secretary of State showing that the Seller is in good standing and authorized to do business and that all state franchise and/or income tax returns and taxes for the Seller for all periods prior to the Closing have been filed and paid.

-27-

- (vii) NO MATERIAL ADVERSE CHANGE. No event or circumstance shall have occurred with respect to the Seller which would constitute a Material Adverse Effect, and the Seller shall not have suffered any Material loss or damages to any of its properties or assets, whether or not covered by insurance, which change, loss or damage Materially affects or impairs the ability of the Seller to conduct its business.
- (viii) CLOSING OF IPO. The closing of the sale of the Parent Shares to the Underwriter of the IPO shall have occurred simultaneously with the IPO Closing Date hereunder.
- (ix) CERTIFICATE. The Buyer shall have received a certificate or certificates, dated the Closing Date and signed by the President and Secretary of the Seller, certifying the truth and correctness of attached copies of the Seller's articles of incorporation, bylaws and resolutions authorizing the consummation of the transactions contemplated hereby.
- (x) GOLDSTEIN EMPLOYMENT AGREEMENT. Goldstein shall have entered into the Goldstein Employment Agreement.
- (xi) REGISTRATION RIGHTS AGREEMENT. Seller shall have entered into the Registration Rights Agreement.
- (xii) BILL OF SALE. Seller shall have delivered to the Buyer instruments of assignment and transfer or bills of sale signed by the Seller as the Buyer reasonably requests, including the Bill of Sale attached as Exhibit C ("Bill of Sale").
- (xiii) INVESTOR REPRESENTATION LETTER. Seller and each of the Stockholders shall have delivered to the Parent at or prior to the signing of the Registration Statement an Investor Representation Letter in the form attached as Exhibit D, with respect to the acquisition of the Parent Shares to be issued to Seller.
- (xiv) STOCK PLEDGE AGREEMENT. Seller shall have delivered to Buyer a Stock Pledge Agreement in the form attached as Exhibit E ("Stock Pledge Agreement") as well as the Parent Shares issuable to the Seller at the

Closing (complete with stock powers executed in blank).

- (xv) GUARANTEE. Goldstein, individually, shall have delivered to Buyer a Guaranty of Performance in the form of Exhibit F.
- (xvi) SATISFACTION. All actions, proceedings, instruments and documents required to carry out the transactions contemplated by this Agreement or incidental hereto and all other related legal matters shall have been approved by counsel to the Parent.
- 6.4 FURTHER ASSURANCES. At and after the Closing, each of the Parties shall take all appropriate action and execute all documents of any kind which may be reasonably necessary or desirable to carry out the transactions contemplated hereby. The Seller, at any time at or after the

-28-

Closing, will execute, acknowledge and deliver any further bills of sale, assignments and other assurances, documents and instruments of transfer, reasonably requested by the Buyer, and will take any other action consistent with the terms of this Agreement that may reasonably be requested by the Buyer, for the purpose of assigning and confirming to the Buyer, all of the Assets. The Buyer shall notify the Seller promptly, and in no event more than ten (10) business days after the Buyer's receipt, of any tax inquiries or notifications thereof which relate to any period prior to the Effective Date, and the Seller shall prepare and deliver responses to such inquiries as the Seller deems necessary or appropriate. In addition, the Seller shall make available the Books and Records of the Business during reasonable business hours and take such other actions as are reasonably requested by the Buyer to assist the Buyer in the operation of the Business.

- 6.5 CONFIDENTIAL INFORMATION. After the Closing and except as otherwise specifically permitted in this Agreement or reasonably required by the Parent to conduct its business and pursue the IPO, each Party to this Agreement agrees, on behalf of itself and ifs Affiliates, to use reasonably efforts not to divulge, communicate, use to the detriment of any other Party to this Agreement or its Affiliates or for the benefit of any other person or persons, any confidential information or trade secrets of such other Party with respect to the Assets or the Business, including personnel information, secret processes, know-how, customer lists, formulae, or other technical data; provided, if any Party to this Agreement or any of its Affiliates is compelled to disclose such information to any tribunal, regulatory or governmental authority or agency or else stand liable for contempt or suffer other censure and penalty, such Party may so disclose such information without any liability hereunder.
- 6.6 ASSIGNMENT OF CONTRACTS. On or before the Effective Date, Seller shall have delivered to Buyer all of the Contracts presently in force and shall have effected a valid assignment of all of Seller's rights and obligations thereunder.

# ARTICLE VII INDEMNIFICATION

#### 7.1 INDEMNIFICATION.

A. BY THE SELLER AND THE STOCKHOLDER. Subject to Section 7.1(E) hereof, the Seller and the Stockholder, individually, jointly and severally, (collectively herein "Seller Indemnitors") shall indemnify, save, defend and hold harmless the Parent and Buyer and their respective shareholders, directors, officers, partners, agents and employees (collectively, the "Buyer Indemnified Parties") from and against any and all costs, lawsuits, losses, liabilities, deficiencies, claims and expenses, including interest, penalties, attorneys' fees and all amounts paid in investigation, defense or settlement of any of the foregoing (collectively referred to herein as "Damages"), (i) incurred in connection with or arising out of or resulting from or incident to any breach of any covenant, breach of warranty as of the Effective Date, or the inaccuracy of any representation as of the Effective Date, made by the Seller in or pursuant to this Agreement or the Ancillary Agreements, or any other agreement contemplated hereby or in any schedule, certificate, exhibit, or other instrument furnished or to be furnished by the Seller or the Stockholder under this Agreement, (ii) based upon, arising out of, or otherwise in respect of any liability or obligation of the

-29-

Business or relating to the Assets (a) relating to any period prior to the Effective Date, other than those Damages based upon or arising out of the Assumed Liabilities, or (b) arising out of facts or circumstances existing prior to the Effective Date, other than those Damages based upon or arising out of the Assumed Liabilities; provided however, that the Seller Indemnitors shall not be liable for any such Damages to the extent, if any, such Damages result from or arise out of a breach or violation of this Agreement by any Buyer Indemnified Parties, and (iii) any liability under the Securities Act, the Exchange Act or other federal or state law or regulation, at common law or otherwise, arising out of or based upon any untrue statement or alleged untrue statement of a Material fact relating to the Seller or the Stockholder, and provided to Parent or its counsel by the Seller or the Stockholder, contained in the Registration Statement or any prospectus forming a part thereof, or any amendment thereof or supplement thereto, or arising out of or based upon any omission or alleged omission to state therein a Material fact relating to the Seller or the Stockholder required to be stated therein or necessary to make the statements therein not misleading, provided however, that such indemnity shall not inure to the benefit of Parent and Buyer to the extent such untrue statement (or alleged untrue statement) was made in, or omission (or alleged omission) occurred in, any preliminary prospectus and Seller or the Stockholder provided, in writing, corrected information to Parent and Parent's counsel for inclusion in the Final Prospectus, and such information was not included or properly delivered.

- B. BY THE BUYER. Subject to Section 7.1(E) hereof, the Parent and Buyer shall indemnify, save, defend and hold harmless the Seller and the Stockholder (collectively, the "Seller Indemnified Parties") from and against any and all Damages (i) incurred in connection with or arising out of or resulting from or incident to any breach of any covenant, breach of warranty as of the Effective Date, or the inaccuracy of any representation as of the Effective Date, made by the Buyer or Parent in or pursuant to this Agreement, the Ancillary Agreements, or any other agreement contemplated hereby or in any schedule, certificate, exhibit, or other instrument furnished or to be furnished by the Buyer under this Agreement, (ii) based upon, arising out of or otherwise in respect of any liability or obligation of the Business or relating to the Assets (a) relating to any period on and after the Effective Date, other than those Damages based upon or arising out of the Retained Liabilities, or (b) arising out of facts or circumstances existing on and after the Effective Date, other than those Damages based upon or arising out of the Retained Liabilities; provided, however, that the Buyer shall not be liable for any such Damages to the extent, if any, such Damages result from or arise out of a breach or violation of this Agreement by any Seller Indemnified Party, (iii) under the Securities Act, the Exchange Act or other federal or state law or regulation, at common law or otherwise, arising out of or based upon any untrue statement or alleged untrue statement of a Material fact relating to Parent, Buyer or any Other Acquired Business contained in any preliminary prospectus, the Registration Statement or any prospectus forming a part thereof, or any amendment thereof or supplement thereto, or arising out of or based upon any omission (or alleged omission) to state therein a Material fact relating to Parent or Buyer or any of the Other Acquired Businesses required to be stated therein or necessary to make the statements therein not misleading.
- C. DEFENSE OF CLAIMS. If any lawsuit or enforcement action is filed against any Party entitled to the benefit of indemnity hereunder, written notice thereof describing such lawsuit or enforcement action in reasonable detail and indicating the amount (estimated, if necessary) or good faith estimate of the reasonably foreseeable estimated amount of Damages (which estimate shall in

-30-

no way limit the amount of indemnification the indemnified Party is entitled to receive hereunder), shall be given to the indemnifying Party as promptly as practicable (and in any event within ten (10) days, after the service of the citation or summons) ("Notice of Action"); provided that the failure of any indemnified Party to give timely notice and an estimated amount of Damages shall not affect its rights to indemnification hereunder to the extent that the indemnified Party demonstrates to the indemnifying Party that the amount of Damages the indemnified Party is entitled to recover has not increased by its failure to so notify the indemnifying Party within ten (10) days and so long as the indemnifying Party is not materially prejudiced by the failure to receive such notice. The indemnifying Party may elect to compromise or defend any such asserted liability and to assume all obligations contained in this Section 7.1 to indemnify the indemnified Party by a delivery of notice of such election ("Notice of Election") within ten (10) days after delivery of the Notice of

Action. Upon delivery of the Notice of Election, the indemnifying Party shall be entitled to take control of the defense and investigation of such lawsuit or action and to employ and engage attorneys of its own choice to handle and defend the same, at the indemnifying Party's sole cost, risk and expense, and the indemnified Party shall cooperate in all reasonable respects, at the indemnifying Party's sole cost, risk and expense (except with respect to the fees and expenses of the indemnified Party's attorney, which shall be borne by the indemnified Party) with the indemnifying Party and such attorneys in the investigation, trial, and defense of such lawsuit or action and any appeal arising therefrom; provided, however, that the indemnified Party may, at its own cost, risk and expense, participate in such investigation, trial and defense of such lawsuit or action and any appeal arising therefrom. If the Notice of Election is delivered to the indemnified Party, the indemnified Party shall not pay, settle or compromise such claim without the indemnifying Party's consent, which consent shall not be unreasonably withheld. If the indemnifying Party elects not to defend the claim of the indemnified Party or does not deliver to the indemnified Party a Notice of Election within ten (10) days after delivery of the Notice of Action, the indemnified Party may, but shall not be obligated to, defend, compromise or settle (exercising reasonable business judgment) the claim or other matter on behalf, for the account, and at the risk, of the indemnifying Party.

- D. THIRD PARTY CLAIMS. The provisions of this Section 7.1 are not limited to matters asserted by the Parties, but cover Damages incurred in connection with third party claims. The indemnity hereunder is in addition to any and all rights and remedies of the Parties in connection herewith.
- E. LIMITATION ON INDEMNIFICATION. Notwithstanding the other provisions of this Section 7.1, Seller Indemnitors shall not be liable to Buyer Indemnified Parties, and Parent and Buyer shall not be liable to Seller Indemnified Parties, for the first \$25,000 in aggregate Damages suffered by such indemnified Parties; provided, however, that once any such indemnified Parties have suffered Damages aggregating in excess of \$25,000, the indemnifying Party shall reimburse the indemnified Parties for the full amount of such Damages, including the \$25,000 in Damages initially excluded. In no event shall the aggregate Damages payable by an indemnifying Party to indemnified Parties exceed the Purchase Price.
- 7.2 SURVIVAL OF REPRESENTATIONS AND WARRANTIES. All of the representations, warranties, covenants and agreements of the Parties made herein and at the time of the Closing or in writing delivered pursuant to the provisions of this Agreement shall survive the consummation of the

-31-

transactions contemplated hereby and any examination on behalf of the Parties until the Expiration Date.

ARTICLE VIII
TERMINATION AND REMEDIES

- 8.1 SPECIFIC PERFORMANCE; REMEDIES. Each of the Parties hereby agrees that the transactions contemplated by this Agreement are unique, and that each Party shall have, in addition to any other legal or equitable remedy available to it, the right to enforce this Agreement by decree of specific performance. If any legal action or other proceeding is brought for the enforcement of this Agreement or because of an alleged dispute, breach, default or misrepresentation in connection with any of the provisions of this Agreement, the successful or prevailing Party or Parties shall be entitled to recover reasonable attorneys' fees and other costs incurred in that action or proceeding in addition to any other remedies to which it, he or they may be entitled at law or equity. The rights and remedies granted herein are cumulative and not exclusive of any other right or remedy granted herein or provided by law.
- To the extent not otherwise prohibited by OFFSET; REMEDIES. applicable law, all amounts due and owing by the Buyer to the Seller or the Stockholder under this Agreement, the Ancillary Agreements, or any other document, instrument, or agreement executed in connection herewith shall be subject to offset by the Buyer to the extent of any Damages incurred by any breach by the Seller or the Stockholder, under this Agreement or any Ancillary Agreement, or any document, instrument, or agreement executed in connection herewith. In the event Buyer elects to offset any Damages incurred as a result of any such breach, Buyer shall furnish Seller or the Stockholder, as appropriate, notice containing detailed information about the breach, the magnitude of the damages that Buyer has or reasonably expects to incur, and whether the offset is against the Parent Shares pledged under the Stock Pledge Agreement or otherwise (the act of offsetting by Buyer shall be referred to as an "Offset"). The Seller and the Stockholder acknowledge and agree that but for the right of Offset contained in this Agreement, the Buyer would not have entered into this Agreement or any of the transactions contemplated herein. If any legal action or other proceeding is brought for the enforcement of this Agreement, any Ancillary Agreement, or any document, instrument, or agreement executed in connection herewith, or because of an alleged dispute, breach, default or misrepresentation in connection with any of the provisions of this Agreement, or any Ancillary Agreement, or any document, instrument, or agreement executed in connection herewith, the successful or prevailing Party or Parties shall be entitled to recover other remedies to which it or they may be entitled at law or equity. The rights and remedies granted herein are cumulative and not exclusive of any other right or remedy granted herein or provided by law. Buyer shall not effect an Offset hereunder without giving Seller or a Stockholder, as appropriate, at least ten (10) days advance written notice of its intent to do so.
- 8.3 TERMINATION. Termination of This Agreement. (a) This Agreement may be terminated at any time prior to the Closing solely:
  - (i) by the mutual written consent of the Parent and the Seller;

- (ii) by the Seller, on the one hand, or by the Parent, on the other hand, if the transactions contemplated by this Agreement to take place at the Closing shall not have been consummated by January 15, 1998, (subject to the terms of the Side Letter) unless the failure of such transactions to be consummated results from the willful failure of the Party seeking to terminate this Agreement to perform or materially adhere to any agreement required hereby to be performed or adhered to by it prior to or at the Closing or thereafter on the IPO Closing Date;
- (iii) by the Seller, on the one hand, or by the Parent, on the other hand, if a Material breach or default shall be made by the other Party in the observance or in the due and timely performance of any of the covenants, agreements or conditions contained herein; or
- (iv) by the Parent if it is entitled to do so as provided in Section 5.11.
  - (b) this Agreement may be terminated after the Closing solely:
- (i) by the Parent or the Seller if the Underwriting Agreement is terminated pursuant to its terms after the Closing and prior to the consummation of the IPO; or
- (ii) automatically and without action on the part of any party hereto if the IPO is not consummated within ten (10) New York City business days after the date of the Closing.
- 8.4 LIABILITIES IN EVENT OF TERMINATION. If this Agreement is terminated pursuant to Section 8.3, there shall be no liability or obligation on the part of any Party hereto except to the extent that such liability is based on the breach by that Party of any of its representations, warranties or covenants set forth in this Agreement.

# ARTICLE IX COVENANTS OF BUYER AND SELLER AFTER CLOSING

# 9.1. PREPARATION AND FILING OF TAX RETURNS.

- (i) The Seller shall file or cause to be filed all federal income tax returns of the Seller for all taxable periods that end on or before the IPO Closing Date, and shall permit the Parent to review all such tax returns prior to such filings.
- (ii) Parent shall file or cause to be filed all separate tax returns of, or that include, any Other Acquired Business for all taxable periods ending after the IPO Closing Date.
- (iii) Each Party shall, and shall cause its Subsidiaries and Affiliates to, provide to each of the other Parties hereto such cooperation

-33-

relevant portions of relevant tax returns, together with relevant accompanying schedules and relevant work papers, relevant documents relating to rulings or other determinations by taxing authorities and relevant records concerning the ownership and tax basis of property, which such Party may possess. Each Party shall make its employees reasonably available on a mutually convenient basis at its cost to provide explanation of any documents or information so provided. Subject to the preceding sentence, each Party required to file tax returns pursuant to this Agreement shall bear all costs of filing such tax returns.

9.2 RESTRICTIVE LEGEND. The Seller consents to the imprinting on all certificates representing Parent Shares issued to it as part of the Purchase Price of the following legend:

THE SHARES OF COMMON STOCK REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED, NOR THE SECURITIES LAWS OF ANY STATE. SUCH SHARES MAY NOT BE SOLD, PLEDGED, HYPOTHECATED OR OTHERWISE TRANSFERRED AT ANY TIME, EXCEPT UPON (1) SUCH REGISTRATION, OR (2) DELIVERY TO THE ISSUER OF SUCH SHARES OF AN OPINION OF COUNSEL REASONABLY ACCEPTABLE TO THE ISSUER, THAT REGISTRATION IS NOT REQUIRED FOR SUCH TRANSFER, OR (3) THE SUBMISSION TO THE ISSUER OF SUCH SHARES OF OTHER EVIDENCE, REASONABLY ACCEPTABLE TO THE ISSUER, TO THE EFFECT THAT ANY SUCH SALE, PLEDGE, HYPOTHECATION OR TRANSFER WILL NOT BE IN VIOLATION OF THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED, OR OTHER APPLICABLE SECURITIES LAWS OF ANY STATE, OR ANY RULES OR REGULATIONS PROMULGATED THEREUNDER.

- 9.3 PLEDGE OF PARENT SHARES. Seller shall deliver all Parent Shares acquired from the Buyer as part of the Purchase Price to Buyer to be held pursuant to the terms of the Stock Pledge Agreement.
- 9.4 DELIVERY OF TAX CERTIFICATE. Within one year from the Closing Date, Seller shall deliver such tax certificate to Buyer as required by Section 2.8 of this Agreement.
- 9.5 SIDE LETTER The Side Letter, by and between Parent and Goldstein, dated September 15, 1997 and attached hereto as Exhibit H, is incorporated herein and made a part of this Agreement for all purposes (the "Side Letter").

ARTICLE X
MISCELLANEOUS

10.1 FEES. Except as expressly set forth herein to the contrary, each Party shall be responsible for all costs, fees and expenses (including attorney and accountant fees and expenses) paid or incurred by such Party in connection with the preparation, negotiation, execution, delivery and performance of this Agreement, or otherwise in connection with the transaction contemplated hereby.

-34-

- 10.2 MODIFICATION OF AGREEMENT. This Agreement may be amended or modified only in writing signed by all of the Parties.
- 10.3 NOTICES. All notices, consents, demands or other communications required or permitted to be given pursuant to this Agreement shall be deemed sufficiently given when delivered personally or telefaxed with receipt of transmission confirmed during regular business hours during a business day to the appropriate location described below, or three (3) business days after posting thereof by United States first-class, registered or certified mail, return receipt requested, with postage and fees prepaid and addressed as follows:

IF TO SELLER OR Lee Goldstein

STOCKHOLDER: Reporting Services Associates

225 South 15th Street, 22nd Floor

Philadelphia, PA 19102

With a copy to: Benjamin S. Ohrenstein

Attorney at Law

354 W. Lancaster Avenue

Haverford, Pennsylvania 19041

IF TO BUYER

OR PARENT: Litigation Resources of America-Northeast, Inc.

Litigation Resources of America, Inc.

650 First City Tower, 1001 Fannin

Houston, Texas 77002 Phone: 713/653-7100 Fax: 713/653-7172

With a copy to: John R. Boyer, Jr.

Boyer, Ewing & Harris Incorporated Nine Greenway Plaza, Suite 3100

Nine Greenway Plaza, Suite 3100 Houston, Texas 77046

Phone: 713/871-2025 Fax: (713) 871-2024

Any addressee at any time by furnishing notice to the other addressees in the manner described above may designate additional or different addresses for

subsequent notices or communications.

- 10.4 SEVERABILITY. The invalidity or unenforceability of any provision of this Agreement shall not invalidate or affect the enforceability of any other provision of this Agreement.
- 10.5 ENTIRE AGREEMENT; BINDING EFFECT. This Agreement and the Ancillary Agreements set forth the entire agreement among the Parties with respect to the subject matter hereof. This Agreement shall be binding upon and shall inure to the benefit of the Parties and their respective successors and assigns.

-35-

- 10.6 WAIVER. No delay in the exercise of any right under this Agreement shall waive such rights. Any waiver, to be enforceable, must be in writing.
- 10.7 GOVERNING LAW. THIS AGREEMENT SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH AND GOVERNED BY THE LAWS OF THE STATE OF TEXAS.
- 10.8 ASSIGNMENT. Neither Seller, nor Parent or Buyer may assign this Agreement or any interest therein; provided that Seller may assign its rights hereunder to Lee Goldstein, individually, and Parent and Buyer may assign their rights hereunder to an Affiliate.
- 10.9 HEADINGS. Headings in this Agreement are for convenience only and shall not affect the interpretation of this Agreement.
- 10.10 SCHEDULES AND EXHIBITS. All Schedules and Exhibits attached to this Agreement or to be delivered by the Seller, upon review and approval by the Buyer, are and shall be hereby incorporated in and made a part of this Agreement. All Schedules to this Agreement must be delivered no later than four (4) days prior to Closing, in order to provide the Buyer ample time to review and evaluate the items described therein and disclosed thereby. Although the Schedules remain subject to the review and approval of the Buyer, no such review or approval shall constitute a waiver by the Buyer of any breach or default caused by the inaccuracy or incompleteness of any Schedule, the accuracy and completeness of the Schedules being the sole responsibility of the Seller.
- 10.11 RIGHTS AND LIABILITIES OF PARTIES. Nothing in this Agreement, whether express or implied, is intended to confer any rights or remedies under or by reason of this Agreement on any persons other than the Parties, the Buyer Indemnified Parties and their respective successors and assigns, nor is anything in this Agreement intended to relieve or discharge the obligation or liability of any third persons to any Party to this Agreement, nor shall any provision give any third person any right of subrogation or action over against any Party to this Agreement.
- 10.12 SURVIVAL. Subject to Section 7.2, this Agreement, including but not limited to all covenants, warranties, representations and indemnities contained herein, shall survive the Closing, and the Bill of Sale and all other documents,

instruments or agreements relating to the Assets and the transactions contemplated herein shall not be deemed merged therein.

- 10.13 COUNTERPARTS. This Agreement may be executed in multiple counterparts, each of which shall have the force and effect of an original, and all of which shall constitute one and the same agreement.
- 10.14 ARBITRATION AND LIMITATION ON CLAIMS. Any controversy, dispute or claim arising out of, in connection with, or in relation to, the interpretation, performance or breach of this Agreement, including, without limitation, the validity, scope and enforceability of this Section which cannot first be settled through ordinary negotiation between the Parties shall be submitted in good faith to mediation by and in accordance with the Commercial Mediation Rules of the American Arbitration Association or any successor organization. In the event that mediation of such controversy, dispute or claim cannot be settled through the mediation proceeding, the Parties agree

-36-

that the controversy, dispute or claim shall be submitted to binding and final arbitration conducted in Houston, Harris County, Texas by and in accordance with the then existing Rules for Commercial Arbitration of the American Arbitration Association or any successor organization. Any such arbitration shall be to a three member panel selected through the rules governing selection and appointment of such panels of the American Arbitration Association or any successor organization. The award rendered by the arbitrators may be confirmed, entered and enforced as a judgment in any court of competent jurisdiction; however, the Parties otherwise waive any rights to appeal the award except with regard to fraud by the panel. Any such action must be brought within two years of the date the cause of action accrues. The arbitrators shall award the Party which substantially prevails in any arbitration proceeding recovery of that Party's attorneys' fees, the arbitrators' fees and all costs in connection with the arbitration from the Party who does not substantially prevail. The Parties' remedies are limited solely to the specific remedies provided in this agreement or in the other. The parties waive any entitlement to punitive damages, consequential damages and lost profits and will limit any damage claim to actual economic damages incurred. Nothing in this Section 10.14 shall restrict any Parties' ability to seek injunctive or other equitable relief in any court of competent jurisdiction prior to initiating mediation or arbitration. In the event that such injunctive or equitable relief is sought by any Party, such Party is specifically entitled to enforce the appropriate provisions of the Agreement in obtaining such relief in any court of competent jurisdiction and, thereafter, submit the remaining controversy, dispute or claim to arbitration in accordance with this Section 10.14.

10.15 DRAFTING. All Parties hereto acknowledge that each Party was actively involved in the negotiation and drafting of this Agreement and that no law or rule of construction shall be raised or used in which the provisions of this Agreement shall be construed in favor or against any Party hereto because one is deemed to be the author thereof.

- (a) Section 203(d) and 207(m)(2) of the Pennsylvania Securities Act of 1972, as amended (the "Act") requires that each person who accepts an offer to purchase securities exempted from registration directly from an issuer, shall receive written notice of such person's right to withdraw his acceptance, without incurring any liability to the seller, the underwriter or any other person, within two business days from the date of receipt by the issuer of such person's written binding contract of purchase.
- (b) By execution of this Agreement, Seller acknowledges that (i) it has received written notice of its rights under Sections 203 and 207 of the Act, and (ii) that Seller's right of withdrawal shall continue until the close of business on the second business day following the execution date of this Agreement.

-37-

IN WITNESS WHEREOF, the undersigned have executed and delivered this Agreement in multiple counterparts effective as of the date first written above.

#### BUYER:

LITIGATION RESOURCES OF AMERICA-NORTHEAST, INC., a New York corporation

By: /s/ Richard O. Looney

Richard O. Looney, Chairman and Chief Executive Officer

# PARENT:

LITIGATION RESOURCES OF AMERICA, INC., a Texas corporation

By: /s/ Richard O. Looney

Richard O. Looney, Chairman and Chief Executive Officer

SELLER:

REPORTING SERVICES ASSOCIATES, INC. a Pennsylvania corporation

By: /s/ Lee Goldstein

-----

Lee Goldstein, President

STOCKHOLDER

/s/ Lee Goldstein

Lee Goldstein, Individually

-38-

# Schedules

2.1(a) -	Equipment
2.1(b) -	Contracts
2.1(c) -	Books and Records
2.1(e) -	Intellectual Property
2.1(g) -	General Intangibles
2.2 -	Excluded Assets
2.7 -	Allocation of Purchase Price
3.3(a) -	Consents and Approvals
3.3(b) -	Breaches or Defaults
3.5 <b>-</b>	Exceptions to Title
3.6 -	Leased Assets
3.13 -	Insurance Policies
3.14 -	Banking
3.16(a) -	Employees
3.16(b) -	Independent Contractors
3.17 -	Employee Benefit Plans
3.18 -	Employment Agreement
3.19 -	Liabilities
3.20 -	Litigation
3.24 -	Certain Changes or Events
3.25 -	Customers
Exhibits	

A - Goldstein Employment Agreement
B - Registration Rights Agreement
C - Bill of Sale
D - Investor Representation Letter
E - Stock Pledge Agreement
F - Guaranty

-39-

#### EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT (this "Agreement"), dated effective the ____ day of October, 1997 (the "Effective Date"), is entered into by and between LITIGATION RESOURCES OF AMERICA-NORTHEAST, INC., a New York corporation (hereinafter called the "Company," which term includes any directly or indirectly controlled subsidiaries or successor entities), and Lee Goldstein, an individual residing in the State of Florida (the "Employee"). The Company may sometimes hereinafter be referred to as "Employer." The Employer and Employee may sometimes hereinafter be referred to singularly as a "Party" or collectively as the "Parties." All capitalized terms not otherwise defined herein shall have the same meaning as contained in that certain Agreement of Purchase and Sale of Assets executed as of September 24, 1997 (the "Purchase Agreement"), by and among the Company, Reporting Services Associates, Inc. a Pennsylvania corporation ("Seller"), the Employee, individually, and Litigation Resources of America, Inc., a Texas corporation (the "Parent").

WITNESSETH:

WHEREAS, Employee has been and employee and the Owner of the Seller and its business known as Reporting Services Associates, Inc. (the "RSA Business"), and his knowledge of the affairs of the RSA Business, particularly its court reporting business in Philadelphia, Pennsylvania, are of great value to the Company; and

WHEREAS, pursuant to the terms of the Purchase Agreement the Company has purchased from the Seller, and the Seller has sold to the Company, all or substantially all of the Assets of the Seller, which required the approval of at least a majority of the shareholders of the Seller; and

WHEREAS, part of the consideration given to the Seller and the Employee under the Purchase Agreement included an agreement by the Company to enter into this Agreement; and

WHEREAS, the Parties would not have entered into the Purchase Agreement without the execution of this Agreement.

NOW THEREFORE, for and in consideration of the mutual covenants, promises and undertakings herein contained and other consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, the Parties hereby undertake and agree as follows:

1. Employment Term. The Employer hereby employs the Employee commencing on

for one or more successive additional one (1) year terms upon mutual agreement of the Parties at least 90 days prior to the expiration of the initial term or any such renewal term. Unless otherwise provided herein, Sections 12 - 26 of this Agreement shall survive the expiration or termination of this Agreement, for any reason whatsoever. The Employee accepts such employment and agrees to perform the services specified herein, all upon the terms and conditions hereinafter stated.

- The Employee shall serve as the President of the RSA Division of 2. Duties. the Company and as Northeast Regional Vice President, and shall report to, and be subject to the general direction and control of the Chief Executive Officer and the Board of Directors of the Company (the "Board"). The Employee shall perform such management and administrative duties, consistent with the Employee's position, as are from time to time reasonably assigned to the Employee by the Chief Executive Officer and the Board including developing local, regional, and national customers for the Company and its Affiliates (defined below). The Employee also agrees to perform, without additional compensation, such other services for the Company, and for any parent, subsidiary or affiliate corporations of the Company and any partnerships in which the Company may from time to time have an interest (herein collectively called "Affiliates"), as the Chief Executive Officer or Board shall from time to time reasonably specify, if such services are of the nature commonly associated with the positions of Division President and of Regional Vice President of a company engaged in activities similar to the activities engaged in by the Company and to perform such other activities as are consistent with the Employee's past responsibilities as an employee of the Seller and the RSA Business; provided, that Employee shall not be required to engage in any business that is not reasonably related to the Business of the Company. purposes of this Agreement, the "Business of the Company" or, alternatively, "Business" shall be defined as the current business of the Company, including, but not limited to, the marketing and providing of court reporting and litigation support services in the Philadelphia, Pennsylvania area. "Company" as used in this Agreement shall be deemed to include and refer to all such Affiliates.
- 3. Extent of Service. The Employee shall devote his full business time, attention and energy to the business of the Employer, and shall not be engaged in any other business activity during the term of this Agreement. The foregoing shall not be construed as preventing the Employee from making passive investments in other businesses or enterprises, if (i) such investments will not require services on the part of the Employee which would in any material way impair the performance of his duties under this Agreement, (ii) such other businesses or enterprises are not engaged in any business competitive with the business of the Company, and (iii) the Employee has complied with Sections 12

and 13 of this Agreement with respect to each such passive investment.

- 4. Compensation. As payment for the services to be rendered by the Employee hereunder during the initial term, the Employee shall be entitled to receive:
  - (a) a salary in the amount of One Hundred Seventy-Five Thousand and No/100 Dollars (\$175,000.00) (representing One Hundred Thousand and No/100 Dollars (\$100,000.00) for services rendered as a Division President and Seventy-Five Thousand and No/100 Dollars (\$75,000.00) for services rendered as a Regional Vice

-2-

President) per year effective as of the date hereof, which shall be payable monthly or in accordance with the payroll policies of the Company in effect from time to time if such policies provide for payment of salary more frequently than monthly, until the termination of this Agreement;

- (b) a monthly car allowance in the amount of Six Hundred and No/100 Dollars (\$600.00), which shall be payable in accordance with the payroll policies of the Company;
- (c) a bonus consisting of two components to be calculated in accordance with Schedule A attached hereto, payable within ninety (90) days after the end of each fiscal year of the Company (the "Annual Bonus") including, without limitation, the first fiscal year of the Company; provided however, that any Annual Bonus calculated with respect to a fiscal year during which the Employee was employed for only a part of such year shall be prorated to account for the number of days during such year in which Employee was employed by the Company; and
- (d) a commission payment to be calculated in accordance with Schedule B attached hereto, based upon the revenues generated from National Account (as herein defined) sales originated by the Employee. The amount of commissions payable will be based upon a formula ("Formula") as it exists on the date of the origination of the National Account. A "National Account" is any account established with an insurance or "Fortune 500" company pursuant to which two or more of the Company's offices render services. Until notice of a change in the Formula applicable to the Employee is given to the Employee by the Company, the Formula applicable to the Employee shall be as set forth on Schedule B; and
- (e) In the event that the Employee identifies a business as an acquisition candidate for the Company or any of its subsidiaries, and such acquisition candidate has not been previously brought to the attention of the Company, and the Company or any of its subsidiaries ultimately acquires such business, the Employee shall be entitled to receive, at the closing of such acquisition, a cash payment equal to 10% of the investment banking fee paid to

The Gulfstar Group, Inc. or such other investment banking firm by the Company or its subsidiary in connection with such acquisition. In no event shall the Company or any subsidiary be obligated to close any acquisition with regard to any business identified by the Employee as an acquisition candidate, and no fee shall be payable hereunder unless and until such acquisition is completed.

5. Expenses. During the term of this Agreement, the Employer shall promptly pay or reimburse the Employee for all reasonable out-of-pocket expenses for travel, meals, hotel accommodations and similar items incurred by him in connection with the Business of the Company and approved by the Board or incurred in accordance with the travel and reimbursement policies of

-3-

the Company as the same shall be in effect from time to time, upon submission by him of an appropriate statement documenting such expenses.

- 6. Employee Benefits. During the term of this Agreement, the Employee shall be entitled to participate in all employee benefit plans from time to time made generally available to the executive employees of the Company, including any stock option plan, retirement plan, profit-sharing plan, group life plan, health or accident insurance or other employee benefit plans as the same shall be maintained in effect, as determined by the Board. Until the Company is able to procure its own insurance coverage, the Company agrees to continue the prior insurance previously provided to the Employee by Seller. The Employer will use commercially reasonably efforts to assist Employee in procuring insurance coverage for any preexisting conditions.
- 7. Vacation. During the term of this Agreement, the Employee shall be entitled to annual vacation time determined in accordance with the vacation policies of the Company in effect from time to time but not less than four (4) weeks per year, during which time his compensation shall be paid in full. Unused vacation time shall not accrue from year to year, unless otherwise required by law.
- 8. Covenants of Employee. For and in consideration of the employment herein contemplated and the consideration paid or promised to be paid by the Company, the Employee does hereby covenant, agree and promise that during the term hereof and thereafter to the extent specifically provided in this Agreement:
  - (a) Except as otherwise specifically permitted by this Agreement, during the term of this Agreement, Employee will not actively engage, directly or indirectly, in any other business other than that of Company, except at the direction or approval of the Company.
  - (b) The Employee will use his best reasonable efforts to truthfully and accurately make, maintain and preserve all records and reports that the Company may from time to time request or require.

- (c) The Employee will fully account for all money, records, goods, wares and merchandise or other property belonging to the Company of which the Employee has custody, and will pay over and deliver same promptly whenever and however he may be reasonably directed to do so by the Company.
- (d) The Employee will obey all rules, regulations and special instructions of the Company applicable to him, and will be loyal and faithful to the Company at all times.
- (e) The Employee will make available to the Company any and all of the information of which he has knowledge relating to the business of the Company, and

-4-

will make all suggestions and recommendations which he feels will be of mutual benefit to the Parties.

- (f) The Employee agrees that upon termination of his employment hereunder he will immediately surrender and turn over to the Company all books, records, forms, specifications, formulae, data, processes, papers and writings related to the Business of the Company and all other property belonging to the Company, together with all copies of the foregoing, it being understood and agreed that the same are the sole property of the Company.
- (q) The Employee agrees that all ideas, concepts, processes, discoveries, devices, machines, tools, materials, designs, improvements, inventions and other things of value relating to the Business of the Company (hereinafter collectively referred to as "intangible rights"), whether patentable or not, which are conceived, made, invented or suggested by him alone or in collaboration with others during the term of his employment, and whether or not during regular working hours, shall be promptly disclosed in writing to the Company and shall be the sole and exclusive property of the Company. The Employee hereby assigns all of his right, title and interest in and to all such intangible rights to the Company, and its successors or In the event that any of such intangible rights shall be deemed by the Company to be patentable or otherwise registerable under any federal, state or foreign law, the Employee further agrees that, at the expense of the Company, he will execute all documents and do all things reasonably necessary, advisable or proper to obtain patents therefor or registration thereof, and to vest in the Company full title thereto.
- 9. Mutual Covenants of the Company and the Employee. For and in consideration of the employment herein contemplated and the compensation, covenants, conditions and promises herein recited, the Company and the Employee do hereby mutually agree that during the term hereof:
  - (a) The Employee shall not, by reason of this Agreement, have any vested interest in, or right, title or claim to, any land, buildings,

equipment, machinery, processes, systems, products, contracts, goods, wares, merchandise, business assets or other things of value belonging to or which may hereafter be acquired or owned by the Company.

(b) In carrying out his duties as President of the RSA Division and Northeast Regional Vice President of the Company, the Employee shall primarily be responsible for making day-to-day decisions in the ordinary course of business of the Company, subject to possible review by the Chief Executive Officer and/or the Board. The responsibility for the Company's plans, properties, contracts, methods, and policies shall be vested in the Board and the Company may, in its sole and

-5-

absolute discretion, give, sell, assign, transfer or otherwise dispose of any or all of its assets or businesses in whole or in part, to any person, firm or corporation, whether or not such person, firm or corporation is in any manner owned by or associated with or affiliated with the Company.

- (c) The Employee acknowledges that because of the nature of the position for which he has been employed, the Employee may be called upon to perform such duties and render such services as are required of him hereunder irregularly, and agrees to perform to the best of his abilities such duties as the business may reasonably demand, and acknowledges that the number of hours per day or per week may vary. Notwithstanding the foregoing, the Employee shall work in a manner that is consistent with his prior customary practice on behalf of the Seller and the RSA Business.
- (d) The Company agrees that it will not terminate any employee of the Company without giving prior notification of such termination to the Employee.
- 10. Termination of Employment for Cause. The Employer may terminate the employment of the Employee if the Employer suffers or may reasonably be expected to suffer any material adverse effect as a result of the Employee (any such termination being a termination for "Cause"):
  - (a) Breaching any material provision of this Agreement and failing to cure such breach within ten (10) days after receipt of written notice thereof;
    - (b) Misappropriating funds or property of the Company;
  - (c) Securing any personal profit not thoroughly disclosed to and approved by the Company in connection with any transaction entered into on behalf of the Company;
  - (d) Engaging in conduct, even if not in connection with the performance of his duties hereunder, which would reasonably be expected to result in a material adverse effect to the interest of the Company if he was retained as an employee, such as his commission of a felony or a crime of

- (e) Becoming and remaining "Disabled," as hereinafter defined (either physically, mentally or otherwise) for a period of one hundred thirty-five (135) days during any consecutive twelve-month time period;
- (f) Failing to carry out and perform the duties assigned to the Employee in accordance with the terms hereof and failing to cure such breach within ten (10) days after written notice thereof; or

-6-

(g) Failing to comply with corporate policies of the Company that are promulgated from time to time and made known to Employee and failing to cure such breach within ten (10) days after written notice thereof.

In the event of the death of the Employee, such occurrence shall immediately constitute a termination for Cause. Except as provided in item (e) above, no termination for Cause shall be effective if the Employee is Disabled.

In the event the Employee is terminated for Cause because he is Disabled, the Employee may be permitted to participate in any disability insurance policy the Company then has in effect.

In the event of termination of his employment for Cause, the Employee shall be entitled to receive his compensation, as determined in Section 4 of this Agreement, due or accrued on a pro rata basis to the date of termination. Any salary or remuneration owed as of the date of termination shall be paid less the amount of damages, if any, caused to the Company by such breach, but no such damages offset shall extend beyond any compensation due and owing under this Agreement.

Notwithstanding the cure provisions provided in Sections 10(a), 10 (f) and 10(g), the Employee shall not have the opportunity to cure any violation of these subsections if such violation cannot reasonably be expected to be cured. In such event, the Company shall be required to furnish the Employee notice of the violation, but the Employee shall not be furnished an opportunity to cure.

"Disabled" shall mean the continuous inability, whether mental or physical, of Employee to perform his normal job functions as determined by at least two of three medical physicians selected as follows: the Employee or his legal designee shall be entitled to appoint one physician, the Company shall be entitled to appoint one physician, and such two appointed physicians shall mutually appoint a third physician. Notwithstanding the foregoing, the Employee, or his designee, and the Company may mutually agree that he is "Disabled" within the meaning of this Agreement.

11. Termination By the Company Without Cause or By the Employee With Good Reason. The Company may terminate the employment of Employee for any

reason other than those for Cause, in which event such termination shall be deemed a "Termination Without Cause." In addition, the Employee shall have the right to terminate this Agreement for any material breach of this Agreement by the Company, which shall include but not be limited to materially changing the duties assigned to Employee beyond those contemplated in Section 2 of this Agreement or causing Employee to relocate his primary residence in violation of Section 2 of this Agreement; provided that the Company shall be furnished ten (10) days notice of such breach and an opportunity to cure (any such termination constituting a "Termination By Employee With Good Reason"). Notwithstanding the cure provisions provided in the preceding sentence, the Employer shall not have the opportunity to cure any violation of this Agreement if such violation cannot reasonably be expected to be cured

-7-

but the Employee shall still furnish notice to the Company. In the event of a Termination Without Cause or a Termination with Good Reason by the Employee, the Company shall continue making payments to Employee, but at a salary level equal to the Division President portion of Employee's compensation only, as set forth in Section 4 of this Agreement, for a period equal to the lesser of (i) one (1) year, or (ii) the remaining term of this Agreement, which amount, in the event of a Termination Without Cause or a Termination By Employee With Good Reason, shall constitute the full and total amount of liquidated damages that the Employee shall be entitled to receive from the Company and its Affiliates for any contractual or tort claims arising out of his employment relationship with the Company.

- 12. Covenant Not to Compete. The Employee recognizes that the Company has business goodwill and other legitimate business interests which must be protected in connection with and in addition to the Information (as defined hereinafter), and therefore, in exchange for access to the Information, the specialized training and instruction which the Company will provide, the Company's agreement to employ the Employee on the terms and conditions set forth herein, the Company's agreement to execute and consummate the Purchase Agreement, and the promotion and advertisement by the Company of Employee's skill, ability and value in the Company's business, subject to the provisions of the next full paragraph of this Section 12, the Employee agrees that in the event (i) Employee is terminated for Cause, or (ii) Employee leaves the employ of the Company other than a Termination By Employee With Good Reason prior to expiration of the term of the Agreement, or (iii) upon the expiration of the term of this Agreement, then for a period of the latest date of (i) five (5) years after the date of this Agreement, or (ii) three (3) years after the date employment is so terminated, except in the event of Termination Without Cause or in the event of Termination By Employee With Good Reason By Employee:
  - (a) Employee will not in any capacity or relationship enter into, engage in, or be connected with any business or business operation or activity within a fifty (50) mile radius of any office location then operated by the Employer at the time of such termination, which consists in whole or in part

- (b) Employee will not call upon any customer whose account is serviced in whole or in part by the Employer or its Affiliates at the time of the termination of Employee's employment, with the purpose of selling or attempting to sell to any such customer any services included within that offered by the Employer or its Affiliates; and
- (c) Employee will not intentionally divert, solicit or take away any customer, supplier or employee of the Employer or its Affiliates, or the patronage of any customer or supplier of the Employer or its Affiliates, or otherwise interfere with or disturb the relationship existing between the Employer or its Affiliates and any of their respective customers, suppliers or employees, directly or indirectly.

-8-

The foregoing restrictive covenants shall apply to the Employee in the event of his Termination Without Cause or in the event of Termination By Employee With Good Reason by the Employee, but only for a period of one (1) year.

In the event the Company ceases operation of the Business of the Company other than in a merger, consolidation, or similar transaction, or upon the filing of a bankruptcy or receivership proceeding against the Employer, or upon the appointment of a liquidator for the Company, the provisions of this Section 12 shall not be applicable to the conduct of Employee subsequent thereto.

It is mutually understood and agreed that if any of the provisions relating to the scope, time or territory in this Section 12 are more extensive than is enforceable under applicable laws or are broader than necessary to protect the good will and legitimate business interests of the Company, then the Parties agree that they will reduce the degree and extent of such provisions by whatever minimal amount is necessary to bring such provisions within the ambit of enforceability under applicable law.

The Parties acknowledge that the remedies at law for breach of Employee's covenants contained in this Section 12 are inadequate, and they agree that the Company shall be entitled, at its election, to injunctive relief (without the necessity of posting bond against such breach or attempted breach), and to specific performance of such covenants in addition to any other remedies at law or equity that may be available to the Company.

13. Business Opportunities. Except for passive investments by the Employee in publicly traded entities, or investments in private ventures which do not compete with, or are not in the same business as, the Company and which come to the attention of the Employee outside of the scope of his employment, for as long as the Employee shall be employed by the Company and thereafter with

respect to any business opportunities learned about during the time of Employee's employment by the Company, the Employee agrees that with respect to any future business opportunity or other new and future business proposal which is offered to, or comes to the attention of, the Employee and which is in any way related to, or connected with, the Business of the Company, the Company shall have the right to take advantage of such business opportunity or other business proposal for its own benefit. The Employee agrees to promptly deliver notice to the Board in writing of the existence of such opportunity or proposal and the Employee may take advantage of such opportunity only if the Employer does not elect to exercise its right to take advantage of such opportunity.

14. Confidential Information. The Employee acknowledges that in the course of his employment with the Company, he will receive certain trade secrets, know-how, lists of customers, employee records and other confidential information and knowledge concerning the Business of the Company (hereinafter collectively referred to as "Information") which the Company desires to protect. The Employee understands that such Information is confidential and he agrees that he will not reveal such Information to anyone outside the Company except (i) for information already known to the public, now or in the future, or (ii) in connection with any legal proceeding

-9-

regarding this Agreement, the Purchase Agreement or the transactions contemplated thereby or as otherwise required by law or judicial order. The Employee further agrees that during the term of this Agreement and thereafter he will not use such Information in competing with the Company. Upon termination of his employment hereunder, the Employee shall surrender to the Company all papers, documents, writings and other property produced by him or coming into his possession by or through his employment hereunder and relating to the information referred to in this Section 14, which are not general knowledge in the industry, and the Employee agrees that all such materials will at all times remain the property of the Company.

15. Notices. All notices, consents, demands or other communications required or permitted to be given pursuant to this Agreement shall be deemed sufficiently given when delivered personally with a written receipt acknowledging delivery or telefaxed with receipt confirmed, or three (3) business days after the posting thereof by United States first class, registered or certified mail, return receipt requested, with postage fee prepaid and addressed as follows:

If to the Company:

Litigation Resources of America-Northeast, Inc. 1001 Fannin, Suite 650 Houston, Texas 77002 Telefax: (713) 653-7172 Attn: Richard O. Looney If to the Employee: Lee Goldstein

Reporting Services Associates
225 South 15/th/ Street, 22/nd/ Floor

Philadelphia, PA 19102

Any Party may change its address for notice hereunder by providing written notice of such change to the other Party hereto.

- 16. Specific Performance. The Employee acknowledges that a remedy at law for any breach or attempted breach of Sections 12, 13 or 14 of this Agreement will be inadequate, the Employee agrees that the Company shall be entitled to specific performance and injunctive and other equitable relief in case of any such breach or attempted breach, and further agrees to waive any requirement for the securing or posting of any bond in connection with the obtaining of any such injunctive or any other equitable relief.
- 17. Severability. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement shall be prohibited by or invalid under applicable law, such provisions shall be ineffective to the extent of such provision or invalidity only, without invalidating the remainder of such provision or the remaining provisions of this Agreement.

-10-

- 18. Assignment. This Agreement may not be assigned by the Employee. Neither the Employee, his spouse nor their estates shall have any right to encumber or dispose of any right to receive payments hereunder, it being understood that such payments and the right thereto are nonassignable and nontransferable.
- 19. Binding Effect. Subject to the provisions of Section 18 of this Agreement, this Agreement shall be binding upon and inure to the benefit of the Parties hereto, the Employee's heirs and personal representatives, and the successors and assigns of the Company.
- 20. Governing Law. This Agreement shall be construed and enforced in accordance with and governed by the laws of the State of Texas.
- 21. Prior Employment Agreements. Employee represents and warrants to the Company that he has fulfilled all of the terms and conditions of all prior employment agreements to which he may be or have been a party, and at the time of execution of this Agreement is not a party to any other employment agreement.
- 22. Parol Evidence. This Agreement constitutes the sole and complete agreement between the Parties hereto with respect to the subject matter hereof, and no verbal or other statements, inducements or representations have been made to or relied upon by either Party, and no modification hereof shall be effective unless in writing signed and executed in the same manner as this Agreement,

provided, however, the amount of compensation to be paid Employee for services to be performed for Company may be changed from time to time by the Parties hereto by written agreement without in any other way modifying, changing or affecting this Agreement and the performance by the Employee of any of the duties of his employment with the Company. Written notification of any modification of compensation paid or payable to the Employee for his services shall be conclusively deemed to be a ratification and confirmation of this Agreement amended by such change in compensation unless the Employee shall object in writing with ten (10) days after such written notification from the Company.

- 23. Waiver. Any waiver to be enforceable must be in writing and executed by the Party against whom the waiver is sought to be enforced.
- 24. Arbitration. Any controversy, dispute or claim arising out of, in connection with, or in relation to, the interpretation, performance or breach of this Agreement, including, without limitation, the validity, scope and enforceability of this Section which cannot first be settled through ordinary negotiation between the Parties shall be submitted in good faith to mediation by and in accordance with the Commercial Mediation Rules of the American Arbitration Association or any successor organization. In the event that mediation of such controversy, dispute or claim cannot be settled through the mediation proceeding, the Parties agree that the controversy, dispute or claim shall be submitted to binding and final arbitration conducted in Houston, Harris County, Texas by and in accordance with the then existing Rules for Commercial Arbitration of the American Arbitration Association or an successor organization. Any such arbitration shall be to a three

-11-

member panel selected through the rules governing selection and appointment of such panels of the American Arbitration Association or any successor organization. The award rendered by the arbitrators may be confirmed, entered and enforced as a judgment in any court of competent jurisdiction; however, the Parties otherwise waive any rights to appeal the award except with regard to fraud by the panel. Any such action must be brought within two years of the date the cause of action accrues. The arbitrators shall award the Party which substantially prevails in any arbitration proceeding recovery of that Party's attorneys fees, the arbitrators' fees and all costs in connection with the arbitration from the Party who does not substantially prevail. The Parties' remedies are limited solely to the specific remedies provided in this agreement. The Parties waive any entitlement to punitive damages, consequential damages and lost profits and will limit any damage claim to actual economic damages incurred. Nothing in this Section 24 shall restrict the Company's ability to seek injunctive or other equitable relief in any court of competent jurisdiction prior to initiating mediation or arbitration for any violation by the Employee of Sections 12, 13, 14 or 16 of this Agreement. In the event that such injunctive or equitable relief is sought by the Company, the Company is specifically entitled to enforce the provisions of these Sections in obtaining

such relief in any court of competent jurisdiction and, thereafter, submit the remaining controversy, dispute or claim to arbitration in accordance with this Section 24.

- 25. Drafting. All Parties hereto acknowledge that each was actively involved in the negotiation and drafting of this Agreement and that no law or rule of construction shall be raised or used in which the provisions of this Agreement shall be construed in favor or against any Party hereto because one is deemed to be the author thereof.
- 26. Multiple Counterparts. This Agreement may be executed in multiple counterparts, each of which shall have the force and effect of an original, and all of which shall constitute one and the same agreement.

-12-

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement as of the date and year first above written.

THE COMPANY:

LITIGATION RESOURCES OF AMERICA-NORTHEAST, INC., a New York corporation

By:
Richard O. Looney

Chief Executive Officer

THE EMPLOYEE:

Lee Goldstein

Schedule A--Calculation of Annual Bonus Schedule B--Calculation of Commissions

-13-

SCHEDULE A

## CALCULATION OF ANNUAL BONUS

Each year the accountants regularly employed by the Company shall determine the amount of EBIT Profit, if any, of (i) the RSA Division and (ii) the Northeast Region of the Company (being that region of the Company which Lee

Goldstein, as RVP, is responsible for managing) during each consecutive twelve (12) month time period ending on the last day of the fiscal year of the Company ("respectively the "Annual RSA EBIT Profits" and the "Annual Northeast Region EBIT Profits"), commencing with the first fiscal year of the Company and continuing each year during the term of this Agreement. Beginning with the first fiscal year of the Company, (x) to the extent that the Annual RSA EBIT Profits for the current year exceed the Annual RSA EBIT Profits for the prior year, the Employee shall be paid an annual bonus equal to ten percent (10%) of the amount of such excess, if any ("Bonus Component 1"), and (y) to the extent that the Annual Northeast Region EBIT Profits for the current year exceed the Annual EBIT Profits for the prior year, the Employee shall be paid an annual bonus equal to ten percent (10%) of the amount of such excess, if any ("Bonus Component 2"); provided that for the first fiscal year of the Company (A)(i) the Bonus Components 1 and 2, respectively shall be calculated for each full month of operations and added together, (ii) the Bonus Components 1 and 2 respectively for any partial month of operations shall be divided by the number of actual days in such month and multiplied by 30 to create a full month and (iii) the sum of (A)(i) and (A)(ii) shall be added together, that result divided by the number of full and partial months of operations and the quotient multiplied by 12 to create the number representing Bonus Components 1 and 2 for the first fiscal year and (B) the Annual Northeast Region EBIT Profits for the prior year shall be deemed to be \$ [NET WORTH ON BALANCE SHEET] and the Annual RSA EBIT Profits for the prior year shall be deemed to be \$ For purposes of this calculation, EBIT Profits for each of the RSA Divisions in the Northeast Region shall mean their gross earnings before income taxes, interest, depreciation and amortization, but excluding the effects of any acquisitions and after deductions for all annual bonuses to be paid to the managers in the Northeast Region.

-14-

SCHEDULE B

-----

<TABLE>
<CAPTION>
Commission Price Structure of Percentage of Gross Sales National Account

<S> <C>

3.00% Market price

2.50% Discount of 5% to 10% from market price rate

2.00% Discount of 10% to 15% from market price

rate

Discount of 15% to 20% from market price rate

1.00%

Discount of 20% to 25% from market price rate

Discount of 25% or more from market price rate

</TABLE>

In certain circumstances, more than one individual may be entitled to receive commissions based upon sales from a National Account. In such circumstances, the commissions described above would be shares by the Employee and the other individuals on such a basis as is determined to be fair by the Company's Board of Directors. All commissions payable hereunder will be paid in cash within 45 days of the end of the fiscal year of the Company in which they are earned.

-15-

EXHIBIT C

DRAFT OF SEPTEMBER 22, 1997

# BILL OF SALE, ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS BILL OF SALE, ASSIGNMENT AND ASSUMPTION AGREEMENT (this Bill of Sale") is entered into effective as of September____, 1997, among REPORTING SERVICES ASSOCIATES, INC., a Pennsylvania corporation ("Seller"), and LITIGATION RESOURCES OF AMERICA-NORTHEAST, INC., a New York corporation ("Purchaser"). Purchaser and Seller may be hereinafter sometimes referred to collectively as the "Parties" or individually as a "Party." All defined terms not otherwise defined herein shall have the meanings ascribed to them in that certain Agreement of Purchase and Sale of Assets of even date herewith (the "Agreement"), executed among Seller, Purchaser and Litigation Resources of America, Inc., a Texas corporation.

W I T N E S S E T H:

WHEREAS, Purchaser has agreed to purchase from Seller, and Seller has agreed to grant, bargain, sell, convey, transfer, assign and deliver to Purchaser, the Assets (but not the Excluded Assets); and

WHEREAS, as partial consideration for the sale and assignment of the Assets, Purchaser has agreed to assume the Assumed Liabilities, on and subject to the terms and conditions set forth in the Agreement.

NOW, THEREFORE, in consideration of the premises and other good and valuable consideration, including the delivery to Seller of the Purchase Price, the receipt and sufficiency of which are hereby acknowledged, Purchaser and Seller hereby agree as follows:

- 1. SALE AND ASSIGNMENT. Seller has granted, bargained, sold, conveyed, transferred, assigned and delivered, and by these presents does grant, bargain, sell, convey, transfer, assign and deliver unto Purchaser, its successors and assigns, the Assets. Seller warrants that Seller is the lawful owner in every respect of all of the Assets and that the Assets are free and clear of any and all liens, security agreements, encumbrances, claims, demands, and charges of every kind and character whatsoever other than as previously disclosed in writing to Purchaser. Seller hereby binds Seller and Seller's successors and assigns to warrant and defend the title to all of the Assets unto Purchaser and Purchaser's successors and assigns forever against every person whomsoever lawfully claiming or to claim the Assets or any part thereof. Purchaser hereby accepts the conveyance, transfer, assignment and delivery of the Assets.
- 2. ASSUMPTION. Subject to the exceptions and exclusions of Section 2.6 of the Agreement, and otherwise on and subject to the terms and conditions of the Agreement, Purchaser hereby assumes and agrees to pay and perform the Assumed Liabilities.
- 3. FURTHER ACTIONS. Seller hereby consents and agrees to any lawful action taken by Purchaser in connection with the enforcement of, or the legal protection of, the Assets, and confers upon Purchaser full right of substitution in any and all such actions. Seller further covenants and agrees to execute such further documents and take such additional actions as may reasonably be requested by Purchaser to vest in Purchaser any and all of the Assets and otherwise to effectuate the intent of this Bill of Sale. Each of the Parties shall perform such actions and deliver or cause to be delivered any and all such documents, instruments and agreements as the other Party may reasonably request for the purpose of fully and effectively carrying out this Agreement and the transactions contemplated hereby.
- 4. GOVERNING LAW; JURISDICTION; VENUE; SERVICE. This Agreement shall be construed and enforced in accordance with and governed by the laws of the State of Texas, without regard to conflicts of law principles, and the laws of the United States applicable in Texas. Venue for any litigation between the Parties hereto with respect to the subject matter of this Agreement shall be Harris County, Texas. Each Party hereby irrevocably submits to personal jurisdiction in Texas. Each Party hereby waives all objections to personal jurisdiction in Texas and venue in Harris County for purposes of such litigation. Each Party waives summons or citation and agrees that delivery of a duly filed complaint or petition as provided in the notice section of this Agreement will suffice as substitute service of summons or citation.
- 5. MODIFICATION OF AGREEMENT. This Agreement may be amended or modified only by written instrument signed by all of the Parties.

6. ENTIRE AGREEMENT; BINDING EFFECT. This Agreement, and the documents,
instruments and agreements executed in connection herewith, set forth the entire
agreement and understanding between the Parties with respect to the subject
matter hereof and thereof. This Agreement shall be binding upon and shall inure
to the benefit of the Parties and their respective successors and assigns.
7. COUNTERPARTS. This Agreement may be executed in multiple counterparts,
each of which shall have the force and effect of an original, and all of which
together shall constitute one and the same agreement.

EXECUTED	AND	DELIVERED	EFFECTIVE	as	of	the	date	first	written	above.

2

Lee Goldstein, President
Ву:
REPORTING SERVICES ASSOCIATES:
SELLER:
Richard O. Looney, President
By:
LITIGATION RESOURCES OF AMERICA-NORTHEAST, INC., a New York corporation
PURCHASER:

A:\BILL OF SALE1.WPD

3

EXHIBIT D

DRAFT OF SEPTEMBER 22, 1997

[FORM OF SELLER'S INVESTOR REPRESENTATION LETTER]

, 1
-----

Litigation Resources of America, Inc.
Litigation Resources of America -- Northeast, Inc.
1001 Fannin, Suite 650
Houston, Texas 77002
Attn: Mr. Richard O. Looney, President

Re: Investor Representations

### Gentlemen:

The purpose of this letter is to evidence certain representations and warranties to be made with respect to certain matters relating to the acquisition by Reporting Services Associates, Inc., a Pennsylvania corporation and/or its sole shareholder, Lee Goldstein, an individual rediding in the State of Florida (the "Seller") of shares of Common Stock issued by Litigation Resources of America, Inc., a Texas corporation (the "Company"), having a par value of one-cent (\$0.01) per share (the "Common Stock"), for and in partial consideration of the sale of certain of the assets of the Seller to Litigation Resources of America -- Northeast, Inc., a New York corporation ("Buyer"), upon the terms and conditions set forth herein and in that certain Agreement of Purchase and Sale of Assets (the "Purchase Agreement"), entered into by and among the Seller, the sole shareholder of the Seller, the Company, and the Buyer. The undersigned Seller or shareholder of Seller is sometimes hereinafter referred to as the "Investor."

The Investor hereby represents and warrants to the Company, the Buyer, and each of the Company's and the Buyer's officers, directors, shareholders, agents, attorneys, employees and representatives as follows:

1. Investment Intent. (i) The Common Stock is being acquired solely for the account of the Seller, for investment and not with a view to or for the resale, distribution, subdivision or fractionalization thereof, (ii) the Seller has no contract, understanding, undertaking, agreement or arrangement with any person to sell, transfer or pledge to any person the Common Stock or any part thereof, (iii) the Seller has no present plans to enter into any such contract, undertaking, agreement or arrangement, (iv) the Seller understands the legal consequences of the foregoing representations and warranties to mean that the Seller must bear the economic risk of the investment in the Common Stock for an indefinite period of time, (v) the Investor has such knowledge and experience in financial and business matters that the Investor is capable of evaluating the merits and

Litigations Resources of America, Inc. Litigations Resources of America -- Northeast, Inc. risks of acquiring the Common Stock, and (vi) the Investor acknowledges that the acquisition of the Common Stock by the Seller involves a high degree of risk which may result in the loss of the total amount of this investment.

- 2. No General Solicitation. The Common Stock has been offered to the Seller without any form of general solicitation or advertising of any type by or on behalf of the Company or any of its officers, directors, shareholders, employees, agents, attorneys or representatives.
- 3. Access to Information. The Seller, and to the extent the Investor is not the Seller, the Investor, has (i) for a reasonable amount of time had an opportunity to ask questions and receive answers concerning the terms and conditions of the issuance of the Common Stock and the proposed business and affairs of the Company, and is satisfied with the results thereof, (ii) has been given access, if requested, to all other documents with respect to the Company or this transaction, as well as to such other information as the Seller or the Investor has requested, and (iii) has relied solely on investigations conducted by the Investor in making the decision to acquire the Common Stock or approve the transactions set forth in the Purchase Agreement.
- 4. Exemption Status. The Investor understands that the Common Stock to be sold hereunder are being issued in reliance upon the exemptions from registration under the Securities Act of 1933, as amended. The Investor understands that the undersigned, the Company, the Company's officers, directors, shareholders, employees, agents, attorneys and representatives are relying on, among other things, the representations and warranties of the Investor set forth herein in issuing the Common Stock to the Seller.
- 5. Securities Compliance. The Investor understands and agrees that (i) no sale, distribution, transfer or other disposition of the Common Stock, or any portion thereof, can be made by the Seller unless the Common Stock has been registered under the Securities Act of 1933, as amended, and applicable securities laws of any other relevant jurisdiction, or exemptions from such registration are available, as evidenced by an opinion of counsel, satisfactory to the Company, with respect to the proposed sale, distribution, transfer or other disposition, and (ii) an appropriate legend will be endorsed on the Common Stock evidencing such restrictions.
- 6. Accredited Investor Status. The Investor is an "accredited investor" within the meaning of Rule 501 of Regulation D promulgated under the Securities Act.
- 7. Representations of Natural Persons. If the Investor is a natural person, the Investor has reached the age of majority in the state in which the Investor resides, has adequate

Litigations Resources of America, Inc. Litigations Resources of America Northeast, Inc, 1997
Page 3
means of providing for the Investor's current financial needs and contingencies, is able to bear the substantial economic risks of an investment in the Common Stock, has no need for liquidity in such investment, and is able to withstand a complete loss of such investment.
8. Representations of Entities. If Investor is a corporation, partnership, trust or other entity, (i) it is authorized and qualified to purchase and hold the Common Stock, (ii) it has not been formed for the purpose of acquiring the Common Stock, (iii) the person executing this Agreement for and on behalf of such entity has been duly authorized by such entity to do so, (iv) it is willing and able to bear the substantial economic risk of an investment in the Common Stock and has no need for liquidity with respect thereto, and (v) it is able to withstand a complete loss of its investment.
9. No Governmental Review. The Investor acknowledges and understands that no federal or state agency has passed on the fairness of the investment in the Common Stock, nor made any recommendation or endorsement of the Common Stock, and that there is a significant risk of loss of all or a portion of the Seller's investment in the Common Stock.
10. State of Residence and Domicile. The Investor is either (i) a permanent resident of the State of Florida, or (ii) not a resident or citizen of the United States.
The Investor acknowledges that the Company and the Company's officers, directors, agents, attorneys and other representatives are relying on the representations and warranties set forth herein, and would not deliver the Common Stock to the Seller but for the execution and delivery of this letter by the Investor.
Very truly yours,
EXHIBIT E DRAFT AS OF NOVEMBER 6, 1997
STOCK PLEDGE AGREEMENT
THIS STOCK PLEDGE AGREEMENT (this "Pledge Agreement") is made effective as of the day of, 1997, by REPORTING SERVICES

ASSOCIATES, a Pennsylvania corporation ("Pledgor"), and LITIGATION RESOURCES OF AMERICA -- NORTHEAST, INC., a New York corporation ("Secured Party"). All capitalized terms contained herein without definition shall have the respective meanings given to them in that certain Agreement of Purchase and Sale of Assets dated of even date herewith (the "Purchase Agreement") by and among the Pledgor, Secured Party, Litigation Resources of America, Inc., a Texas corporation and the parent company of the Secured Party (the "Parent"), and the stockholder of the Pledgor.

W I T N E S S E T H:

WHEREAS, Pledgor has agreed to sell substantially all of its assets to Secured Party upon the terms and conditions contained in the Purchase Agreement; and

WHEREAS, Pledgor has certain obligations under the Purchase Agreement, including, but not limited to, the obligation of Pledgor to indemnify Secured Party for any breaches of representations and warranties of Pledgor contained in the Purchase Agreement; and

WHEREAS, pursuant to the terms of the Purchase Agreement and as partial consideration for the purchase of the stock of the Company by the Secured Party, the Pledgor has been issued an aggregate of _____ shares of common stock, \$.01 par value per share (the "Stock"), of Parent; and

WHEREAS, the terms of the Purchase Agreement provide for the Pledgor to pledge the Stock, whether now owned or hereinafter acquired, to the Secured Party to partially secure the obligations of Pledgor under the Purchase Agreement.

NOW, THEREFORE, in consideration of the premises and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties agree as follows:

1. Pledge of Stock. Pledgor hereby pledges and grants to Secured Party a security interest in the Stock, which shall attach immediately upon each issuance of Stock to all shares of Stock issued to Pledgor in accordance with the terms of the Purchase Agreement. Immediately upon receipt of any shares of Stock, Pledgor shall be required to deliver to Secured Party the certificate or certificates representing the Stock in order that Secured Party might perfect its security interest therein. The Pledgor and the Secured Party hereby acknowledge and agree that the value of the Stock ("Agreed Value") shall be deemed to be (i) the IPO Price if shares are being surrendered hereunder in order to effect an adjustment in the Purchase Price and (ii) if shares are being surrendered hereunder for any other reason, the average public trading price of each share of Stock

over the five (5) most recent business days falling prior to the date of delivery by the Secured Party to the Pledgor of the notice of an event requiring an Offset, as such term is defined in the Purchase Agreement. Pledgor shall possess all voting rights pertaining to the Stock, so long as an Event of Offset, as hereinafter defined, has not occurred, or if an Event of Offset has allegedly occurred but is being disputed by the parties hereto prior to submission to arbitration in accordance with Section 10.14 of the Purchase Agreement, and Secured Party shall have no voting rights that may be presently or hereafter attributable to the Stock. In addition, so long as an Event of Offset has not occurred, or if an Event of Offset has allegedly occurred but is being disputed by the parties hereto prior to submission to arbitration in accordance with Section 10.14 of the Purchase Agreement, then Pledgor shall have the right to receive all dividends, if any, on the Stock, and Pledgor shall be entitled to receive all proceeds upon liquidation of the Stock, if any, as well as all other rights with respect to the Stock except for the right to transfer title thereto. Notwithstanding the foregoing, if an Event of Offset has occurred and (i) has been resolved, either by failure to timely dispute it as required by Section 10.14 of the Purchase Agreement, by agreement or by arbitration decided in favor of Secured Party (a "Resolved Event of Offset") or (ii) has been submitted to arbitration in accordance with Section 10.14 of the Purchase Agreement which arbitration is still pending or in process (a "Continuing Event of Offset"), then Secured Party shall have the right to designate a representative or trustee to vote those shares of Stock covered by or subject to the Resolved Event of Offset or Continuing Event of Offset (the "Offset Shares"), to receive all dividends and liquidation proceeds with respect to the Offset Shares, and to receive all other rights with respect to the Offset Shares.

- 2. Representations and Warranties. Pledgor hereby represents, warrants and covenants to and with Secured Party that:
  - (a) Pledgor will not, without the written consent of Secured Party, sell, contract to sell, encumber, or dispose of the Stock or any interest therein until this Pledge Agreement and all obligations under the Purchase Agreement have been fully satisfied.
  - (b) No consent of any party is necessary for the Pledgor to perform its obligations hereunder, or if any such consent is required, such consent has been received prior to the execution of this Pledge Agreement.
- 3. Event of Offset. Each delivery by Secured Party to the Pledgor of a notice of a claim of offset shall constitute an Event of Offset ("Event of Offset") under this Pledge Agreement.

#### 4. Remedies.

(a) Upon the occurrence of a Resolved Event of Offset, Secured Party may, at its option, exercise with reference to the Stock any and all of the rights and remedies of a secured party under the Uniform Commercial Code as adopted in the State of Texas and as otherwise granted therein or under any

other applicable law or under any other agreement executed by Pledgor, including, without limitation, the right and power to sell, at public or private sale(s), or otherwise dispose of or keep the Stock and any part or parts thereof, or interest or interests therein owned by Pledgor, in any manner authorized or permitted under

-2-

this Pledge Agreement or under the Uniform Commercial Code, and to apply the proceeds thereof toward payment of any costs and expenses and attorneys' fees and legal expenses thereby incurred by Secured Party, and toward payment of the obligations under the Purchase Agreement in such order or manner as Secured Party may elect. Notwithstanding anything to the contrary contained herein, the Secured Party shall only foreclose on that portion of the Stock that is reasonably necessary in the reasonable good faith judgment of the Secured Party in order to satisfy the amount of the claim constituting the Resolved Event of Offset. For purposes hereof, the Agreed Value of the Stock shall be deemed to be the value that the Secured Party is receiving on the foreclosure of the Stock and Secured Party shall not be entitled to foreclose on more Stock than is necessary to recover all of its damages resulting from the Resolved Event of Offset.

- (b) Secured Party is hereby granted the right, at its option, after a Continuing Event of Offset, to transfer at any time to itself or its nominee the securities or other property hereby pledged, or any part thereof, and to thereafter exercise all voting rights with respect to such Stock so transferred and to receive the proceeds, payments, monies, income or benefits attributable or accruing thereto and to hold the same as security for the obligations hereby secured, or at Secured Party's election, to apply such amounts to the obligations, only if due, and in such order as Secured Party may elect or Secured Party may, at its option, without transferring such securities or property to its nominee, exercise all voting rights with respect to the securities pledged hereunder and vote all or any part of such securities at any regular or special meeting of shareholders.
- (c) Pledgor hereby agrees to cooperate fully with Secured Party in order to permit Secured Party to sell, at foreclosure or other private sale, Pledgor's interest in the Stock pledged hereunder as provided in this Pledge Agreement. Specifically, Pledgor agrees to deliver to Secured Party the certificate or certificates representing the Stock if Pledgor has possession at that time, to fully comply with the securities laws of the United States and of the State of Texas and to take such other action as may be necessary to permit Secured Party to sell or otherwise transfer the securities pledged hereunder in compliance with such laws.
- 5. Termination. This Pledge Agreement shall continue as security for the payment or satisfaction of the obligations under the Purchase Agreement until the earliest to occur of (i) termination of this Pledge Agreement by written

notice of the Secured Party to the Pledgor or (ii) the date upon which none of the representations and warranties of Pledgor contained in the Purchase Agreement survive and all covenants and obligations of Pledgor under said Purchase Agreement have been fully and properly performed, or (iii) three (3) years after the date hereof, provided that Secured Party has not given Pledgor notice of an Event of Offset which has not been satisfied by Pledgor, or if there is an Event of Offset, the pledge shall continue only to the extent of the number of Shares based on the Agreed Value equal to the amount of damages reasonably expected to be caused by the Event of Offset; provided however, upon the expiration of one year and three years after the date of this Pledge Agreement (each a Release Date), the following provisions shall apply: (x) if no Event of Offset exists on such Release Date, the Secured Party shall release one-half (1/2) of the number of Shares initially pledged hereunder from the pledge established hereunder, and the remaining Shares shall remain pledged under the terms and conditions of this Pledge Agreement;

-3-

or (y) if an Event of Offset exists, the amount of Damages resulting from such Event of Offset shall be determined, and on the first and second Release Date, one-half and all of the remaining Shares, respectively, that have not been Offset against shall be released from the pledge established hereby and delivered to the Pledgor and the remaining Shares shall remain pledged under the terms and conditions of this Pledge Agreement. If such a Continuing Event of Offset exists, the pledge shall continue only to the extent of the amount of Stock (based on the Agreed Value) equal to the amount of Damages claimed in the Offset Claim or the amount of damages reasonably expected to be caused by the Event of Offset, as applicable.

- 6. Release from Pledge. Upon the termination of this Pledge Agreement, Secured Party shall immediately release its security interest in the Stock. In addition, Secured Party shall deliver the certificate or certificates representing the Stock to Pledgor if Secured Party has possession of such certificates at that time. Upon such occurrence, the security interest of Secured Party shall automatically terminate and Secured Party shall thereafter have no interest whatsoever in the Stock.
- 7. Notices. All notices, requests, demands, claims, and other communications hereunder will be in writing. Any notice, request, demand, claim, or other communication hereunder shall be deemed duly given if (and then two business days after) it is sent by registered or certified mail, return receipt requested, postage prepaid, and addressed to the intended recipient as set forth below:

If to Pledgor: Reporting Services Associates, Inc.

225 South 15/th/ Street Philadelphia, PA. 19102 Telephone: (215) 735-2332 Telefax: (215) 735-1695 Attn. Lee Goldstein

Copy to: Benjamin S. Ohrenstein

354 West Lancaster Avenue

Suite 212

Haverford, PA. 19041

Telephone: (610) 649-1268 Telefax: (610) 642-6553

If to the

Secured Party: Litigation Resources of America-Northeast, Inc.

c/o Litigation Resources of America, Inc.

650 First City Tower, 1001 Fannin

Houston, Texas 77002

Attn: President

Copy to: Boyer Ewing & Harris Incorporated

Nine Greenway Plaza, Suite 3100

-4-

Houston, Texas 77046 Attn: John R. Boyer

- 8. Successors. This Pledge Agreement shall be binding upon, and inure to the benefit of the parties hereto and their successors and assigns. Any assignee whatsoever will be bound by the obligations of the assigning party under this Pledge Agreement, and any assignment shall not diminish the liability or obligation of the assignor under the terms of this Pledge Agreement unless otherwise agreed.
- 9. Severability. In the event that any one or more of the provisions contained in this Pledge Agreement or in any other instrument referred to herein, shall, for any reason, be held to be invalid, illegal, or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision of this Pledge Agreement or any such other instrument.
- 10. Paragraph Headings. The paragraph headings used herein are descriptive only and shall have no legal force or effect whatsoever.
- 11. Gender. Whenever the context so requires, the masculine shall include the feminine and neuter, and the singular shall include the plural and conversely.
- 12. Survival of Warranties. All representations, warranties, and agreements made by the parties in this Pledge Agreement or in any certificates delivered pursuant hereto will survive the execution date hereof.

- 13. Applicable Law. This Pledge Agreement shall be construed and interpreted in accordance with the laws of the United States of America and the State of Texas, and is intended to be performed in accordance with and as permitted by such laws.
- 14. Definitions. All terms and definitions used herein shall have the same meaning as in the Purchase Agreement unless otherwise indicated.
- 15. Drafting. The parties hereto acknowledge that each party was actively involved in the negotiation and drafting of this Pledge Agreement and that no law or rule of construction shall be raised or used in which the provisions of this Pledge Agreement shall be construed in favor or against either party hereto because one is deemed to be the author thereof.
- 16. Attorneys' Fees. If any litigation is instituted to enforce or interpret the provisions of this Pledge Agreement or the transactions described herein, the prevailing party in such action shall be entitled to recover its reasonable attorneys' fees from the other party hereto.
- 17. Arbitration. The arbitration provisions contained in Section 10.14 of the Purchase Agreement shall govern this Pledge Agreement.

-5-

18. Multiple Counterparts. This Pledge Agreement may be executed in multiple counterparts each of which shall be deemed an original and all of which shall constitute one instrument.

-6-

IN WITNESS WHEREOF, this Pledge Agreement has been executed effective as of the date first above written.

PLEDGOR:

REPORTING SERVICES ASSOCIATES, INC.

By:
Lee Goldstein, President

SECURED PARTY:

LITIGATION RESOURCES OF AMERICA -- NORTHEAST, INC.

By:

-7-

EXHIBIT F

DRAFT OF SEPTEMBER 25, 1997

# GUARANTY OF PERFORMANCE

THIS GUARANTY OF PERFORMANCE (the "Guaranty") is executed effective the ___ day of _____, 1997, by Lee Goldstein, an individual (the "Guarantor"), pursuant to the terms of that certain Agreement of Purchase and Sale of Assets (the "Agreement") entered into effective as of the even date herewith, by and among Litigation Resources of America -- Northeast, Inc., a New York corporation ("Buyer"), Reporting Services Associates, Inc., a Pennsylvania corporation ("Seller"), Lee Goldstein, an individual ("Goldstein"), and Litigation Resources of America, Inc., a Texas corporation and the parent company of Buyer ("Parent") (Buyer and the Parent are sometimes hereinafter referred to collectively as the "LRA Companies"). All defined terms contained herein shall have the meanings ascribed thereto in the Agreement.

WITNESSETH:

WHEREAS, pursuant to the terms of the Agreement, Buyer shall purchase substantially all of the Assets of Seller, upon the terms and conditions contained in the Agreement; and

WHEREAS, Goldstein is the sole shareholder of Seller; and

WHEREAS, Guarantor acknowledges that he is receiving numerous and substantial benefits from the consummation of the transactions contemplated in the Agreement, and that he is willing to execute this Guaranty for and in consideration of such benefits, and in order to induce the LRA Companies to enter into the Agreement and consummate the transactions contemplated therein; and

WHEREAS, the LRA Companies would not have executed and consummated the transactions described in the Agreement but for the agreement of Guarantor to execute and deliver to the LRA Companies this Guaranty;

NOW, THEREFORE, for and in consideration of the terms and conditions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by Guarantor, and in order to induce the LRA Companies to enter into the Agreement and consummate the transactions contemplated therein, the Guarantor hereby covenants and agrees as follows:

1. Guarantor hereby absolutely, unconditionally and irrevocably

guarantees the full and punctual payment and performance by the Seller of all of the obligations, duties, covenants, agreements and conditions provided in the Agreement to be paid or performed by the Seller thereunder, including without limitation all indemnification obligations of the Seller set forth in Article VII of the Agreement.

2. This Guaranty is unconditional and the liability of Guarantor shall be primary and absolute, in the same manner as if Guarantor were named in and had signed the Agreement individually as the Seller thereunder. Guarantor agrees that neither bankruptcy, insolvency, lack of

capacity nor any other disability or impediment against enforcement of the liability of the Seller shall in any manner impair or affect Guarantor's liabilities and obligations hereunder.

- 3. It shall not be necessary or required in order to maintain and enforce Guarantor's liability hereunder that demand be made upon the Seller or that action be commenced or prosecuted against the Seller or that any effort be made to enforce the liability or responsibility of the Seller for performance of the Seller's obligations or duties under or in connection with the Agreement, and it shall not be required that the Seller or any other party liable on the Agreement be joined in any action brought against Guarantor for enforcement of Guarantor's liability and responsibility under this Guaranty, or that judgment have theretofore been obtained against the Seller or any other party liable therefor or in connection with any such claim.
- 4. Guarantor agrees that no waiver by the LRA Companies or forbearance or delay by the LRA Companies in asserting or enforcing any rights or remedies of either of the LRA Companies or with respect to the Seller or any other party who may be or become responsible for performance of any of Seller's obligations or duties shall in any wise affect, impair, or release Guarantor's liability hereunder.
- 5. Guarantor expressly waives and agrees that no notice of default by the Seller or other notice or demand need be given by the LRA Companies to Guarantor as a condition of maintaining or enforcing Guarantor's liabilities and obligations under this Guaranty. Likewise, Guarantor agrees that the LRA Companies' release or subordination of, or failure or delay to enforce or seek to realize upon, any security now or hereafter held or acquired by the LRA Companies for performance of any of the obligations or duties of the Seller under or in connection with the Agreement, or any other action which either of the LRA Companies may take or fail to take with respect to or against the Seller, shall not impair, affect, or release Guarantor's liability hereunder.
- 6. In the event default is made in the prompt payment of amounts due, or performance of obligations due, under this Guaranty, and the same is placed in the hands of an attorney for collection or enforcement, or suit is brought on same, and the same is collected or enforced through any judicial proceeding whatsoever, then Guarantor agrees and promises to pay all of the attorneys' and collection fees, costs and expenses incurred by the LRA Companies in enforcing

their rights hereunder.

- 7. The obligations of Guarantor shall continue in full force and effect against Guarantor until all of the obligations guaranteed hereunder have been paid in full, and fully and finally performed. This Guaranty covers any and all of such obligations, whether presently outstanding or arising subsequent to the date hereof. This Guaranty is valid and binding upon and enforceable against Guarantor and the successors and assigns of Guarantor.
- 8. All rights of the LRA Companies hereunder or otherwise arising under any documents executed in connection with or as security for the obligations guaranteed hereby, are separate and cumulative and may be pursued separately, successively, or concurrently.

2

- 9. Notwithstanding any provision of this Guaranty to the contrary, the liability of Guarantor hereunder shall not exceed the liability which Guarantor would have under the Agreement if it had signed the Agreement as the Seller thereunder.
- 10. This instrument may be amended or modified only by written instrument signed by Guarantor and the LRA Companies.
- 11. This instrument sets forth the entire agreement and understanding between Guarantor and the LRA Companies with respect to the subject matter hereof, and may be executed in multiple counterparts, each of which shall have the force and effect of an original, and all of which together shall constitute one and the same agreement.

THE UNDERSIGNED ACKNOWLEDGES THAT THE EXECUTION OF THIS GUARANTY RESULTS IN LIABILITY ON THE PART OF THE UNDERSIGNED FOR THE REPAYMENT OF THE DEBTS AND THE PERFORMANCE OF THE OBLIGATIONS HEREIN DESCRIBED, AND COULD RESULT IN THE ATTACHMENT OF THE UNDERSIGNED'S ASSETS. THE UNDERSIGNED ACKNOWLEDGES HAVING RECEIVED THE ADVICE OF LEGAL COUNSEL PRIOR TO EXECUTION OF THIS DOCUMENT.

IN WITNESS WHE	REOF,	the	undersigned	has	executed	this	Guaranty	of
Performance effecti	ve as	of _		,	1997.			

REPORTING SERVICES ASSOCIATES, INC. 225 South 15th Street--22nd Floor Philadelphia, Pennsylvania 19102

September 15, 1997

VIA FACSIMILE ONLY

Mr. Richard Looney Chief Executive Officer Litigation Resources of America, Inc. 1001 Fannin Suite 650 Houston, Texas 77002

Dear Mr. Looney:

In furtherance of our conversations of earlier today we have agreed to revise to some extent the two (2) letters which I faxed to you on September 12, 1997. This letter is intended to replace both of those letters, and neither of those letters shall now be considered viable or of any effect.

A copy of this letter, when executed by you, on behalf of Litigation Resources of America, Inc. ("LRA"), shall be considered a legally binding agreement between LRA and Reporting Services Associates, Inc. ("RSA").

This will confirm that the following accurately sets forth the terms and provisions of the agreement which you and I have negotiated the past few days on behalf of LRA and RSA:

- --In view of the failure of LRA and RSA to execute a Purchase Agreement and/or other agreements and related instruments prior to September 1, 1997 as contemplated by the letter of intent dated July 16, 1997 between LRA and RSA, RSA is no longer required to deal exclusively with LRA regarding the sale of its assets, and that said letter of intent has been terminated.
- --Notwithstanding the termination of the aforesaid letter of intent and provided that all of the terms and provisions set forth herein are satisfied (it being agreed that time is of the essence as to the satisfaction of all such terms and conditions), RSA agrees that as of the execution date of this letter, it, RSA, again will exclusively deal with LRA in good faith negotiations regarding the sale of its assets or its business through the period or periods of time hereinafter.

Page Two Mr. Richard Looney

- --Upon the execution by LRA and RSA of a mutually acceptable definitive purchase agreement as defined hereinbelow. LRA will pay RSA the sum of Four Hundred Thousand Dollars (\$400,000.00) in cash, or the acceptable equivalent to RSA. If the closing of the transaction does not take place by the close of business on January 15, 1998, then RSA, at its option, may elect to extend the termination date of the said period of time to March 31, 1998; it may do so, provided that on or before January 15, 1998 it notifies LRA in writing of its election to do so and, provided further, that within 72 hours of receipt of said written notice, LRA pays RSA the additional sum of Eighty Seven Thousand Five Hundred Dollars (\$87,500.00) in cash, or the acceptable equivalent to RSA. In the event the IPO contemplated by the purchase agreement as defined hereinbelow has not come to fruition by January 15, 1998 and in the further event RSA does not exercise its option to extend the termination date to March 31, 1998, it, RSA, shall repay LRA the sum of Two Hundred Thousand Dollars (\$200,000.00) on or before January 16, 1998.
- --On or before Noon (C.D.S.T.), Wednesday, September 17, 1997 LRA and RSA shall execute a mutually acceptable definitive purchase agreement ("Purchase Agreement") for the sale of RSA's assets or its business and all other appropriate related agreements and instruments in form and terms similar to those documents which have been negotiated to date pursuant to the aforementioned terminated letter of intent dated July 16, 1997, except that no payments of cash thereunder shall be made to RSA at the time of execution of the Purchase Agreement but, rather, said payments shall be promptly made to RSA from the proceeds of the IPO contemplated by the Purchase Agreement, and the payments to RSA shall consist of cash in the sum of Seven Million Dollars (\$7,000,000.00) and stock with an aggregate value of Two Million Five Hundred Thousand Dollars (\$2,500,000.00) at 90% of the IPO price. The payments to RSA pursuant to the Purchase Agreement shall not be reduced or offset by any payments to RSA for its agreement to deal exclusively with LRA as set forth hereinabove.
- --Lee R. Goldstein, on behalf of RSA, shall have the right and opportunity to meet with the representatives of LRA, its underwriters, accountants, and other individuals who are involved in preparing for its IPO, and to continue to meet with them, observe their preparations, and review their work products at any and all times during Monday, September 15, 1997 and Tuesday, September 16, 1997.

Page Three Mr. Richard Looney September 15, 1997

- Within, and no later than three (3) months from the date of execution of this letter, and at no additional cost to RSA, RSA shall be provided with full and complete copies of certified audited reports, including signed appropriate certified opinion letters, prepared by Coopers & Lybrand, LLP, for the calendar years 1995 and 1996 and for the twelve (12) months ended June 30, 1997 of the operations of RSA.
- A copy of this letter or its counterpart, duly signed by an authorized officer of LRA, must be returned to RSA at its offices in Philadelphia, Pennsylvania, or personally handed to Lee R. Goldstein, not later than 2:00 P.M. (C.D.S.T.), Tuesday, September 16, 1997.

This Agreement may be executed in two (2) or more counterparts, each of which shall be deemed to be an original, but all of which shall constitute the same Agreement.

Very truly yours,

/s/ Lee R. Goldstein

LEE R. GOLDSTEIN President

LRG:lds

Intending to be legally bound hereby, agreed to and accepted by the undersigned this 16th day of September, 1997.

LITIGATION RESOURCES OF AMERICA, INC.

By: /s/ Richard Looney

Richard Looney
Chief Executive Officer

#### FIRST AMENDMENT TO EMPLOYMENT AGREEMENT

THIS FIRST AMENDMENT TO EMPLOYMENT AGREEMENT (the "Agreement"), dated effective the 1st day of October, 1997 (the "Effective Date"), is entered into by and among U.S. LEGAL SUPPORT, INC., a Texas corporation ("USLS"), LOONEY & COMPANY, a Texas corporation (the "Company") and RICHARD O. LOONEY, an individual residing in the State of Texas (the "Employee"). The Company and USLS may sometimes hereinafter be referred to collectively as "Employers" or singularly as "Employer." The Employers and Employee may sometimes hereinafter be referred to singularly as a "Party" or collectively as the "Parties."

W I T N E S S E T H:

WHEREAS, Litigation Resources of America, Inc., the Company and Employee entered into that certain employment agreement ("Employment Agreement") dated January 17, 1997;

WHEREAS, Litigation Resources of America, Inc. changed its name to "U.S. Legal Support, Inc." effective September 25, 1997;

WHEREAS, the Employee has been the President and Chief Executive Officer of the Employers under the Employment Agreement since January 17, 1997 and his knowledge of the affairs of the Company, particularly the court reporting business in Texas and nationwide, are of great value to the Employers;

WHEREAS, in consideration of the services and knowledge of the Employee and in consideration of Employee's agreement to an increase in the scope of the covenant not to compete contained herein, the Company has agreed to increase the compensation of Employee hereunder;

NOW THEREFORE, for and in consideration of the mutual covenants, promises and undertakings hereinafter contained, the Parties hereby agree as follows:

- 1. In the event of the successful consummation of a public offering by USLS of its securities for cash in an underwritten public offering registered on the appropriate form with the SEC, then effective 12:01a.m. on the date of such public offering, the Employment Agreement shall be amended as follows:
  - A. Section 4(a) of the Employment Agreement is hereby deleted in its entirety and substituted therefore is the following:
    - "(a) a salary in the amount of Two Hundred Fifty Thousand and No/100 Dollars (\$250,000.00) per year

effective as of the date hereof which shall be payable during the term of this Agreement in accordance with the payroll policies of the Employers in effect from time to time (but in no event less frequently than monthly); and,"

- B. Section 12(a) of the Employment Agreement is hereby deleted in its entirety and substituted therefore is the following:
  - "(a) Employee will not in any capacity or relationship enter into, engage in, or be connected with any business or business operation or activity within a fifty (50) mile radius of any office location then operated by the Employers or their Affiliates at the time of such termination, which consists in whole or in part of the Business of the Employers (as defined hereinafter). For purposes of this Agreement, the "Business of the Employers" shall be defined as the current business and all future business of the Employers and their Affiliates, including, but not limited to, the providing of court reporting, litigation support services, and legal staffing services; and,"
- 2. Drafting. Both Parties hereto acknowledge that each Party was actively involved in the negotiation and drafting of this Agreement and that no law or rule of construction shall be raised or used in which the provisions of this Agreement shall be construed in favor or against either Party hereto because one is deemed to be the author thereof.
- 3. Employment Agreement. All other terms and conditions contained in the Employment Agreement shall remain in full force and effect except as otherwise specifically amended thereby.

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement as of the date and year first above written.

THE COMPANY:
LOONEY & COMPANY,
a Texas corporation

By: /s/ Richard O. Looney
-----Richard O. Looney, President

-2-

U.S. LEGAL SUPPORT, INC., a Texas corporation

By: /s/ David W. Pfleghar

._____

David W. Pfleghar Chief Financial Officer

THE EMPLOYEE:

/s/ Richard O. Looney

Richard O. Looney

#### FIRST AMENDMENT TO EMPLOYMENT AGREEMENT

THIS FIRST AMENDMENT TO EMPLOYMENT AGREEMENT (the "Agreement"), dated effective the 1st day of October 1997, is entered into by and among KLEIN, BURY AND ASSOCIATES, INC., a Florida corporation (the "Company"), MICHAEL KLEIN (the "Employee") and U.S. LEGAL SUPPORT, INC., a Texas corporation and parent corporation of the Company (formerly "Litigation Resources of America, Inc.") ("USLS"). The Company, Employee and the USLS may sometimes hereinafter be referred to singularly as a "Party" or collectively as the "Parties."

W I T N E S S E T H:

WHEREAS, the Company, Employee, and Litigation Resources of America, Inc. entered into that certain employment agreement ("Employment Agreement") dated January 17, 1997;

WHEREAS, Litigation Resources of America, Inc. changed its name to "U.S. Legal Support, Inc." effective September 25, 1997;

WHEREAS, the Employee has been the President of the Company under the Employment Agreement since January 17, 1997 and his knowledge of the affairs of the Company, particularly the court reporting business in Florida and nationwide, continue to be of great value to the Company and USLS;

WHEREAS, in consideration of the services and knowledge of the Employee and in consideration of Employee's agreement to an increase in the scope of the covenant not to compete contained herein, the Company has agreed to increase the compensation of Employee hereunder.

NOW THEREFORE, for and in consideration of the mutual covenants, promises and undertakings hereinafter contained, the Parties hereby agree as follows:

- 1. Section 4(a) of the Employment Agreement is hereby deleted in its entirety and substituted therefore is the following:
  - "(a) a salary in the amount of One Hundred Seventy Five Thousand and No/100 Dollars (\$175,000.00) per year effective as of the date hereof which shall be payable during the term of this Agreement in accordance with the payroll policies of the Company in effect from time to time (but in no event less frequently than monthly); and,"
- 2. Drafting. Both Parties hereto acknowledge that each Party was actively involved in the negotiation and drafting of this Agreement and that no law or rule of construction shall be raised or used in which the provisions of this Agreement shall be construed in favor or against either Party hereto because one is deemed to be the author thereof.

3.	Employment	Agreer	ment.	All	other	terms	and	condi	tions	conta	ained	in	the
Employment	. Agreement	shall	remain	in	full	force	and	effect	excep	ot as	other	cwis	se
specifical	Lly amended	therek	oy.										

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement as of the date and year first above written.

THE COMPANY: KLEIN, BURY & ASSOCIATES, INC., a Florida corporation

By: /s/ Richard O. Looney

Richard O. Looney

Chief Executive Officer

U.S. LEGAL SUPPORT, INC., a Texas corporation

By: /s/ Richard O. Looney

Richard O. Looney
Chief Executive Officer

THE EMPLOYEE:

/s/ Michael Klein
----Michael Klein

## CONFIDENTIAL

-----

Mr. Richard Looney President Looney & Company 650 First City Tower 1001 Fannin Houston, Texas 77002

Dear Mr. Looney:

The GulfStar Group, Inc. ("GulfStar") welcomes the opportunity to assist Looney & Company ("Looney" or the "Company") in the acquisition of the stock of and/or the substantial portion of the assets of up to three companies that provide litigation support services similar to those provided by Looney (individually and collectively hereinafter referred to as the "Acquisition") and in the private placement of equity financing and/or senior debt financing ("Private Placement") to be used to fund acquisitions, expansion and working capital. GulfStar anticipates the Private Placement to be for an aggregate amount of approximately \$5 million to \$10 million, the amount of which shall be mutually agreed upon.

Set forth below are (i) the services GulfStar will perform in the course of this assignment, (ii) the total fees and expenses payable to GulfStar in exchange for those services and (iii) general terms and conditions of the engagement.

### 1. SERVICES TO BE PROVIDED BY GULFSTAR

During the course of the assignment, we will perform the following private placement services on behalf of the Company:

. GulfStar will assist the Company in determining an appropriate value and structure for each acquisition.

Mr. Richard Looney March 4, 1996 Page 2

. We will also assist the negotiation of the attendant letters of intent and in the subsequent due diligence process.

- . If satisfactory terms are agreed upon, GulfStar will serve as the Company's financial advisor throughout the process of drafting and negotiating the necessary definitive purchase agreements.
- . With the assistance of Looney, we will prepare a Confidential Memorandum describing the Company and outlining the desired terms of the financing.
- . Once the Confidential Memorandum has been reviewed and approved by Looney, we will contact a focused group of institutional investors who will then receive a copy of the Confidential Memorandum only if they express preliminary interest in the transaction.
- . Subsequent to the initial circulation of the Confidential Memorandum, GulfStar will arrange for prospective investors to meet with the Company's management and will solicit preliminary commitments.
- . We will assemble an investor group and assist the Company in negotiating the financing on the most favorable terms available.
- . If satisfactory terms are agreed upon, we will serve as the Company's financial advisor through the process of drafting and executing a definitive funding agreement and related documents.
- . If GulfStar is requested to provide other services (e.g. additional merger and acquisition advisory services, debt placement services, etc.) such assignments and fees will be negotiated separately on mutually agreeable terms.

# II. FEE AND EXPENSE ARRANGEMENTS

The professional fees and expense reimbursements payable to GulfStar with respect to this assignment are set forth below:

. GulfStar shall be paid a retainer of \$25,000 ("Retainer Fee") payable \$15,000 in cash upon execution of this agreement and \$10,000 upon execution of a letter of intent to complete the Acquisition. Upon the successful completion

Mr. Richard Looney March 4, 1996 Page 3

of the financing contemplated herein, the Retainer Fee shall be credited against the Placement Fee (as hereinafter defined).

GulfStar shall be paid a private placement fee ("Placement Fee") equal to 3.0% of the principal amount of capital (equity and debt) arranged, irrespective of the amount funded at closing, provided, however, the minimum Placement Fee in the event that a Private Placement

transaction closes shall be \$150,000. Such fee shall be payable by Looney only in the event of a successful placement of new capital and shall be due in cash at closing.

- GulfStar shall be paid a merger and acquisition advisory fee equal to 1.0% of the total amount of consideration paid in each Acquisition ("M&A Advisory Fee"). Such fee shall be payable in cash upon the successful completion of the Acquisition as contemplated herein. Notwithstanding anything to the contrary herein, the foregoing M&A Advisory Fee shall not be payable by the Company in connection with any transaction involving the sale of the assets or stock of Looney & Company to an affiliate, or the purchase by Looney & Company of the assets or stock of an affiliate. For purposes of calculating the M&A Advisory Fee, consideration is hereby understood to include cash, stock, long-term debt assumed, earnouts, contingent payments and seller financing. Notwithstanding anything to the contrary, GulfStar shall have the right, but not the obligation, to invest up to 100% of the M&A Advisory Fee in the Company on terms identical to those of the Private Placement contemplated herein.
- In the event that all or a substantial portion of the assets, stock or business of the Company is acquired by or merged into another unaffiliated company (the "Sale"), GulfStar shall be paid a sale advisory fee equal to 2.0% of the total amount of consideration paid in such transaction ("Sale Advisory Fee"). For purposes of calculating the Sale Advisory Fee, consideration is hereby understood to include cash, stock, long-term debt assumed, earnouts, contingent payments and seller financing. Such fee shall be payable in cash upon the successful completion of the Sale as contemplated herein. The obligation to pay the Sale Advisory Fee shall expire in accordance with the provisions contained in Section III(B) provided hereinbelow.

Mr. Richard Looney March 4, 1996 Page 4

- Upon closing a transaction involving the successful placement of new capital, GulfStar shall also receive five year warrants to purchase an amount of the Company's fully-diluted Common Stock equal to [2.5% multiplied by (100% minus the percentage of the Company sold in the Private Placement)]. The purchase price of these warrants will be equal to 110% of the price used for purposes of this transaction. The terms of such warrants will generally be equivalent to the terms used in the equity placement contemplated herein.
- . GulfStar shall be reimbursed on a monthly basis for our direct out-of-pocket expenses reasonably incurred in connection with this assignment, such amount not expected to exceed \$10,000 in the aggregate. In connection with the foregoing, GulfStar shall furnish the Company written documentation of all out-of-pocket expenses that

it has incurred together with such other information that is reasonably necessary to satisfy then applicable expense deduction reporting requirements of the Internal Revenue Service.

#### III. GENERAL TERMS AND CONDITIONS

#### A. DEFINITION OF LOONEY & COMPANY

The terms "Looney" and the "Company", as employed herein, are understood to include Looney & Company and all of its subsidiaries, and affiliated companies.

#### B. TERM OF ENGAGEMENT

GulfStar will have the exclusive right, for an initial period of 150 days following the execution date, to serve as Looney's investment banking representative with respect to the matters set forth herein. Upon expiration of 150 days, the engagement shall be automatically extended unless terminated in writing by either GulfStar or Looney. Further, any entity contacted during the course of this assignment will be deemed to be an interested party ("Interested Party"). The Interested Parties will be identified by us in writing upon the earlier of the termination or successful consummation of this assignment. Should this assignment be terminated prior to the completion of the transaction(s) contemplated hereby, and should the Company subsequently complete such transaction(s) with an Interested Party within a 12 month period following such termination, then GulfStar

Mr. Richard Looney March 4, 1996 Page 5

shall be due full compensation under II with respect to the participation of any Interested Parties in such financing.

#### C. INDEMNIFICATION

Looney agrees to indemnify and hold GulfStar (and each of its partners, officers and employees) harmless against any losses, claims, damages or liabilities which GulfStar may be subject to in connection with services performed in its capacity as advisor as described in this engagement letter and to periodically reimburse GulfStar for any legal and other expenses reasonably incurred by GulfStar in connection with investigating or defending any action or claim in connection therewith, provided however, that Looney shall not be obligated under the foregoing indemnity agreement with respect to any loss, claim, damage or liability (or action in connection therewith) to the extent that a court of competent jurisdiction shall have determined by final judgment that such loss, claim, damage or liability resulted from the

willful misfeasance or gross neglect of GulfStar, and provided further that if Looney assumes the defense of GulfStar then Looney shall not be obligated to pay any legal fees or expenses thereafter incurred by GulfStar, so long as there are no conflicting legal defenses or interests between Looney and GulfStar. Looney shall not be liable for any action settled without its consent. The indemnification and reimbursement provided to GulfStar hereunder shall be applicable whether or not negligence of GulfStar is alleged or proven.

#### D. LACK OF INDEPENDENT VERIFICATION

During the course of this assignment, GulfStar may rely upon the opinions of experts (including, but not limited to, independent public accounting firms) with respect to the accuracy of certain data provided by Looney. GulfStar will make no effort to independently verify the accuracy of any expert opinions so relied upon.

#### E. CONFIDENTIALITY

All non-public information supplied to GulfStar by Looney with respect to the Company will be held in strict confidence, as we understand that much of this information is treated as highly confidential by you and is not normally

Mr. Richard Looney March 4, 1996 Page 6

divulged to outside sources.

#### F. AMENDMENTS

Both parties agree that this document can be modified or amended only through the written agreement of GulfStar and Looney.

If the foregoing accurately sets forth your understanding of our agreement, please so indicate by signing, dating and returning one of the enclosed copies of this letter, while retaining the other copy for your records.

We are delighted to have the opportunity to assist you in this important matter and we look forward to working with you in the successful completion of this assignment.

Sincerely,

THE GULFSTAR GROUP, INC.

By: /s/ Daryl R. Swarts

# Daryl R. Swarts Managing Director

ACCEPTED AND AGREED TO:

LOONEY & COMPANY

By: /s/ Richard O. Looney

_____

Title: CEO

_____

Date: March 4, 1996

#### CONSENT OF INDEPENDENT ACCOUNTANTS

We consent to the inclusion in this registration statement on Form S-1 (File No. 333-36575) (i) of our report dated December 15, 1997, which includes an explanatory paragraph regarding the Company's ability to continue as a going concern, on our audit of the consolidated financial statements and financial statement schedules of U.S. Legal Support, Inc., (ii) of our report dated September 5, 1997, on our audits of the financial statements and financial statement schedule of Looney & Company, (iii) of our report dated August 15, 1997 of our audits of the financial statements of Klein, Bury & Associates, (iv) of our report dated September 4, 1997, of our audits of the financial statements of G&G Court Reporters, (v) of our report dated September 19, 1997, of our audits of the financial statements of San Francisco Reporting Service, (vi) of our report dated September 5, 1997, of our audits of the financial statements of Legal Enterprise, Inc., (vii) of our report dated August 29, 1997, of our audits of the financial statements of Elaine P. Dine, Inc., (viii) of our report dated September 5, 1997, of our audits of the financial statements of Burton House, Inc. d.b.a. Ziskind, Greene, Watanabe, & Nason, (ix) of our report dated September 5, 1997 of our audits of the financial statements of Jilio & Associates, (x) of our report dated August 29, 1997, except as to the information presented in Notes 4 and 6, for which the date is September 12, 1997, of our audits of the financial statements of Reporting Service Associates, Inc., (xi) of our report dated August 29, 1997, of our audits of the financial statements of Kirby A. Kennedy & Associates, (xii) of our report dated September 19, 1997 of our audits of the financial statements of Johnson Court Reporting Group, (xiii) of our report dated October 21, 1997, of our audits of the financial statements of Amicus One Legal Support Services, Inc., (xiv) of our report dated September 19, 1997 of our audits of the financial statements of Block Court Reporting, Inc., and (xv) of our report dated September 22, 1997, of our audits of the financial statements of Commander Wilson, Inc. We also consent to the references to our firm under the caption "Experts."

COOPERS & LYBRAND L.L.P.

Houston, Texas

December 17, 1997